Mental Health Stigma in the Military



Joie D. Acosta, Amariah Becker, Jennifer L. Cerully, Michael P. Fisher, Laurie T. Martin, Raffaele Vardavas, Mary Ellen Slaughter, Terry L. Schell



ELECTRONIC COPIES OF RAND RESEARCH ARE PROVIDED FOR PERSONAL USE; POSTING TO A NONRAND WEBSITE IS PROHIBITED. THIS PUBLICATION IS AVAILABLE FOR LINKING OR FREE DOWNLOAD AT **WWW.rand.org**

	Report Docume	entation Page			Form Approved 1B No. 0704-0188
maintaining the data needed, and c including suggestions for reducing	lection of information is estimated to completing and reviewing the collect this burden, to Washington Headqu uld be aware that notwithstanding ar DMB control number.	ion of information. Send comments arters Services, Directorate for Info	regarding this burden estimate rmation Operations and Reports	or any other aspect of the s, 1215 Jefferson Davis	is collection of information, Highway, Suite 1204, Arlington
1. REPORT DATE 2014		2. REPORT TYPE		3. DATES COVE 00-00-2014	RED to 00-00-2014
4. TITLE AND SUBTITLE				5a. CONTRACT	NUMBER
Mental Health Stigma in the Military			5b. GRANT NUN	IBER	
				5c. PROGRAM E	LEMENT NUMBER
6. AUTHOR(S)				5d. PROJECT NU	JMBER
				5e. TASK NUME	SER
				5f. WORK UNIT	NUMBER
RAND Corporation	ZATION NAME(S) AND AE n,National Defense I Box 2138,Santa Mor	Research Institute (8. PERFORMINO REPORT NUMB	G ORGANIZATION ER
9. SPONSORING/MONITO	RING AGENCY NAME(S) A	AND ADDRESS(ES)		10. SPONSOR/M	ONITOR'S ACRONYM(S)
				11. SPONSOR/M NUMBER(S)	ONITOR'S REPORT
12. DISTRIBUTION/AVAII Approved for publ	LABILITY STATEMENT ic release; distributi	ion unlimited			
13. SUPPLEMENTARY NO	OTES				
14. ABSTRACT					
15. SUBJECT TERMS					
16. SECURITY CLASSIFIC	CATION OF:		17. LIMITATION OF	18. NUMBER	19a. NAME OF
a. REPORT unclassified	b. ABSTRACT unclassified	c. THIS PAGE unclassified	ABSTRACT Same as Report (SAR)	OF PAGES 329	RESPONSIBLE PERSON

Standard Form 298 (Rev. 8-98) Prescribed by ANSI Std Z39-18 For more information on this publication, visit www.rand.org/t/RR426

Library of Congress Cataloging-in-Publication Data is available for this publication.

ISBN: 978-0-8330-8504-7

Published by the RAND Corporation, Santa Monica, Calif. © Copyright 2014 RAND Corporation **RAND**^{*} is a registered trademark.

Limited Print and Electronic Distribution Rights

This document and trademark(s) contained herein are protected by law. This representation of RAND intellectual property is provided for noncommercial use only. Unauthorized posting of this publication online is prohibited. Permission is given to duplicate this document for personal use only, as long as it is unaltered and complete. Permission is required from RAND to reproduce, or reuse in another form, any of its research documents for commercial use. For information on reprint and linking permissions, please visit www.rand.org/pubs/permissions.html.

The RAND Corporation is a research organization that develops solutions to public policy challenges to help make communities throughout the world safer and more secure, healthier and more prosperous. RAND is nonprofit, nonpartisan, and committed to the public interest.

RAND's publications do not necessarily reflect the opinions of its research clients and sponsors.

Support RAND Make a tax-deductible charitable contribution at www.rand.org/giving/contribute

www.rand.org

Preface

Despite the efforts of both the U.S. Department of Defense (DoD) and the Veterans Health Administration to enhance mental health services, many service members are not regularly seeking needed care when they have mental health symptoms or disorders. The research team hypothesized that mental health stigma may be a barrier to mental health treatment-seeking among military service members. Without appropriate treatment, these mental health symptoms or disorders can have wide-ranging and negative impacts on the quality of life and the social, emotional, and cognitive functioning of affected service members.

The RAND National Defense Research Institute (NDRI) was asked to inventory and assess stigma-reduction strategies across both the services and DoD as a whole, to identify strengths and gaps that should be addressed. Informed by this inventory and feedback from an expert panel, NDRI developed a set of recommended priorities for stigma reduction. These recommendations answered such questions as "Where are there gaps in stigma-reduction strategies?" "What stigma-reduction strategies seem particularly promising?" "Which of the current stigma-reduction strategies should be continued or enhanced?" and "Where is there duplication or overlap, or alternatively, conflicting messages among current strategies?" This report summarizes the findings of this assessment. The contents of this report will be of particular interest to policymakers in DoD, other command and line leadership, and mental health providers and other professionals.

This research was sponsored by the Office of the Assistant Secretary of Defense for Health Affairs and the Defense Centers of Excellence for Psychological Health and Traumatic Brain Injury and conducted within the Forces and Resources Policy Center of NDRI, a federally funded research and development center sponsored by the Office of the Secretary of Defense, the Joint Staff, the Unified Combatant Commands, the Navy, the Marine Corps, the defense agencies, and the defense Intelligence Community.

For more information on the RAND Forces and Resources Policy Center, see http://www.rand.org/nsrd/ndri/centers/frp.html or contact the director (contact information is provided on the web page).

Contents

Preface	iii
Figures	ix
Tables	xi
Summary	xiii
Acknowledgments	. xxv
Abbreviations	xxvii
CHAPTER ONE	
Introduction	1
Need for an Assessment of the U.S. Department of Defense Approach to Stigma	
Reduction	1
Purpose of This Research	2
Methods	2
Literature Review	2
Microsimulation Modeling of Costs	3
Interviews with Program Staff	3
Prospective Policy Analysis	4
Expert Panel	4
Organization of This Report	5
CHAPTER TWO	
Defining Stigma in the Military Context	7
Importance of a Clear Operational Definition and Conceptual Model of Stigma	7
Definition of Mental Health Stigma in the Military Context	8
Conceptual Model of Mental Health Stigma in the Military Context	8
Military Context Is Unique	11
Public, Institutional, and Social Contexts Are Interconnected but Not Well	
Understood	12
Related Terms Are Often Used Interchangeably but Are, in Fact, Distinct from Stigma.	13
Stereotype, Prejudice, and Discrimination	13

Barriers to Care	14
Stigma Reduction Is One Strategy Used to Promote Treatment-Seeking and Well-Bei	ng 14
CHAPTER THREE	
Prevalence of Mental Health Stigma in the Military	17
Prevalence of Stigma in the Military	
Joint Mental Health Advisory Team 7	
Marines Attending the Combat Operational Stress Control Program	20
Navy Quick Poll	20
2011 Department of Defense Health Related Behaviors Survey of Active Duty Military	
Personnel	
Hoge et al. Study of Barriers to Care Among Military Personnel Experiencing	
Combat Duty	22
Survey of Individuals Previously Deployed for Operation Enduring Freedom or	
Operation Iraqi Freedom	
Limitations of Military Measures of Stigma	23
Challenges Comparing the Prevalence of Stigma in the Military with That in the	
General U.S. Population	24
Conclusion	27
CHAPTER FOUR	
Societal Costs of Mental Health Stigma in the Military	
Literature Review to Estimate Stigma's Impact on Treatment-Seeking	
Regression Analyses to Estimate Stigma's Impact on Treatment-Seeking	
Exploratory Analyses Examining Other Barriers to Care	
Expert Panel to Vet Model Assumptions and Parameters	
Modeling Societal and Medical Costs	53
Conclusion	54
CHAPTER FIVE	
Promising Programmatic and Policy Approaches to Reducing Stigma	
Intervening in the Public and Social Contexts	55

Public and Social Context Interventions Tested in Civilian Populations	56
Contact with Individuals with Mental Illness	56
Training on How to Help Someone in Emotional Distress	
Multimedia Campaigns	
Public and Social Context Interventions Tested in Military Populations	
Intervening in the Institutional Context	60
Intervening with People with Mental Health Concerns	61
Summary of Possible Interventions	63
Public and Social Contexts	

Military Context	64
People with Mental Health Disorders	65
CHAPTER SIX	
U.S. Department of Defense Programs to Reduce Mental Health Stigma	67
The Primary Approach Is a Universal Culture Shift to Promote Mental Health and	
Treatment-Seeking	67
More-Indicated or Targeted Military Stigma-Reduction Programs Are Also Being	
Implemented	70
Identifying and Interviewing Representatives from Stigma-Reduction Programs	70
U.S. Department of Defense Programs and Policies Target the Military Treatment	
System	71
U.S. Department of Defense Programs Target Military Norms, Culture, and Social	
Context	72
U.S. Department of Defense Programs Target People with Mental Health Disorders	73
Most Stigma Programs Are Department-Wide	
Few Stigma-Reduction Programs Are Being Evaluated	
Suggestions to Augment the U.S. Department of Defense's Approach to Reducing	
Mental Health Stigma	75
Lessons Learned from Effective Civilian Programs May Help Improve the U.S.	
Department of Defense Programs Targeting Military Norms, Culture, and	
Social Context	75
Programs Targeting the Individual Context May Help Broaden the U.S.	
Department of Defense's Current Approach	76
Conclusion	
CHAPTER SEVEN	
U.S. Department of Defense Policies Related to Stigma	79
Approach to Prospective Policy Analysis	
Literature Review	
Systematic Assessment of Policies	
Key Findings from the Content Analysis of U.S. Department of Defense Policies	
Conclusion	
CHAPTER EIGHT	
Key Findings and Priorities for Improving the U.S. Department of Defense's	
Approach to Stigma Reduction	93
Key Findings from the Report	
Priorities to Improve Stigma-Reduction Interventions	
1. Explore Interventions That Directly Increase Treatment-Seeking	
2. Consider Evidence-Based Approaches to Empowering Service Members with	
Mental Health Concerns to Support Their Peers	95

3. I	Design New or Adapt Existing Intervention-Delivery Mechanisms to Minimize	
	Operational Barriers for Service Members Seeking Treatment	95
4.1	Embed Stigma-Reduction Interventions in Clinical Treatment	96
5. I	Implement and Evaluate Stigma-Reduction Programs That Target Service	
	Members Who Have Not Yet Developed Symptoms of a Mental Health	
	Disorder	96
Prior	ities to Improve Policies That Contribute to Stigma Reduction	97
1. I	Provide Better Guidance for Policies in Which a Mental Health Condition or	
	Treatment Prohibits Job Opportunities or Actions	97
2.1	Review the Stigmatizing Language Identified in Policies to Determine Whether	
	It Should Be Removed	97
3. (Offer Incentives for Positive Behaviors That Promote Mental Well-Being	98
Prior	ities to Improve Research and Evaluation Related to Stigma Reduction	98
1. (Continue to Improve and Evaluate the Modifications Made to Existing Programs	
	That Begin to Address Stigma and Other Barriers to Care	98
2.1	Examine the Dynamic Nature of Stigma and How It Interacts with Internal and	
	External Conditions over Time	99
3. I	Improve Measures of Prevalence to Improve Tracking of Stigma and Other	
	Barriers to Care	. 100
4.1	Review Classified Department-Wide and Service-Specific Policies to Determine	
	Potential Implications for Mental Health Stigma and Discrimination	
	arching Priority	. 101
1. (Convene a Task Force to Explore the Tensions Between a Command's Need to	
	Know a Service Member's Mental Health Status and Treatment History and	
	the Service Member's Need for Privacy	
Conc	clusion	. 103
	NDIXES Methods for Literature Review	105
	Definitions of <i>Mental Health Stigma</i>	
	Prevalence of Stigma in the General U.S. Population Detailed Methods for the Modeling Approach	
	Program Descriptions and Analysis	
	Policy-Analysis Methods	
	Policies with Implications for Stigma	
	Policies That Contain Negative Terminology with Implications for Stigma	
	Methods Used to Conduct the Expert Panel to Refine and Vet Priorities for	. 221
	Mental Health Stigma Reduction in the U.S. Department of Defense	225
Г	vientai i reatin Stigina Reduction in the 0.5. Department of Defense	. 44)
Bibli	iography	. 231
		_

Figures

S.1.	Conceptual Model of Stigma Reduction in the Military	xv
2.1.	Conceptual Model of Stigma Reduction in the Military	10
3.1.	Stigma Prevalence as Reported in the Mental Health Advisory Team	
	Surveys, 2003 Through 2010	19
5.1.	Interventions in the Public and Social Contexts That May Affect Stigma	59
5.2.	Interventions in the Institutional Context That May Affect Stigma	62
5.3.	Individual-Level Interventions That May Affect Stigma	63
6.1.	Approaches to Address Stigma, Promote Treatment-Seeking, and	
	Encourage Continued Engagement with Treatment Once Initiated	69
7.1.	Evidence-Informed Policy-Analysis Process	80
A.1.	Flow Chart for Literature Search	110
C.1.	Desired Social Distance from People with Schizophrenia or Depression	134
C.2.	Endorsement of Treatment-Seeking for People with Depression or	
	Schizophrenia	135
F.1.	Policy-Analysis Decision Tree	183

Tables

S.1.	U.S. Department of Defense Priorities for Enhancing and Refining Stigma-	
	Reduction Efforts	
1.1.	Methods Used to Answer Each Question	3
2.1.	Examples of Educational Approaches to Promote Treatment-Seeking and	
	Well-Being.	
3.1.	Factors That Affect Decision to Receive Mental Health Services	24
3.2.	Studies on Service Members Who Need Mental Health Treatment but Do	
	Not Seek It	
3.3.	Surveys Assessing Mental Health Stigma	26
4.1.	Studies Identified as Most Likely to Have Empirical Evidence of the	
	Association Between Stigma and Treatment-Seeking Behavior	
4.2.	Barrier-to-Care Items in the Invisible Wounds Survey	
4.3.	Total Aggregate Costs for Each Simulation	
4.4.	Aggregate Suicide Attempts and Deaths by Suicide for Each Simulation	
7.1.	Number of Policies That May Reduce Stigma	
7.2.	Number of Policies That May Contribute to Stigma	82
A.1.	Specific Searches	
A.2.	Web-Based Search-Result Tallies	
A.3.	Inclusion and Exclusion Criteria	. 109
A.4.	Data Abstraction Form	. 111
B.1.	Definitions of Mental Health Stigma	
B.2.	Definitions of Institutional Stigma and Structural Stigma	. 125
B.3.	Definitions of Individual Stigma and Self-Stigma	. 126
B.4.	Definitions of <i>Public Stigma</i>	
C.1.	Surveys Assessing Mental Health Stigma in the U.S. Population	. 133
D.1.	Population Groups in the Microsimulation Model	. 138
D.2.	Parameters Ranked According to Their Impact on Costs	. 142
D.3.	Stigma Regression Model	. 148
D.4.	Estimate of Yearly Mental Health Treatment Utilization After Removing	
	the Effect of Stigma	. 148
D.5.	Results of the Regression Model with Modification 1	. 148
D.6.	Results of the Regression Model with Modification 2	. 148

D.7.	Odds Ratio for the Barrier "My Family and Friends Would Be More	1/0
Da	Helpful Than a Mental Health Professional"	. 149
D.8.	Change in Treatment Utilization Based on a Reduction and Elimination	
	of the Barrier "My Family and Friends Would Be More Helpful Than a	150
Da	Mental Health Professional"	. 150
D.9.	Suicide Attempts and Deaths by Suicide When the Probability of	150
D 10	Treatment Is Zero.	
D.10.	Mean Cost Output When the Probability of Treatment Is Zero	
D.11.	Suicide Attempts and Deaths by Suicide at Baseline	
D.12.	Cost Output at Baseline	. 155
D.13.	Suicide Attempts and Deaths by Suicide When the Barrier of Friends and	
	Family Is Reduced by 50 Percent	. 156
D.14.	Cost Output When the Barrier of Friends and Family Is Reduced by	
	50 Percent	. 157
D.15.	Suicide Attempts and Deaths by Suicide When the Barrier of Friends and	
	Family Is Reduced by 100 Percent	. 158
D.16.	Cost Output When the Barrier of Friends and Family Is Reduced by	
	100 Percent	. 159
D.17.	Total Aggregate Costs for Each Simulation	. 159
D.18.	Aggregate Suicide Attempts and Deaths by Suicide for Each Simulation	. 160
E.1.	Programs Contacted and Decisions About Inclusion	. 167
E.2.	Afterdeployment.org	. 169
E.3.	Breaking the Stigma	. 170
E.4.	Embedded Behavioral Health	
E.5.	Military Pathways	. 173
E.6.	Real Warriors Campaign	. 175
F.1.	Summary of Dynamics Research Corporation Policy Search Results	
F.2.	Number of Policies That May Contribute to or Reduce Stigma	
G.1.	U.S. Department of Defense Policies That May Either Contribute to or	
	Reduce Stigma	. 187
G.2.	Army Policies That May Either Contribute to or Reduce Stigma	
G.3.	Navy and U.S. Navy Bureau of Medicine and Surgery Policies That May	
-	Either Contribute to or Reduce Stigma	. 201
G.4.	Marine Corps Policies That May Either Contribute to or Reduce Stigma	
G.5.	Air Force Policies That May Either Contribute to or Reduce Stigma	
G.6.	National Guard Policies That May Either Contribute to or Reduce Stigma	
G.7.	Coast Guard Policies That May Either Contribute to or Reduce Stigma	
H.1.	Policies That Contain Negative Terminology with Implications for Stigma	
I.1.	Expert Ratings of Stigma-Reduction Priorities	
		0

Despite the efforts of both the U.S. Department of Defense (DoD) and the Veterans Health Administration to enhance mental health services, many service members are not regularly seeking needed care when they have mental health symptoms or disorders. Without appropriate treatment, these mental health symptoms or disorders can have wide-ranging and negative impacts on the quality of life and the social, emotional, and cognitive functioning of affected service members.

The services have been actively engaged in developing policies, programs, and campaigns designed to reduce stigma and increase service members' help-seeking behavior. However, there has been no comprehensive assessment of these efforts' effectiveness and the extent to which they align with service members' needs or evidence-based practices. To help address this gap, the Office of the Assistant Secretary of Defense for Health Affairs and the Defense Centers of Excellence for Psychological Health and Traumatic Brain Injury asked the RAND National Defense Research Institute (NDRI) to inventory and assess stigma-reduction strategies both across the services and within DoD as a whole, to identify programmatic strengths, as well as gaps that should be addressed.

Purpose of This Report

The goal of this research was to assess DoD's approach to stigma reduction—how well it is working and how it might be improved. Our assessment focused on efforts that were active from January to June 2013. To accomplish this goal, we addressed the following research questions:

- 1. What does *mental health stigma* mean in the military context?
- 2. What is the prevalence of mental health stigma in the military, and what are its medical and societal costs?
- 3. What does the scientific evidence base show about the most-promising program and policy options for reducing stigma?

- 4. How well do DoD's programs and policies align with what the evidence base shows?
- 5. What priorities should DoD consider to enhance and refine stigma-reduction efforts?

To address these questions, RAND researchers used five complementary methods: (1) literature review, (2) a microsimulation modeling of costs, (3) interviews with program staff, (4) prospective policy analysis, and (5) an expert panel.

Findings

The findings from our study of mental health stigma in the military answer each of the above questions.

What Does Mental Health Stigma Mean in the Military Context?

To avoid the lack of conceptual clarity that accompanies literature without a clear definition of *mental health stigma*, our first objective was to develop a working definition of *stigma* and a conceptual model showing the factors that influence stigma and its possible outcomes. The purpose of the definition and conceptual model (see Figure S.1) was to guide the identification of promising intervention strategies and to provide a foundation for our assessment of DoD's current approach to reducing stigma. Though we acknowledge that there are many facets of stigma, for the purposes of this project, we define *mental health stigma* as a dynamic process by which a service member perceives or internalizes this brand or marked identity about himself or herself or people with mental health disorders (PWMHDs). This process happens through an interaction between a service member and the key contexts in which the service member resides.

The conceptual model operationalizes this definition of *stigma* by linking it to the key contexts that create stigma—the public context, institutional context, social context, and individual context—and the empirically and theoretically derived impacts of stigma. These include four immediate outcomes that we found to be empirically linked to stigma (coping mechanisms [e.g., hide, withdraw], interpersonal outcomes [e.g., self-esteem], attitudes toward treatment-seeking, and intentions to seek treatment) and four long-term outcomes that literature has theoretically linked to stigma (well-being, quality of life [e.g., productivity], treatment initiation, and treatment success). We were unable to empirically link these long-term outcomes directly to stigma. Despite popular opinion and a strong theoretical base that stigma deters treatment-seeking, we were unable to identify empirical literature to support this link. However, a variety of other factors (e.g., availability of providers, time off of work to seek care) may affect whether intentions to seek treatment translate into actual behavior.

Figure S.1 Conceptual Model of Stigma Reduction in the Military



What Is the Prevalence of Mental Health Stigma in the Military, and What Are Its Medical and Societal Costs?

We examined the historical prevalence of various stigma measures among deployed Operation Iraqi Freedom (OIF) soldiers screening positive for mental health symptoms or disorders using publicly available data contained in the Army's Mental Health Advisory Team (MHAT) reports. Collectively, these data indicate possible changes in stigma over time and differences across populations. Consistent with the literature was our finding that people seeking mental health treatment reported higher perceived levels of stigma. Although we were unable to definitely conclude that declines in stigma (particularly in the public, institutional, and social contexts that are assessed by the MHAT) resulted in increased treatment-seeking, further research could help determine the impact of the declines in stigma. Additionally, several limitations preclude drawing conclusive results from the stigma data. No single measure is being used to assess stigma. Most of the military measures assess stigma in the public, institutional, or social context and do not assess stigma within the individual context. These challenges, among others, pose a major limitation for the advancement of our understanding of stigma and strategies to address it, both within and outside the military context. Notably, we are unable to link military mental health stigma to changes in treatmentseeking behaviors, and any patterns and trends in stigma found in military populations are not directly comparable to U.S. prevalence estimates. Also, little is known about the extent to which stigma prevalence varies across different military populations (e.g., officers versus enlisted personnel).

These same measurement challenges also created difficulties in estimating the costs resulting from stigma. However, given the prevalence of mental health stigma in the military, we next developed a microsimulation model to estimate the costs resulting from stigma. These costs included treatment costs, costs of lost productivity, and suicide attempts and deaths by suicide. We based our model on an existing one that estimates the costs of untreated mental health symptoms or disorders among service members deployed as part of Operation Enduring Freedom (OEF) or OIF and developed by Kilmer et al. (2011) as part of the RAND Invisible Wounds Project (Tanielian and Jaycox, 2008). To adapt the model to include mental health stigma, we reviewed the literature, conducted a series of regression analyses, and convened an expert panel. Regression analyses of longitudinal data on stigma and treatment-seeking were run because we could not find a robust empirical literature directly linking to treatmentseeking and stigma (e.g., because stigma was defined in a variety of ways, because attitudes and intentions were measured rather than actual behaviors, because stigma was captured only in a public context and not an individual one). The regression analyses revealed that stigma did not predict initiation of treatment-seeking. When we input data from the regression analyses and the literature into our microsimulation models, we found that decreasing (or completely eliminating) stigma would not increase the number of service members seeking mental health treatment (i.e., did not significantly increase the probability that a service member would initiate treatment). The lack of data and the finding that decreasing stigma would not increase treatment-seeking were consistent with our review of empirical published literature and supported by an expert panel. One expert panelist stated that it was not a surprise that "changes in knowledge and attitudes [which are two of the key outcomes targeted by stigma-reduction programs] do not result in changes in behavior." The expert referenced a large body of research on prevention programs that suggest that explicit behavioral changes need to be specified, modeled, and practiced before behavior can be expected to change. There may have been reasons that we found no evidence that stigma decreases treatmentseeking behavior, which goes against theoretical underpinnings and popular opinion. First, measures have neither fully captured the contexts that affect stigma nor appropriately differentiated between stigma directed at PWMHDs and stigma directed at the act of mental health treatment-seeking. The public, institutional, and social contexts are more fully assessed, but, because they may be filtered by the individual context, these may have the most-distal impacts on stigma. Similarly, stigma may predict other variables that more directly influence behavior. Because stigma is so distal from the

outcome, it may be difficult to empirically link. Finally, stigma may be less directly linked to the decision to seek help than to treatment success.

To understand what some of the more-proximal factors are that affect treatmentseeking, we conducted some exploratory regression analyses of other logistical, institutional, and cultural barriers to care, as well as beliefs and preferences for treatment. Our analyses suggested that other barriers to care (not stigma) may influence treatment utilization—in particular, the perception that support from family and friends provides a more helpful alternative to professional mental health treatment. Reducing this barrier by 50 percent would increase treatment costs by just under \$3 million but would result in more than \$9 million in savings in lost productivity and aggregated costs. Cost savings more than doubles if we eliminate this barrier, falling by more than \$32 million from the baseline.

What Does the Scientific Evidence Base Show About the Most-Promising Program and Policy Options for Reducing Stigma?

Even though we were unable to find empirical evidence that stigma *directly* affects treatment-seeking (possibly due to limited and inconsistent measurement), the literature theorizes that stigma may *indirectly* affect treatment-seeking, affecting coping styles, attitudes and intentions toward help-seeking, and interpersonal outcomes, such as self-esteem. Given this finding, we identified the most-promising programmatic and policy approaches to reducing stigma. Within the military context, these involved educating key power groups and changing policy to reduce discriminatory behavior among individual service members and military leadership, who often set the climate within units and the military institution as a whole. Contact-based programs (i.e., exposing service members to a fellow service member in recovery from a mental health disorder [MHD]), education and training programs, and multimedia campaigns have been shown to reduce stigma within the public context. Cognitive techniques (e.g., psychoeducation, cognitive restructuring) to teach people strategies to better control or accept their thoughts, feelings, sensations, and memories have been shown to be effective at reducing stigma. We used this literature as a basis for comparison in our assessment of DoD programs and policies.

How Well Do U.S. Department of Defense Programs and Policies Align with What the Evidence Base Shows?

Despite the availability of a wide range of evidence-based treatments for MHDs, the proportion of service members who seek needed treatment remains low. In response, DoD and individual branches of service have made a concerted effort to promote treatment-seeking through specific programs to reduce stigma, as well as through a widespread culture shift, in which mental health is discussed in the context of readiness and resilience and in which help-seeking is redefined as a sign of strength. In addition, DoD is implementing a range of programs within the military and public contexts, including the following:

- efforts to make the act of seeking care less stigmatizing (e.g., by embedding behavioral health providers in brigade combat teams)
- educating service members and military leaders to improve their mental health literacy (e.g., through trainings and media campaigns, such as the Real Warriors Campaign)
- providing opportunities for these same groups to interact with service members in recovery from MHDs.

These approaches are generally aligned with the promising approaches described in the scientific literature and may have contributed, at least in part, to the declining trends in perceived stigma among service members. Current DoD stigma-reduction efforts primarily target the public context. Strategies targeting the individual context were focused on education and training. However, we were unable to identify any systematic intervention targeted at people who are in need of care but may or may not be seeking it (e.g., psychotherapeutic approaches to stigma reduction among PWMHDs). Few strategies targeted the military or institutional context. Most of these programs were not being evaluated.

We also worked with the Defense Centers of Excellence for Psychological Health and Traumatic Brain Injury (DCoE) to conduct an extensive search for and a systematic assessment of existing DoD policies to identify how these policies might be contributing to or reducing mental health stigma. DCoE conducted the search for policies using a search strategy developed in partnership with the study team, and the study team conducted the systematic policy assessment. Our assessment of the policy context found that ambiguities in policy language might contribute to concerns. Despite the presence of equal-opportunity policies, wide variability and ambiguity in policies that prohibit service members with MHDs from career opportunities might inadvertently create opportunities for discrimination. These policies do not define triggers for opportunity limitations (e.g., any history of mental illness, anyone engaged in treatment) and do not acknowledge a threshold of symptomology or continuum of recovery. Additionally, conflicting language and intentions of policy highlight key tensions between the privacy of service members seeking mental health treatment and the need for commanders to assess unit fitness. These tensions will need to be addressed if DoD is to be successful in encouraging treatment-seeking among more of its service members.

We also identified policies that support universal educational stigma-reduction programs but not more-targeted programs for those in mental health treatment. This is a key gap in programs, given the higher prevalence of stigma among service members in treatment than among those not in treatment. Finally, we identified three areas of policy that DoD should consider reviewing. First, some policies use negative terminology and reinforce stereotypes about PWMHDs. Revising this language may help to minimize the likelihood that service members would feel stigmatized as a result. Second, we identified policies that allow nonprofessionals to determine mental health fitness and that support the use of mandated mental health screening for specific individuals or groups. Although these practices may be important to protect unit fitness in the military context, these practices could put some service members at risk for stigma and discrimination. Third, mental health screening and evaluation programs may be used inappropriately, and careful consideration of the implementation of such programs is necessary to ensure that they promote positive, rather than negative, attitudes toward treatment-seeking.

What Priorities Should the U.S. Department of Defense Consider to Enhance and Refine Stigma-Reduction Efforts?

Given these strengths and gaps in DoD's current approach to stigma reduction, we developed a set of priorities for improving DoD's approach to stigma reduction.

			Priority	
Priority Category	Description	Within Category	By Importance	By Validity
Improving stigma-reduction interventions	Explore interventions that directly increase treatment-seeking. Focusing primarily on a single barrier to care, such as stigma, may obfuscate other potential interventions to promote help-seeking. There are many potential approaches to promote help-seeking along the continuum of stress that have proven effective. In addition to targeting stigma, DoD should explore other mechanisms for increasing treatment-seeking and reducing barriers to mental health care.	1	1	1
	Consider evidence-based approaches to empowering service members who have mental health concerns to support their peers. Expert panelists suggested that promoting the empowerment of PWMHDs to provide peer support for one another is an important approach that DoD should consider to reduce stigma. Military-affiliated panelists suggested that peer-support programs were already occurring to some degree throughout DoD; however, the evidence base behind these programs was limited.	2	2	3

Table S.1

U.S. Department of Defense Priorities for Enhancing and Refining Stigma-Reduction Efforts

Table S.1—Continued

		Priority		
Priority Category	Description	Within Category	By Importance	By Validity
	Design new or adapt existing intervention- delivery mechanisms to minimize operational barriers for service members seeking treatment. Expert panelists also discussed how service members' preference for self- management might be a key barrier to their accessing services. Improving the complement of alternative mechanisms for treatment delivery that create fewer operational impediments to service members could appeal to service members with a preference for self- management.	3	3	5
	Embed stigma-reduction interventions in clinical treatment. Because stigma is a potential clinical risk factor, it should be assessed during routine clinical examination and monitored throughout treatment so that it can be addressed as a part of a comprehensive treatment strategy.	4	10	10
	Implement and evaluate stigma-reduction programs that target service members who have not yet developed symptoms of mental illness. DoD should draw on the evidence base summarized in this report to identify programs that may translate effectively to the military context and adapt them for use within the military. These additional programs are intended to complement existing efforts to change the culture within the military to increase help-seeking behavior.	5	11	11
Improving policies that contribute to stigma reduction	Provide better guidance for policies in which an MHD or treatment prohibits job opportunities or actions. A large number of the policies we reviewed prohibited specific job opportunities or actions if a service member had an MHD or sought mental health treatment. For many of these policies, the language is unclear, stating only that a service member is prohibited if he or she has a mental health issue. It is imperative that DoD provide additional guidance that clarifies what is meant by having a mental health issue and that is more attentive to the continuum of mental health.	1	6	2

		Priority			
Priority Category	Description	Within Category	By Importance	By Validity	
	Review the stigmatizing language identified in policies to determine whether to remove it. In 12% of policies, we identified language that was pejorative and characterized MHDs and treatment in a negative light. Editing these policies to remove this stigmatizing language may help to reduce the likelihood that this language contributes to stigma and could improve the clarity of the policies.	2	8	7	
Improving research and evaluation related to stigma reduction	Continue to improve and evaluate the modifications made to existing programs that begin to address stigma and other barriers to care. DoD is already implementing modifications to existing initiatives that begin to address barriers to care and may contribute to a larger culture shift in the military. To ensure that these efforts are appropriately assessed for their effectiveness, DoD should improve evaluations of these programs to ensure that they assess behavioral impacts.	1	4	9	
	Examine the dynamic nature of stigma and how it interacts with internal and external conditions over time. Much of the stigma research focuses on schizophrenia or general mental health concerns, rather than PTSD, anxiety, or depression, the disorders that may be of most interest to DoD. More research to understand how stigma differs among these disorders and whether there are differential beliefs, attitudes, or knowledge about treatment efficacy for these disorders would help DoD better target stigma-reduction efforts. Additionally, because stigma is not static, more research on how stigma manifests based on level of mental health symptomology and individual interactions with various external conditions (e.g., family members, unit commanders) is needed to improve understanding of stigma's impact and identify the optimal intervention points, especially for interventions that require multiple boosters to maintain their effectiveness.	2	5	5	

Table S.1—Continued

Table S.1—Continued

		Priority			
Priority Category	Description	Within Category	By Importance	By Validity	
	Improve measures of prevalence to improve tracking of stigma and other barriers to care. Instituting common tracking measures would allow for research on the extent to which the institutional and public contexts affect stigma and how those effects may vary by demographics, such as rank, race, age, or gender. Understanding how stigma differentially affects specific populations, as well as identifying consistent effects across populations, will be important for developing interventions that are tailored to specific populations.	3	9	8	
	Review classified DoD and service-specific policies to determine potential implications for mental health stigma and discrimination. The priorities presented here are based on a review of policies that are accessible without clearance. A military-affiliated panelist recommended obtaining and reviewing classified policies to determine whether to develop additional policy priorities based on the implications of those policies for mental health stigma and discrimination.	4	12	14	
Overarching	Convene a task force to explore the tensions between a command's need to know a service member's mental health status and treatment history and the need for privacy. A task force of experts could play an important role in assessing what type of information mental health providers should and should not share with commanders and in developing clear communications and processes for these exceptions.	1	7	6	

NOTE: PTSD = posttraumatic stress disorder.

Experts rated the validity and importance of these priorities and agreed that the priorities shown in Table S.1 were critical for DoD. We considered a priority valid if adequate scientific evidence or professional consensus exists to support a link between the proposed priority and reducing stigma or improving service members' help-seeking. A priority was considered important if addressing or undertaking the priority has a critical influence on reducing stigma or improving service members' help-seeking and there are serious adverse consequences of not addressing or undertaking the priority. Appendix I lists the participating experts.

Conclusion

There is still much unknown about the influence that stigma has for PWMHDs on initiation of treatment, treatment success (e.g., retention), and, ultimately, their quality of life. These priorities represent a first step for where additional program and policy development and research and evaluation are needed to improve understanding of how best to get service members with mental illness the needed treatment as efficiently and effectively as possible. Additional research and evaluation are also needed to more fully understand barriers to care among service members and which of these barriers most affect treatment initiation, treatment success, and overall quality of life.

We gratefully acknowledge the assistance of our expert panelists:

- Thomas Britt of Clemson University College of Business and Behavioral Science
- Rebecca L. Collins of RAND
- Patrick Corrigan of the Lewis College of Human Sciences at the Illinois Institute of Technology
- Kristie Gore of RAND
- Charles W. Hoge of the U.S. Army (retired)
- Bernice A. Pescosolido of Indiana University and the Indiana Consortium for Mental Health Services Research
- John Roberts of the Wounded Warrior Project
- Nancy A. Skopp of Madigan Army Medical Center
- Tracy Stecker of the Dartmouth Psychiatric Research Center
- David L. Vogel of Iowa State University Department of Psychology
- Nathaniel G. Wade of Iowa State University
- Christopher Warner of the U.S. Army Medical Department Activity—Alaska.

We also thank Blair Smith of RAND for the administrative support she provided in preparing this document. In addition, we thank our project monitor at the Defense Centers of Excellence for Psychological Health and Traumatic Brain Injury, Lt. Monique Worrell, for her support of our work.

We benefited from the valuable insights we received from M. Audrey Burnam of RAND and from David Vogel. We addressed their constructive critiques as part of RAND's rigorous quality assurance process and improved the quality of this report. Finally, we thank Blair Smith and Jody Larkin of the RAND Library, both of whom provided important insight and support for this project.

5AFI	Fifth Air Force instruction
5R0X1C1	Air Force specialty code for chaplain assistant
ADRP	Army doctrine reference publication
AETC	Air Education and Training Command
AFDWSUP	Air Force District of Washington supplement
AFGM	Air Force guidance memorandum
AFGSCGM	Air Force Global Strike Command guidance memorandum
AFI	Air Force instruction
AFJI	Air Force joint instruction
AFMAN	Air Force manual
AFPAM	Air Force pamphlet
AFPD	Air Force policy directive
AFR	Air Force Reserve
AFRCI	Air Force Reserve Command instruction
AI	administrative instruction
AIEP	American Indian Alaskan Native Employment Program
ANG	Air National Guard
ANGDIR	Air National Guard directory
ANGI	Air National Guard instruction
AOF	airfield operations flight

AR	Army regulation
ARC	Air Reserve Command
ARNG	Army National Guard
ARNGUS	Army National Guard of the United States
ATTP	Army tactics, techniques, and procedures
BFV	Bradley Fighting Vehicle
BRFSS	Behavioral Risk Factor Surveillance System
BUMED	U.S. Navy Bureau of Medicine and Surgery
BUMEDINST	U.S. Navy Bureau of Medicine and Surgery instruction
BUMEDNOTE	U.S. Navy Bureau of Medicine and Surgery note
CalMHSA	California Mental Health Services Authority
CFETP	career-field education and training plan
CGFAP	Coast Guard Family Advocacy Program
CI	confidence interval
CLFS	civilian labor-force status
CNGBI	Chief, National Guard Bureau instruction
CNO	chief of naval operations
COMDTINST	commandant instruction
CONUS	continental United States
COSC	Combat Operational Stress Control
СҮР	Child and Youth Program
DA	Department of the Army
DCoE	Defense Centers of Excellence for Psychological Health and Traumatic Brain Injury
DEOC	Defense Equal Opportunity Council
DLAR	Defense Logistics Agency regulation
DoD	U.S. Department of Defense

DoDEA	Department of Defense Education Activity
DoDI	Department of Defense instruction
DoDM	Department of Defense manual
DON	Department of the Navy
DRC	Dynamics Research Corporation
DTM	directive-type memorandum
EA	executive agent
EAD	extended active duty
EBH	Embedded Behavioral Health
EBT	evidence-based treatment
EEO	equal employment opportunity
EEOC	Equal Employment Opportunity Commission
EO	executive order
ESSP	Expeditionary Site Survey Process
FAP	Family Advocacy and General Counseling Programs
FM	field manual
FY	fiscal year
GSS	General Social Survey
HBM	health belief model
HHQ	higher headquarters
HIPAA	Health Insurance Portability and Accountability Act
HQMC	Headquarters, U.S. Marine Corps
IACP	International Association of Chiefs of Police
IAW	in accordance with
ICV	infantry carrier vehicle
IG	inspector general
IHA	indirect hire agreement

IMHS	Integrated Mental Health Strategy
INCIRLIKABSUP	Incirlik Air Base Supplement
ING	Inactive Army National Guard
IOOV	In Our Own Voice
J-MHAT 7	Joint Mental Health Advisory Team 7
LL	lower limit
L&O	law and order
LPSP	Limited Privilege Suicide Prevention
MAJCOM	major command
МСО	Marine Corps order
MDD	major depressive disorder
MEDCOM	U.S. Army Medical Command
MEO	Military Equal Opportunity
MHAT	Mental Health Advisory Team
MHD	mental health disorder
MHFA	Mental Health First Aid
Mil. R. Evid.	Military Rule of Evidence
MKTS	Military Knowledge and Training System
MP	military police
MTF	medical treatment facility
NAF	nonappropriated fund
NASD	National Alcohol Screening Day
NATO	North Atlantic Treaty Organization
NAVMC	Navy and Marine Corps form
NAVMED	Naval Medicine
NCS	National Comorbidity Survey
NCS-R	National Comorbidity Survey Replication

NDAA	National Defense Authorization Act
NDRI	RAND National Defense Research Institute
NDSD	National Depression Screening Day
NEC	Navy enlisted classification
NGR	National Guard regulation
NSDUH	National Survey on Drug Use and Health
OCS	operational contract support
OEF	Operation Enduring Freedom
OEH	occupational and environmental health
OIF	Operation Iraqi Freedom
OPM	Office of Personnel Management
OPNAVINST	Office of the Chief of Naval Operations instruction
OR	odds ratio
OSCAR	Operational Stress Control and Readiness
PCS	permanent change of station
PES	Performance Evaluation System
PILOTS	Published International Literature on Traumatic Stress
PRP	Personnel Reliability Program
PSP	Personnel Security Programs
PTSD	posttraumatic stress disorder
PWMHD	person with a mental health disorder
RP	religious program specialist
SAAC	Sexual Assault Advisory Council
SAMHSA	Substance Abuse and Mental Health Services Administration
SAPR	Sexual Assault Prevention and Response
SBCT	Stryker brigade combat team
SD	standard deviation

SECNAVINST	Sometomy of the Nervy instruction
	Secretary of the Navy instruction
SEP	special-emphasis program
SF	security force
SOF	special operations forces
SOP	standard operating procedure
SOS	Signs of Suicide
STANAG	standardization agreement
T2	National Center for Telehealth and Technology
TBI	traumatic brain injury
TDRL	temporary disability retired list
Tprod	total productivity lost
TRiM	Trauma Risk Management
TSR	Traumatic Stress Response
Tsui	total cost of a suicide
TsuiV	total cost of suicide and value of life lost
Ttreat	total treatment cost
Tx	treatment
UC	usual care
UCLA	University of California, Los Angeles
UCMJ	Uniform Code of Military Justice
UL	upper limit
USAFESUP	U.S. Air Forces in Europe supplement
USASOC	U.S. Army Special Operations Command
VA	U.S. Department of Veterans Affairs

Need for an Assessment of the U.S. Department of Defense Approach to Stigma Reduction

Despite the efforts of both the U.S. Department of Defense (DoD) and the Veterans Health Administration to enhance mental health services, many service members are not regularly seeking needed care when they have mental health symptoms or disorders. Without appropriate treatment, these mental health symptoms or disorders can have wide-ranging and negative impacts on the quality of life and the social, emotional, and cognitive functioning of affected service members.

Both the DoD Task Force on Mental Health (2007) and the DoD Task Force on the Prevention of Suicide Among Members of the Armed Forces (2010) identified the stigma of mental illness as a significant issue preventing service members from seeking help for mental health symptoms or disorders. They defined the stigma of mental illness as the negative attitudes and beliefs about or associated with people labeled as mentally ill. The 2010 survey from the Army Office of the Surgeon General's Mental Health Advisory Team (Office of the Surgeon General, 2011) found that, although the prevalence of stigma among service members is decreasing, it remains high and is actually highest among people who screen positive for mental health symptoms or disorders. These advisory bodies all concluded that addressing the stigma of mental illness is critical to ensuring that service members seek needed mental health care, both to facilitate problem resolution and to prevent more-serious negative outcomes.

The services have been actively engaged in developing policies, programs, and campaigns designed to reduce stigma and increase service members' help-seeking behavior. However, there has been no comprehensive assessment of these efforts' effectiveness and the extent to which they align with service members' needs or evidence-based practices. To help address this gap, the Office of the Assistant Secretary of Defense for Health Affairs and the Defense Centers of Excellence for Psychological Health and Traumatic Brain Injury asked the RAND National Defense Research Institute (NDRI) to inventory and assess stigma-reduction strategies both across the services and within DoD as a whole, to identify strengths, as well as gaps in the strategies that should be addressed.

Purpose of This Research

The goal of this research was to assess DoD's approach to stigma reduction—how well it is working and how it might be improved. Our assessment focused on efforts that were active from January to June 2013. To accomplish this goal, we addressed the following research questions:

- 1. What does *mental health stigma* mean in the military context?
- 2. What is the prevalence of mental health stigma in the military, and what are its medical and societal costs?
- 3. What does the scientific evidence base show about the most-promising program and policy options for reducing stigma?
- 4. How well do DoD's programs and policies align with what the evidence base shows?
- 5. What priorities should DoD consider to enhance and refine stigma-reduction efforts?

Methods

To address these questions, NDRI used five complementary methods: (1) literature review, (2) microsimulation modeling of costs, (3) interviews with program staff, (4) prospective policy analysis, and (5) an expert panel. Table 1.1 shows which methods we used to address which specific aim (with the aims corresponding to the research questions above). Each method is then described briefly below and in more detail in the report appendixes.

Literature Review

We conducted a systematic literature review of theoretical works on stigma and prior studies of stigma-reduction programs. We began by reviewing the resources used in two earlier literature reviews that were related to our own study: RAND's stigma-reduction work for the California Mental Health Services Authority (Collins et al., 2012) and the Defense Centers of Excellence for Psychological Health and Traumatic Brain Injury's (DCoE's) report "Behavioral Health Stigma and Access to Care" (DCoE, 2012). We then supplemented the literature from these works by performing our own web-based search of peer-reviewed literature in content-relevant databases. These sources underwent successive rounds of screening, including a title and abstract review followed by a full-text review, to exclude irrelevant and unsuitable articles. We then coded and reviewed articles selected for inclusion, and we abstracted details relating to the focus of our study.

	Method				
- Research Question	Literature Review	Program Interview	Policy Analysis	Microsimulation	Expert Panel
1. What does <i>mental health stigma</i> mean in the military context?	х				х
2. What is the prevalence of mental health stigma in the military, and what are its medical and societal costs?	x			x	
3. What does the scientific evidence base show about the most-promising program and policy options for reducing stigma?	х				х
4. How well do DoD's programs and policies align with what the evidence base shows?		x	х		х
5. What priorities should DoD consider to enhance and refine stigma-reduction efforts?	х	х	х	х	х

Table 1.1 Methods Used to Answer Each Question

Microsimulation Modeling of Costs

To calculate the costs associated with mental health stigma, we constructed a microsimulation model that assesses the societal costs within the United States of service members who do not seek appropriate mental health care because of public or selfstigma. These costs include treatment and rehabilitation costs for service members with posttraumatic stress disorder (PTSD) and depression, medical costs associated with suicide attempts and completions, value of lives lost to suicide, and lost productivity stemming from PTSD and depression. The model takes a representative cohort of service personnel and models their life courses over two years, taking into account probabilistic events that may occur as a result of a mental health disorder (MHD). We used 16 parameters to determine overall societal costs.

Interviews with Program Staff

Identifying programs funded by DoD that focus on stigma reduction was not a straightforward task. As a result, we used a multifaceted approach to identifying programs whose staff we sought to interview for this report. Our general approach was to identify as many potential programs as possible in order to ensure that we had not omitted any and to apply the exclusion criteria only after we had adequate information about each potential program, generally obtained through an interview with a program representative. The methods we used to identify programs were web and other media
searching; scanning DoD program materials in the public domain (e.g., websites, brochures); reviewing relevant documents in the public domain; consulting with military personnel; obtaining DoD lists of programs; consulting with topic-area experts within and outside of RAND; and snowball sampling. We identified 26 potential programs, five of which met our inclusion criteria. For those meeting inclusion criteria, we then used information from the interviews to categorize these programs activities into those that address stigma in the institutional context, those that address stigma in the public context, and those that support people with mental health concerns at the individual level.¹ A program may contribute to stigma reduction in more than one context.

Prospective Policy Analysis

We identified and analyzed policies in DoD that could either reduce or increase the stigmatization of those who experience mental health symptoms or disorders or access mental health care. This process involved three steps. First, DCoE supplied a list of 3,558 policies that could potentially have implications for mental health stigma or discrimination. To generate that list, DCoE conducted a prospective policy search using terms related to mental health (e.g., *suicide-, mental, psych-, emotion, counseling*) and to stigma (e.g., *stigma, access, barriers, help*) and then removed any counseling policies not directly related to mental health care (e.g., financial counseling). Then, we created a decision tree and used it to systematically determine whether a policy was likely to reduce stigma and discrimination, contribute to it, or not have implications. We identified 323 policies as relevant to mental health stigma (9 percent of the 3,558 identified). We then analyzed the content of these 323 policies deemed relevant to stigma and discrimination and summarized potential implications for DoD.

Expert Panel

We utilized a modified version of the RAND/University of California, Los Angeles (UCLA) Appropriateness Method (Fitch et al., 2001) to develop a list of recommended priorities. First, we convened a panel consisting of 12 experts in two key areas:

- mental health stigma
- mental health in the military (PTSD, deployment psychology).

We sent experts a summary of the findings from the report and a set of proposed priorities based on these findings. We presented proposed priorities as affirmative statements about what DoD should do to reduce mental health stigma among service members. We presented a short rationale for each priority and possible short- and long-term steps to achieve that priority. During the meeting, panelists discussed the strengths and

¹ Here we mean people with mental health disorders (PWMHDs) but also those experiencing clinical and subclinical symptoms.

weaknesses of each proposed priority, as well as brainstormed on any priorities that were missing from the list. Panelists then rated each proposed priority on its validity and its importance to DoD.

Organization of This Report

Chapter Two presents our definition of *mental health stigma* in the military context, summarizing key findings about how stigma is defined in the literature (research question 1). Chapter Three describes the prevalence of stigma in the general population, as well as the military (research question 2). Chapter Four presents estimates of the medical and societal costs associated with stigma, as well as with other several common barriers to mental health care (research question 2). Chapter Five describes the most-promising program and policy options for reducing stigma (research question 3). Chapters Six and Seven present findings from our assessment of the alignment between DoD programs and policies, respectively, and the current scientific evidence base (research question 4). Recommendations of our assessment are presented in Chapter Eight (research question 5). We also provide eight appendixes: our methods for literature review, definitions of mental health stigma, prevalence of stigma in the general population, the details of our methods for our modeling approach, program descriptions and analysis, our methods for policy analysis, policies with implications for stigma, policies with negative terminology with implications for stigma, and methods used with our expert panel.

This chapter describes the process that we used to derive a definition and conceptual model of mental health stigma in the military context and summarizes key theoretical approaches to understanding stigma. We derived these through the literature review and consultation with our expert panel. The purpose of the definition and conceptual model is to guide the identification of promising intervention strategies and to provide a foundation for our assessment of DoD's current approach to reducing stigma.

Importance of a Clear Operational Definition and Conceptual Model of Stigma

The variations in definition are very important because the way in which *stigma* is defined dictates the types of stigma-reduction strategies that are pursued. For example, if *stigma* is defined as occurring within the public context, is based on the perception that someone is seeking treatment for an MHD, and reduces treatment-seeking, stigma-reduction strategies might focus on improving attitudes of units, peers, and family members toward mental health treatment. However, if *stigma* is defined as occurring in the individual context because someone fears discrimination if he or she is diagnosed with an MHD, stigma-reduction strategies may focus on educating people at risk for MHDs about privacy policies associated with being diagnosed with MHDs.

More than half of the articles we reviewed (55 percent) did not define *stigma*. Many others used imprecise definitions characterizing stigma by the context in which it occurs (e.g., institutional, public); the impacts of stigma (e.g., discrimination against people with mental health disorders [PWMHDs], decrease in treatment-seeking); and the identifying characteristics associated with stigma (e.g., diagnosis of an MHD, act of treatment-seeking). In total, we identified 98 distinct definitions of *stigma* (see Appendix B). This lack of conceptual clarity makes it difficult to understand what construct was actually being measured or discussed and consequently makes it difficult to identify how best to intervene to reduce stigma.

To avoid the lack of conceptual clarity that accompanies literature without a clear definition of *mental health stigma*, our first objective was to develop a working defini-

tion of *stigma* and a conceptual model showing the factors that influence stigma and its possible outcomes. The purpose of the definition and conceptual model was to guide the identification of promising intervention strategies and to provide a foundation for our assessment of DoD's current approach to reducing stigma. Although our focus was on mental health stigma in a military setting, we considered a broad range of definitions spanning beyond the military-specific or even mental health–specific context in order to develop our own definition.

Definition of Mental Health Stigma in the Military Context

The literal definition of *stigma* is a "brand" or "mark of infamy" (Sadow and Ryder, 2008) associated with a specific subgroup or identity. This "marked identity" (J. Phelan and Link, 2011) indicates that one is outside of what is normal or acceptable, which, in turn, allows for a differentiation process. This differentiation or labeling process is then used to separate or isolate people with the undesirable characteristic (i.e., "us' from 'them'" [Link and Phelan, 2001]). Through this process, a group (e.g., the general population) does more than simply identify these people as different; members of the group consider such people inferior, and this differentiation may result in discrediting or a loss of status, which "reduces the bearer from a whole and usual person to a tainted, discounted one" (Goffman, 1963).

Though we acknowledge that there are many facets of stigma, for the purposes of this project, we define *mental health stigma* as a dynamic process by which a service member perceives or internalizes this brand or marked identity about himself or herself or PWMHDs. This process happens through an interaction between a service member and the key contexts in which the service member resides. According to Major and O'Brien (2005), stigma is "relationship- and context-specific. . . . [I]t does not reside in the person but rather within a specific social context" (Dalky, 2012, p. 4).

Conceptual Model of Mental Health Stigma in the Military Context

The conceptual model operationalizes this definition of *stigma* by linking it to the key contexts that create stigma—the public context, institutional context, social context, and individual context—and the empirically and theoretically derived impacts of stigma. These include four immediate outcomes we found to be empirically linked to stigma (coping mechanisms [e.g., hide, withdraw], interpersonal outcomes [e.g., self-esteem], attitudes toward treatment-seeking, and intentions to seek treatment) and four long-term outcomes that literature has theoretically linked to stigma (well-being, quality of life [e.g., productivity], treatment initiation, and treatment success). We were unable to empirically link these long-term outcomes directly to stigma. Despite popu-

lar opinion and a strong theoretical base that stigma deters treatment-seeking, we were unable to identify empirical literature to support this link. However, a variety of other factors (e.g., availability of providers, time off of work to seek care) may affect whether intentions to seek treatment translate into actual behavior.

Many sources delineated three key contexts that promote stigma: institutional, public, and self. According to the literature, *institutional stigma* has been defined as arising from the "policies of private and governmental institutions" that either intentionally or unintentionally "restrict opportunities [and] hinder the options of people with mental illness" (Corrigan and O'Shaughnessy, 2007, p. 90). Public stigma reflects the knowledge, attitudes, and beliefs about PWMHDs or about mental health treatment and the prejudicial and discriminatory behaviors described above, coming from friends, family, co-workers, the public, and even health care providers. *Self-stigma* was most commonly defined as the "internalization of public stigma" and was often associated with a "loss of self-esteem and self-efficacy" (Corrigan, Thompson, et al., 2003). We believe, however, that the common conceptualizations of institutional and public stigma can arise. This is a subtle but very important distinction because stigma is not a characteristic or object that one has or gives; it is a *process* by which someone perceives or internalizes interactions with specific people in specific contexts.

Figure 2.1 (adapted from Bronfenbrenner, 1979) shows an ecological model of stigma, with the person (service member or not) at the center. The service member is surrounded directly by

- the social context, made up of key relationships someone has with others (e.g., family, friends, unit members, command leadership)
- the institutional context, which is made up of the broader policies and systems within which someone operates
- the public context, which represents the military norms and culture in which the person operates.

Together, the institutional and public contexts make up the broader military context. It is important to note that there are also national norms and an economic and cultural environment of the broader public outside of the military that may affect perceptions of stigma; however, for the purposes of this report, we focus on the contexts specified in the conceptual model.

Within the public and social contexts, several major factors may produce stigma, including the knowledge, attitudes, and beliefs that service members hold about PWMHDs and about seeking mental health treatment. These negative knowledge, attitudes, and beliefs about PWMHDs can translate into discriminatory behavior toward PWMHDs, including withholding help from PWMHDs, avoiding them entirely, segregating PWMHDs from those without, and forcing treatment or criminal



Figure 2.1 Conceptual Model of Stigma Reduction in the Military

justice actions on PWMHDs. However, there are limited data on the military service members' knowledge, attitudes, and beliefs toward treatment; most of the literature in this area measures perceived stigma toward *people seeking treatment* rather than stigma toward the *treatment itself*. Few studies speak directly to the extent of discriminatory behavior directed toward PWMHDs in a military context.

The majority of literature examining contextual factors at the institutional level focuses on *discrimination*. Institutional discrimination toward PWMHDs manifests in a range of policies and practices. Some policies or practices may *contribute to* discrimination toward PWMHDs, while others may be in place to *prevent or lessen* discrimination. Some research suggests that the military's strict emphasis on fulfilling one's military duties may be a factor leading to stigma and discrimination toward PWMHDs (Gibbs et al., 2011; Barrett, 2011). However, the laws and policies concerning psychological disability and treatment confidentiality in the military are quite different from those in civilian environments, and research has only begun to touch on the range of policies that could contribute to the stigmatization of or discrimination against service members with mental illness.

Recent research suggests that stigma at the individual level almost fully mediates the relationship between negative attitudes, beliefs, behaviors, and policies of the public and attitudes toward and intentions to seek treatment by PWMHDs (Ludwikowski, Vogel, and Armstrong, 2009; Vogel, Shechtman, and Wade, 2010; Vogel, Wade, and Hackler, 2007; Ægisdóttir et al., 2011; Wade and Hackler, 2008). Consistent with our definition of stigma is the finding that such stigmatization at the individual level occurs when PWMHDs internalize the negative attitudes and turn stereotypes about mental illness toward themselves (K. Fung, Tsang, and Cheung, 2011; Corrigan and Watson, 2002; K. Fung, Tsang, Corrigan, et al., 2007; Livingston and Boyd, 2010; Corrigan and Rao, 2012; Corrigan, 2005). As illustrated in our conceptual model (Figure 2.1), this process is often proven or theorized to result in some negative consequences, including lower self-esteem, reduced treatment-seeking, and poor adherence to treatment (Corrigan, 2005; Vogel, Wade, and Haake, 2006; Shechtman, Vogel, and Maman, 2010; Vogel, Shechtman, and Wade, 2010; Barney, Griffiths, Jorm, et al., 2006; Evans-Lacko, Brohan, et al., 2012). For example, there is significant evidence that experiencing and internalizing negative attitudes, beliefs, behaviors, and policies directed toward PWMHDs can have negative consequences for interpersonal outcomes, and some evidence that stigma may influence this population's long-term quality-of-life outcomes, including whether treatment is effective. However, there is not compelling evidence that these same internalizations negatively affect whether service members initiate treatment-seeking, despite the fact that such a relationship is often theorized.

One theoretical approach to help explain these associations put forth by Bruce Link and his colleagues is the modified labeling theory, which emphasizes the interactions between people and their environments (Link, Cullen, Frank, et al., 1987; Link, Cullen, Struening, et al., 1989). According to this theory, negative conceptualizations of mental illness in the form of devaluation of PWMHDs develop early in life. When someone is officially labeled as having a mental illness, the societal meaning associated with a mental illness label becomes personally relevant (Kondrat and Early, 2011). As a result, labeled people come to believe that they personally will be rejected by members of society because of their labeled status and its associated meaning (Kondrat and Early, 2011).

It is important to also note that defining *stigma* as a process helps to emphasize the dynamic nature of stigma. Stigma is not a static concept that is either present or absent but a complex process that can change day-to-day and minute-to-minute based on changes in the relationships and context. The arrows in Figure 2.1 signify the dynamic interaction between the individual and the public and military contexts.

Military Context Is Unique

The military context has some unique aspects that we should also consider when defining *mental health stigma* in the military. First, the military screens service members for MHDs upon entry. Therefore, the incidence of some MHDs, such as schizophrenia and bipolar disorder, are not as common among military populations as they are among civilian populations, whereas other MHDs, such as PTSD, are of great concern. Unfortunately, much of the scientific literature has focused on stigma associated with schizophrenia and depression, with little focusing on those PTSD, a primary concern in the military. Second, service members' home lives and work lives are less separate than for their counterparts in the civilian sector. Service members' health insurance provides them access to military service providers, which could help contribute to the perception that leadership will find out if a service member has an MHD. Third, the demographics of the military are unique. The military has a high number of young men. Research has shown that men perceive greater stigma associated with seeking help than women do (Vogel, Wade, and Hackler, 2007), seeking care as a last resort (Angermeyer, Matschinger, and Riedel-Heller, 1999) because they are expected to be stoic, controlled, and self-sufficient (Hammen and Peters, 1978). Finally, the norms and values of unit culture (e.g., shared mission, leave no soldier behind) are a part of the public context that is unique to the military. In the military context, then, we consider mental health stigma to be

- the experiences of service members in response to military institutional factors (i.e., DoD and service-specific policies) leading to discriminatory treatment
- the attitudes, beliefs, and behaviors of others (e.g., family, spouse, unit members) toward service members exhibiting symptoms of or diagnosed with MHDs or who seek treatment.

Again, we refer to *experiences* in a broad sense, encompassing the knowledge, attitudes, beliefs, feelings, intentions, and behaviors of those who are stigmatized.

Of particular importance to the military is the theoretical link suggesting that stigma may affect treatment-seeking and, ultimately, mental health recovery. Recent task force recommendations calling for stigma reduction in the military identify the need to reduce stigma because it serves as a key barrier to help-seeking among service members in need of mental health treatment. These task force recommendations suggest that stigma-reduction efforts are a primary strategy for DoD to increase helpseeking of service members. Improving service members' help-seeking has the longerterm goal of promoting quality of life and well-being among service members and ensuring a mission-ready force.

Public, Institutional, and Social Contexts Are Interconnected but Not Well Understood

Some models of stigma, such as the one in Figure 2.1, touch on the intersecting nature of public, institutional, and social contexts with the individual experience (e.g., Link and Phelan, 2001; Corrigan, Markowitz, and Watson, 2004; Corrigan, 2004b;

Pescosolido, 2008). However, the extent to which public, institutional, and social factors affect each other and the resulting experience of stigma by the PWMHD is not known. Researchers theorize that the relationship between institutional and public factors related to stigma is bidirectional (e.g., Link and Phelan, 2001; Corrigan, Markowitz, and Watson, 2004). Institutional factors leading to discrimination likely shape public perceptions—for example, if policies and practices within the health care system frame a specific MHD as "legitimate," then it may improve the way those exhibiting symptoms of the disorder are viewed in society (Pescosolido, 2008). Conversely, members of the general public may hold stigmatizing perceptions of PWMHDs that could influence institutional policies and procedures. For example, people holding stigmatizing views of PWMHDs may vote for policies that result in discrimination, or they may hold positions of influence that allow them to shape the policies of an institution. Public factors influence stigma in the individual context because stigmatizing normative attitudes and beliefs can result in the internalization of such attitudes and beliefs by those who are in need of help (Essler, Arthur, and Stickley, 2006; Perry, 2011). It is also possible that those who have internalized negative attitudes and beliefs about PWMHDs would then go on to perpetuate the normative stigmatizing attitudes and beliefs. To our knowledge, however, little research explores the potential bidirectional relationship between public factors and individual stigma experience or institutional policies and individual stigma experiences (i.e., the arrows in Figure 2.1), suggesting that these areas are in need of future research.

Related Terms Are Often Used Interchangeably but Are, in Fact, Distinct from *Stigma*

One of the key challenges we faced, when reviewing the stigma literature, was that many authors used key terms related to stigma interchangeably with *stigma*. This suggests that *stigma* is equivalent to these other concepts and dilutes our ability to conceptually understand how stigma is related. In this section, we briefly summarize how stigma is conceptually linked to stereotypes, prejudice, discrimination, and barriers to care. The purpose of this section is to illuminate how these concepts relate to one another to prevent them from their labels being inappropriately used interchangeably in future DoD literature on stigma.

Stereotype, Prejudice, and Discrimination

Across the literature, the term *stigma* was commonly used to describe people's attitudes and behaviors with respect to stereotypes, prejudice, and discrimination. Stereotypes are "beliefs about a stigmatized group," while *prejudice* is defined as the "agreement with stereotypes leading to emotional responses." Discrimination is the "behavioral result of prejudice" (Corrigan, Powell, and Rüsch, 2012, p. 381). The process of moving from holding stereotypes to having an emotional response to them to acting on those responses turns thoughts and beliefs into feelings and attitudes that lead to actions. Although stigma is undoubtedly linked to this process, stereotypes, prejudice, and discrimination are not components of stigma; they are the results of it. "Stigmatizing attitudes . . . can lead to negative feelings, stereotyping and discriminatory behaviors" (Reavley and Jorm, 2011b, p. 1083). This distinction is an important one.

Barriers to Care

Although we can use the term *stigma* in a variety of contexts (e.g., physical disability, drug addiction, religious affiliation), we were interested specifically in understanding stigma related to mental health. Several definitions specific to mental health stigma defined *stigma* as a barrier to care. For example, one article defined *stigma* as "a barrier that discourages individuals and their families from seeking help" (Rae Olmsted et al., 2011). Though stigma may affect treatment-seeking behaviors, these behaviors should not be included in the definition of *stigma*; rather, they are better conceptualized as an outcome of stigma. Reducing the scope of mental health stigma to attitudes or behaviors *that impede treatment-seeking* disregards other important, stigma-like processes that may be equally harmful. Additionally, predefining *stigma* as a barrier to care minimizes the importance of studying the strength of the relationship between stigma and treatment-seeking because it presupposes that such a relationship exists.

Stigma Reduction Is One Strategy Used to Promote Treatment-Seeking and Well-Being

Stigma reduction is but one of the approaches to promoting treatment-seeking and well-being among service members. Focusing on stigma only, without considering the multitude of approaches to promoting treatment-seeking and well-being, is problematic for the following reasons. First, some experts have postulated that talking about stigma actually creates more stigma. That is, to claim that there is stigma associated with mental illness is to establish and perpetuate an association between mental illness and the negative connotations of stigma. In searching for literature related to this argument, however, we were unable to find any empirical data that supported the idea that talking about stigma worsens or perpetuates it. Some leaders in mental health, however, did assert that stigma—either the concept or the word itself—was problematic.

Second, *stigma* is often used to cover a broad range of conditions and qualities. This can be problematic, however, in that the distinct components of *stigma* are rendered indistinguishable, making discourse on specific aspects of stigma and their relation to one another ambiguous (Sayce, 1998). In much of the literature we reviewed, *stigma* was only implicitly defined; however, it is frequently an explicit focus of studies and policy (Manzo, 2004).

Third, *stigma* has an individualistic focus. Most definitions of *stigma* focus either on the stigmatized person (e.g., as being marked or as feeling shame) or on "microlevel interpersonal interactions" (the way someone thinks about or acts toward another person). This fails to capture patterns of social and economic exclusion at the institutional level. Stigma, therefore, is not as effective of a concept when trying to collectively address organizational or societal prejudice as discrimination or oppression is (Oliver, 1990; Sayce, 1998).

Finally, *stigma* focuses on the recipient rather than the promoter of discrimination. Saying that a certain group carries a stigma or is stigmatized suggests that there is something wrong with the members of the group rather than with those who perpetuate prejudice and discrimination (Chamberlin, 1997). By attaching stigma to PWMHDs, we are "colluding with unfairness" by putting expectations for change on the PWMHDs and letting the rest of society "off the hook" (Sayce, 1998). Furthermore, in attributing stigma to PWMHDs, unfairness is often described as "felt" or "perceived," which belittles the presence of actual disparities in treatment. For these reasons, many critics advocate using the word *discrimination* rather than *stigma*.

Because of the multitude of factors that affect treatment-seeking and well-being, it may be important for DoD to use several complementary approaches to encourage treatment-seeking behaviors and retention in care. Additionally, by using a multitude of strategies, DoD can minimize criticisms of any one approach. Each approach should be tied to different *barriers* that discourage service members from obtaining and remaining in care or *facilitators* that encourage service members to obtain and remain in care. Table 2.1 presents six different educational approaches to promoting help-seeking, describes the stigma-reduction strategies and relevant targets for change, and provides some sample media campaign messages to help distinguish among these approaches.

Approach	Intervention Strategy	Target for Change	Sample Media Campaign Message
Reduce stigma	Expose service members (1) to messages that combat myths and discrimination related to mental health messages of recovery and (2) to messages from others who are in recovery from MHDs.	Negative attitudes or beliefs about PWMHDs; discrimination; social isolation	 (1) It is time to talk; it is time to change; let's end mental health discrimination (Time to Change, undated). Mental health symptoms or disorders do not discriminate; people do (Amnesty International, undated). Teenage depression was devastating; childish reactions made it worse (Amnesty International, undated). (2) I am glad I failed at suicide because my life is so amazing now (Safe Schools Coalition, 2013) (typically individual stories or features).
Change military norms	Promote the belief that seeking help is a sign of strength.	Norms encouraging self-reliance, emotional control, and power	Reaching out is a sign of strength; it takes courage to ask for help (Real Warriors Campaign, undated).
Change perceptions about the effectiveness of care	Promote the belief that seeking help is helpful and that service providers can help more than families and friends can.	Perceptions that friends and family are more- effective supports	You cannot fix your mental health with duct tape (SpeakUp ReachOut, 2012). People recover; treatment is effective (Substance Abuse and Mental Health Services Administration, 2013).
Improve peer supports	Educate service members and families on the symptoms of mental illness and available resources.	Social support or encouragement from social network; awareness of available resources; perceptions of need	Back each other up (Back Each Other Up, undated). Look after your mate (University of Bristol Students' Union, undated). Know the signs (Know the Signs, undated).
Reduce access barriers	Educate service members on how to access mental health care.	Logistical and administrative barriers (e.g., transportation, cost, provider availability)	If you or someone you know is in crisis, call (National Institute of Mental Health, undated) (typically messages with information about where to get help, such as crisis lines and websites, often for free).

 Table 2.1

 Examples of Educational Approaches to Promote Treatment-Seeking and Well-Being

Given the important and unique impact that military culture may have on stigma, our intent in this chapter is to answer a set of key questions useful to DoD. These questions include "How does the prevalence of stigma in the military compare to that in the general population?" "Is stigma declining in the military?" "Is stigma more prevalent in certain branches of the military or among certain ranks of service members?" and "Are there data available that can be used for longitudinal tracking of stigma prevalence and to evaluate stigma-reduction programs?" As noted in Chapter Two, there are multiple definitions of *stigma*, which constrained us from identifying a single useful measure to serve as a point of comparison between civilian and military populations and between branches of the military. This chapter summarizes the data we were able to locate on trends in reported stigma in military populations and concludes with a brief discussion of limitations and challenges to current stigma measurement and surveillance.

Prevalence of Stigma in the Military

Several surveys have attempted to measure the impact that stigma has on mental health treatment-seeking in the military. Although the studies are not directly comparable to the U.S. prevalence estimates or to each other given differences in measurement tools, respondent demographics, rank, service, component, and period of deployment, they do collectively provide insight into the prevalence of stigma in the military. Given that perceptions of stigma are more relevant for, and higher among, people who screen positive for MHDs or who have a possible need for services, many sources report perceptions of stigma separately for those who do and do not screen positive for MHDs.

Joint Mental Health Advisory Team 7

The Mental Health Advisory Team (MHAT) reports provide one of the few consistent sources of publicly available information about stigma in the Army and, in some cases, the Marine Corps. However, the MHAT is applicable only to active-duty deployed soldiers and marines (in some cases) and varies in the ranks assessed (e.g., at times, assesses only E1 to E4). The most recent survey, Joint MHAT 7 (J-MHAT 7) (Office

of the Surgeon General, 2011), was conducted in 2010 in support of Operation Enduring Freedom (OEF) and was led by the Office of the Surgeon General of the Army with support from the Offices of the Surgeons General of the Navy and Air Force and the Office of the Medical Officer of the Marine Corps. The majority of J-MHAT 7 data comes from anonymous surveys collected from land combat service members assigned to maneuver unit platoons. Within every maneuver battalion in theater, three line companies are randomly selected; within those companies, three platoons are randomly selected to make up the cluster-based study sample. J-MHAT 7 presents sample-adjusted values based on male respondents and adjusted for demographic sample differences in rank and months deployed. Although the J-MHAT 7 had several objectives, one primary purpose was to assess behavioral health in land combat forces in Army and Marine Corps maneuver units. As part of that effort, soldiers and marines reported the extent to which they agreed with some factors that affected their decision to receive mental health services. We describe the results in this section, separately for the Army and Marine Corps.

Army

Data from E1 through E4 soldiers who had been in theater for nine months show that, of those who screen positive for MHDs, between 28.6 percent ("It would be too embarrassing") and 48.9 percent ("I would be seen as weak") report that stigma-related factors affect their decision to receive mental health services. Other factors included "it would harm my career" (29.2 percent), "my leaders would blame me for the problem" (33.9 percent), "members of my unit might have less confidence in me" (41.8 percent), and "my unit leadership might treat me differently" (46.0 percent). Among those who did not screen positive for any MHDs, the proportion endorsing these items was cut by approximately half, ranging from 13.0 percent ("my leaders would blame me for the problem") to 25.8 percent ("I would be seen as weak").

Marine Corps

In 2010 (Office of the Surgeon General, 2011), between 19.7 percent ("my leaders would blame me for the problem") and 33.3 percent ("members of my unit might have less confidence in me") reported agreeing or strongly agreeing with such sentiments if they were to seek care for MHDs. As with the Army, the proportion endorsing each of these factors was lower among marines who did not screen positive for MHDs, ranging from 12.6 percent ("it would harm my career") to 22.8 percent ("I would be seen as weak").

We also looked at the historical prevalence of various stigma measures among deployed Operation Iraqi Freedom (OIF) soldiers screening positive for mental health symptoms or disorders (Figure 3.1). We base this figure on publicly available data contained in the Army's MHAT reports. We focused on data from OIF because only two MHAT surveys included OEF data. As such, we omitted data from the J-MHAT 7, which focused only on OEF service members. Similarly, we focused primarily on sol-



Figure 3.1 Stigma Prevalence as Reported in the Mental Health Advisory Team Surveys, 2003 Through 2010

SOURCES: Operation Iraqi Freedom Mental Health Advisory Team, 2003; Operation Iraqi Freedom Mental Health Advisory Team II, 2005; Mental Health Advisory Team III, 2006; Mental Health Advisory Team IV, 2006; Mental Health Advisory Team V, 2008; Mental Health Advisory Team VI, 2009; Office of the Surgeon General, 2011. RAND *RR426-3.1*

diers, although we included data on marines whenever available. Overall, these data illustrate decreasing trends in reported stigma among soldiers and marines. For example, within the Army, the proportion of E1 through E4 soldiers in theater for 4.5 months who reported concern that "it would harm my career" as a barrier to treatment-seeking declined by about 6 percent between 2006 and 2010. Similarly, a comparable group of E1 through E4 marines declined in reporting this by 9 percent.

These data should be interpreted with some caveats in mind. First, the stigmaspecific results presented in MHATs I through IV appear to pertain to soldiers across a range of military ranks, while the stigma-specific data in MHATs V through VII relate specifically to soldiers ranking from E1 to E4. Therefore, these two sets of results (i.e., MHATs I through IV and MHATs V through VII) may pertain to different populations and should be interpreted distinctly from one another. Second, MHATs II and III do not present data for certain measures (e.g., harm to one's career), so these data points are missing in Figure 3.1. Third, the stigma-specific data reported in MHATs V are presented as "adjusted," but no information is offered to explain the procedures for adjustment. Therefore, it is unclear whether we can appropriately compare these data with MHAT data from other years. Fourth, it appears that the MHAT definition of mental health problems may have changed slightly over time. MHATs I and II define this category as soldiers screening positive for "depression, anxiety, or traumatic stress." MHATs III, IV, V, and VI define mental health problems as "depression, anxiety, or acute stress." Meanwhile, the J-MHAT 7 stated that it included soldiers screening positive for "any mental health problem." Fifth, the nature of certain stigma measures appears to have changed slightly over time and may not be comparable from year to year. Notably, the measure pertaining to differential treatment by members of one's unit is phrased as "my unit *membership* might treat me differently" in the MHAT V and as "my unit leadership might treat me differently" in other MHATs. Finally, stigma-specific data in MHAT VI were not reported in aggregate. Rather, it was reported among two different groups of soldiers: those in maneuver units and those in sustain and support units. To derive percentages reflective of the entire study sample, we estimated these numbers based on the sample sizes of each of the two subpopulations. However, our estimate could not account for survey respondents who may have skipped specific survey questions. Therefore, our estimates may vary slightly from the actual percentages.

Marines Attending the Combat Operational Stress Control Program

In addition to data on the prevalence of stigma among marines in the J-MHAT 7, a 2012 study examined barriers to help-seeking behavior among 533 marines, drawn from all three communities: infantry (18 percent), logistics (38 percent), and air (44 percent) who attended the Combat Operational Stress Control (COSC) program. Respondents were largely male (92 percent), white non-Hispanic (56.3 percent) or Hispanic (23.7 percent), and 25 or older (66.4 percent). About one-third of the sample were officers (O1–O4; 34.1 percent), and about one out of ten were enlisted (E4–E6, 9.1 percent). The study was conducted "to gather data that can be applied to enhance combat stress briefings to increase help-seeking behavior by targeting the specific concerns" (Momen, Strychacz, and Viirre, 2012, p. 1144). These data can be used as a base-line to track changes over time. Marines reported on factors that affected their decision to seek help for MHDs. These included fear of their commands losing trust in them (49.8 percent), fear of being treated differently (45 percent), concern about lack of confidentiality (37 percent), and a fear of negative effects on their careers (36.5 percent).

Navy Quick Poll

Navy quick polls are brief, periodically administered surveys designed to capture a snapshot, or "quick pulse" view of Navy personnel-related issues. The behavioral health quick poll has been administered annually since 2009, with the most-recent published data from 2011. The prevalence of stigma can be inferred from responses to questions

about whether sailors feel that their command and co-workers would be supportive if they sought help for stress or suicidal thoughts. In both 2009 and 2010, most sailors felt that their command and peers would be supportive if they asked for help, though those who reported "a lot" of stress were more likely to perceive negative outcomes from treatment-seeking. These outcomes included "Chain of Command would be less confident in me," "I would be embarrassed," and "People would treat me differently." Data from 2011 suggest some reductions in stigma, but the improvements were not significant. In 2011, 40 percent of officers and 38 percent of enlisted sailors believed that their commands would treat a person differently if they sought treatment, down from 45 percent and 42 percent, respectively, in 2010. There was also a small decrease among officers who believed that treatment-seeking would have a negative impact on their careers (down to 33 percent in 2011, compared with 37 percent in 2010). When asked about maintaining a security clearance after treatment-seeking, 18 percent of officers and 13 percent of enlisted reported believing that they would be able to keep their clearances, up from 14 percent and 9 percent, respectively, in 2010. Though an increase, the prevalence remains low, suggesting that this is an important barrier to care.

2011 Department of Defense Health Related Behaviors Survey of Active Duty Military Personnel

The 2011 Department of Defense Health Related Behaviors Survey of Active Duty Military Personnel (Barlas et al., 2013) includes collected data from 39,877 active-duty members of the services who were not deployed at the time of the survey:

- Army (15.2-percent response rate)
- Navy (22.3-percent response rate)
- Marine Corps (21.3-percent response rate)
- Air Force (32.9-percent response rate)
- Coast Guard (32.3-percent response rate).

Analyses were weighted to be representative of the DoD services and the Coast Guard separately.

More than one-third of respondents felt that seeking mental health treatment would harm their careers, with active-duty Navy personnel most likely to endorse this sentiment (42.1 percent). As in other studies, people who perceived a need for mental health care were more likely to believe that seeking treatment would damage their careers. Of those who perceived a need but did not seek care, 53.0 percent felt that it would damage their careers, while 40.5 percent of those who did seek care felt that it would damage their careers but sought treatment anyway. Among those who did seek care, 21.3 percent reported that treatment-seeking did have a negative effect on their careers, further reinforcing this concern. Service members in the Marine Corps

(26.2 percent) and the Navy (24.3 percent) who sought treatment were more likely to report that such treatment had a negative effect on their careers.

Hoge et al. Study of Barriers to Care Among Military Personnel Experiencing Combat Duty

This study examined barriers to mental health care among members of three U.S. Army combat infantry units and one Marine Corps combat infantry unit in 2003 (Hoge, Castro, et al., 2004). Though Hoge and his colleagues conducted the study more than a decade ago, the findings remain salient. Comparison of responses reported in this study and more-current studies supports the notion that there has been a reduction in stigma over time. Anonymous surveys were collected from service members either before their deployment to Iraq or three to four months after their return from combat duty in Iraq or Afghanistan. The surveys asked soldiers and marines about their use of professional mental health services and about perceived barriers to mental health treatment, including stigmatization. Like in J-MHAT 7 and other studies, people meeting screening criteria for MHDs were about twice as likely as those who did not to report concern about being stigmatized as a result of their disorder. Among those who screened positive,

- 65 percent reported that they would be seen as weak
- 63 percent believed that others would treat them differently
- 59 percent felt that members of their units would have less confidence in them
- 51 percent felt that leaders would blame them for the problem
- 50 percent felt that it would harm their careers
- 41 percent reported that seeking care would be too embarrassing.

Survey of Individuals Previously Deployed for Operation Enduring Freedom or Operation Iraqi Freedom

In 2007–2008, RAND researchers conducted a large, population-based survey of service members previously deployed as part of OEF or OIF (Schell and Marshall, 2008). The survey was designed to create a broadly representative sample of the population of those who have been deployed as part of OEF or OIF and targeted 24 geographic areas of the United States that encompass the domestic military bases with the largest overall number of deployed personnel. Overall, 1,938 interviews were conducted, and results were weighted to improve the representativeness of the analytic sample and account for the nonresponse in the sampling. The sample was divided as follows: 48.9 percent Army, 18.6 percent Navy, 19.8 percent Air Force, and 12.7 percent Marine Corps.

Barriers to seeking health care for mental health concerns were assessed through a single question: "If you wanted help for an emotional or personal problem, which of the following would make it difficult?" Statements posed as potential barriers to treatment followed the question, and respondents responded "yes" or "no." Several of these barriers were drawn from Hoge, Castro, et al. (2004) and represented institutional and cultural barriers to care. Among those who screened positive for MHDs, 43.6 percent reported that "it could harm my career" and 43.6 percent also felt that treatment-seeking could result in being "denied a security clearance in the future." Other barriers included

- "my coworkers would have less confidence in me if they found out" (38.4 percent)
- "I would think less of myself if I could not handle it on my own" (29.1 percent)
- "I do not think my treatment would be kept confidential" (29.0 percent)
- "my commander or supervisor might respect me less" (23.0 percent)
- "my commander or supervisor has asked us not to get treatment" (7.8 percent).

Table 3.1 maps similarly worded items across studies to facilitate examination of the prevalence of stigma in the military.

Limitations of Military Measures of Stigma

Table 3.1 displays the types of measures currently used by the military to assess the impacts of stigma. These questions focus primarily on the institutional (e.g., concerns around security clearance), public, and social contexts (e.g., members of my unit might have less confidence in me). Fewer surveys included such questions as "I would be seen as weak," which assess self-stigma (individual context). Clearly assessing each of these unique contexts is important because research has suggested that different contexts may have different effects on stigma (e.g., Vogel, Wade, and Hackler, 2007, found that the individual context moderates the effects of the public, institutional, and social contexts). Additionally, few measures separately assess perceptions of PWMHDs and treatment for MHDs. Assessing these separately is important because research suggests that the perceptions of PWMHDs and toward seeking treatment are distinct and can have possible unique and additive effects (Tucker et al., 2013).

Given the declining trend in stigma found through the MHAT surveys, we tried to identify data that described treatment-seeking to see whether the trends in treatment-seeking increased as stigma (particularly at the public and institutional contexts) declined. However, we were unable to locate a single source with longitudinal data on treatment-seeking. We did identify a series of studies that assessed treatment-seeking behaviors in a subset of service members with a need for mental health care. These studies reported data in 2004, 2007, 2008, and 2011. In 2004, Hoge et al. found that 72.9 percent of service members in need of mental health care were not seeking care. We analyzed data collected as part of the Invisible Wounds longitudinal survey of service members in 2011 and found that 54 percent of service members in need of mental health care were not seeking care. Because the sources varied in terms of their

Factor	Marine Corps COSC Study (2012)	DoD Health Related Behaviors Survey (2011)	Navy Quick Poll (2010)	Navy Quick Poll (2011)	Hoge, Castro, et al. (2004)	Invisible Wounds of War (2008)
lt would harm my career.	36.5	37.7	Officers: 37	Officers: 33	50	43.6
Members of my unit might have less confidence in me.	49.8	—	—	—	59	38.4
My unit	45	—	Officers: 45	Officers: 40	63	23.0
leadership might treat me differently.			Enlisted: 42	Enlisted: 38		
l would not be able to keep	—	—	Officers: 14	Officers: 18	_	43.6
my security clearance (quick polls) or I will not be able to obtain a security clearance in the future (Invisible Wounds).			Enlisted: 9	Enlisted: 13		
l do not think that my treatment would be kept confidential.	37	_	_	_	_	29.0

Table 3.1
Factors That Affect Decision to Receive Mental Health Services (%)

SOURCES: Momen, Strychacz, and Viirre, 2012; Barlas et al., 2013; Newell, Whittam, and Uriell, 2010, 2011; Hoge, Castro, et al., 2004; Schell and Marshall, 2008.

samples (e.g., service branch, mental health disorder, help-seeking period), we cannot definitely say that treatment-seeking behavior among service members is improving. However, further research on treatment-seeking is critical to help determine whether declines in stigma correlate with increases in treatment-seeking.

Challenges Comparing the Prevalence of Stigma in the Military with That in the General U.S. Population

Comparing data on the prevalence of stigma in the general U.S. population to prevalence within the military poses challenges for several reasons. First, stigma is measured differently in U.S. general population studies and in military studies of stigma. Outside of the military, stigma-related measures encompass a wide array of constructs (see Tables 3.2 and 3.3). Within the military, stigma is largely conceptualized and assessed as a barrier to seeking care. Because of these differences in measurement, making direct comparisons between levels of stigma in the military and in the U.S. population is difficult. Also, national studies among the general population have largely explored stigmatizing attitudes and beliefs toward people with schizophrenia or depression. Depression is commonly studied among military populations as well. However, attitudes and beliefs toward other disorders of interest to the military community, such as anxiety and PTSD, are absent from national studies.

We also point out that the prevalence of stigma varies from country to country (Evans-Lacko, London, et al., 2012; Pescosolido, 2013; Pescosolido, Medina, et al., 2013). Given the amount of U.S. military activity that takes place in countries outside of the United States, it is possible that cultural norms within those countries could affect the well-being of service members deployed to those areas. To our knowledge,

Source	Sample	MHD	Percentage with Clinically Significan Symptoms but Not Seeking Help	
Hoge, Castro, et al., 2004	Members of four U.S. combat infantry units (three Army units and one Marine Corps unit)	Major depression, generalized anxiety, or PTSD	72.9	Past year (postdeployment)
Milliken, Auchterlonie, and Hoge, 2007	U.S. soldiers returning from Iraq who completed both a postdeployment health assessment and a postdeployment health reassessment	People referred to mental health treatment (disorders not specified)	45.2	90 days
Invisible Wounds (Schell and Marshall, 2008; Tanielian and Jaycox, 2008)	Representative of deployed forces	PTSD or depression	47.3	Past year
Invisible Wounds (unpublished results from 2011) (see Chapter Four)	Guard and Reserve members in Western Pennsylvania who had been deployed since 2003	Major depression, PTSD, alcohol problems, or suicidal ideation	54.0	Past year

Table 3.2Studies on Service Members Who Need Mental Health Treatment but Do Not Seek It

Study	Year	Selected Stigma Items Measured in the Study
U.S. population		
NSDUH	Annually since 1990	Reasons for not seeking mental health treatment, including opinions of neighbors; fear, shame, or embarrassment; effect on job
BRFSS	2007, 2009	Beliefs about caring and sympathy toward people with mental illness; treatment efficacy
GSS	1996, 1998, 2002, 2006	Social distance, perceived dangerousness, treatment endorsement, treatment efficacy ^a
NCS	1990–1992	Treatment-seeking intentions, comfort with talking to a professional, embarrassment about seeking help
NCS-R	2001–2003	Treatment-seeking intentions, comfort with talking to a professional, embarrassment about seeking help
Military populations		
Marine Corps COSC study	2011, 2012	Concerns about career, concerns about treatment confidentiality, concerns about losing unit confidence, concerns about being treated differently by leadership
DoD Health Related Behaviors Survey	2011	Concerns about career
Navy quick poll	2010, 2011	Concerns about career, concerns about losing security clearance, concerns about losing leader confidence
Hoge, Castro, et al., study	2004	Concerns about career, concerns about losing unit confidence, concerns about being treated differently by leadership
Invisible Wounds (Schell and Marshall, 2008; Tanielian and Jaycox, 2008)	2008	Concerns about career, concerns about treatment confidentiality, concerns about losing security clearance, concerns about losing unit confidence, concerns about being treated differently by leadership

Table 3.3 Surveys Assessing Mental Health Stigma

SOURCES: Substance Abuse and Mental Health Services Administration (SAMHSA), 2012; Centers for Disease Control and Prevention et al., 2012; Pescosolido, Martin, et al., 2010; Schnittker, 2008; Kessler, 2002; National Comorbidity Survey, undated.

NOTE: NSDUH = National Survey on Drug Use and Health. BRFSS = Behavioral Risk Factor Surveillance System. GSS = General Social Survey. NCS = National Comorbidity Survey. NCS-R = NCS Replication. ^a These stigma items were in reference to a character depicted as having an MHD in a vignette read by respondents.

no research has explored the interplay between U.S. and military norms about mental health and the norms in countries where service members are deployed. Information about stigma trends in the U.S. general population is available in Appendix C.

Conclusion

Collectively, these data indicate possible changes in stigma over time and differences across populations. Consistent with the literature, we also found that people seeking mental health treatment reported higher perceived levels of stigma. Although we were unable to definitely conclude that declines in stigma (particularly in the public, institutional, and social context, which are assessed by the MHAT) resulted in increased treatment-seeking, further research is needed to determine the impact of the declines in stigma. Additionally, some limitations preclude drawing conclusive results from the stigma data. As demonstrated by the research summarized in this chapter, no single measure is being used to assess stigma, which likely stems from the multitude of ways stigma has been defined in the mental health literature (see Chapter Two). Most of the military measures assess stigma in the public, institutional, or social context but not within the individual context. Assessing stigma within the individual context is important because it moderates the impacts of the other contexts. Despite the availability of published measures, researchers continue to capture the construct of stigma by pulling out items from existing indexes or scales and creating new items (see, e.g., Tables 3.1 and 3.2). In this light, stigma is not being measured by a single, comparable scale or index but by a series of different single-item questions that are grouped together. Further discrepancies exist in the labels of these groupings, with some researchers referring to the items as stigma while others use different constructs (e.g., cultural barriers to care). Finally, some of the research on stigma prevalence among different populations (e.g., military officers versus enlisted personnel) reports findings that are not statistically significant or for which the significance of differences is unknown.

These challenges pose a major limitation for the advancement of our understanding of stigma and strategies to address it, both within and outside the military context. Notably, any patterns and trends in stigma found in military populations are not directly comparable to those in U.S. prevalence estimates. Among military service members, stigma appears to be declining (according to, for example, MHAT and Navy quick poll data). However, because of the limitations discussed, data on this topic are difficult to interpret. In addition, the extent to which factors at the institutional, public, or individual level contribute to these changes is unknown. Also, little is known about the extent to which stigma prevalence various across different military populations. For example, there is some indication that stigma may vary between officers versus enlisted personnel (see, for example, Navy quick poll data), but these data do not appear to be statistically significant. In the 2011 Department of Defense Health Related Behaviors Survey of Active Duty Military Personnel (Barlas et al., 2013), higher levels of stigma prevalence have been noted among certain branches of service (i.e., active-duty Navy and Marine Corps), but only for specific measures of perceived harm to one's career. Amid these inconclusive results, we note one reasonably consistent finding: People screening positive for MHDs tend to report greater concern about being stigmatized because of their disorders than people without MHDs.

Given the prevalence of mental health stigma in the military, what are its medical and societal costs? In this chapter, we present a microsimulation model developed to estimate the costs resulting from mental health stigma.¹ We based our model on an existing one that estimates the costs of untreated mental health symptoms or disorders among service members deployed as part of OEF or OIF and developed by Kilmer et al. (2011) as part of the RAND Invisible Wounds project (Tanielian and Jaycox, 2008). To adapt the model to include mental health stigma, we started by reviewing scientific literature to get an estimated value for how stigma may affect the probability of treatment-seeking. As mentioned previously, there was limited empirical literature that assessed direct impacts of stigma on *actual* help-seeking behaviors—most literature focused on how stigma affects knowledge, attitudes, and *intentions* to seek treatment. The literature we did identify had conflicting findings about stigma's impact on treatment-seeking. Some studies showed that stigma increased the probability of treatment-seeking; others showed that it decreased the probability of treatmentseeking; and most showed that it had no impact on treatment-seeking. We input the range of values from the literature review into the model, but there was no significant impact on medical and societal costs.

To augment the literature, we conducted a series of regression analyses using longitudinal data from service members collected as part of the RAND Invisible Wounds project (Tanielian and Jaycox, 2008). Our regression showed that stigma did not have a significant impact on probability to seek treatment and therefore would have no significant impact on medical and societal costs.

There may have been some reasons for this finding, which, as mentioned previously, goes against theoretical underpinnings and popular opinion. First, measures have neither fully captured the contexts that affect stigma nor appropriately differenti-

¹ Microsimulation models are event-driven simulations that generate individual life histories that can vary by individual socioeconomic and health-related individual characteristics. Such models are appropriate when needing to capture the complex set of behavioral responses that exist for unique individuals. An advantage of the microsimulation approach is that it can treat MHDs as recurring conditions, allowing for both remission and relapse over time.

ated between stigma directed at PWMHDs and stigma directed at the act of mental health treatment-seeking. In particular, the individual context is the least fully assessed by current military measures of stigma despite research that suggests that it may moderate the affects of the public, institutional, and social contexts. The public, institutional, and social contexts are more fully assessed, but, because they may be moderated by the individual context, they may have the most-distal impacts on stigma. Similarly, stigma may predict other variables that more directly influence behavior. Research has suggested, for example, that greater stigma is linked with more negative attitudes about therapy, which, in turn, leads to decreased intention to seek therapy (e.g., Vogel, Wade, and Hackler, 2007). Intentions then theoretically lead to behavior (e.g., Ajzen et al., 1980; Ajzen, 1991). Because stigma is so distal from the outcome, it may be difficult to empirically link. Finally, stigma may be less directly linked to the decision to seek help than to treatment success. Research suggests that stigma may contribute to decreased compliance with therapeutic interventions (K. Fung, Tsang, Corrigan, Lam, et al., 2007; Sirey, Bruce, Alexopoulos, Perlick, Friedman, et al., 2001); missed appointments (Vega, Rodriguez, and Ang, 2010); early termination of treatment (Sirey, Bruce, Alexopoulos, Perlick, Friedman, et al., 2001); and decreased intention to return for subsequent sessions (Wade, Post, et al., 2011). For example, Wade, Post, et al. (2011) found that self-stigma decreased after an initial session of group therapy but that it still predicted participants' willingness to come back for a second session (i.e., those who were higher on self-stigma scales reported less willingness to return). As previously mentioned, the limited empirical literature in this area challenged our ability to include these long-term outcomes in our model. To truly assess the impacts of stigma, improved research is needed that fully explore the empirical links among the public, institutional, social, and individual contexts and with stigma and to fully explore the full range of potential impacts (e.g., treatment success).

To understand what some of the more-proximal factors are that affect treatmentseeking, we conducted some exploratory regression analyses of other logistical, institutional, and cultural barriers to care, as well as beliefs and preferences for treatment. We found that the perception that friends and family are more helpful than mental health professionals was found to significantly decrease the probability of treatment-seeking. We plugged this into the microsimulation model to examine how reducing this barrier to care by 50 percent and 100 percent affects societal and medical costs. We then convened an expert panel to review and provide feedback on our approach to adapting the model. We describe each of these steps in more detail in this chapter.

Literature Review to Estimate Stigma's Impact on Treatment-Seeking

To estimate stigma's effect on medical and societal costs, we needed to run the model representing a world in which stigma was absent and then assessing the difference

between this ideal condition and the actual conditions modeled in the simulation. We intended to use evidence-based estimates from peer-reviewed scientific literature to assess the added costs introduced by stigma, so we conducted a literature review to derive these estimates. Appendix D presents a list of the literature we reviewed and a summary of findings.

Although one would expect that greater perceptions of stigma would be clearly associated with lower probabilities of treatment-seeking propensity, the findings in the literature have been mixed. Some studies have suggested that stigma reduces the likelihood that someone will initiate mental health treatment (e.g., Pietrzak et al., 2009; Barney, Griffiths, et al., 2006). However, the majority of studies support a finding that, although stigma is widely reported as a barrier to care, there is no significant evidence that it affects actual mental health care utilization (e.g., Clement, Schauman, et al., 2014; Gould, Adler, et al., 2010; Schomerus and Angermeyer, 2008; S. Brown, 2010; Vogel, Wade, and Hackler, 2007). We were unable to find robust empirical published literature on the direct impact of stigma on mental health treatment-seeking. For a summary of the literature that prior studies cited as empirical evidence or that we identified in our literature review as possibly containing empirical evidence between stigma and treatment-seeking, see Table 4.1. Therefore, we derived an estimate using data from a RAND military survey conducted as part of the Invisible Wounds project that included questions on mental health utilization, as well as stigma and other barriers to care (Schell and Marshall, 2008).

Regression Analyses to Estimate Stigma's Impact on Treatment-Seeking

RAND researchers conducted a longitudinal large population-based survey on service members previously deployed as part of OEF or OIF (Schell and Marshall, 2008). This survey offers the ability to assess the extent to which specific barriers to care are associated with subsequent mental health treatment in a sample of previously deployed U.S. service members who have been identified as having a need for mental health treatment. To identify the possible effect of mental health stigma on service utilization, we constructed a logistic regression model in which a range of factors assessed at baseline predicted minimally adequate care at follow-up (18 months after baseline). The final analytic sample included 279 active-duty service members who had a need for treatment at follow-up. The analytic strategy was to combine the barriers that were thought to be directly assessing the stigma of mental health symptoms or disorders or treatment into a measure of stigma. Five items were identified as relatively direct measures of a concern that other people might evaluate or treat one poorly because of their stereotypes or prejudice against those with mental health symptoms or disorders (see Table 4.2), which is consistent with our definition of *stigma* (Chapter Two). Other bar-

	-			-		•
Study	Population	Was the Survey a Longitudinal or Cross-Sectional Study?	Does the Study Report How Many Respondents Are Receiving Treatment for Their MHDs? Is This Adjusted to Reflect Overall Prevalence in the Population?	How Are the Results Showing the Dependency of Treatment Propensity on	How Is the Confidence Level Reported?	Within the Reported Cls, Does the Study Show That Stigma Has a Significant Impact in Reducing the Treatment Probability?
Aromaa et al., 2011a	Civilian Finnish community	Cross-sectional	39.6% of respondents with major depression had used health services because of mental health symptoms or disorders in the past year, 25% with primary care health center, 18% with outpatient specialist mental health care, and 10% with private practitioner	OR	<i>P</i> -value	The ORs of desire for social distance (0.81***) and antidepressant attitudes (0.63***) are significantly different from 1. The study assesses attitudes only and does not assess treatment-seeking behavior.

Table 4.1 Studies Identified as Most Likely to Have Empirical Evidence of the Association Between Stigma and Treatment-Seeking Behavior

Study	Population	Was the Survey a Longitudinal or Cross-Sectional Study?		Does the Study Report How Many Respondents Are Receiving Treatment for Their MHDs? Is This Adjusted to Reflect Overall Prevalence in the Population?	How Are the Results Showing the Dependency of Treatment Propensity on	How Is the Confidence Level Reported?	Within the Reported Cls, Does the Study Show That Stigma Has a Significant Impact in Reducing the Treatment Probability?
Barney, Griffiths, Jorm, et al., 2006	Civilian Australian community	Cross-sectional	1,312	Yes, but only for the respondents (i.e., not adjusted to the population level). This ranges from psychiatrists (34%) to general practitioners (73%).	OR	<i>P</i> -value	The OR of perceived stigma (1.28**) is significantly different from 1, indicating an impact of stigma on intentions to seek treatment. The study did not assess treatment- seeking behavior.
C. Brown et al., 2010	Civilian black and white U.S. adults	Cross-sectional	449	Although 50% of the sample showed symptoms of depression, only about 20% were currently being treated for depression.	Correlation matrix	<i>P</i> -value	Neither internalized nor public stigmas were significantly associated with intention to seek treatment for depression or current treatment for depression.

Study	Population	Was the Survey a Longitudinal or Cross-Sectional Study?		Does the Study Report How Many Respondents Are Receiving Treatment for Their MHDs? Is This Adjusted to Reflect Overall Prevalence in the Population?	How Are the Results Showing the Dependency of Treatment Propensity on		Within the Reported Cls, Does the Study Show That Stigma Has a Significant Impact in Reducing the Treatment Probability?
Diala et al., 2000	Civilian U.S. community	Cross-sectional	5,877	No prevalence values reported	OR	<i>P</i> -value	The OR of embarrassment if friends knew (0.3**) and comfort with seeking care (0.1** for those with depression; 0.2** for entire sample) are significantly different from 1. The study does not assess treatment- seeking behavior.

34 Mental Health Stigma in the Military

Study	Population	Was the Survey a Longitudinal or Cross-Sectional Study?		Does the Study Report How Many Respondents Are Receiving Treatment for Their MHDs? Is This Adjusted to Reflect Overall Prevalence in the Population?	How Are the Results Showing the Dependency of Treatment Propensity on	How Is the Confidence Level Reported?	Within the Reported Cls, Does the Study Show That Stigma Has a Significant Impact in Reducing the Treatment Probability?
Downs and Eisenberg, 2012	Civilian students at U.S. universities	Cross-sectional	8,487	Of the respondents with SI, 51% received treatment in the past year, including therapy (40.9%) and medication (35.8%). Of the same sample, 31.6% were currently receiving treatment, including therapy (19.1%) and medication (24.1%).	OR	<i>P</i> -value and 95% Cl	ORs are 1.190** for perceived stigma and 0.725** for personal stigma. The study found that stigma was correlated with treatment; however, because the study was cross-sectional, we were unable to establish a causal link between stigma and treatment-seeking behavior.

Study	Population	Was the Survey a Longitudinal or Cross-Sectional Study?		Does the Study Report How Many Respondents Are Receiving Treatment for Their MHDs? Is This Adjusted to Reflect Overall Prevalence in the Population?	How Are the Results Showing the Dependency of Treatment Propensity on		Within the Reported Cls, Does the Study Show That Stigma Has a Significant Impact in Reducing the Treatment Probability?
Eaton et al., 2008	General population of U.S. military spouses	Cross-sectional	940	Of the spouses who screened positive for an MHD, 68% received mental health care (41% specialty mental health care, 19% from primary care physician, 8% pastoral counseling), compared with 22% for those who screened negative.	N/A	N/A	N/A (The study measured the prevalence of perceived barriers to care for spouses who screened positive for an MHD but did not assess treatment- seeking behavior.)

Study	Population	Was the Survey a Longitudinal or Cross-Sectional Study?		Does the Study Report How Many Respondents Are Receiving Treatment for Their MHDs? Is This Adjusted to Reflect Overall Prevalence in the Population?	How Are the Results Showing the Dependency of Treatment Propensity on Stigma Level Reported?	How Is the Confidence Level Reported?	Within the Reported Cls, Does the Study Show That Stigma Has a Significant Impact in Reducing the Treatment Probability?
Edlund et al., 2008	U.S. veterans with depression	Longitudinal	395	All veterans in the sample were currently receiving treatment for a chronic health condition, so the sample was not representative of the general population.	OR	<i>P</i> -value and 95% Cl	OR is 1.15** for a summary measure of (a) perceived need for depression treatment; (b) believing that treatment for depression will be helpful; and (c) treatment barriers. This summary significantly predicted whether a veteran would initiate and adhere to the use of an antidepressant. However, the subscales were not significant independent predictors.

Study	Population	Was the Survey a Longitudinal or Cross-Sectional Study?		Does the Study Report How Many Respondents Are Receiving Treatment for Their MHDs? Is This Adjusted to Reflect Overall Prevalence in the Population?	How Are the Results Showing the Dependency of Treatment Propensity on	How Is the Confidence Level Reported?	Within the Reported Cls, Does the Study Show That Stigma Has a Significant Impact in Reducing the Treatment Probability?
Eisenberg, Downs, Golberstein, and Zivin, 2009	Civilian students at U.S. universities	Cross-sectional	5,555	No prevalence values were reported.	OR	<i>P</i> -value and 95% Cl	The OR of personal stigma predicting receiving nonclinical support (0.80**) and use of medication (0.57**) and therapy (0.57**) are significantly different from 1, indicating an association between personal stigma and treatment- seeking.

Study	Population	Was the Survey a Longitudinal or Cross-Sectional Study?	Does the Study Report How Many Respondents Are Receiving Treatment for Their MHDs? Is This Adjusted to Reflect Overall Prevalence in the Population?	How Are the Results Showing the Dependency of Treatment Propensity on Stigma Level Reported?	How Is the Confidence Level Reported?	Within the Reported Cls, Does the Study Show That Stigma Has a Significant Impact in Reducing the Treatment Probability?	
Eisenberg, Downs, Golberstein, and Zivin, 2009, continued						Perceived public stigma was not associated with a lower likelihood of receiving nonclinical support, therapy, or medication. Stigma did not have a significant association with willingness to discuss problem with academic staff. Because the study was cross-sectional, we were unable to establish a causal relationship between stigma and treatment- seeking.	
Study	Population	Was the Survey a Longitudinal or Cross-Sectional Study?		Does the Study Report How Many Respondents Are Receiving Treatment for Their MHDs? Is This Adjusted to Reflect Overall Prevalence in the Population?	How Are the Results Showing the Dependency of Treatment Propensity on Stigma Level Reported?	How Is the Confidence Level Reported?	Within the Reported Cls, Does the Study Show That Stigma Has a Significant Impact in Reducing the Treatment Probability?
------------------------	--	--	-----	---	--	---	---
Givens et al., 2007	Civilian black and white U.S. adults	Cross-sectional	490	Of the respondents, one-third had a history of depression, of whom 90% had prior treatment with either prescription medication or counseling.	Adjusted OR	<i>P</i> -value and 95% Cl	Stigma was not associated with acceptability of prescription medication, but two stigma items were associated with intentions to use mental health counseling: feeling ashamed (adjusted OR = 0.43*) and discomfort telling family and friends (adjusted OR = 0.42*). The study did not assess treatment- seeking behavior.

Study	Population	Was the Survey a Longitudinal or Cross-Sectional Study?		Does the Study Report How Many Respondents Are Receiving Treatment for Their MHDs? Is This Adjusted to Reflect Overall e Prevalence in the Population?	How Are the Results Showing the Dependency of Treatment Propensity on	How Is the Confidence Level Reported?	Within the Reported Cls, Does the Study Show That Stigma Has a Significant Impact in Reducing the Treatment Probability?
Golberstein, Eisenberg, and Gollust, 2009	Civilian students at U.S. universities	Cross-sectional	2,782	Of the respondents who screened positive for depression or anxiety, 65% perceived a need for help, of whom 52% utilized mental health treatment.	OR	<i>P-</i> value	The OR of perceived stigma's association with perceiving a need for help (0.86**) was significant only for those age 18–22 but not for older students or overall. Perceived stigma was not associated with treatment-seeking behavior.
Gorman et al., 2011	U.S. National Guard members and their significant others	Cross-sectional	332 National Guard members and 212 significant others	Of the respondents screening positive for an MHD, 53% utilized services (50% of guard members, 61% of significant others).	N/A	N/A	N/A (The study compared stigma for people with mental health diagnoses and those without but did not measure the association between stigma and treatment- seeking behavior.)

Study	Population	Was the Survey a Longitudinal or Cross-Sectional Study?		Does the Study Report How Many Respondents Are Receiving Treatment for Their MHDs? Is This Adjusted to Reflect Overall Prevalence in the Population?	How Are the Results Showing the Dependency of Treatment Propensity on		Within the Reported CIs, Does the Study Show That Stigma Has a Significant Impact in Reducing the Treatment Probability?
Gould, Adler, et al., 2010	U.S., UK, Australian, New Zealand, and Canadian armed forces	Cross-sectional	12,469	No prevalence values were reported.	N/A	N/A	N/A (The study compared perceived barriers for people with MHDs and those without but did not measure the association between stigma and treatment-

and treatmentseeking behavior.)

Study	Population	Was the Survey a Longitudinal or Cross-Sectional Study?		Does the Study Report How Many Respondents Are Receiving Treatment for Their MHDs? Is This Adjusted to Reflect Overall Prevalence in the Population?	How Are the Results Showing the Dependency of Treatment Propensity on	How Is the Confidence Level Reported?	Within the Reported CIs, Does the Study Show That Stigma Has a Significant Impact in Reducing the Treatment Probability?
Hoerster et al., 2012	U.S. veterans of Iraq and Afghanistan with symptoms of depression, PTSD, or alcohol misuse	Cross-sectional	305	37% of respondents (all of whom had symptoms of depression, PTSD, or alcohol misuse) reported at least one stigma-related barrier. Both PTSD-symptom severity and depression- symptom severity were significantly reduced in patients who received "adequate" mental health treatment (at least nine visits).	N/A	<i>P</i> -value	In bivariate tests, respondents with at least nine health care visits (the cutoff used for "adequate care") had significantly higher endorsement of stigma-related barriers (mean \pm SD of 3.0 \pm 1.1 versus 2.6 \pm 1.1). When more variables were entered into the model, endorsement of stigma-related barriers was not associated with receiving adequate mental health care.

Study	Population	Was the Survey a Longitudinal or Cross-Sectional Study?		Does the Study Report How Many Respondents Are Receiving Treatment for Their MHDs? Is This Adjusted to Reflect Overall Prevalence in the Population?	How Are the Results Showing the Dependency of Treatment Propensity on	How Is the Confidence Level Reported?	Within the Reported Cls, Does the Study Show That Stigma Has a Significant Impact in Reducing the Treatment Probability?
Hoge, Castro, et al., 2004	U.S. combat infantry units	Cross-sectional	6,201	Of the respondents screening positive for an MHD, only 38– 45% expressed an interest in receiving help, and 23–40% reported having received professional treatment.	N/A	N/A	N/A (The study did compare stigma for people with mental health diagnoses and those without but did not measure utilization.)
lversen et al., 2011	UK military personnel	Cross-sectional	821	No prevalence values were reported.	N/A	N/A	N/A (The study compared stigma for people with mental health diagnoses and those without but did not measure the association between stigma and treatment- seeking behavior.)

Study	Population	Was the Survey a Longitudinal or Cross-Sectional Study?	What Was the Sample Size of the Study?	Does the Study Report How Many Respondents Are Receiving Treatment for Their MHDs? Is This Adjusted to Reflect Overall Prevalence in the Population?	How Are the Results Showing the Dependency of Treatment Propensity on	How Is the Confidence Level Reported?	Within the Reported Cls, Does the Study Show That Stigma Has a Significant Impact in Reducing the Treatment Probability?
Kessler, Berglund, et al., 2001	U.S. general population	Cross-sectional	5,000	Of the respondents who met the criteria for serious MHDs, 46.2% received professional treatment in the past year, compared with 18.3% of those with other MHDs.	N/A	N/A	N/A (Although this study did not measure the association of stigma with utilization, it did report prevalence of reasons for not seeking care. One of the least common reasons was "concerned about what others might think.")
Kim, Thomas, et al., 2010	U.S. active-duty and National Guard soldiers	Cross-sectional		Of the respondents reporting mental health symptoms or disorders, 13% of active- duty and 17–27% of National Guard soldiers used some type of professional mental health care.	N/A	N/A	N/A (The study measured the prevalence of stigma for soldiers who screened positive for a mental health symptom or disorder but not the association between stigma and treatment- seeking behavior.)

Study	Population	Was the Survey a Longitudinal or Cross-Sectional Study?		Does the Study Report How Many Respondents Are Receiving Treatment for Their MHDs? Is This Adjusted to Reflect Overall Prevalence in the Population?	How Are the Results Showing the Dependency of Treatment Propensity on	How Is the Confidence Level Reported?	Within the Reported Cls, Does the Study Show That Stigma Has a Significant Impact in Reducing the Treatment Probability?
Link, Struening, Rahav, et al., 1997	Civilian U.S. men with MHDs and substance abuse	Longitudinal	84	No prevalence values were reported.	N/A	N/A	N/A (The study measured the prevalence of stigma for men with MHDs and substance abuse, but all of the respondents were undergoing treatment, so the association between stigma and treatment- seeking behavior was not measured.)
Markowitz, 1998	Civilian U.S. PWMHDs	Longitudinal	610	No prevalence values were reported.	N/A	N/A	N/A (The study measured the prevalence of stigma for PWMHDs but not the association between stigma and treatment- seeking behavior.)

Study	Population	Was the Survey a Longitudinal or Cross-Sectional Study?		Does the Study Report How Many Respondents Are Receiving Treatment for Their MHDs? Is This Adjusted to Reflect Overall e Prevalence in the Population?	How Are the Results Showing the Dependency of Treatment Propensity on	How is the Confidence Level Reported?	Within the Reported Cls, Does the Study Show That Stigma Has a Significant Impact in Reducing the Treatment Probability?
Perlick et al., 2001	Civilian U.S. people diagnosed with bipolar affective disorder	Cross-sectional	264	No prevalence values were reported.	N/A	N/A	N/A (The study did measure stigma for people with bipolar but did not measure utilization.)
Pietrzak et al., 2009	U.S. veterans of OEF and OIF	Cross-sectional	272	26% of respondents had treatment in past 6 months, and PTSD was positively associated with counseling (OR = 0.83).	N/A	N/A	N/A (The study measured the association of negative beliefs with utilization and with perceived stigma but did not report the association between stigma and treatment- seeking behavior.)

Study	Population	Was the Survey a Longitudinal or Cross-Sectional Study?		Does the Study Report How Many Respondents Are Receiving Treatment for Their MHDs? Is This Adjusted to Reflect Overall Prevalence in the Population?	How Are the Results Showing the Dependency of Treatment Propensity on	How Is the Confidence Level Reported?	Within the Reported Cls, Does the Study Show That Stigma Has a Significant Impact in Reducing the Treatment Probability?
Sansone and Sansone, 2013	U.S. military parents of children with psychiatric illness	Cross-sectional	67	No prevalence values were reported.	Correlation	<i>P</i> -value	Although this study did not suggest that stigma affects utilization, it did show a significant relationship between illness severity (as measured by frequency of treatment utilization) of the children and career stigma of the parents.

Study	Population	Was the Survey a Longitudinal or Cross-Sectional Study?		Does the Study Report How Many Respondents Are Receiving Treatment for Their MHDs? Is This Adjusted to Reflect Overall Prevalence in the Population?	How Are the Results Showing the Dependency of Treatment Propensity on	How Is the Confidence Level Reported?	Within the Reported Cls, Does the Study Show That Stigma Has a Significant Impact in Reducing the Treatment Probability?
Schomerus and Angermeyer, 2008	Civilian Germans with untreated depression	Cross-sectional	25	28% of respondents (with depression) had used mental health services in the past.	N/A	N/A	N/A (The study found a significant association between personal stigma and lower perceived need for professional help; $\beta = -0.59$). However, the study did not assess the association between stigma and treatment- seeking behaviors.)

Study	Population	Was the Survey a Longitudinal or Cross-Sectional Study?		Does the Study Report How Many Respondents Are Receiving Treatment for Their MHDs? Is This Adjusted to Reflect Overall Prevalence in the Population?	How Are the Results Showing the Dependency of Treatment Propensity on	How Is the Confidence Level Reported?	Within the Reported Cls, Does the Study Show That Stigma Has a Significant Impact in Reducing the Treatment Probability?
Stone, 1998	Members of the U.S. Air Force	Cross-sectional	391	21% of entire sample had voluntarily used mental health services (did not report for those with MHDs specifically).	N/A	<i>P</i> -value	No significant differences were found between voluntary help- seekers and non-help-seekers or between forced help- seekers or help- contemplators and the rest of the sample in their appraisal of stigma or intentions to seek treatment. The study did not assess treatment- seeking behavior.

NOTE: OR = odds ratio. CI = confidence interval.

Domain	Item
Stigma	My friends and family would respect me less. My spouse or partner would not want me to get treatment. My co-workers would have less confidence in me if they found out. My commander or supervisor has asked us not to get treatment. My commander or supervisor might respect me less.
Other barriers to care	It could harm my career. I do not think my treatment would be kept confidential. I would not know where to get help or whom to see. It would be difficult to arrange transportation to treatment. It would be difficult to schedule an appointment. Mental health care would cost too much money. Even good mental health care is not very effective. The medications that might help have too many side effects. It would be difficult to get child care or time off of work. My family or friends would be more helpful than a mental health professional. Religious counseling would be more helpful than mental health treatment. I could lose contact or custody of my children. I could lose medical or disability benefits. I could be denied a security clearance in the future. I have received treatment before and it did not work.

Table 4.2 Barrier-to-Care Items in the Invisible Wounds Survey

SOURCE: Schell and Marshall, 2008.

riers to treatment assessed at baseline were also included in the model as covariates (see Table 4.2). In order to maximize the power to detect an effect of stigma, the model was pruned using a parsimony criterion (Akaike's information criterion; see Akaike, 1987).

Consistent with our review of the empirical published literature is the finding that stigma did not significantly predict subsequent treatment utilization at the p < 0.05 level. Before inserting stigma into the model, analysis of the data found that the yearly probability of treatment initiation by service members with a probable need for mental health treatment is 26.8 percent. The regression model shows that, if we were to completely eliminate the effect of stigma, the yearly probability of seeking treatment would still remain 26.8 percent and, with a 95-percent CI of the effect of stigma, we can be relatively confident that the true effect would result in rates of utilization between 21.5 percent and 32.9 percent.

We also conducted sensitivity tests in which the effect of stigma was estimated with and without two other barriers to care in the model that may be partially affected by concerns about stigma (e.g., it could harm my career, I do not think that my treatment would be kept confidential). We found that decreasing stigma would not increase the number of service members seeking mental health treatment. The sensitivity analyses also found that stigma did not significantly predict subsequent treatment utilization at the p < 0.05 level, regardless of whether we include as covariates the two barriers that may be indirect measures of stigma. In all cases, the effect size of stigma as a predictor of mental health treatment is descriptively small.

Exploratory Analyses Examining Other Barriers to Care

However, the data did suggest that utilization depends mostly on age, branch component, PTSD status, and two other barriers to care:

- I could lose medical or disability benefits.
- My family or friends would be more helpful than a mental health professional.

However, the first of these barriers was reported as a barrier only among those not in treatment; no one in treatment endorsed this barrier. Therefore, we were not able to estimate an OR for this first barrier, the fear that medical or disability benefits would be lost if a service member seeks treatment. However, we proceeded in constructing a regression model that predicts utilization based on the second barrier—namely, that support from family and friends provides a more helpful alternative to professional mental health treatment. The OR for this barrier was significantly less than 1, showing a statistically significant effect of this barrier on treatment utilization; people who believed that mental health professionals were not particularly helpful were much less likely to subsequently seek mental health care. Using the treatment probabilities from the microsimulation model (described in detail in Appendix D), our findings suggested that the number of service members seeking mental health treatment would increase approximately 7 percent if this barrier could be eliminated. We then presented these findings to an expert panel.

Expert Panel to Vet Model Assumptions and Parameters

In September 2013, we convened a panel of ten experts whom we asked to provide feedback on

- model assumptions and parameters, particularly for those parameters that have the greatest impact on cost outcomes
- our approach to quantifying the effect of stigma on mental health treatment utilization
- our analysis of other barriers to care that may affect mental health treatment utilization.

In summary, the expert panel indicated agreement with the model assumptions and parameters and made some recommendations about how model findings should be presented, which we have implemented in this report. A summary of the expert panel discussion is contained in Appendix I.

Modeling Societal and Medical Costs

Following recommendations made by the expert panel, we first modeled medical and societal costs for the hypothetical case in which PWMHDs do not enter treatment. We then modeled baseline costs, which represent suicide and cost outcomes without stigma or any other barriers to care removed. Finally, we modeled costs with a single barrier to care (perception that friends and family are more helpful than a mental health professional) reduced by 50 percent and by 100 percent. More information on the micro-simulation model, methods used to adapt the model (literature review and regression analyses), and the findings from the model runs are included in Appendix D.

As the probability that a service member will seek treatment increases (going down the rows in Tables 4.3 and 4.4), total treatment costs increase. The elimination of the barrier "I think my friends and family are more helpful than a mental health professional" does not produce a dramatic increase in treatment costs. Reducing this barrier by 50 percent would increase treatment costs by just under \$3 million. However, looking at total productivity lost and aggregated cost, we see cost savings. Reducing this barrier by 50 percent would result in more than \$9 million in savings in lost productivity and aggregated costs. Not surprisingly, the cost savings are even greater if this barrier is completely eliminated (100-percent reduction). In this case, the aggregated cost would fall by more than \$32 million from the baseline. Notice that, if we eliminate this barrier, cost savings are more than double those produced by only 50-percent reduction (Table 4.3). This is determined by both larger cost savings in total productivity and a smaller number of suicide attempts and, although a marginal reduction, a smaller number of deaths by suicide (Table 4.4).

Simulation	Total Treatment Cost	Total Productivity Lost	Total Cost of Suicide	Total Cost of Suicide and Value of Life Lost	Aggregated Cost
No treatment	0	741.69	2.55	313.47	1,055.16
Baseline treatment	21.65	650.98	2.26	277.64	950.28
50% reduction in barrier	24.42	641.39	2.23	275.23	941.04
100% reduction in barrier	27.32	624.71	2.18	265.9	917.94

Table 4	1.3
---------	-----

Total Aggregate	Costs for Each	Simulation (in	millions of dollars)
-----------------	----------------	----------------	----------------------

NOTE: The barrier to care referred to in the table is the perception that friends and family are more helpful than a mental health professional.

Simulation	Number of Suicide Attempts	Number of Deaths by Suicide		
No treatment	482	37		
Baseline treatment	425	33		
50% reduction in barrier	420	33		
100% reduction in barrier	410	32		

 Table 4.4

 Aggregate Suicide Attempts and Deaths by Suicide for Each Simulation

NOTE: The barrier to care referred to in the table is the perception that friends and family are more helpful than a mental health professional.

Conclusion

Our regression analyses showed that stigma did not have a significant impact on probability to seek treatment and therefore would have no significant impact on medical and societal costs (i.e., when entered into the microsimulation model, had no effect on initiation of treatment-seeking). We discussed that there may have been some reasons for this finding, which goes against theoretical underpinnings and popular opinion. These include limitations in the way in which stigma is measured. Measures have neither fully captured the contexts that affect stigma nor appropriately differentiated between stigma directed at PWMHDs and stigma directed at the act of mental health treatment-seeking. The public, institutional, and social contexts—which are more fully assessed—may have the most-distal impacts on stigma because they are moderated by the individual context. Similarly, stigma may predict other variables that more directly influence behavior. Because stigma is so distal from the outcome, it may be difficult to empirically link. Finally, stigma may be less directly linked to the decision to seek help than to treatment success.

To understand what some of the more-proximal factors are that affect treatmentseeking, we conducted some exploratory regression analyses of other logistical, institutional, and cultural barriers to care, as well as beliefs and preferences for treatment. Our analyses suggested that other barriers to care (not stigma) may influence treatment utilization—in particular, the perception that support from family and friends provides a more helpful alternative to professional mental health treatment. Reducing this barrier by 50 percent would increase treatment costs by just under \$3 million but would result in more than \$9 million in savings in lost productivity and aggregated costs. Cost savings more than double if we eliminate this barrier, falling by more than \$32 million from the baseline.

CHAPTER FIVE Promising Programmatic and Policy Approaches to Reducing Stigma

Given the conceptual model described in Chapter Two, what are the most-promising programmatic and policy approaches to reducing stigma? This chapter presents a review of the literature on evidence-based programs and policies to reduce stigma. We review the factors associated with mental health stigma that can be changed through intervention. To help distinguish factors, we organize them by each of the four key contexts that might be targeted through intervention (i.e., public, institutional, social, and individual) as specified in the conceptual model in Chapter Two. Within each context, we first review the literature, paying particular attention to the strength of evidence for each factor. Then we describe implications for the intervention strategies in the military context.

Intervening in the Public and Social Contexts

We group together the literature on interventions for the public and social contexts because these interventions are largely overlapping in the literature and vary only based on the population to which the interventions are delivered. For interventions in the public context, these are delivered to broad populations or a group (e.g., an entire installation), whereas interventions in the social context are targeted at friends, family, and unit members. A large majority of the articles we identified on stigma-reduction interventions targeting the public and social contexts were published post-2000, suggesting that *the literature on effective stigma-reduction interventions is a growing area*. About one-quarter of the articles we reviewed described a stigma-reduction intervention that was tested and found to be effective in reducing stigma (n = 62) or review articles summarizing effective stigma interventions (n = 6).

The literature on stigma-reduction interventions varies from large-scale randomized controlled trials to small-scale pre/post-test designs; however, *the majority of the literature is made up of studies that assessed intervention effectiveness through a pre/post approach with limited or no longer-term follow-up and that did not use a control or comparison group.* To assess the quality of the literature, we abstracted information about the methodology used in these studies to determine

- whether intervention participants had been randomly assigned to the intervention group
- whether there was a control or comparison group
- the type and timing of data collected (cross-sectional versus longitudinal)
- the type of data analyses used to determine intervention effectiveness (e.g., descriptive, bivariate, multivariate).

Only 19 articles used random assignment to establish an intervention group and control group, which is considered to be the least biased way to test an intervention. Among those that used random assignment, about half (n = 10) collected longitudinal data across more than two time points (e.g., pre-intervention, postintervention, six months postintervention) ranging from one week postintervention to six months postintervention. Collecting data longitudinally is critical for stigma-reduction efforts because many of the studies we reviewed found that stigma-reduction interventions had short-term effects that did not last over time (K. Anderson and Austin, 2012; Castro, Adler, et al., 2012; Rusch, Kanter, Angelone, et al., 2008; Jorm, Kitchener, Fischer, et al., 2010; Corrigan, Larson, et al., 2007; Adler et al., 2011). Among those that both used random assignment and collected longitudinal data, about half (n = 6) used multivariate analyses to detect significant differences between their intervention and control groups.

Public and Social Context Interventions Tested in Civilian Populations

From this literature base, we were able to identify a limited number of well-established interventions tested in civilian populations, each of which we describe in more detail below. We describe interventions that had more than one article published demonstrating their effectiveness or interventions that used random assignment in their testing and collected longitudinal data to demonstrate whether the intervention had lasting impact on participants. We found that interventions ranged widely in terms of the target audience from the public to more-specific populations, such as pharmacy or medical students. We identified no effective interventions focusing on friends, families, or co-workers.

Contact with Individuals with Mental Illness

The National Alliance on Mental Illness developed the In Our Own Voice (IOOV) program, which features two people with mental illness interacting with participants for 90 minutes to share their initial experiences with mental illness and how they came to terms with it, effective treatments and daily coping mechanisms, and their successes, hopes, and dreams. The in-person interaction is accompanied by standardized videos or PowerPoint presentations that share more educational information. Several studies of IOOV have been conducted, primarily with undergraduate psychology students. Wood and Wahl (2006) initially tested the effectiveness of this intervention and found

significant decreases in social distance and significant increases in knowledge and attitude scores from pre-test to posttest among the intervention group. Scores did not change significantly for those in the control group. Mann and Himelein (2008) replicated those results. Pittman, Noh, and Coleman (2010) tested the intervention with social work students and found similar impacts on stigma. Rusch, Kanter, Angelone, et al. (2008) followed with a more focused study using a sample of 43 undergraduate psychology students and found that IOOV was more effective than psychoeducation at reducing stigma; however, the impact was short term. IOOV was also tested with medical residents and found to increase their perception that mental health treatment is worthwhile (Schmetzer and Lafuze, 2008).

Using a sample of 200 undergraduate psychology students, Corrigan, Rafacz, Hautamaki, et al. (2010) compared three variations on IOOV to determine relative effectiveness of each:

- the full 90-minute interaction
- an abbreviated 30-minute interaction that features only one speaker and does not use the videos
- an education-only component that does not include any interaction with anyone with mental illness.

Participants in the 90- and 30-minute conditions recalled more positive than negative statements about a person with a mental illness than did participants in the education-only condition, and there were no significant differences between the 30and 90-minute conditions, suggesting that a more streamlined version of this intervention may be as effective as a longer version.

Training on How to Help Someone in Emotional Distress

Mental Health First Aid (MHFA) is a nine-hour training to teach participants how to help someone who may be exhibiting suicidal thoughts or behaviors, an acute stress reaction, panic attacks, or acute psychotic behaviors. Several studies (Kitchener and Jorm, 2002, 2004, 2006) found that MHFA decreased social distance, improved perceptions of mental health treatment as effective, increased confidence in providing help to someone with a mental health symptom or disorder, and increased the actual amount of help provided to others. O'Reilly et al. (2011) tested MHFA with pharmacy students and found similar results. Additionally, after the training, student participants were significantly more confident in their ability to provide pharmaceutical services to consumers with mental illness. When they tailored the training to a specific MHD (schizophrenia or depression), Jorm, Kitchener, Fischer, et al. (2010) found variations in effectiveness of MHFA:

- Beliefs about the effectiveness of depression treatment improved for all participants regardless of whether they received information in a paper manual or through e-learning and continued to remain improved six months after the intervention.
- Beliefs about the effectiveness of schizophrenia treatment improved only for those who received the information through e-learning and was short term (i.e., was not detectable at the six-month follow-up).

Multimedia Campaigns

Changing Minds was a media campaign in England implemented between 1998 and 2003 using websites, campaign videos in cinemas, leaflets for the public and health care professionals, and reading materials for young people for use in school. Mehta et al. (2009) found that Changing Minds was effective in initially decreasing some negative attitudes toward people with mental illness (e.g., mental illness is a lack of self-discipline and willpower), particularly among people who knew someone with a mental illness (Crisp et al., 2005), but that the impact diminished over time and, in fact, rates of stigma in 2003 were higher than in 1998.

Changing Minds was one of three components in England's overall stigmareduction strategy, Time to Change. Time to Change also includes mass physical exercise events and an online resource on mental health and employment. An assessment conducted immediately after the campaign launched found that fewer people said that they had often or very often worried that others would view them unfavorably because they use mental health services (Schneider, Beeley, and Repper, 2011). Although not a direct measure of stigma, Henderson, Evans-Lacko, et al. (2012) assessed whether people with mental illness experienced more or less discrimination in a range of 21 life behaviors (e.g., from their employers, neighbors, landlords). Twelve months after the campaign launched, respondents reported significantly less discriminatory behavior in about one-third of the life behaviors—specifically, they reported fewer instances of being shunned and fewer instances of being discriminated against because of their mental illness by friends, family, social acquaintances, and employers (Henderson, Evans-Lacko, et al., 2012).

Figure 5.1 shows the conceptual model introduced in Chapter Four with the public-level intervention targets described above. The lines in the model leading from public interventions to treatment-seeking behavior are dashed to indicate that we were not able to locate any empirical evidence that these interventions would increase treatment-seeking, although theoretical models suggest that these factors may be important.

Public and Social Context Interventions Tested in Military Populations

We were unable to locate a robust literature base specifically testing mental health stigma-reduction interventions in the military context. However, we did identify five



Family, friends, and unit Social context

Reduced discriminatory behavior toward PWMHDs

Improved public mental health literacy

Intervention outcomes
 Reduced social distance

Figure 5.1 Interventions in the Public and Social Contexts That May Affect Stigma

RAND RR426-5.1

studies focusing specifically on two types of interventions: Battlemind and Trauma Risk Management (TRiM).¹ Both of these interventions intend to train the general population of military service members on how to identify and help a person in emotional distress.

Battlemind encourages early identification of PWMHDs by providing education on resiliency and suicide prevention and offering specific interventions, such as suicide incident-response teams during each deployment phase (i.e., predeployment, deployment, redeployment, reintegration). Adler et al. (2011) compared the effects of the Battlemind debriefing with the effects of a standard stress-education course and found that, for participants with high levels of combat exposure, large-group Battlemind training led to lower stigma levels than stress education did (Adler et al., 2011). Similarly, Castro, Adler, et al. (2012) reported that soldiers' changes in attitudes about the stigma of seeking mental health care were found immediately posttraining, but these changes did not carry over at follow-up, suggesting that the training's impact may be

¹ The Battlemind intervention has been integrated in Comprehensive Soldier Fitness. However, we refer to Battlemind throughout the report because that is the name of the intervention used in published literature.

short term. A descriptive study (Warner, Appenzeller, Parker, et al., 2011) looked more broadly at the Battlemind program's impact on deaths by suicide among active-duty Army soldiers who had deployed. Among soldiers who received Battlemind training, the authors found an annual suicide rate of 16 in 100,000, compared with a theater rate of 24 in 100,000 (Warner, Appenzeller, Parker, et al., 2011). Because of the challenges in studying suicide trends (see Ramchand et al., 2011), it is not clear whether this is a significant decrease in deaths by suicide, so it is difficult to determine the effectiveness of the Battlemind program. There has been no research assessing whether Battlemind influences actual treatment-seeking behavior or adherence to treatment.

TRiM is a 2.5-day training in which mental health practitioners in the UK Royal Navy who have been designated as TRiM practitioners teach nonmedical personnel to offer basic skills in psychological first aid and how to identify people at risk of psychological injury to facilitate early referral. A 2007 study found that TRiM improved participants' attitudes to stress and PTSD, general mental health, and the likelihood that participants would seek help from TRiM practitioners (Gould, Greenberg, and Hetherton, 2007). However, a later randomized controlled trial that used more-rigorous research methods and compared TRiM with usual care found no significant differences in terms of stigma or psychological health (Greenberg, Langston, Everitt, et al., 2010). Although the Gould et al. study showed that participants reported that they would be more likely to seek help from TRiM practitioners, there is no research showing whether TRiM influences whether service members actually seek treatment or adhere to their treatment regimes.

Intervening in the Institutional Context

Little research tests interventions at the institutional level. However, numerous researchers have theorized or discussed possible intervention targets. Some researchers have suggested that *individual factors are not sufficient targets* when seeking to change institutional policies or practices (Hill, 1988; Pincus and Ehrlich, 1999). Although educating key power groups could, in theory, be impactful, *changing policy or legislation* to modify or counter the stigma or discrimination toward PWMHDs is often viewed as a more appropriate solution (Corrigan, 2005).

Corrigan (2004b) has proposed a "target-specific" stigma change model that focuses on the ability of people in power (via their institutional or organizational positions) to influence stigma and discrimination toward PWMHDs. People in power include employers, health care providers, policymakers, landlords, criminal justice professionals, and representatives of the media. Examples of possible intervention strategies include changing law to affect behavior in the workplace and educating people in power within institutional settings in order to counter negative beliefs about mentally ill people—for example, beliefs that they are not capable of working (Corrigan, 2004b). Likewise, Link and Phelan (2001) suggest that targeting beliefs, attitudes, and behaviors of specific people (e.g., employers), alone, is insufficient. They further emphasize the need to change the structural conditions leading to stigma. Stigma-reduction efforts, they argue, must change the "fundamental causes" of stigma—i.e., the attitudes and beliefs of people in power or the societal structures that enable these people to impose their attitudes and beliefs onto others. Arboleda-Flórez and Stuart (2012) offer "legislative reform" as a strategy to prohibit discrimination against PWMHDs and ensure appropriate employment, education, and housing. Moreover, Pescosolido (2008) presents a theoretical model encapsulating a range of micro-, macro-, and mesolevel characteristics that may lead to stigmatization or discrimination, including that which occurs within systems of health care.

In the U.S. military context, Kelley et al. (2014) found that perceived organizational support (i.e., the belief that one's organization values his or her contributions and cares about his or her well-being) may be able to lessen stigma among soldiers returning from combat. Perceived organizational support may also be a protective factor for soldiers. Barnes et al. (2013) found that higher levels of perceived organizational support predicted lower levels of PTSD before, during, and after deployment among service members on a peacekeeping mission. Some researchers have also noted the opportunity for stigma intervention in theater by deploying mental health providers with military units and enabling them to become credible "known entities" among the command and particularly command leadership (Hoyt, 2006). Although this strategy is seeking primarily to decrease barriers to care, this may also affect potentially stigmatizing beliefs, attitudes, and behaviors at the military unit level and is one of the strategies of the Navy's Operational Stress Control and Readiness program (Hoyt, 2006). In addition, government policies concerning the disclosure of treatment information may be another target for stigma intervention. As mentioned previously in this report, the government modified its security-clearance application (required for many military occupations) in 2008 amid research highlighting the salience of career concerns among military service members (Schell and Marshall, 2008). This change enabled service members not to disclose the receipt of mental health counseling for combat-caused disorders or for family problems. To our knowledge, neither the stigma nor discrimination outcomes of either policy change have been measured.

Figure 5.2 shows the conceptual model introduced in Chapter Four with the institutional-level intervention targets described above.

Intervening with People with Mental Health Concerns

To date, the focus of stigma reduction has been on reducing public stigma using the approaches described above. There has been comparatively little research examining the effectiveness of intervention strategies designed to reduce or eliminate self-stigma





(Dickstein et al., 2010; Mittal et al., 2012). Of 14 published studies to date (Luoma et al., 2008; Link, Struening, et al., 2002; McCay et al., 2007; Shin and Lukens, 2002; Alvidrez et al., 2009; Hammer and Vogel, 2010; Griffiths et al., 2004; MacInnes and Lewis, 2008; Knight, Wykes, and Hayward, 2006; Aho-Mustonen et al., 2011; K. Fung, Tsang, and Cheung, 2011; Lucksted et al., 2011; Wade, Post, et al., 2011; Adler et al., 2009), most utilized cognitive techniques (e.g., psychoeducation, cognitive restructuring) to teach people strategies to better control their thoughts, feelings, sensations, and memories. However, some included elements of acceptance and commitment therapy, which focuses on teaching people to notice, accept, and embrace their thoughts, feelings, and memories, rather than actively work to control them. A limited number included more-complex or multimodal interventions, including group discussion, social skills training, goal attainment, and problem-solving skills, in addition to the more-traditional psychoeducation and cognitive behavioral therapy and exercises (K. Fung, Tsang, and Cheung, 2011; Lucksted et al., 2011; Mittal et al., 2012).

Eight of the 14 studies (57 percent) (Luoma et al., 2008; Shin and Lukens, 2002; Hammer and Vogel, 2010; Griffiths et al., 2004; MacInnes and Lewis, 2008; Lucksted et al., 2011; Wade, Post, et al., 2011; Adler et al., 2009) demonstrated significant reductions in self-stigma among those who participated in the intervention. Interventions to reduce self-stigma may be less effective for people with schizophrenia or a psychotic disorder, however; only two of seven studies (28 percent) targeting these populations demonstrated a reduction in self-stigma. Though these types of therapies and interven-

tions hold promise for reducing self-stigma, particularly among people with less severe MHDs, the vast majority of these studies were exploratory in nature with small sample sizes, did not include a comparison group, and did not follow anyone over time to assess whether any observed reductions in self-stigma were maintained postintervention (Dickstein et al., 2010; Mittal et al., 2012). More work is needed to identify the most-effective strategies for reducing self-stigma among PWMHDs.

Figure 5.3 shows the conceptual model introduced in Chapter Four with the individual-level intervention targets described above. The intervention targets currently described in the literature primarily support the development of cognitive coping skills. There is, however, limited evidence that these interventions affect someone's initiation of mental health treatment or retention or adherence to a treatment regime.

Summary of Possible Interventions

Collectively, interventions designed to date have attempted to change and improve the public's knowledge about MHDs and mental health treatment (e.g., mental health literacy) and minimize negative attitudes and behaviors, such as social distancing. Other



Figure 5.3 Individual-Level Interventions That May Affect Stigma

RAND RR426-5.3

interventions have focused more on helping PWMHDs cope with difficult public and institutional contexts and increase someone's likelihood of help-seeking.

Although such strategies may be promising, more research is needed to fully vet the above approaches and build a consistent evidence base among military-connected individuals. It is also important to note that effectiveness across interventions varied based on type of intervention strategy and by type of MHD (e.g., schizophrenia versus PTSD), and several of the interventions discussed had only short-term impacts.

According to our conceptual model, individual interventions to build cognitive coping skills in PWMHDs may be the most direct way to reduce the negative impacts of stigma. However, building the public and PWMHDs' mental health literacy, increasing positive attitudes and reducing negative attitudes toward PWMHDs and mental health treatment, improving self-efficacy and skills to intervene with someone in emotional distress, and decreasing social distancing may help reduce the negative influences of the public context. Within the institutional context, reviewing policies and practices to ensure that they protect PWMHDs from discrimination and are not likely to inadvertently result in discriminatory treatment of PWMHDs may also help reduce stigma.

If the primary goal is to increase the likelihood that someone initiates treatment, stigma-reduction interventions may not be the only choice. Numerous barriers to treatment-seeking have been identified in the literature, many of which pose a more significant barrier than stigma. Although stigma-reduction efforts may increase treatment-seeking, a multipronged approach to removing barriers may be warranted.

Public and Social Contexts

Contact-based programs, such as the National Alliance on Mental Illness' IOOV program, that bring together people in recovery from MHDs and their peers without MHDs aim to improve knowledge about MHDs and mental health treatment, also called mental health literacy, and decrease discriminatory behaviors and social distance between PWMHDs and their peers. Similarly, education and training programs intend to teach participants how to help someone in emotional distress and consequently reduce social distance and improve attitudes toward mental health treatment. These programs also target self-efficacy to provide help to someone with an MHD and encourage individual help-seeking behaviors. Multimedia campaigns primarily intend to improve attitudes toward PWMHDs. Again, we were not able to locate any empirical evidence that these interventions would increase treatment-seeking; however, theoretical models suggest that these factors may be important.

Military Context

There is little research to guide the selection of intervention targets. Perceived organizational support (i.e., belief that one's organization values his or her contributions and cares about his or her well-being) may be able to lessen stigma among soldiers returning from combat and could serve as a protective factor for soldiers (Kelley et al., 2014; Barnes et al., 2013). Researchers have also suggested that educating key power groups and changing policy or legislation to modify or counter the stigma or discrimination toward PWMHDs could, in theory, be impactful by reducing discriminatory behavior toward PWMHDs and improving the mental health literacy of military leaders who often set the climate within units and the military institution as a whole (Corrigan, 2005). Although we were not able to locate any empirical evidence that these changes affect stigma, theoretical models suggest that these changes may be important.

People with Mental Health Disorders

Cognitive techniques (e.g., psychoeducation, cognitive restructuring) to teach people strategies to better control or accept their thoughts, feelings, sensations, and memories have been shown to be effective at reducing stigma. Evidence supports that these techniques aid in the development of cognitive coping skills. There is, however, limited evidence that these interventions affect someone's likelihood of initiating or remaining in mental health treatment. It is also important to note that effectiveness across interventions varied based on type of intervention strategy and by type of MHD (e.g., schizophrenia versus PTSD), and several of the interventions discussed demonstrated only short-term impacts, or evaluators measured impacts only over the short term.

In summary, our literature review revealed that individual interventions to build cognitive coping skills in PWMHDs may be the most direct way to reduce the negative impacts of stigma. However, building the public mental health literacy, improving positive attitudes and reducing negative attitudes toward PWMHDs and mental health treatment, improving self-efficacy and skills to intervene with someone in emotional distress, and decreasing social distancing may help reduce the negative influences of the public context. Within the institutional context, reviewing policies and practices to ensure that they protect PWMHDs from discrimination and are not likely to inadvertently result in discriminatory treatment of PWMHDs may also help reduce stigma.

U.S. Department of Defense Programs to Reduce Mental Health Stigma

Despite the availability of a wide range of evidence-based treatments for MHDs, the proportion of service members who seek needed treatment remains low. In response, DoD and individual branches of service have made a concerted effort to promote treatment-seeking through specific programs to reduce stigma, as well as through a widespread culture shift in which mental health is discussed in the context of readiness and resilience and in which help-seeking is redefined as a sign of strength. These approaches are generally aligned with the promising approaches described in Chapter Five and may have contributed, at least in part, to the declining trends in perceived stigma described in Chapter Three. This chapter provides an overview of this culture shift, highlights the success of this approach for reducing public stigma, and describes programs that have activities designed to more specifically address mental health stigma. It concludes with a discussion of approaches that may help augment stigma-reduction efforts in the military and potential lessons learned from civilian approaches.

The Primary Approach Is a Universal Culture Shift to Promote Mental Health and Treatment-Seeking

In August 2011, DoD mandated that all services "foster a culture of support in the provision of mental healthcare and voluntarily sought substance abuse education to military personnel in order to dispel the stigma of seeking mental healthcare and/ or substance misuse education services" (DoD Instruction [DoDI] 6490.08, 2011, p. 209). Individual services recognize psychological health as an essential component to total force fitness and readiness and have acted on the sentiments in this command notification "by encouraging prevention, early intervention, and help-seeking behaviors" (Nathan, 2012, p. 10). For example, Army Regulation (AR) 600-63 mandates that leaders receive training on how to create an "atmosphere within their commands that reduces stigma and encourages help-seeking behavior" (pp. 64–65). Structural and logistical barriers to receiving mental health care have also been addressed through policies requiring after-duty hours for mental health care, adaptations to psychological health support across traditional and nontraditional health care systems, and imple-

mentation of campaigns aimed at destigmatizing behavior seeking. Chapter Seven contains a more detailed discussion of military policies related to stigma reduction.

Within the military, the culture shift around the acceptability of MHDs and support for mental health treatment-seeking is apparent in some contexts, which is to be commended given that such mandates have been in existence for a relatively short period of time. In addition to the DoD-wide and service-specific policies described above, service members and, in many cases, military leadership are coming forward to tell their stories in different venues and as features of specific programs and media campaigns in the hopes that other service members may relate and see the benefits of mental health treatment. Such efforts also help to dispel the myth that seeking treatment will harm one's military career. This culture shift is also apparent in the large number of programs that educate participants on mental health, review signs and symptoms of MHDs, share mental health resources (e.g., where can one go for help), or encourage treatment-seeking more generally. Examples of these programs include Comprehensive Soldier and Family Fitness, the COSC programs of the Marine Corps and Navy, and the Yellow Ribbon program. These programs collectively contribute to a necessary shift in perceptions about the value of mental health treatment and a reduction in public stigma.

This collective culture shift across DoD programs and policies is a critically important component of a comprehensive strategy to share key messages about mental health, reduce stigma, and increase treatment-seeking. Studies have demonstrated, for example, that culture shapes someone's willingness to seek treatment, as well as his or her feelings of self-worth (Pescosolido, 2013). It has also been argued that more targeted programs designed to promote treatment-seeking among those in need of services, or programs designed to combat stigma among those currently in treatment, may be less successful and positive outcomes may not be sustained if someone seeking treatment consistently receives negative or counterproductive messages from the community (Pescosolido, 2013). Though it is difficult to attribute the decline in public stigma over time to a single program or effort, evidence supporting the effectiveness of this culture shift overall to reduce public stigma can be found in the MHAT data (see Chapter Three), in which opinions about the negative implications of treatmentseeking have generally declined over time.

Though we acknowledge the critical importance of this culture shift and of the wide range of programs and activities that contribute to it to reduce public stigma, identifying every program that referenced mental health or promoted treatment-seeking for inclusion in this report was beyond the scope of this project. Given that the culture shift is occurring and appears to be relatively successful at reducing public stigma, it became apparent that a closer look at some of the more-targeted stigma-reduction programs was warranted and may be beneficial to DoD. It is worth noting that, though public stigma has declined overall over time, the perceived negative implications of treatment-seeking remain about twice as high for people who screen positive

for MHDs (e.g., who would benefit from treatment) than for those who did not screen positive. Findings such as these suggest the need for more targeted interventions and programs that not only promote treatment-seeking but also reduce self-stigma and encourage continued engagement with treatment once initiated. Without such supports, people who seek services may find it difficult to stay in treatment.

In Figure 6.1, we adapt the Combat Operational Stress Continuum to help conceptualize the portfolio of DoD programs that address stigma and the approaches used to support a culture shift (discussed above) to promote treatment-seeking. The left side of the blue bar represents the programs that take a broader approach, reaching a much larger population but having less of an emphasis on stigma-reduction strategies specifically. Programs and policies on this end of the continuum are important for spreading messages about mental health and the value of treatment-seeking and contribute to the culture shift described above.

Toward the right side of the blue bar are programs that more specifically target stigma, including self-stigma among those currently in treatment. These programs are important because they not only may help to promote treatment-seeking among those with highest need but also can help people cope with any public or self-stigma received because of treatment-seeking, resulting in a higher likelihood that someone will stay in treatment. Targeted programs, however, are not limited to people with mental health concerns but may include, for example, programs to improve knowledge and reduce stigmatizing attitudes and behaviors among service members who come in contact with or oversee PWMHDs. In general, these programs have a smaller reach and place a greater emphasis on stigma reduction than education about mental health and available resources more broadly. Programs falling in the middle of the bar take a more

Figure 6.1

Approaches to Address Stigma, Promote Treatment-Seeking, and Encourage Continued Engagement with Treatment Once Initiated

Ready (Green)		Reacting (Yellow)	Injured (Orange))	lll (Red)	
 Optimal functioning Adaptive coping Well trained and prepared Fit, tough, and focused Behaving ethically Cohesive units, ready families 	or imp • Anxiou	r transient distress airment ıs or irritable ed self-control ocus	 More severe or persistent distress or impairment Leaves lasting evidence Loss of control or moral values Persistent shame, guilt, or blame 		 Persistent and disabling distress or loss of function Diagnosable clinical disorder Symptoms and disability persist over many weeks Symptoms and disability get worse over time 	
Emphasis on stigma Population reached						
Universal		Indicated			Targeted	

RAND RR426-6.1

indicated approach, in which the target audience may include people who have mental health concerns but have yet to seek treatment.¹

More-Indicated or Targeted Military Stigma-Reduction Programs Are Also Being Implemented

We sought to identify and interview representatives from all DoD-sponsored programs that directly supported stigma reduction in the military. Appendix E describes detailed methods used in our approach, as well as program summaries for all included programs.

Identifying and Interviewing Representatives from Stigma-Reduction Programs

RAND staff identified 26 programs and conducted 30- to 60-minute interviews with each program to gain a deeper understanding of their activities in stigma reduction to determine whether they met our inclusion criteria:

- The program actively focuses on stigma reduction.
- The program focuses on mental health and not treatment for drug or alcohol problems.
- The activity is sponsored or funded by DoD.
- The activity has a target audience that includes, at a minimum, active-duty, National Guard, or Reserve Component service members or their family members.
- The activity conducts its efforts either in theater or out of theater and was in operation at some point between January 2013 and July 2013.

Telephone interviews were conducted between January 2012 and July 2013 and covered the following topics:

- background of program (mission and goals, population served, length of time in existence, funding source, staffing model and program leadership)
- program alignment with theoretical model or definition of *stigma*
- detailed discussions of components or activities of program designed to address stigma (e.g., how does activity affect stigma, mode of delivery, dose or intensity of activity)
- current enrollment of program, capacity
- program evaluation, evidence of effectiveness.

 $^{^{1}}$ The mission of indicated programs is to identify individuals who are exhibiting early warning signs or symptoms of MHDs and to target them with special programs (Institute of Medicine, 1994).

Of 26 programs, five met our inclusion criteria. Table D.2 in Appendix D provides a summary of which programs we excluded and the rationale for exclusion:

- Five were excluded because they were not or are no longer DoD-funded.
- Three were excluded because they were no longer in existence or were no longer stand-alone programs (e.g., subsumed by another program).
- One program was excluded because it focused exclusively on alcohol treatment.²
- Nine programs were excluded because they did not or do not directly or primarily target stigma reduction. For these programs, the only activities related to stigma were encouraging treatment as part of broader training, developing communication plans with the goal of normalizing treatment-seeking, describing the availability of mental health services, and promising privacy to service members. For two reasons, we decided to exclude these programs from further review and focus instead on programs that directly and actively addressed stigma:
 - Because these types of activities are implemented across a range of programs, activities, and policies to promote a culture shift around treatment-seeking, it is very difficult to ensure a complete census of all relevant efforts. Identification of every program that mentioned mental health in some capacity was beyond the scope of our project.
 - A recognition remains, as evidenced by the J-MHAT 7 and other data, that these more universal approaches will create some incremental change and that more-targeted programs may be warranted to reach those who are in need of treatment and less likely to respond to more universal approaches.

Because mental health stigma may emerge and present differently in different settings, maintaining the relationship- and context-specific understanding of stigma is critical to our analysis. To ensure that we attend to context, we describe our findings, highlighting program activities that contribute to stigma reduction in the broader military context (i.e., institutional context), the public context (including social networks), and the context of working directly with PWMHDs.

U.S. Department of Defense Programs and Policies Target the Military Treatment System

See Chapter Eight for an analysis of DoD-wide and service-specific policies that may contribute to or help reduce mental health stigma. However, we highlight one program that targets the military treatment system. The Embedded Behavioral Health (EBH) program employs 13 behavioral health and support personnel per brigade combat team, effectively changing the support structure of the team. The program establishes working relationships between behavioral health providers and key battalion personnel. As

 $^{^2}$ For the purposes of this report, we focused only on mental health, not substance abuse.

a result, program staff can observe trends in attitudes toward behavioral health within a given unit and can prompt an intervention if prejudicial attitudes or discriminatory behaviors are found to exist. Given that this program results in a change to the staffing and support structure of a brigade combat team, it is included within the military context. Of note, the program has other activities that contribute to stigma reduction in the public context, described in the next section.

U.S. Department of Defense Programs Target Military Norms, Culture, and Social Context

We identified five programs that specifically target the knowledge, attitudes, and beliefs that service members, social networks (e.g., family and friends), and providers hold about service members with MHDs and about seeking mental health treatment. Prejudicial negative knowledge, attitudes, and beliefs about PWMHDs can translate into discriminatory behavior, including withholding help from PWMHDs, avoiding them entirely, segregating PWMHDs from those without, and forcing criminal justice actions against them.

EBH, mentioned in the previous section, includes activities that encourage soldiers to talk with peers about behavioral health issues and actively prompts commanders to be proponents of behavioral health and treatment-seeking.

Breaking the Stigma is a U.S. Army Special Operations Command (USASOC) command-driven stigma-reduction training program that aims to build resiliency and optimize performance by reinforcing the importance of maintaining psychological fitness and seeking mental health care when needed. The training is built around a 24-minute video in which a series of senior-ranking and respected special operations forces (SOF) personnel share their stories of dealing with combat- and deployment-related issues and the consequences (to their units, careers, and families) of seeking or not seeking help. The training consists of viewing the video, a presentation by two of the video participants, an overview of available resources, a briefing from unit providers, and a briefing from commanders (who share their own connections and experiences).

The Military Pathways program seeks to promote the improvement of mental health though a range of activities. The program offers a web portal that includes selfdirected online mental health screenings, a learning section, a military mental health blog, and a postassessment resource and referral section. The program also conducts an educational program called Signs of Suicide (SOS) for military middle and high school children attending school within the Department of Defense Education Activity (DoDEA) system. Military Pathways also disseminates some educational and promotional materials at a range of events and national screening days. In addition to these activities, the program has some activities more targeted toward people in need of mental health services (a description of these activities is included in the next section).

Afterdeployment.org delivers web-based applications to the military community targeting psychological health and traumatic brain injury (TBI). At the time of this

writing (September 2013), stigma was a separate topic page on its website where a reader could find an interactive resource booklet that addresses myths about MHDs and treatment, gives suggestions for changing misconceptions and provides information on how to overcome prejudicial attitudes in oneself and others. Afterdeployment. org also contains eight videos of people sharing their personal stories about their experiences with about toward help-seeking. Program staff reported plans to fold stigma into some clinical topics on the website so that it has a more pervasive presence, rather than isolating it to a single topic page on the website.

The Real Warriors Campaign is a multimedia public awareness campaign designed to encourage help-seeking behavior among service members, veterans, and military families. The campaign's core messages are specifically designed to address the barriers and motivators to seeking care for psychological health concerns that exist within the military community. The core messages let service members, veterans, and their families know the following key things:

- Reaching out for help is a sign of strength that benefits the individual, his or her family, and the entire military community.
- It is possible to seek care for psychological health concerns and maintain a successful military or civilian career.
- Warriors are not alone in coping with psychological health concerns, and every service member, veteran, and his or her family members should feel comfortable reaching out to his or her unit, chain of command, fellow warriors, and community resources for support.
- Experiencing psychological stress as a result of deployment is common, and successful care and positive outcomes are greatly assisted by early intervention.

The campaign features video profiles of real service members of varying ranks, services, and components candidly sharing their stories of coping with and successfully seeking care for psychological health concerns. It also features perspectives from leaders, peers, and family members. The Real Warriors Campaign targets the public context through its outreach efforts and dissemination (e.g., social and traditional media with board audience reach leaders, families, providers, and the public at large).

U.S. Department of Defense Programs Target People with Mental Health Disorders

The five programs mentioned above that address public context also address the individual context to some extent. Resources, videos, and other educational efforts that may be useful for shaping the attitudes and perceptions of mental health and treatment-seeking of the public in general may also be helpful in shaping similar attitudes and perceptions among people with mental health concerns. Though programs may address both public and individual contexts, some activities, such as online videos of success stories, and key messages around seeking help as a sign of strength that are

offered by the Real Warriors Campaign are designed to more directly address stigma at the individual level.

Military Pathways, for example, offers Video Doctor, an interactive video program designed to walk users through stages of change to assist people in seeking treatment or in caring for themselves. Military Pathways also includes a video, "A Different Kind of Courage," in conjunction with a two-workbook curriculum. The video features current and former service members and spouses who share their personal stories of seeking treatment for mental health issues.

In addition to the online videos described earlier, Afterdeployment.org offers a self-administered assessment to identify one's perceived comfort level with seeking treatment and, as mentioned above, has an interactive resource booklet that includes information on how to overcome perceptions of stigma.

Most Stigma Programs Are Department-Wide

Though the culture shift around perceptions of mental health and acceptability of treatment-seeking is occurring DoD-wide, it is also likely that there will be some service variability in their relative emphases on stigma reduction and their support of more-targeted stigma-reduction programs specifically. Of the five programs identified, three are DoD-wide: Military Pathways, Afterdeployment.org, and the Real Warriors Campaign. The remaining two, Breaking the Stigma and EBH, are Army programs. We did not identify targeted stigma-reduction programs within the Navy, Marine Corps, or Air Force, though we do acknowledge the importance of more-widespread education and messaging around mental health and treatment-seeking and the support of people seeking mental health care.

Few Stigma-Reduction Programs Are Being Evaluated

The programs above represent an important step forward in an overall stigma-reduction strategy for DoD. This portfolio of programs could be further strengthened, however, by further building the evidence base for their effectiveness—in both reach and impact. We found, for example, that the evaluations conducted on many of these programs lacked the rigor, comprehensiveness, or specificity needed to determine whether these programs are effective. Although each of the five programs collects data on metrics, these metrics are largely process measures, such as website metrics and satisfaction surveys. These are helpful in ensuring that the programs are being used and resonate with the target audience, but they do not provide information on whether the program is effective in reducing stigma or increasing treatment-seeking among those in need of services. Strengthening the evidence base for these programs will provide helpful information about which of these programs should be scaled up or implemented in other locations.

Suggestions to Augment the U.S. Department of Defense's Approach to Reducing Mental Health Stigma

Though the activities and programs identified above likely contribute to stigma reduction in the military, there are still program areas that DoD may consider augmenting to improve its approach to reducing mental health stigma. To identify these gaps, we compared the programs used by DoD and the portfolio of evidence-based programs we identified as part of our literature review in Chapter Five. Appendix E contains a detailed description of our literature review findings on evidence-based stigmareduction programs.

Lessons Learned from Effective Civilian Programs May Help Improve the U.S. Department of Defense Programs Targeting Military Norms, Culture, and Social Context

Though all five programs addressed stigma reduction in the public context, lessons can be learned from approaches in the civilian population to strengthen such efforts. Perhaps one of the best-studied and best-regarded stigma-reduction programs in the civilian context is IOOV, developed by the National Alliance on Mental Illness. Program features include two PWMHDs interacting with participants for 90 minutes to share their initial experiences with MHDs and how they came to terms with those conditions, effective treatments and daily coping mechanisms, and their successes, hopes, and dreams. The in-person interaction is accompanied by standardized videos or PowerPoint presentations that share more educational information. Although there are numerous examples across DoD of people coming forward to tell their stories either in person or in video format (e.g., Afterdeployment.org, Real Warriors Campaign), these videos or encounters are often brief, lasting only a few minutes. A unique aspect of IOOV, and perhaps why it is effective, is that the two people telling their stories are with the participants for 90 minutes and provide the education themselves such that, by the end of the program, participants' perceptions of PWMHDs have changed as they realize that people with mental health concerns are not stereotypically violent, childish, or unpredictable but are people who are of value to society. Although people repeatedly hear messaging about the value of a resilient and ready service member to the military, the implication is that, if a service member is not ready, he or she is of less value to the organization. Messaging is less clear around what value a PWMHD is to the military. Existing programs may benefit from strengthening messaging around success stories of people currently in treatment to help debunk some of the morecommon myths around mental health recovery.

In addition to efforts that are designed to educate the public generally, other stigma-reduction programs have focused on training people to help someone in emotional distress. MHFA is a nine-hour training to teach participants how to help someone who may be exhibiting suicidal thoughts or behaviors, an acute stress reaction,
panic attacks, or acute psychotic behaviors. Several studies (Kitchener and Jorm, 2002, 2004, 2006) found that MHFA decreased social distance, improved perceptions of the effectiveness of mental health treatment, increased confidence in providing help to someone with a mental health symptom or disorder, and increased the actual amount of help provided to others. The Royal Navy adapted MHFA to create TRiM, though findings to date have been mixed. As of October 2013, a study was in progress to test the effectiveness of MHFA among U.S. military populations. Though findings have not yet been released, if effective, such programs may provide a fresh perspective on stigma-reduction strategies.

Finally, there may be ways to strengthen the military's antistigma campaign, the Real Warriors Campaign, by drawing on examples from other campaigns that have sought to reduce stigma. Time to Change, for example, was a large social marketing campaign that occurred in England between 2009 and 2011. The campaign deployed messages intended to combat myths about PWMHDs, change attitudes toward PWMHDs, and change behavior toward PWMHDs. Messages were disseminated through many channels, including radio and television ads, websites, online videos, and posters. The campaign is regarded as successful because it resulted in positive shifts in attitudes toward PWMHDs and intentions to be more socially inclusive of PWMHDs across the general population of England (Evans-Lacko, Henderson, and Thornicroft, 2013). Knowledge about mental health and reported social interaction with PWMHDs did not change in the general population. PWMHDs in England reported experiencing less discrimination during the campaign time period (Corker et al., 2013). The Real Warriors Campaign and other stigma-reduction campaigns may be strengthened by adapting messages shown to be successful in other contexts.

Programs Targeting the Individual Context May Help Broaden the U.S. Department of Defense's Current Approach

To date, there has been comparatively little research examining the effectiveness of intervention strategies designed to reduce or eliminate self-stigma (Dickstein et al., 2010; Mittal et al., 2012). Although reducing self-stigma may be relevant for initiation of treatment, it is also important to encourage continued engagement in care once the person has reached out. Most civilian approaches to overcoming self-stigma for people engaged in care have included cognitive techniques (e.g., psychoeducation, cognitive restructuring) to teach participants strategies to better control their thoughts, feelings, sensations, and memories. Some included elements of acceptance and commitment therapy focus on teaching participants to notice, accept, and embrace their thoughts, feelings, feelings, and memories rather than actively work to control them. A limited number of interventions include more-complex or multimodal interventions, including group discussion, social skills training, goal attainment, and problem-solving skills, in addition to the more-traditional psychoeducation and cognitive behavioral therapy and exercises (K. Fung, Tsang, and Cheung, 2011; Lucksted et al., 2011; Mittal et al.,

2012). However, new models for intervention are emerging (e.g., prototype/willingness model by Hammer and Vogel, 2013) that try to change willingness to seek care (a spontaneous decision) in addition to the attitudes and norms that predict intentions to seek care. Although military mental health providers may be providing one or more of these cognitive techniques or approaches to their patients to support their recovery, there is not a standardized program or intervention to address these concerns. Though this may not be a central component of an overall DoD strategy to stigma reduction, it is an important one if the ultimate goal is not just initiation of treatment but successful resolution of the MHD and return to duty.

Conclusion

It is clear that a concerted effort is being made within DoD to reduce stigma around mental health with the goal of increasing treatment-seeking among those who would benefit. In recent years, there has been a culture shift in which mental health is discussed more openly, people are coming forward to tell their personal stories of mental health and the benefits of treatment, and a vast array of programs provide high-level education about mental health and discuss the availability of treatment. In addition, we identified five programs that include more-targeted activities designed to reduce stigma in the broader military and public contexts and with PWMHDs. This complement of activities generally aligns with promising practices described in Chapter Five and may have contributed in part to the declines in perceived stigma described in Chapter Three. Though these programs fill important needs, gaps remain in terms of evaluation and establishment of program effectiveness. Although there are models of stigma-reduction programs in the civilian population that could be drawn on to either supplement or bolster existing programs, it is clear from the paucity of research and lack of effective programs that there is no "magic bullet" and that a multipronged strategy is needed. Despite some examples, there is not a lot in the civilian context on which to build. However, DoD may wish to explore ways in which programs or component pieces of those programs may be adopted or adapted to a military setting to augment their ongoing efforts as the military improves efficiencies across medical commands.

In this chapter, we aim to explore the DoD policy landscape as it pertains to mental health stigma. To identify the institutional policies that might either contribute to or reduce stigma toward service members with mental health challenges or toward those seeking treatment, we conducted a systematic assessment of policies in DoD and the services using a prospective policy-analysis methodology. We first describe the methods of the analysis, followed by a discussion of an important policy tension that emerged from our analysis—the tension between the need for commanders to assess the fitness of service members under their command and the need for privacy among service members with MHDs. Then, we present a summary of the methods and a summary of the institutional policies, highlighting our key findings. Detailed methods of the policy identification and review process are available in Appendix F. A full list of DoD and service-specific policies related to stigma appears in Appendix G.

Approach to Prospective Policy Analysis

To assist DoD in its mission of reducing mental health stigma and promoting treatmentseeking for service members who need it, we opted to take an evidence-informed approach to analyzing DoD policies related to mental health stigma (Lavis et al., 2009); see Figure 7.1. To develop a list of recommended policy priorities, our approach focuses on using evidence from two sources: (1) our systematic assessment of policies (outlined in this chapter) and (2) the literature review exploring stigma, stigma-reduction strategies, and the relationship between stigma and treatment-seeking. Experts in mental health stigma and military mental health then reviewed and rated this list of recommended priorities. Experts rated them on their importance to DoD and their validity (i.e., whether there was evidence to suggest that, if implemented, the action prioritized would reduce mental health stigma). Appendix I contains detailed description of the expert panel methods and describes the expertise of each panel member, as well as the final ratings of each priority. The final priority policy recommendations and the shortand long-term steps to achieve these priorities are described in more detail in Chapter



Figure 7.1 Evidence-Informed Policy-Analysis Process

Eight. We hope that these policy priorities are useful in helping DoD to make wellinformed policy decisions.

Literature Review

Although programs targeting individuals can be one effective strategy for reducing stigma and discrimination (see Chapter Six), our literature review revealed that policy change is a tool for countering stigma toward PWMHDs (Corrigan, 2005). Changing policies is one way to change the structural conditions that can lead to stigma and discrimination within institutions (Link and Phelan, 2001).

Systematic Assessment of Policies

In order to determine which DoD policies might be relevant to mental health stigma, we began with a list of DoD policies provided by DCoE. These policies were determined to be potentially relevant to stigma because they contained words related to stigma or mental health (see Appendix G for the full list). We then reviewed each policy to determine whether it was actually relevant to mental health (e.g., some policies contained the term *discrimination* but about discriminating between stimuli on a test, not about discrimination because of mental health). If the policy was relevant to mental health, we did a more thorough content analysis of the policy to identify key policy issues that could have an impact on stigma by either contributing to or reducing stigma. Content from each policy was assigned to the following ten categories, if applicable, which were generated based on our literature review:

- Six categories focused on policies that may *reduce* military mental health stigma.
 - mandating a stigma-reduction intervention, such as education, training, or a media campaign (see the literature review for a description of stigma-reduction interventions)

- mandating that mental health providers utilize a psychotherapeutic approach (e.g., cognitive restructuring) for all service members seeking treatment to improve their ability to cope with stigma
- ascribing **consequences for behavior that is prejudicial or discriminatory** against PWMHDs. These could include personnel actions or legal sanctions.
- mandating that service members with MHDs be treated no differently from those without MHDs. Some examples of this would be policies that mandate that service members with MHDs are to stay with their units and are not prohibited from deploying.
- making the process for accessing mental health care less stigmatizing. This could include offering confidential care or placing mental health providers in less distinct locations.
- protecting the privacy or confidentiality of service members' mental health information.
- Four categories focused on policies that may *contribute* to military mental health stigma.
 - using terminology that is outside the range of standard psychiatric or psychological practice and may be considered to negatively portray PWMHDs (e.g., "bizarre," "temper tantrums," or "childish outbursts")
 - prohibiting certain actions of PWMHDs (e.g., exclusion from certain roles or specialties) or implying incompetence of PWMHDs
 - mandating or implying a lack of privacy or confidentiality of one's mental health information
 - involving **non-mental health professionals in determining mental fitness** or interpreting the implications of mental health symptoms or disorders.

We also identified a set of policies related to **mental health screening** in which implications for stigma were unclear. Mandatory screening can be beneficial if it can promote early intervention for those who may display symptoms. However, if screening or evaluation is perceived as a punishment or means for retaliation, then it could be perceived as stigmatizing and may result in negative mental health outcomes.

Policies could have provisions that fell across multiple categories and provisions that both contribute to and reduce stigma. In total, we identified 323 unique policies with implications for stigma. It is important to note that none of the DoD policies mentioned in this chapter has been evaluated to determine its impact on stigma, and it was outside the scope of this study to conduct comprehensive outcome evaluations for each policy.

One hundred twenty unique policies could reduce stigma. The numbers of policies in Table 7.1 do not sum to 120 because some policies had multiple implications for stigma reduction (i.e., one policy could have some aspects that could reduce stigma but also have aspects that contribute to it).

Policy Owner	Mandates Stigma- Reduction Intervention	Mandates Psychotherapeutic Approach	c Prejudice or			
DoD	2	0	3	8	11	12
Army	5	0	2	11	10	7
Navy	1	0	2	3	3	4
BUMED	0	0	1	1	1	3
Marine Corps	2	0	0	9	4	0
Air Force	3	0	0	4	4	13
National Guard	0	0	0	0	2	0
Total	13	0	8	36	35	39

Table 7.1 Number of Policies That May Reduce Stigma

NOTE: The numbers of policies that may contribute to or reduce stigma are not discrete. Some policies may include provisions that could both contribute to and reduce stigma. BUMED = U.S. Navy Bureau of Medicine and Surgery.

Two hundred three policies may contribute to stigma. Table 7.2 illustrates the number of policies that could potentially increase the stigmatization of PWMHDs,

	-	5		
Policy Owner	Uses Negative Terminology	Prohibits Actions or Implies Incompetence	Limits Privacy	Has Non–Mental Health Professionals Determine Fitness
DoD	1	21	14	1
Army	4	28	7	10
Navy	0	4	10	8
BUMED	0	1	2	2
Marine Corps	1	23	0	4
Air Force	8	29	16	5
National Guard	0	3	5	1
Total	14	109	54	32

Table 7.2 Number of Policies That May Contribute to Stigma

NOTE: The numbers of policies that may contribute to or reduce stigma are not discrete. Some policies may include provisions that could both contribute to and reduce stigma.

based on our policy assessment. It is important to note that the numbers of policies in the table do not sum to 203 because some policies had multiple implications for stigma.

Key Findings from the Content Analysis of U.S. Department of Defense Policies

We conducted a detailed content analysis of DoD policies in each of the ten categories in order to summarize similarities and differences across policies that represent trends in DoD policies or areas with clear disagreement or ambiguities. Informed by that content analysis, we identified six key findings:

- Key tensions exist between the privacy of service members seeking mental health treatment and the need for commanders to assess unit fitness.
- Despite the presence of equal-opportunity policies, wide variability and ambiguity in policies that prohibit service members with MHDs from career opportunities may inadvertently create opportunities for discrimination.
- Policies support universal educational stigma-reduction programs but not moretargeted programs for those in (or in need of) mental health treatment.
- Some policies could potentially reinforce stereotypes through the use of negative terminology.
- Other policies may expose service members to stigma or discrimination because they allow non-mental health professionals to determine mental health fitness.
- Mental health screening is beneficial for early intervention but, if used improperly, may be stigmatizing and have negative effects on mental health (e.g., reduce self-esteem).

In the rest of this section, we discuss each of these findings in detail, referencing specific policies as examples.

Key Tensions Exist Between the Privacy of Service Members Seeking Mental Health Treatment and the Need for Commanders to Assess Unit Fitness

Our analysis identified a key policy tension between the need for commanders to assess the fitness of service members under their command and the need for privacy among service members with MHDs. The nature of the military requires commanders to be aware of factors that affect the ability of service members under their command to execute their required duties. However, one key strategy for reducing stigma as a barrier to care is to allow those with MHDs to seek care privately, under the premise that seeking care in private reduces the likelihood of being exposed to others' stigmatizing attitudes and behavior. Several DoD policies, detailed in this section, illustrate this tension between protecting privacy and keeping commanders appropriately informed.

Military Rule of Evidence (Mil. R. Evid.) 513 details psychotherapist/patient privilege in the military court system. The rule states,

A patient has a privilege to refuse to disclose and to prevent any other person from disclosing a confidential communication made between the patient and a psychotherapist or an assistant to the psychotherapist, in a case arising under the UCMJ [Uniform Code of Military Justice], if such communication was made for the purpose of facilitating diagnosis or treatment of the patient's mental or emotional condition.

However, the following exceptions to this privilege exist (Mil. R. Evid. 513):

1. when the patient is dead; 2. when the communication is evidence of child abuse or of neglect, or in a proceeding in which one spouse is charged with a crime against a child of either spouse; 3. when federal law, state law, or service regulation imposes a duty to report information contained in a communication; 4. when a psychotherapist or assistant to a psychotherapist believes that a patient's mental or emotional condition makes the patient a danger to any person, including the patient; 5. if the communication clearly contemplated the future commission of a fraud or crime or if the services of the psychotherapist are sought or obtained to enable or aid anyone to commit or plan to commit what the patient knew or reasonably should have known to be a crime or fraud; 6. when necessary to ensure the safety and security of military personnel, military dependents, military property, classified information, or the accomplishment of a military mission; 7. when an accused offers statements or other evidence concerning his [or her] mental condition in defense, extenuation, or mitigation, under circumstances not covered by [Rule of Courts-Martial] 706 or Mil. R. Evid. 302. In such situations, the military judge may, upon motion, order disclosure of any statement made by the accused to a psychotherapist as may be necessary in the interests of justice; or 8. when admission or disclosure of a communication is constitutionally required.

Note that this policy applies only to service members who have been court-martialed.

Although most states specify that mental health professionals have a "duty to warn" or "duty to report" if a client or patient poses an imminent danger to him- or herself or others (i.e., exception 4), exception 6 is unique to the military context and provides a broader set of circumstances in which psychotherapist/patient privilege does not hold. Mil. R. Evid. 513 allows for some confidentiality of patients' interactions with their psychotherapists but provides an exception for when the patient disclosed information that could harm military resources or missions.

Mental health professional/patient privilege is further discussed in Air Force Instruction (AFI) 44-109 ("Mental Health, Confidentiality, and Military Law"), which explicitly acknowledges that the policy is intended to balance "the commander's need to know the mental well being of members of his/her command with the mental health patient's need for confidentiality and privacy." This policy reiterates Mil. R. Evid. 513 for cases initiated under the UCMJ but goes further in specifying the Air Force Limited Privilege Suicide Prevention (LPSP) program. The objective of the LPSP program

is to "identify and treat those Air Force members who, because of the stress of impending disciplinary action under the Uniform Code of Military Justice (UCMJ), pose a genuine risk of suicide" (AFI 44-109). Under the program, any official involved in the impending disciplinary action who has a "good faith belief" that the service member in question is at risk of attempting suicide can notify the service member's commander that the service member "be referred for a mental health evaluation and possible placement in the LPSP Program." If a mental health professional conducts an evaluation and believes that there is a suicide risk, the service member is placed in the program and receives "limited protection" (that is, limited ability to invoke the privilege in military court) until a mental health professional believes that there is no longer a continuing risk of suicide. Limited protection refers to "information revealed in, or generated by [the service member's] clinical relationship with [mental health professionals]. Such information may not be used in the existing or any future UCMJ action or when weighing characterization of service in a separation." The parties to whom mental professionals can disclose case-file information for a service member in the LPSP program are the following (AFI 44-109):

Other medical personnel directly engaged in evaluating and treating program participants. This would include [mental health] professional staff at other facilities to which the member may be referred.

VA [U.S. Department of Veterans Affairs] treatment personnel when members are transferred directly to a VA facility.

The confinement facility commander when members are transferred to a confinement facility as a result of an ongoing court-martial.

Other authorized personnel with a need to know in the official performance of their duties. . . . However, if the disclosure is for the purpose of a criminal investigation or proceeding under the UCMJ, the privilege in Military Rule of Evidence 513 . . . will preclude disclosure unless an exception applies. [Mental health] professionals should consult with the staff judge advocate before any release made under this provision.

The policy also refers to a later section indicating when a mental health professional is required to contact the service member's commander. These circumstances are the following (AFI 44-109):

In the [mental health professional's] opinion, the member is a danger to self or others, or poses a threat to security.

The member is admitted to or discharged from a mental health unit or ward, or when a member is referred for admission to a medical unit by [a mental health

professional] for a mental health related concern (e.g., detoxification, treatment of a self-inflicted injury, assessment of neurological impairment, etc.);

In the [mental health professional's] opinion, the member's mental status has deteriorated to the degree that it may significantly affect work or family functioning. (NOTE: "Significantly affect" is defined as posing a risk to self, others, property, security, or the accomplishment of the military mission.)

The [mental health professional] suspects the existence of family maltreatment, substance abuse, or other child abuse. (Notification to [the] Family Advocacy [Program] is also required IAW [in accordance with] AFI 40-301, *Family Advocacy*)

As with Mil. R. Evid. 513, this policy aims to balance the service member's need for confidentiality with the military's need to mitigate effects of the service member's MHD on military readiness.

DoDI 6490.08 ("Command Notification Requirements to Dispel Stigma in Providing Mental Health Care to Service Members") aims to dispel stigma through the protection of the privacy of those seeking mental health care by requiring that care not be reported to a service member's commander unless a specific duty-to-report event occurs. Although this protects privacy, it also indicates that there are indeed circumstances in which service members' mental health status and treatment-seeking are not kept private. These circumstances are as follows (DoDI 6490.08):

Harm to Self. The provider believes there is a serious risk of self-harm by the Service member either as a result of the condition itself or medical treatment of the condition.

Harm to Others. The provider believes there is a serious risk of harm to others either as a result of the condition itself or medical treatment of the condition. This includes any disclosures concerning child abuse or domestic violence consistent with DoD Instruction 6400.06...

Harm to Mission. The provider believes there is a serious risk of harm to a specific military operational mission. Such serious risk may include disorders that significantly impact impulsivity, insight, reliability, and judgment.

Special Personnel. The Service member is in the Personnel Reliability Program as described in DoD Instruction 5210.42 . . . or is in a position that has been preidentified by Service regulation or the command as having mission responsibilities of such potential sensitivity or urgency that normal notification standards would significantly risk mission accomplishment. Inpatient Care. The Service member is admitted or discharged from any inpatient mental health or substance abuse treatment facility as these are considered critical points in treatment and support nationally recognized patient safety standards.

Acute Medical Conditions Interfering With Duty. The Service member is experiencing an acute mental health condition or is engaged in an acute medical treatment regimen that impairs the Service member's ability to perform assigned duties.

Substance Abuse Treatment Program. The Service member has entered into, or is being discharged from, a formal outpatient or inpatient treatment program consistent with DoD Instruction 1010.6 . . . for the treatment of substance abuse or dependence.

These conditions overlap with those in Mil. R. Evid. 513, and, in these cases, DoDI 6490.08 indicates that the provider is to disclose the minimum amount of necessary information to the commander.

In addition to limited privilege, some services offer special privacy protections after a disaster or other traumatic event. DoDI 6200.03 (*Public Health Emergency Management Within the Department of Defense*) outlines the procedures for a disaster mental health response team to address the psychological aspects of a public health incident. This policy states that the help and resources provided by this team are not to be documented in medical or mental health records. As a result, mental health may be perceived as easier to access at a time (i.e., during a disaster) when people may be in need. Similarly, the support services offered by Air Force Traumatic Stress Response (TSR) teams to service members (AFI 44-153, *Traumatic Stress Response*) are not documented in medical or mental health records, protecting the privacy of airmen seeking prevention or early intervention services after a traumatic event. TSR teams offer education, screening, and referrals for treatment to airmen after a traumatic event occurs. They do not provide treatment.

In nonmedical settings, DoDI 6490.06 ("Counseling Services for DoD Military, Guard and Reserve, Certain Affiliated Personnel, and Their Family Members") protects the confidentiality of those seeking nonmedical counseling through Military OneSource or Military Family Life Consultants unless a need-to-disclose situation arises (e.g., indicating intent to harm self or others).

Despite Equal-Opportunity Policies, Wide Variability and Ambiguity in Policies That Prohibit Opportunities May Inadvertently Create Opportunities for Discrimination

Most of these are policies we identified that mandate equal treatment for people with mental health challenges present or reference equal-opportunity statements. DoD Directive (DoDD) 1350.2 (*Department of Defense Military Equal Opportunity [MEO] Program*) states that DoD policy is to

ensure that all on-base activities and, to the extent of the ability of the Department of Defense, any off-base activities available to military personnel are open to all military personnel and their family members regardless of race, color, religion, age, physical or *mental disability*, sex, or national origin. (emphasis added)

Thus, this and other policies explicitly name service members and their family members who have MHDs as a protected group.

Despite the emphasis on equal opportunity, we identified 97 policies that prohibited the actions of PWMHDs or seeking treatment. For most of these policies, action was prohibited in the form of restricting service members with MHDs from serving in certain positions. The nature of the positions with mental health restrictions varies greatly. For example, someone with a history of mental health symptoms or disorders may be disqualified from positions as a chaplain assistant (AFI 52-102V2, "Chaplain Assistant Professional Development"), nuclear weapon personnel (AR 50-5, *Nuclear Surety*), drill sergeant (AR 614-200, *Enlisted Assignments and Utilization Management*), or postal clerk (Marine Corps Order [MCO] 1200.17D, *Military Occupational Specialties Manual*, parts 1, 2, and 4), to name a few. Although some of these limitations may be reasonable (e.g., if the position is one in which the consequences of not performing it properly could result in direct harm to others), other limitations may unnecessarily eliminate service members with past mental health symptoms or disorders who have recovered and could competently fulfill their duties in the position.

Another policy, DoDI 6130.03 (*Medical Standards for Appointment, Enlistment, or Induction in the Military Services*) contains a list of mental and behavioral health disorders that preclude joining the military. The list contains a large number of disorders, including psychotic disorders, bipolar, major depression with care for more than two years, adjustment disorder, history or disturbance of conduct, impulse control, other behavior and personality disorders, eating disorders, suicidal behavior, psychiatric hospitalization, PTSD, obsessive-compulsive disorder, and anxiety disorder. The implication is that people with any of these disorders (regardless of treatment or recovery status) are not able to fulfill the duties required of them in the military. This may be true for some people, but, for others, those who have experienced MHDs and recovered from them, the assumption of incompetence or inability to fulfill military duties may be incorrect. Revising these policies may help signal the commitment of DoD and each service to reducing the stigmatization of PWMHDs and may implicitly signal that such stigmatization is not acceptable.

Wide variability and ambiguity in the discussion of MHDs complicate the issue of restricting someone with an MHD or a history of MHDs from serving in the military or fulfilling certain roles. *Having an MHD* could refer to having received a diagnosis of the disorder in the past, currently having a diagnosis, or currently having the diagnosis and managing it adequately with treatment. Because interpretation of what *having an MHD* means can affect the jobs and roles service members may hold, clear

guidance on what constitutes having a mental health symptom or disorder needs to be provided. Understanding what is meant by saying that a service member has an MHD and where he or she is in the recovery process is relevant to determining how policies apply to these people. Although many of these policies may be appropriate, others may be unnecessarily restrictive and eliminate opportunities for service members with mental health challenges. Many of these policies would benefit from clarifying language around exactly what mental health experience (e.g., current diagnosis managed with treatment, former diagnosis) precludes a service member from filling a role.

Policies Support Universal Educational Stigma-Reduction Programs but Not More-Targeted Programs for Those in Mental Health Treatment

We identified 13 policies that enacted a formal program that may support stigma reduction, most typically as part of a broader educational program about suicide, mental health, or redeployment. The policies did not describe stigma-reduction messaging in great detail, and further evaluation of each stigma-reduction program would be necessary to determine its effectiveness. For example, AFI 90-505 (*Suicide Prevention Program*) includes a suggested suicide prevention training curriculum that addresses stigma through the presentation of information about common concerns (e.g., treatment-seeking affecting ability to gain security clearances, confidentiality of treatment) and career effects of help-seeking. In another example, AR 600-63 (*Army Health Promotion*) describes leadership training required of all Army leaders. This training covers such topics as

current Army policy toward suicide prevention, suicide-risk identification, and early intervention with at-risk personnel. This includes how to refer subordinates to the appropriate helping agency, and how to create an atmosphere within their commands that reduces stigma and encourages help-seeking behavior.

We were unable to identify any policies that explicitly discussed the use of psychotherapeutic approaches to aid service members in coping with the experience of stigma. It is possible that military behavioral health personnel use this technique as part of their treatment strategies. However, none of the policies that we reviewed explicitly documented it. Careful consideration of how policies may protect these people or allow them to cope with stigmatizing and discriminatory behavior from others would help bridge a gap in the current landscape of policies addressing stigma. This is especially important given the findings described in Chapter Two, that perceptions of stigma were twice as high among those seeking treatment for MHDs as among those not seeking treatment.

We were also unable to identify policies that supported the adaptation or use of documented, efficacious stigma-reduction programs designed for the general population. However, DoD is currently pilot-testing MHFA in a military context.

Some Policies Reinforce Stereotypes Through the Use of Negative Terminology

Fourteen policies were identified that could potentially reinforce negative stereotypes about PWMHDs. These policies used negative terminology to refer to PWMHDs, their capabilities, or actions (see Appendix H for these policies). Referrals to PWMHDs as being unsuitable or including MHDs in a list of characteristics that make people a risk for dangerous or violent behavior promote stereotypes of PWMHDs being unable to recover, unable to execute their duties, or being dangerous. Other language also implies childishness, using such phrases as "acting out" or having "temper tantrums." In addition to the 14 policies identified, many more policies contained terminology that is standard within the legal or medical context (e.g., "mental incompetence," "mental defects") but that can be in conflict with a focus on stigma reduction because they promote negative attitudes and beliefs about PWMHDs.

Policies Allow Nonprofessionals to Determine Mental Health Fitness, Which Could Promote Stigma

We identified 32 policies that allow people who are not mental health professionals to determine mental fitness or interpret the implications of mental health symptoms or disorders. Having people without adequate or appropriate mental health training make determinations about mental health status allows for the possibility that someone's lack of knowledge or potentially stigmatizing attitudes can result in discrimination toward the person experiencing mental health challenges. For example, consider Army Tactics, Techniques, and Procedures (ATTP) 3-21.9 (SBCT [Stryker Brigade Combat Team] *Infantry Rifle Platoon and Squad*). This policy states that "platoon and squad leaders are active participants in the areas of hygiene, sanitation, counseling, and *the treatment of stress and combat and operational stress reactions*" (emphasis added). Including treatment in the platoon and squad leader's role could potentially lead to negative outcomes for soldiers. Leaders who have inaccurate knowledge or who hold stigmatizing attitudes toward PWMHDs could negatively affect the lives of soldiers experiencing mental health symptoms.

Policies That Support the Use of Mandated Mental Health Screening for Specific Individuals or Groups May Have Implications for Stigma

We identified 33 policies related to mental health screening that neither clearly contributed to nor clearly promoted stigma. Mental health screening can be a positive step to early intervention, identifying service members in distress or at risk of experiencing MHDs and referring them to appropriate resources. However, inappropriate use of mental health screening or evaluation (e.g., as a punishment or means of retaliation, by non-mental health professionals) could lead to negative attitudes toward early intervention and treatment and, in the case of screening that utilizes self-reports of symptoms, efforts by service members to avoid screening may occur. For example, BUMED Instruction 5041.6 ("Navy Medicine Hotline Program") details a hotline and protocols for reporting retaliatory mental health evaluations (e.g., ordered for a service member as a punishment). We also note that screening could result in stigmatization if it is carried out in such a way that some groups of service members are referred for screening more than others, implicitly creating a distinction between the groups based on actual or perceived potential for MHDs. Because inappropriate use could outweigh the positive benefits of screening, careful consideration of implementation of screening policies (e.g., providing safeguards, such as BUMED Instruction 5041.6, thoughtfully selecting time points for screening) is necessary.

Conclusion

In summary, we were unable to identify any tracking or measurement of institutional context, so it is difficult to know the impacts of the policies we describe above. However, survey items that assess concerns that seeking mental health treatment may negatively affect military careers, a highly endorsed concern in the military, may provide useful context for interpreting the impact of these institutional factors. Our review of the policy context found that ambiguities in policy language might contribute to concerns. Despite the presence of equal-opportunity policies, wide variability and ambiguity in policies that block service members with MHDs from career opportunities may inadvertently create opportunities for discrimination. These policies do not define triggers for opportunity limitations (e.g., any history of an MHD, anyone engaged in treatment) and do not acknowledge a threshold of symptomology or continuum of recovery. Additionally, conflicting language and intentions of policy highlight key tensions between the privacy of service members seeking mental health treatment and the need for commanders to assess unit fitness. These tensions will need to be addressed if DoD is to be successful in encouraging treatment-seeking among more of its service members.

We also identified policies that support universal educational stigma-reduction programs but not more-targeted programs for those in mental health treatment. This is a key gap in programs, given the higher prevalence of stigma among service members in treatment than among those not in treatment.

Finally, we identified three areas of policy that DoD should consider reviewing. First, some policies used negative terminology and reinforced stereotypes about PWMHDs. Revising this language may help to minimize the likelihood that service members would feel stigmatized as a result. Second, we identified policies that allow nonprofessionals to determine mental health fitness and that support the use of mandated mental health screening for specific individuals or groups. Although these practices may be important to protect unit fitness in the military context, these practices may put some service members at risk for stigma and discrimination. Third, mental health screening and evaluation programs may be used inappropriately, and careful consideration of the implementation of such programs is necessary to ensure that these programs promote positive, rather than negative, attitudes toward treatment-seeking.

CHAPTER EIGHT

Key Findings and Priorities for Improving the U.S. Department of Defense's Approach to Stigma Reduction

In this chapter, we summarize key study findings, outline priorities for improving DoD's approach to stigma reduction, and include both short-term (in the next one to two years) and long-term (in the next three to five years) steps DoD can take to achieve these priorities. These priorities are ranked according to the findings described in this report and have been further refined and vetted through an expert panel. Appendix I contains more details on the expert-panel process used to finalize these priorities. In this chapter, we organize priorities into three categories: enhancing or improving stigma-reduction intervention, policies, and research and evaluation related to mental health stigma in the military. In addition, there is a single overarching priority. For each priority ranked in comparison with others. The experts rated 14 priorities. They agreed that 13 of the priorities were critical to include in the report; we describe these in this chapter.

Key Findings from the Report

DoD's current approach to reducing stigma appears to have reduced reports of perceived stigma, as described in Chapter Three. However, there are several areas in which program, policy, and research can be improved to enhance the effectiveness of DoD's approach. Our analysis revealed several key findings that helped to frame the need for the priorities described in this section.

The difficulty linking stigma to mental health treatment-seeking in both Chapters Three and Four suggests the need for improvements in research and evaluation of stigma and stigma-reduction initiatives.

The assessment of stigma-reduction programs found that DoD should continue its comprehensive approach to stigma reduction that blends a culture shift with several stigma-reduction programs. Lessons learned from effective civilian programs may help improve the DoD programs targeting military norms, culture, and social context. These include the use of more contact-based intervention, educational efforts targeting people in distress, and effective messaging. Programs targeting the individual context may help broaden DoD's current approach. These include programs for those in need of mental health treatment and those currently receiving treatment. Finally, more evaluation of stigma-reduction programs is needed.

The policy analysis revealed six findings. First, key tensions exist between the privacy of service members seeking mental health treatment and the need for commanders to assess unit fitness. Second, despite the presence of equal-opportunity policies, wide variability and ambiguity in policies that block service members with MHDs from career opportunities may inadvertently create opportunities for discrimination. Third, policies support universal educational stigma-reduction programs but not moretargeted programs for those in or in need of mental health treatment. Fourth, some policies could potentially reinforce stereotypes through the use of negative terminology. Fifth, other policies may expose service members to stigma or discrimination because they allow non–mental health professionals to determine mental health fitness. Finally, mental health screening is beneficial for early intervention but, if used improperly, may be stigmatizing and have negative effects on mental health (e.g., reduce self-esteem).

These findings and the feedback of the expert panel guided the development of the 13 priorities described in the remainder of this chapter.

Priorities to Improve Stigma-Reduction Interventions

1. Explore Interventions That Directly Increase Treatment-Seeking

Focusing primarily on a single barrier to care, such as stigma, may obfuscate other potential interventions to promote help-seeking. As mentioned previously, there are many potential approaches to promote help-seeking along the continuum of stress that have proven effective. In addition to targeting stigma, DoD should explore other mechanisms for increasing treatment-seeking and reducing barriers to mental health care. Research has specifically supported interventions in four areas: changing perceptions about the effectiveness of care, reducing access barriers, changing masculine norms, and increasing peer support. No one factor appears to be *the* key target for interventions focused on barriers to mental health care. Instead, several possible intervention targets—over and above those focused on stigma—appear promising for reducing barriers to care and ultimately increasing treatment-seeking. DoD should explore in depth the evidence pertaining to these four different types of barriers to mental health treatment and consider developing or bolstering effective interventions in these areas. Expert panelists also suggested that DoD explore stigma-reduction interventions at the unit and family levels, citing that the unit and family are critical influences over whether a service member in need accesses care. Panelists also stressed the need for these interventions to be sensitive to culture, defined broadly (e.g., military and racial/ ethnic culture). Expert panelists rated this priority at the top for both importance and validity.

- Short-term steps: Comprehensively review and assess the research pertaining to five types of barriers to mental health care: mental health literacy, beliefs about treatment effectiveness, access barriers, masculine norms, and peer support. The assessment should examine both the impact of these barriers and how these barriers vary by racial/ethnic group, rank, gender, and other factors.
- Long-term steps: Develop, validate, and implement interventions that have proven effective and show promise for reducing barriers to mental health care among military populations.

2. Consider Evidence-Based Approaches to Empowering Service Members with Mental Health Concerns to Support Their Peers

Expert panelists suggested that promoting the empowerment of PWMHDs to provide peer support for one another is an important approach that DoD should consider to reduce stigma. Military-affiliated panelists suggested that peer-support programs were already going on to some degree throughout DoD; however, the evidence base behind these programs was limited. One panelist referred to the empowerment focus of the gay rights movement (e.g., gay pride, proud to come out as gay) and suggested that a similar movement among PWMHDs would help to generate momentum to address the negative perceptions that promote stigma. This was the second-highest-rated priority in terms of importance and the third-highest in terms of validity.

- Short-term steps: Identify evidence-based approaches to peer support.
- Long-term steps: Implement and evaluate the effectiveness of peer-support programs (both existing and new).

3. Design New or Adapt Existing Intervention-Delivery Mechanisms to Minimize Operational Barriers for Service Members Seeking Treatment

Expert panelists also discussed how service members' preferences for self-management may be a key barrier to their accessing services. Improving the complement of alternative mechanisms for treatment delivery that create fewer operational impediments to service members could appeal to service members with a preference for self-management. For example, offering Internet-based treatment or brief sessions would help decrease disruption to service members' workdays while still allowing service members to seek the help they need to ensure that they can meet the demands of their role. Research is needed on the effectiveness of these alternative mechanisms to ensure that they deliver treatment with effectiveness equivalent to more-traditional treatment delivery mechanisms (e.g., one-on-one sessions with a counselor). Experts rated this priority as thirdmost important and fifth-most valid (out of 14 total priorities).

• Short-term steps: Generate a list of possible delivery mechanisms and get line leadership and service member feedback to identify the most promising.

• Long-term steps: Evaluate effectiveness of the promising delivery mechanisms, in terms of both getting service members to use the mechanisms and the effectiveness of the intervention delivered through the mechanisms.

4. Embed Stigma-Reduction Interventions in Clinical Treatment

Stigma is a potential clinical risk factor, and research has suggested that it may delay treatment-seeking, worsen treatment outcomes, reduce treatment compliance, and increase the risk of relapse (Shrivastava et al., 2013). These types of negative treatment outcomes can perpetuate poor functioning and further isolate service members who have accessed mental health services. In order to address this potential clinical risk factor, stigma should be assessed during routine clinical examination and monitored throughout treatment so that it can be addressed as a part of a comprehensive treatment strategy. Evaluations to validate existing clinical antistigma interventions for a military population and innovative antistigma interventions, relevant for a military clinical population, are needed to improve outcomes of service members in treatment. Experts ranked this priority as tenth out of 14 in terms of both importance and validity.

- Short-term steps: Assess for stigma during initial clinical examinations and regularly during treatment.
- Long-term steps: Validate existing clinical antistigma interventions for military populations or develop new and innovative antistigma interventions that can be used for a military clinical population during treatment.

5. Implement and Evaluate Stigma-Reduction Programs That Target Service Members Who Have Not Yet Developed Symptoms of a Mental Health Disorder

Our literature review identified some evidence-based programs, such as MHFA (the SAMHSA-adapted version), that target stigma reduction among people who may be at high risk for developing MHDs. DoD should draw on this evidence base to identify programs that may translate effectively to the military context and adapt them for use within the military. In particular, the expert panel suggested as potential priorities programs that seek to address stigma at the unit level or change the culture of the small unit. These additional programs are intended to complement existing efforts to change the culture within the military to increase help-seeking behavior. Experts ranked this priority as 11th out of 14 in terms of both importance and validity.

- Short-term steps: Adapt evidence-based stigma-reduction programs for people at high risk for mental health conditions for the military context.
- Long-term steps: Implement and evaluate the effectiveness of these programs.

Priorities to Improve Policies That Contribute to Stigma Reduction

1. Provide Better Guidance for Policies in Which a Mental Health Condition or Treatment Prohibits Job Opportunities or Actions

A large number of the policies we reviewed prohibited specific job opportunities or actions if a service member had an MHD or sought mental health treatment. For many of these policies, the language is unclear, stating only that a service member is prohibited if he or she has a mental health issue. For example, to qualify as an Army recruiter, a candidate must have "no record of emotional or mental instability" (see AR 601-1). It is unclear whether a service member responding well to treatment for an MHD (e.g., being successfully managed with medication) would be prohibited from this opportunity. It is also not clear whether there are certain mental health conditions that merit different considerations (e.g., depression may occur only once in a lifetime, whereas schizophrenia is a more persistent MHD). Given the large number of actions prohibited in these policies, it is imperative that DoD provide additional guidance that clarifies what is meant by potentially ambiguous phrases, such as "have no history of a psychiatric disorder, alcoholism or drug abuse unless a medical evaluation determines the condition no longer exists" (MCO 1200.17D). For example, does having no history of a psychiatric disorder mean no diagnosis at any point in a service member's life? Does psychiatric disorder encompass any mental health diagnosis (e.g., depression, PTSD, schizophrenia)? Can a condition "no longer exist" if no symptoms are displayed while someone is undergoing treatment? Or does the phrase refer to experiencing no symptoms without being in treatment? Clarifying such phrases is more attentive to the continuum of mental health. This guidance would also allow DoD to further define which job categories and duties require certain levels of mental health functioning. Experts ranked this priority sixth-most important and second-most valid.

- Short-term steps: Create guidance to support leaders' decisions related to mental health exclusions for each of the policies that prohibit job opportunities or actions.
- Long-term steps: Monitor enforcement of that guidance.

2. Review the Stigmatizing Language Identified in Policies to Determine Whether It Should Be Removed

In 12 percent of the policies we examined, we identified language that characterized mental health issues in a negative light. Editing these policies to remove this stigmatizing language may help to reduce the likelihood that this language contributes to stigma, and it would improve the clarity of the policies. For example, one DoD policy (DoD Manual [DoDM] 5105.21-V1) indicates that anyone with a certain type of security clearance must report to authorities if he or she becomes aware of "any information that could reflect on their trustworthiness or on that of [others]." The subsequent list of information that might fall into this category includes "apparent mental or emotional problems." This language is stigmatizing in that it implies that mental or

emotional health symptoms or disorders render someone untrustworthy. Expert panelists noted that changing stigmatizing language in policy is not a sufficient condition for reducing stigma because it may not necessarily translate to changes in behavior. Although this and other policy changes may be important, expert panelists recommended that policy actions be accompanied by a more multifaceted approach, including program development and research and evaluation. This priority was ranked by experts as eighth-most important and seventh-most valid.

- Short-term steps: Revise policies to remove stigmatizing language.
- Long-term steps: Create guidance to avoid potentially stigmatizing language in future policies.

3. Offer Incentives for Positive Behaviors That Promote Mental Well-Being

In addition to identifying consequences for prejudicial or discriminatory behaviors toward service members with MHDs, experts emphasized the importance of recognizing and rewarding behaviors that promote mental well-being and are supportive of service members with MHDs. DoD could define these positive and affirming behaviors related to mental health (e.g., positive coping) and help-seeking (e.g., helping a fellow service member access care) and assign merit (e.g., via an award or consideration during promotion decisions) to the process of performing these behaviors. One expert suggested that focusing on the top performers (e.g., 10 to 15 percent "best" service members) in modeling these positive behaviors might help create momentum needed for a wide-scale adoption of positive and affirming behaviors that promote mental wellbeing. This priority was one of the lowest ranked by experts, ranking second to last in terms of both importance and validity.

- Short-term steps: Identify behaviors that promote mental well-being and create a supportive environment for help-seeking. Identify the incentives that are most compelling to service members.
- Long-term steps: Begin offering incentives and evaluate their impact on helpseeking and on stigma.

Priorities to Improve Research and Evaluation Related to Stigma Reduction

1. Continue to Improve and Evaluate the Modifications Made to Existing Programs That Begin to Address Stigma and Other Barriers to Care

DoD is already implementing some modifications to existing initiatives that begin to address barriers to care and may contribute to a larger culture shift in the military. To ensure that these efforts are appropriately assessed for their effectiveness, DoD should improve evaluations of these programs to ensure that they measure behavioral impacts. Including measures of behavior, such as changes in the initiation of treatment or treatment adherence or, in the case of the public, actual social distancing and discriminatory behavior, rather than changes in knowledge, attitudes, beliefs, or intentions, is likely to yield more-compelling evidence to support the effectiveness of interventions. Evaluations should also assess both the cumulative impacts of multilevel initiatives that promote help-seeking and the contribution that each specific intervention makes. Given the range of context factors at the public and institutional levels that may influence stigma specifically and help-seeking more broadly, as well as the variation in effectiveness of interventions by specific MHD, it is likely that population-level stigmareduction interventions will need to occur at multiple levels (e.g., public, institutional, and individual). Although research assessing the effectiveness of these multilevel interventions presents a complex challenge, future studies should be sure to consider both the cumulative effects of interventions at all levels and the specific impacts of each intervention. Experts ranked this priority as fourth-most important and ninth-most valid.

- Short-term steps: Develop guidance for behavioral impact measures that can be used for evaluation of programs that address stigma and other barriers to care, as well as programs that promote help-seeking or a shift in military culture more broadly and improve the number of programs using these measures.
- Long-term steps: Ensure that all applicable programs are using behavioral impact measures to assess effectiveness.

2. Examine the Dynamic Nature of Stigma and How It Interacts with Internal and External Conditions over Time

Much of the stigma research focuses on schizophrenia or general mental health concerns, rather than PTSD, anxiety, or depression—the disorders that may be of most interest to DoD. More research to understand how stigma differs between these disorders and whether there are differential beliefs, attitudes, or knowledge about treatment efficacy for these disorders would help DoD better target stigma-reduction efforts. Additionally, because stigma is not static, more research on how stigma manifests based on level of mental health symptomology and individual interactions with various external conditions (e.g., family members, unit commanders) is needed to improve understanding of the impact of stigma and identify the optimal intervention points, especially for interventions that require multiple boosters to maintain their effectiveness. Though an intervention may result in immediate change, its effectiveness diminishes significantly if those changes are not maintained over time. Studies should therefore follow participants for several months or years to examine the true impact of the intervention. Experts ranked this priority as the fifth-most important and the fifthmost valid.

- Short-term steps: Commission a longitudinal research study to assess the dynamic nature of stigma. Ensure that evaluation studies track impacts over time.
- Long-term steps: Implement longitudinal research and evaluation studies and adapt policies and programs based on findings.

3. Improve Measures of Prevalence to Improve Tracking of Stigma and Other Barriers to Care

Instituting common tracking measures would allow for research on the extent to which the institutional and public contexts affect stigma and how those effects may vary by demographics, such as rank, race, age, or gender. Much of the research we reviewed considered stigma discretely within a specific context or for a specific relationship. Understanding how stigma differentially affects specific populations, as well as identifying consistent effects across populations, will be important for developing interventions tailored to specific populations or applicable across the general population. Because stigma is not static, stigma should be tracked regularly to better understand the varied internal and external factors that affect its manifestation. Currently, tracking efforts happen annually or less often (e.g., behavioral health poll, MHAT, status-offorces surveys) and do not include family units. Understanding how families (spouses, children, significant others) affect stigma is also an important aspect of tracking that should be considered for future measurement efforts. Experts ranked this priority as the ninth-most important and the eighth-most valid.

- Short-term steps: Develop common measures for tracking stigma and other barriers to care that can be implemented regularly without testing biases and that include measures related to familial and significant relationships (e.g., spouse, children).
- Long-term steps: Regularly implement a set of common measures of stigma and other barriers to care force-wide and analyze the data to understand how stigma varies by contextual and demographic differences.

4. Review Classified Department-Wide and Service-Specific Policies to Determine Potential Implications for Mental Health Stigma and Discrimination

We base the priorities presented here on a review of policies that are accessible without clearance. A military-affiliated panelist recommended obtaining and reviewing classified policies to determine whether additional policy priorities should be developed based on the implications of those policies for mental health stigma and discrimination. This priority was among the lowest ranked by experts, ranking 12th-most important and least valid.

• Short-term steps: Work with DoD to identify potentially relevant classified policies.

• Long-term steps: Conduct analysis of these policies and identify any new policy priorities.

Overarching Priority

The priority described in this section cuts across programs, policies, and research and evaluation so did not fit in any one of the preceding priority categories. Therefore, we have created a separate section for this overarching priority. *Overarching* does not mean "most important," as evidenced by the expert ratings.

1. Convene a Task Force to Explore the Tensions Between a Command's Need to Know a Service Member's Mental Health Status and Treatment History and the Service Member's Need for Privacy

In the civilian sector, the Americans with Disabilities Act (Pub. L. 101-336, 1990) and the Health Insurance Portability and Accountability Act (HIPAA) (Pub. L. 104-191, 1996) protect the confidentiality and rights of PWMHDs. The former legislation, however, does not apply in the military context, wherein exists a unique set of demands not commonly encountered in civilian environments. The latter policy contains exemptions when applied in a military context. Specifically, HIPAA exemptions are made in the following situations (Pub. L. 104-191, 1996):

To determine a Service member's fitness to perform any particular mission, assignment, order or duty, including compliance with any actions required as a precondition to performance of such mission, assignment, order or duty; To assess medical readiness and fitness for deployability (e.g., immunization status, temporary or permanent profile status, Medical Evaluation Board (MEB)/Physical Evaluation Board (PEB) related data, allergies, blood type, flight status); To initiate Line of Duty (LOD) determinations and to assist investigating officers in accordance with (IAW) AR 600-8-4 (*Line of Duty Policy, Procedures, and Investigations*); To carry out Soldier Readiness Program and mobilization processing requirements IAW AR 600-8-101 (*Personnel Processing [In-, Out-, Soldier Readiness, Mobilization, and Deployment Processing]*); To monitor the Army Weight Control Program; To provide initial and follow-up reports IAW AR 608-18 (*The Army Family Advocacy Program*).

Provisions also allow mental health care providers to give commanders minimum necessary details about the condition or care of soldiers in their command under certain circumstances, including the following:

To avert a serious and imminent threat to health or safety of a person, such as suicide, homicide or other violent action; To warn commanders of medications that could impair the ability to perform assigned duties (e.g., drowsiness, altered

alertness, slowed cognition); To warn commanders of conditions that can impair the Soldier's performance of duty; To recommend a command-referral to a substance abuse treatment program. (Headquarters, Department of the Army, 2012a, pp. 64–65)

A department-level task force of experts could play an important role in assessing what type of information should and should not be shared in each of the above circumstances and in developing clear communications and processes for these exceptions. For example, the Air Force has a policy allowing service members limited privilege with a mental health professional for short-term care (that is, the short-term care does not become part of the medical record). The task force could review this policy to determine whether it should extend to additional service branches. Experts rated this priority as the seventh-most important and the sixth-most valid.

- Short-term steps: Convene the task force.
- Long-term steps: Implement policy and procedural changes recommended by the task force, and monitor enforcement and impact of these changes.

Examining how other professions with similar unique demands (e.g., police, intelligence community) approach privacy policies related to mental health care may provide some insights and lessons learned that would be useful to DoD. Although the law enforcement context varies from the military context, it bears some similarities in that it is a profession involving service to one's community, the use of force or firearms in the line of duty, the ability to operate successfully in emotionally charged situations, and an increased risk of trauma exposure. In law enforcement, as in DoD, a balance must be struck between protecting the confidentiality of personnel seeking help and having sufficient information about personnel psychological health status to determine whether personnel can adequately fulfill their duties. The International Association of Chiefs of Police (IACP), in conjunction with the U.S. Department of Justice's Bureau of Justice Assistance, established the IACP National Law Enforcement Policy Center, through which it develops model law enforcement policies on a wide range of difficult issues. Subject-matter experts create these model policies, incorporating both research findings and field experience. The models provide a valuable resource for law enforcement agencies that are developing or updating their own policies (IACP, undated [b]). We identified two IACP model policies and accompanying policy papers on the topics of providing mental health services to law enforcement employees and for procedures for personnel support after a shooting event.

Conclusion

There is still much unknown about the influence that stigma has for PWMHDs on initiation of treatment, treatment success (e.g., retention), and, ultimately, quality of life. These priorities represent a first step for where additional program and policy development and additional research and evaluation are needed to improve understanding of how best to get service members with MHDs the needed treatment as efficiently and effectively as possible. Additional research and evaluation are needed to more fully understand barriers to care among service members and which of these barriers most affect treatment initiation, treatment success, and overall quality of life.

We conducted a systematic literature review of theoretical works on stigma and prior studies of stigma-reduction programs. We began by reviewing the resources used in two earlier literature reviews related to our own, RAND's stigma-reduction work for the California Mental Health Services Authority (CalMHSA) (Collins et al., 2012) and the DCoE report "Behavioral Health Stigma and Access to Care" (DCoE, 2012). We then supplemented the literature from these works by performing our own web-based search of peer-reviewed literature in content-relevant databases. These sources underwent successive rounds of screening, including a title and abstract review followed by a full-text review, to exclude irrelevant and unsuitable articles. We then reviewed and coded articles selected for inclusion, and we abstracted details relating to the focus of our study.

Identifying Articles for Review

Literature Search

To identify relevant sources, we conducted a literature search in two parts. Part 1 consisted of combining the references used in two prior literature reviews, one review for CalMHSA and another in the DCoE report on behavioral health stigma and access to care. This step facilitated the literature search by allowing us to quickly identify relevant works and narrow the focus of our subsequent searches. Part 2 was a more comprehensive literature search on mental health stigma in 11 databases that focused on substantive areas pertaining to health (psychology and medicine), defense, and the social sciences broadly:

- PsycINFO (psychology)
- PubMed (medicine)
- MEDLINE (medicine)
- CINAHL (health care)
- EconLit (economics)
- Social Sciences Abstracts (social sciences)

- ProQuest Military Collection (defense)
- Sociological Abstracts (ProQuest) (social sciences)
- Social Services Abstracts (ProQuest) (social services)
- Published International Literature on Traumatic Stress (PILOTS) (mental health)
- Web of Science (general).

Our search consisted of four queries, each aimed at answering different questions related to our study (see Table A.1). We first identified the study questions that the literature review was intended to answer (based on the overall study aims), then consulted with a librarian to identify appropriate search strategies for each study question. We conducted a search in each of the aforementioned databases for each of the four study questions, to enhance the breadth of results. We restricted our search to articles written in English and published in peer-reviewed journals. Studies were limited to those involving human subjects. Additionally, we narrowed our search dates to the past 11 years (since 2002) if military-specific terms were searched and to the past two years (since 2011) for non-military-specific searches. We searched general literature since 2011 to pick up any new articles published since the CalMHSA and DCoE literature searches. Because the military-specific literature was of most interest to the current project and CalMHSA did not specifically cull any military-specific literature, we expanded the dates of that search to begin in 2002. In general,

- query 1 searches looked for articles related to the concepts of "mental health," "stigma," and either "discrimination" or "treatment seeking"
- query 2 looked for articles related to the concepts of "mental health," "stigma," "program," and "evaluation"
- query 3 was a military-specific search that looked for articles related to the concepts of "mental health," "stigma," and "cost"
- query 4 was a military-specific search that looked for articles related to the concepts of "mental health," "stigma," and "careers."

We chose specific keywords following consultation with the study team and with a librarian. Table A.1 presents details of the search strategies for all searches conducted. We identified additional articles by reviewing the references in articles identified for inclusion in the review.

Title, Abstract, and Full-Text Review

After combining all search results, we removed duplicates between databases and then between queries. Additionally, we removed any sources that we had already collected via the CalMHSA and DCoE reports (part 1 of our literature review). To help ensure that all sources identified in the literature searches were relevant, we reviewed the arti-

Table A.1 Specific Searches

Query	Related Study Question	Search Terms	Search Limits
1	What is stigma? What conceptual or theoretical models exist? How has it been defined? How is stigma encompassing of or related to barriers of care, discrimination, psychological resilience, career trajectories, and other factors?	Concept 1: ("mental health" or "mental illness" or "behavioral health" or "mental disorder" or "psychiatric disorder") and Concept 2: (stigma* or self-stigma*) and Concept 3: (literac* or "barriers to care" or resilienc* or "help seeking" or "help-seeking" or "treatment seeking" or "treatment-seeking" or "care utilization" or "treatment utilization" or discriminat* or stereotyp* or career*)	January 2011– November 2012
2	What interventions that target mutable factors are most effective in reducing stigma in a military context?	Concept 1: ("mental health" or "mental illness" or "behavioral health" or "mental disorder" or "psychiatric disorder") and Concept 2: (stigma* or self-stigma* or discrimin* or prejud* or stereotyp*) and Concept 3: (educate or education or program* or intervention* or prevent* or treatment or treatments or campaign* or policy or policies) and Concept 4: (evaluat* or assess* or measur*)	January 2011– November 2012
3	What are the economic implications or societal costs of stigma?	Concept 1: ("mental health" or "mental illness" or "behavioral health" or "mental disorder" or "psychiatric disorder") and Concept 2: (stigma* or self-stigma* or discrimin* or prejud* or stereotyp*) and Concept 3: (econom* or cost*) and Concept 4: (military or "armed services" or "armed forces" or army or navy or marines or "air force" or "coast guard" or "national guard" or soldier or serviceman or servicewoman or airman or sailor or marine or guardsman or guardsmen or warrior or warriors or combatant or combatants or veteran or veterans or "department of defense" or	January 2002– November 2012

Table A.1—Continued

Query	Related Study Question	Search Terms	Search Limits
4	What impact does seeking mental health treatment have on military careers?	Concept 1: ("mental health" or "mental illness" or "behavioral health" or "mental disorder" or "psychiatric disorder") and Concept 2: (impact* or effect* or hurt* or harm* or detriment*) and Concept 3: (career* or employ* or job* or promotion*) and Concept 4: (military or "armed services" or "armed forces" or army or navy or marines or "air force" or "coast guard" or "national guard" or soldier or soldiers or servicemen or servicewomen or serviceman or servicewoman or airman or sailor or marine or guardsman or guardsmen or warrior or warriors or combatant or combatants or veteran or veterans or "department of defense" or reservist*)	January 2002– November 2012

NOTE: Each query was searched in each of the following: PsycINFO, PubMed, MEDLINE, CINAHL, EconLit, Social Sciences Abstracts, ProQuest Military Collection, Sociological Abstracts, Social Services Abstracts, PILOTS, and Web of Science.

Database	Query 1	Query 2	Query 3	Query 4
PsycINFO	319	591	26	314
PubMed	210	280	15	102
MEDLINE	218	251	16	103
CINAHL	81	83	3	43
EconLit	2	9	3	0
Social Sciences Abstracts	40	65	2	19
ProQuest Military Collection	21	57	253	398
Sociological Abstracts	143	67	2	9
Social Services Abstracts	37	53	1	4
PILOTS	39	39	9	56
Web of Science	311	331	11	97
Total results	1,421	1,826	341	1,145
Number of duplicates	692	847	48	295
Nonrelevant records	278	546	242	582
Total unique results	451	433	51	268

Table A.2 Web-Based Search-Result Tallies

cle titles and abstracts to remove articles that were clearly irrelevant to the current project. We then removed articles from consideration as appropriate and obtained others for full-text review. We subjected all articles identified for full-text review to the inclusion and exclusion criteria, as detailed in Table A.3. We then coded articles that met the inclusion criteria according to the process outlined in the next section.

Articles Identified During the Literature Search

Combining the literature from the CalMHSA (125 articles) and DCoE (99 articles) stigma reports and removing duplicates yielded 214 unique sources, and combining the four queries of the web-based search yielded 1,025 unique results. In total, 1,209 unique sources were identified. We excluded 886 articles during our title and abstract review and obtained full-text versions of the remaining 323 articles. We excluded another 71 articles during the full-text review. The primary reason for exclusion during the full-text review was that the article fell outside the scope of our research (e.g., an article on mental health care that mentions stigma in passing but did not directly discuss it). An additional 78 articles were found during the full-text review in the references of other articles reviewed and were added for inclusion. This resulted in 330 articles included in our review. Figure A.1 is a flow chart depicting the identification and exclusion of references. The reference list at the end of this report includes all the articles reviewed.

Criterion Type	Considerations		
Inclusion	Content: Articles that provided a theoretical model or framework for stigma Articles that described the evaluation of a program whose primary goal is to reduce or prevent mental health stigma Articles that reported prevalence rates of stigma Articles that discussed the effects of stigma on treatment-seeking and utilization Articles that reported the associated costs of stigma, both financial and social Articles that reported the effects of stigma on careers and promotions		
	General: Articles that employed a range of experimental strategies Articles written outside the United States Both civilian- and military-specific sources		
Exclusion	Articles published in a language other than English Editorials, letters, and commentaries Dissertation papers and master's theses		

Table A.3 Inclusion and Exclusion Criteria



Figure A.1 Flow Chart for Literature Search

Abstracting Consistent Information from Each Source

Article Coding and Data Abstraction

We reviewed each article for the following types of information, where relevant:

- general information: the citation information for the article, approach (e.g., experimental, review, meta-analysis), and population described in the article (e.g., military or nonmilitary)
- theoretical description
 - the article's definition of stigma and any theoretical model of stigma provided
 - descriptions of how knowledge, attitudes, and behaviors related to stigma
 - whether stigma was described at the individual, public, or organizational level
 - the relationship of stigma to treatment-seeking
- prevalence data: statistics on the prevalence of stigma, including the population, methods, and measure used to assess prevalence
- effects of stigma
 - *on treatment:* treatment type, treatment population, treatment effects, and effect size

- *on costs:* any costs associated with stigma reported in the article and how these costs were defined and modeled
- on careers: observed effects of stigma on careers reported in the article, the specific population described, and how these career effects were defined and modeled
- evaluations: for those articles describing interventions, a description of the stigmareduction intervention, the MHD targeted by the intervention, the specific intervention population, outcomes tested in the evaluation, and findings of effectiveness reported in the article
- empirical study design (if applicable): elements of the study design, including whether the study was cross-sectional or longitudinal, frequency of follow-up, use of a comparison group, random assignment to study group, measures used, and method of data collection.

Further detail on each of these abstracted pieces of information is available in the data abstraction form, shown in Table A.4.

Element Abstracted from Each Article	Brief Description of the Element		
General information			
Coder	Initials of the reviewer		
Citation	Reference information for the article reviewed		
Article approach	A classification of the article type (e.g., experimental, quasi- experimental, review, synthesis, meta-analysis)		
Population described	A classification of the population described in the article (e.g., Army, Navy, Marine Corps, Guard, Reserve, international military, veteran, not military)		
Theoretical description			
Definition	The verbatim definition of <i>stigma</i> provided in the article		
Description of model	A summary of the theoretical model of stigma, if included		
Component			
Knowledge and beliefs	A description of the knowledge or beliefs associated with stigma, if included		
Attitudes	A description of the attitudes associated with stigma, if included		
Behaviors	A description of the behaviors associated with stigma, if included		

Table A.4 Data Abstraction Form
Table A.4—Continued

Element Abstracted from Each Article	Brief Description of the Element
Level	
Individual level	Whether or not individual or self-stigma was described in the article
Public level	Whether or not public stigma was described in the article
Organizational level	Whether or not organizational or structural stigma was described in the article
Relationship to treatment- seeking	A description of the relationship of stigma to the construct of treatment-seeking, if available
Prevalence data	
Prevalence reported	A stigma prevalence rate or percentage given in the article
Prevalence population	A description of the population included in the prevalence statistic (e.g., age group, military or nonmilitary, location, sex, race)
Citation	(If prevalence is cited) a full reference of the original source of the statistic
Measure	(If prevalence is primary) a description of the measure used to determine prevalence
Sample size	(If prevalence is primary) the size of the sample used to determine prevalence
Covariates	(If prevalence is primary) a description of any covariates reported
Effect of stigma	
On treatment	
Treatment type	A classification of the type of treatment reported (e.g., individual therapy, group therapy, drug use)
Treatment population	A description of the population described (e.g., age group, military or nonmilitary, location, sex, race)
Treatment effects	A description of the treatment effects reported
Effect size	The effect size reported for treatment effects
Standard deviation	Standard deviation
Model	The type of model used (e.g., one or two step, specific subpopulation)
On costs	
Model	The type of model used to determine costs
Findings	A description of the reported costs associated with stigma

Element Abstracted from Each Article	Brief Description of the Element
On careers	
Definition	A description of how career effects were defined in the article
Population	A description of the population described (e.g., age group, military or nonmilitary, location, sex, race)
Findings	A description of the career effects reported
Model	The type of model used (e.g., one or two step, specific subpopulation)
Program evaluation	
MHD targeted	A classification of the MHD targeted by the intervention or program (e.g., general, depression, anxiety, PTSD, schizophrenia, other)
Intervention description	A one- to two-sentence overview of the program or intervention
Intervention population	A description of the population targeted by the program (e.g., age group, military or nonmilitary, location, sex, race)
Outcomes tested	A description of the outcomes tested in the program evaluation
Findings	A brief summary of the findings for each outcome tested
Empirical study design	
Random assignment	Whether or not participants were randomly assigned to a program or intervention
Measure timing	A description of the timing of measure assessment relative to the program (e.g., pre/post assessment, post-only)
Measure frequency	A description of the frequency of measure assessment (e.g., longitudinal, cross-sectional)
Control group	Whether a control or comparison group was included in the study
Data-collection method	A classification of the method used to collect data for the study (e.g., interview, focus group, survey, observational)
Analysis type	A description of the type of statistical analysis conducted (e.g., qualitative synthesis, descriptive analysis, bivariate analysis, multivariate analysis)

Table A.4—Continued

Our goal in creating our abstraction form was to provide a standardized way to review our sources and extract relevant details to help inform our study. Because our literature review encompassed many different types of sources (e.g., reviews, studies), many fields or even entire sections were often not applicable to the article at hand. For example, an experimental study evaluating a stigma-reduction program may not describe the effects or costs of stigma, report prevalence data, or provide a theoretical model but would provide information that was abstracted for the "Program evaluation" and "Empirical study design" sections. Alternatively, a purely theoretical article may provide a detailed model of stigma without describing a scientific study, making the "Empirical study design" section irrelevant.

Procedure for Coding Articles

Coders received initial instruction on the use of the data-abstraction form and the content to be included. We distributed the articles among the team members for independent coding. The team regularly reviewed questions about coding to ensure reliability and consistency among members. Once we had coded all the articles, we reviewed all the data-abstraction forms for completeness and clarity and then compiled them into a single form for analysis. The tables in this appendix provide verbatim definitions from the literature of various kinds of stigma (mental health stigma in Table B.1, institutional and structural stigma in Table B.2, individual and self-stigma in Table B.3, and public stigma in Table B.4) and indicate certain areas of commonality that we found:

- stigma as label, mark, attribute, or identity
- stigma as otherness, separation, or differentiation
- stigma as a loss of status or being disgraced, discredited, or dehumanized
- stigma as a stereotype or belief
- stigma as prejudice or attitude
- stigma as discrimination or behavior
- stigma as a barrier to care.

Table B.1 Definitions of *Mental Health Stigma*

Source	Definition	Label ^a	Otherness ^b	Loss of Status ^c	Stereotype or Belief	Prejudice or Attitude	Discrimination or Behavior	Barrier to Care
Arboleda- Flórez and Stuart, 2012	"Stigma has been defined as a feeling of being negatively differentiated owing to a particular condition, group membership, or state in life. The process of stigmatization occurs when there is a power differential, as only powerful groups can create social inequities."	x	x					
Beltran et al., 2007	"Attitudes towards people with mental illness within the general population are generally negative. This negative attitude is manifested in bias, distrust, stereotyping, fear, embarrassment, anger and/or avoidance."		x		x	x		
Britt, Wright, and Moore, 2012	"Stigma refers to service members believing that seeking treatment would be embarrassing, cause harm to their career, and/or cause their fellow unit members to have less confidence in them."			х				х
K. Brown and Bradley, 2002	"A mark of disgrace or reproach. Stigma is not simply the use of negative labels or wrong words; it is disrespectful to the individual who has mental illness. It further discourages the individual from seeking the help needed for fear of discrimination. Furthermore, stigma encourages fear, mistrust and violence against people with mental illness."	x						x
S. Brown et al., 2010	"Negative attitudes and beliefs [held by the general public] towards individuals with mental illness."				х	x		
Byrne, 1999 ^d	"A mark of disgrace or discredit that sets a person aside from others."	х	х	х				

Table B.1—Continued

Source	Definition	Label ^a	Otherness ^b	Loss of Status ^c	Stereotype or Belief	Prejudice or Attitude	Discrimination or Behavior	Barrier to Care
Corrigan, 2011	"The prejudice and discrimination that affect people with mental illness when appreciable segments of the general population endorse negative stereotypes about psychiatric disorders."				x	x	x	
Corrigan, 2012b	"Stigma has been described in terms of prejudice (agreement with stereotypic beliefs leading to hostile emotional responses, such as fear and anger) and discrimination (the behavioral consequence of prejudice, which leads to social distance and the loss of opportunity, such as a good job or nice place to live)."		x		x	x	x	
Corrigan and Penn, 1999 ^e	"Stigmas are negative and erroneous attitudes about these persons [with mental illness]. Stigma is another term for prejudice or negative stereotyping. In terms of mental illness, stigmas represent invalidating and poorly justified knowledge structures that lead to discrimination."				x	x	x	
Corrigan, Powell, and Rüsch, 2012	"Stigma has been defined as stereotypes (beliefs about a stigmatized group), prejudice (agreement with stereotypes leading to emotional responses), and discrimination (the behavioral result of prejudice)."				x	x	x	
Corrigan, Rafacz, and Rüsch, 2011	"Comprising stereotypes, prejudice, and discrimination."				X	х	x	
Corrigan, Roe, and Tsang, 2011	"Prejudice and discrimination with which the public has branded people labeled with mental illness."	х				х	x	

Table B.1—Continued

Source	Definition	Label ^a	Otherness ^b	Loss of Status ^c	Stereotype or Belief	Prejudice or Attitude	Discrimination or Behavior	Barrier to Care
Crocker, Major, and Steele, 1998 ^f	"Stigma occurs when a person possesses or is believed to possess some attribute or characteristic that conveys a social identity that is devalued in a particular context."	х		х				
Dalky, 2012	"People who are stigmatized have (or are believed to have) an attribute that marks them as different and causes them to be denigrated. Stigma is believed to be relationship- and context-specific It does not reside in the person but rather within a specific social context."	х		х				
Eisenberg, Downs, and Golberstein, 2012	"Mental illness stigma refers to negative stereotypes and prejudices about people with mental illness, and is a widespread phenomenor with damaging social, psychological, and economic consequences."	١			Х	x		
Essler, Arthur, and Stickley, 2006	"Stigma is the way certain attributes are socially agreed as worthy of devaluation and social avoidance[It is caused by] an automatic urge to discriminate against others to increase their own self-esteem [as] people inevitably focus upon the weaknesses of others to make themselves feel better."	х	x	х				
Farina and Feliner, 1973	"Along with being held in low esteem, someone who is known to have been mentally ill is, as a result, perceived as being more inadequate and incompetent in [his or her] behavior than a control person."			х				
Goffman, 1963 ^g	"Stigma is a relationship between an attribute and a stereotype."	х			х			

Table B.1—Continued

Source	Definition	Label ^a	Otherness ^b	Loss of Status ^c	Stereotype or Belief	Prejudice or Attitude	Discrimination or Behavior	Barrier to Care
Goffman, 1963 ^h	"An attribute that is deeply discrediting and that reduces the bearer from a whole and usual person to a tainted, discounted one."	x		х				
Hayward and Bright, 1997 ⁱ	"The negative effects of a label placed on any group."	х						
Idemudia and Matamela, 2012	"An attribute or quality that significantly discredits an individual in the eyes of others, who is seen by them as having an illness that is socially unacceptable, therefore he or she must be isolated or ostracized."		x	х				
Jones, 1984 ^j	"Stigma takes place when the mark (condition considered deviant by society) links the person via attributional processes to undesirable characteristics that discredit him or her in the eyes of others [There are] six dimensions of stigma: concealability, course, disruptiveness, aesthetics, origin, peril."	х		x				
Kim et al., 2011	"Stigma is operationalized by way of public beliefs, prejudices, and stereotypes that, when internalized, may damage self-esteem and impede treatment seeking."	х	x		x	x		х

Table	B.1—	Continued
-------	------	-----------

Source	Definition	Label ^a	Otherness ^b	Loss of Status ^c	Stereotype or Belief	Prejudice or Attitude	Discrimination or Behavior	Barrier to Care
Link and Phelan, 2001 ^k	"In our conceptualization, stigma exists when the following interrelated components converge. In the first component, people distinguish and label human differences. In the second, dominant cultural beliefs link labeled persons to undesirable characteristics to negative stereotypes. In the third, labeled persons are placed in distinct categories to as to accomplish some degree of separation of 'us' from 'them'. In the fourth, labeled persons experience status loss and discrimination that lead to unequal outcomes. Finally, stigmatization is entirely contingent on access to social, economic, and political power that allows the identification of differentness, the constructions of stereotypes, the separation of labeled persons into distinct categories, and the full execution of disapproval, rejection, exclusion and discrimination. Thus, we apply the term stigma when elements of labeling, stereotyping, separation, status loss, and discrimination co-occur in a power situation that allows the components of stigma to unfold."	x	x	x	x	x	x	
Vlagaña et al., 2007	"A source of shame that is cast onto individuals with mental illness by society."			х				
Masuda, Anderson, and Edmonds, 2012	"Stigma is conceptualized as a set of negative attitudes toward people with a psychological disorder, such as that they are unpredictable or hopeless in recovery."					x		х
Masuda and .atzman, 2011	"Mental health stigma can be roughly defined as a multidimensional process of objectifying and dehumanizing a person because of being labeled as 'having a mental disorder'."	х		х				

Table B.1—Continued

Source	Definition	Label ^a	Otherness ^b	Loss of Status ^c	Stereotype or Belief	Prejudice or Attitude	Discrimination or Behavior	Barrier to Care
Morrison, Becker, and Bourgeois, 1979	"Public fear and subsequent rejection of people with mental illness."		x					
Naylor et al., 2009	"Negative attitudes towards mental health difficulties among peers."					x		
Perry, 2011	"Labeled individuals are subjected to differential treatment by others. As they attempt to cope with discrimination and loss of status associated with their new identity, people with mental illness engage in strategies such as secrecy and withdrawal that exacerbate their social isolation."	x	x	х	x		x	
Pescosolido, 2008	"A mark separating individuals from one another based on a socially conferred judgment that some persons or groups are tainted and 'less than.' Stigma often leads to negative beliefs (stereotypes) the endorsement of those stereotypes (prejudice) and a desire to avoid or exclude persons who hold stigmatized statuses."	х	x		x	х		
J. Phelan and Link, 2004	"Negative perceptions of people with mental illness."					x		
S. Phelan et al., 2011	"Stigma refers to the marked identity of people with traits that are different than what is considered normal or ideal and encompasses beliefs and attitudes (prejudice) as well as behavior (enacted stigma or discrimination) toward the person with the stigmatized trait."	x	x		x	x	x	
Pope, 2011	"Stigma is defined as a mark of disgrace or infamy; a stain or mark of reproach."	х						

Table B.1—Continued

Source	Definition	Label ^a	Otherness ^b	Loss of Status ^c	Stereotype or Belief	Prejudice or Attitude	Discrimination or Behavior	Barrier to Care
Price, 2011	"Stigma is generally thought of as a stereotype that is attached to people who have a characteristic or identity, such as a mental health disorder, that is thought to be flawed or undesirable. Stigmatization is thought to occur when this characteristic is so negatively regarded that alienation is the inevitable result."	x	x		x			
Rae Olmsted et al., 2011	"A barrier that discourages individuals and their families from seeking help, or an issue that may cause shame or discredit; an enduring condition, status, or attribute that is negatively valued by a society and whose possession consequently discredits and disadvantages an individual."	x		х		x		x
Reavley and Jorm, 2011b	"Stigmatizing attitudes towards people with mental disorders are common in adolescents and can lead to negative feelings, stereotyping and discriminatory behaviors."				x	x	х	
Rosen et al., 2011	"Mental health stigma [is] a potentially negatively predisposing attitude."					x		
Rukavina et al., 2012	"Negative portrayals of people with mental illness and the public's negative attitudes."					х		
Sadow and Ryder, 2008	"A brand, a mark of infamy."	х						
Sansone et al., 2008	"In the context of mental illness, the word stigma refers to the negative value judgment associated with having a psychiatric disorder."					х		
Schneider, Beeley, and Repper, 2011	"The subjective state of being embarrassed about a mental health problem or feeling discriminated against on account of it."						x	

Table B.1—Continued

Source	Definition	Label ^a	Otherness ^b	Loss of Status ^c	Stereotype or Belief	Prejudice or Attitude	Discrimination or Behavior	Barrier to Care
Schomerus and Angermeyer, 2008	"Stigmatisation of those with mental illness has been conceptualised as a process ultimately resulting in status loss and discrimination."			х			х	
Skopp et al., 2012	"Social-cognitive processes [that] motivate people to avoid label of mental illness that results when people are associated with mental health care."	x						х
Spagnolo, Murphy, and Librera, 2008	"Stigma is experienced both externally and internally. The external effects of stigma refer primarily to discrimination against people with mental illnesses with regard to housing, work, and social interactions."						x	
Stromwall, Holley, and Bashor, 2011	"Negative implicit or explicit attitudes about people perceived to have a mental illness, and discrimination to overt behaviors (e.g., failure to hire) theorized to result from stigmatizing attitudes."	х		х		x	x	
Stuart, 2004	"In modern times, stigma is understood as an invisible mark that signifies social disapproval and rejection Stigma is deeply discrediting and isolating, and it causes feelings of guilt, shame, inferiority and a wish for concealment. Stigma has been variously understood as a consequence of the visible signs or symptoms of a disorder; a result of having received a psychiatric label, regardless of whether visible signs or symptoms are present; or as a consequence of having received psychiatric treatment, particularly if the locus of care was a psychiatric hospital or if treatment was legally mandated."	x	X	x				

Table B.1—Continued

Source	Definition	Label ^a	Otherness ^b	Loss of Status ^c	Stereotype or Belief	Prejudice or Attitude	Discrimination or Behavior	Barrier to Care
Stull, 2012	"Stigma is a complex construct that involves many factors, including attributes, stereotypes, prejudice, and discrimination."	х	х	х	х	х	x	
Sudom, Zamorski, and Garber, 2012	"Being discriminated against unfairly."						x	
Wahl, 1999	"Negative responses to people who have been identified as having a mental illness, limiting opportunities and undermining self-esteem."	х					x	
Wahl, 2012	"The prejudice and discrimination faced by people when others learn that they have been diagnosed with, and/or treated for, a mental disorder."					x	x	

^a Label, mark, attribute, or identity.

^b Otherness, separation, or differentiation.

^c Loss of status or being disgraced, discredited, or dehumanized.

^d Also cited by MacInnes and Lewis, 2008; Yau, Pun, and Tang, 2011.

^e Also cited by Greene-Shortridge, Britt, and Castro, 2007.

^f Also cited by Corrigan and Shapiro, 2010.

^g Also cited by Dalky, 2012; Link, Yang, et al., 2004.

^h Also cited by Arboleda-Flórez and Stuart, 2012; Corrigan and Shapiro, 2010; Gould, Greenberg, and Hetherton, 2007; Heijnders and Van Der Meij, 2006; Kassam, Glozier, et al., 2011; Kassam, Williams, and Patten, 2012; Lakeman et al., 2012; Quinn et al., 2011; Sadow and Ryder, 2008; Stromwall, Holley, and Bashor, 2011; Stull, 2012; Thoits, 2011.

ⁱ Also cited by MacInnes and Lewis, 2008.

^j Also cited by Kim et al., 2011; Link, Yang, et al., 2004.

^k Also cited by Barke, Nyarko, and Klecha, 2011; Corrigan and Shapiro, 2010; Dalky, 2012; Hackler, 2011; Heijnders and Van Der Meij, 2006; Kumar, 2011; Link, Yang, et al., 2004; Phelan, 2005.

Table B.2 Definitions of Institutional Stigma and Structural Stigma

Source	Definition	Label ^a	Otherness ^b	Loss of Status ^c	Stereotype or Belief	Prejudice or Attitude	Discrimination or Behavior	Barrier to Care		
Corrigan and O'Shaughnessy, 2007 ^d	"Structural stigma [is] policies of private and governmental institutions that intentionally restrict opportunities of people with mental illness, and the policies of institutions that yield unintended consequences that hinder the options of people with mental illness."						х			
Laraway, 2007	"Structural Stigma [is] institutional policies or practices that unnecessarily restrict opportunities because of psychological health issues."						х			
^a Label, mark, at	^a Label, mark, attribute, or identity.									
^b Otherness, sep	^b Otherness, separation, or differentiation.									
^c Loss of status c	Loss of status or being disgraced, discredited, or dehumanized.									

^d Also cited by Wright, Jorm, and Mackinnon, 2011.

Table B.3 Definitions of *Individual Stigma* and *Self-Stigma*

Source	Definition	Label ^a	Otherness ^b	Loss of Status ^c	Stereotype or Belief	Prejudice or Attitude	Discrimination or Behavior	Barrier to Care
Ægisdóttir et al., 2011	"Self-stigma is the internalization of the negative stereotyped messages that are given persons who seek such [mental health] services."				х			х
Boardman et al., 2011	"Self stigma [is the state in which] stigmatised individuals internalise their negative identity and thus come to 'self-stigmatise', incorporating stigmatised views in their self-perception."	х			x			
Brohan et al., 2011	"Self-stigma is a personal response to perceived mental illness stigma. It can be considered a transformative process wherein a person loses his or her previously held or desired identities, e.g., as a parent employee, friend, partner etc. to adopt a stigmatised and devalued view of themselves."	x		x				
Corrigan, Markowitz, et al., 2003 ^d	"Self-stigma [is] defined as the loss of self- esteem and self-efficacy experienced by some people with mental illness, resulting in part from the internalization of public stigma."			x				
Corrigan, Morris, et al., 2012	"Self-stigma [is] the harm that occurs when the person internalizes the prejudice."				х	х		
Corrigan and O'Shaughnessy, 2007	"Self stigma [is] the prejudice individuals turn against themselves because they are members of a stigmatized group."	x				x		
Corrigan and Rüsch, 2002	"Self-stigma [is] the reactions which individuals turn against themselves because they are members of a stigmatized group."	x						
Corrigan and Wassel, 2008	"Self-stigma is internalization of public stigma."	х						

Table B.3—Continued

Source	Definition	Label ^a	Otherness ^b	Loss of Status ^c	Stereotype or Belief	Prejudice or Attitude	Discrimination or Behavior	Barrier to Care
Dalky, 2012	"'Self-stigma' or 'stigma perception,' is the extent to which an individual believes others stigmatize him or her because of mental illness."				х			
Dickstein et al., 2010	"Self-stigma refers to the internalization of [these] negative beliefs."				х			
Evans-Lacko, Brohan, et al., 2012	"Self-stigma [is] a process in which a person with a mental illness applies and internalizes stigmatizing attitudes and beliefs held by the public [and is linked to] lower self-efficacy, less treatment seeking, and higher rates of hospitalizations."			х	x	x		x
K. Fung, Tsang, and Cheung, 2011	"Self-stigmatization is regarded as the self- discredit of individuals via the internalization of negative stereotypes towards themselves and/or their social group."			х	x			
Kranke, Floersch, et al., 2011	"Self stigma pertains to the individual with mental illness internalizing rejection from society, and often leads to lower self-esteem and shame."		x	x				
Laraway, 2007 ^e	"Self-Stigma [is] an individuals' perception of themselves."				х			
Lucksted et al., 2011	"Self-stigma [is the process in which] a person internalizes stigmatizing societal messages about mental illness. Self-stigma can lead to depression, demoralization, poorer illness management, social avoidance, and obstruct the pursuit and achievement of recovery goals."		x		x			
MacInnes and Lewis, 2008	"Self-stigma [is] the reactions of stigmatized individuals towards themselves."					х		

Table B.3—Continued

Source	Definition	Label ^a	Otherness ^b	Loss of Status ^c	Stereotype or Belief	Prejudice or Attitude	Discrimination or Behavior	Barrier to Care
Mittal et al., 2012	"Self-stigma is defined as the perception of oneself as inadequate or weak if one were to seek professional help [and the] shame, evaluative thoughts, and fear of enacted stigma that results from individuals' identification with a stigmatized group that serves as a barrier to the pursuit of valued life goals."	х		х	x			x
Momen, Strychacz, and Viirre, 2012	"Self-stigma reflects an individual's internalization of the cultural beliefs about mental illness which leads to feelings of shame and inadequacy."				x	x		
Muñoz et al., 2011	"Self-stigma or internalized stigma [is] the stigma endorsed by the people with a mental illness about themselves just for having their mental illness."				x			
Parle, 2012	"Self-stigma or internalised stigma is the process in which people with mental health problems turn the stereotypes about mental illness adopted by the public, towards themselves. They assume they will be rejected socially and so believe they are not valued."		x	x	x			
Price, 2011	"Internalized stigma describes the individual's view of self due to the experience of mental illness."				х			
Wade, Post, et al., 2011	"Self-stigma is the fear of losing self-respect or self-esteem as a result of seeking help [and] a person's negative perceptions of him- or herself as a result of having a mental illness."			x	x			х

Table B.3—Continued

Source	Definition	Label ^a	Otherness ^b	Stereotype or Belief	Prejudice or Attitude	Discrimination or Behavior	Barrier to Care
Wright, Jorm, and Mackinnon, 2011	"Self-stigma—the stigmatizing views individuals have in regard to themselves."			x			

^a Label, mark, attribute, or identity.

^b Otherness, separation, or differentiation.

^c Loss of status or being disgraced, discredited, or dehumanized.

^d Also cited by Ben-Zeev et al., 2012; Greene-Shortridge, Britt, and Castro, 2007; Reavley and Jorm, 2011b.

^e Also cited by Yap, Wright, and Jorm, 2011.

Table B.4 Definitions of *Public Stigma*

Source	Definition	Label ^a	Otherness ^b	Loss of Status ^c	Stereotype or Belief	Prejudice or Attitude	Discrimination or Behavior	Barrier to Care
Ægisdóttir et al., 2011	"Public stigma refers to the negative social labels attached to persons who seek mental health services."	х						х
Ben-Zeev et al., 2012	"Public stigma is the phenomenon of the social groups endorsing stereotypes about, and subsequently acting against, individuals who report mental distress and seek treatment."				х		х	х
Chan, Mak, and Law, 2009	"Public stigma refers to general public's endorsement of a set of prejudicial attitudes, negative emotional responses, discriminatory behaviors, and biased social structures towards members of a subgroup."					x	x	
Corrigan, 2004a	"Public stigma [is the] ways in which the public reacts to a group based on stigma about that group."						x	
Corrigan, Markowitz, et al., 2003	"Three components of public stigma: stereotypes, prejudice and discrimination. Stereotypes are collectively held beliefs about the members of social groups—efficient means of categorizing information allowing people to quickly generate impressions and expectations of individuals who belong to a group. People who are prejudiced endorse these negative stereotypes. Discrimination is a behavioral response based on prejudice towards a minority group that may result in harm towards the members of that group (e.g., coercion in terms of mandatory treatment or segregation or unwillingness to help/active avoidance, social distancing). Stigma also stems from perceptions of cause and controllability (attribution theory)."	Х ,	x		x	x	X	

Table B.4—Continued

Source	Definition	Label ^a	Otherness ^b	Loss of Status ^c	Stereotype or Belief	Prejudice or Attitude	Discrimination or Behavior	Barrier to Care
Corrigan, Kerr, and Knudsen, 2005	"Public stigma is the phenomenon of large social groups endorsing stereotypes about and acting against a stigmatized group: in this case, people with mental illness."				х		x	
Corrigan, Rafacz, et al., 2010	"Public stigma is the prejudice and discrimination that occur when the general population endorses the stigma."					х	x	
Corrigan and Rao, 2012	"Public stigma refers to the negative attitudes held by members of the public about people with devalued characteristics."	x		х		x		
Corrigan and Wassel, 2008	"Public stigma represents what the public does to people who are marked with a mental illness."	x					х	
Dalky, 2012	"Public stigma encompasses reactions of the general public toward a group based on individual stigma directed toward that group."		х				x	
Dickstein et al., 2010	"Public stigma entails invalidating and unjustified beliefs (i.e., prejudices and endorsed stereotypes) about others."			х	х	x		
Greene- Shortridge, Britt, and Castro, 2007	"Public stigma is the reaction of the general public toward people with mental illness [and is] composed of stereotypes, prejudice, and discrimination."				х	x	x	
Kranke, Floersch, et al., 2011	"Public stigma pertains to negative social behaviors, reactions, and beliefs directed toward people with mental illness."				Х	х	x	
Laraway, 2007	"Public Stigma [is] public (mis)perceptions of individuals with mental illnesses."				x			

Table B.4—Continued

Source	Definition	Label ^a	Otherness ^b	Loss of Status ^c	Stereotype or Belief	Prejudice or Attitude	Discrimination or Behavior	Barrier to Care
Momen, Strychacz, and Viirre, 2012	"Public stigma relates to beliefs held by the general public about the attributes of those with mental illness that can consequently lead to prejudice and discrimination."	х			x	х	х	
Muñoz et al., 2011	"Social or public stigma [is] the stigmatizing attitudes about mental illness endorsed by the general population."					x		
Price, 2011	"Public stigma stems from society's negative beliefs and attitudes about mental illness and results in stereotyping and prejudice against those with psychiatric disorders."				x	x		
Rüsch, Angermeyer, and Corrigan, 2005	"Public stigma consists of these three elements—stereotypes, prejudice and discrimination—in the context of power differences and leads to reactions of the general public towards the stigmatised group."				x	x	x	
Wade, Post, et al., 2011	"Public stigma [is] the general public's negative reactions to those with a mental illness that can lead to avoidance, discrimination, and/or stereotyping."		x		x	х	x	

^a Label, mark, attribute, or identity.

^b Otherness, separation, or differentiation.

^c Loss of status or being disgraced, discredited, or dehumanized.

The stigmatization of PWMHDs and treatment seeking is a concern both in and out of the military. Several studies have assessed the prevalence of mental health stigma at the level of the U.S. general population. These studies have used a variety of items to measure stigma (see Table C.1).

Social Distance

A recent review (Parcesepe and Cabassa, 2012) of national stigma studies found that members of the general population vary in their desire for social distance from PWMHDs. Social distance is measured by asking respondents how they would feel about engaging in multiple social relationships varying in intimacy with people with MHD. In the 2006 GSS, respondents read a vignette depicting a person with either schizophrenia or depression. Respondents then answered the social distance questions, in response to one of the characters in the vignette. Figure C.1 shows that, in general,

Table C.1 Surveys Assessing Mental Health Stigma in the U.S. Population

Study	Year	Selected Stigma Items Measured in the Study
NSDUH	Annually since 1990	Reasons for not seeking mental health treatment, including opinions of neighbors; fear, shame, or embarrassment; effect on job
BRFSS	2007, 2009	Beliefs about caring and sympathy toward people with mental illness; treatment efficacy
GSS	1996, 1998, 2002, 2006	Social distance, perceived dangerousness, treatment endorsement, treatment efficacy ^a
NCS	1990–1992	Treatment-seeking intentions; comfort with talking to a professional; embarrassment about seeking help
NCS-R	2001–2003	Treatment-seeking intentions; comfort with talking to a professional; embarrassment about seeking help

^a These stigma items were in reference to a character depicted as having an MHD in a vignette read by respondents.



Figure C.1 Desired Social Distance from People with Schizophrenia or Depression

SOURCE: Davis and Smith, 2006. Figure adapted from Collins et al., 2012. RAND RR426-C.1

respondents desired greater social distance from people with schizophrenia, relative to people with depression (Pescosolido et al., 2010; Schnittker, 2008). In addition, willingness to engage in social relationships varied based on the intimacy of the possible relationship. In general, respondents were least willing to have someone with an MHD marry into their family or to have to work closely with him/her. Respondents were slightly more willing to spend an evening socializing with, be friends with, or move next door to a person with an MHD.

Stigmatizing Attitudes and Beliefs About People with Mental Health Disorders

A prevalent belief about people with MHD is that they are more dangerous than the rest of the population (Parcesepe and Cabassa, 2012). The 2006 GSS included measures of the perceived dangerousness of PWMHDs. This item referenced a vignette character described as having either depression or schizophrenia. Thirty-three percent of respondents believed that people with depression were likely to be dangerous, and 63 percent believed the same about people with schizophrenia (Pescosolido, Martin, et al., 2010; Schnittker, 2008).

The BRFSS included an item measuring attitudes of the public toward PWMHDs. This item assessed level of agreement with the statement "people are generally caring and sympathetic toward people with mental illness." Across 35 U.S. states, the percentage of respondents agreeing with this item varied from 35 percent to 67 percent, depending on the state (Centers for Disease Control and Prevention et al., 2012).

Beliefs About Treatment-Seeking

The 2006 GSS included several measures of beliefs about treatment-seeking. Respondents were asked what treatments they would recommend for vignette characters with schizophrenia or depression, as well as their perceptions of the efficacy of treatment, in general, for the characters with schizophrenia or depression. A very high proportion of respondents believed that treatment would be effective for both depression and schizophrenia (96 percent and 98 percent of respondents, respectively). Respondents frequently endorsed going to a medical doctor, psychiatrist, or other mental health professional as treatment options, along with taking a prescription medication (see Figure C.2). Smaller numbers endorsed checking into a mental hospital as a treatment option.

In addition to the GSS data, the BRFSS measured level of agreement with the statement "treatment can help people with mental illness leave normal lives." More than 80 percent of U.S. adult respondents agreed with this statement, with levels of agreement varying across the 35 U.S. states sampled (Centers for Disease Control and





SOURCE: Davis and Smith, 2006. Figure adapted from Collins et al., 2012. RAND *RR426-C.2*

Prevention et al., 2012). The NCS-R asked what percentage of people who see a professional for serious emotional problems are helped. A large proportion of the sample thought that the majority of people would be helped by treatment. Of the sample, 45.6 percent indicated that 50 to 74 percent would be helped, and 26.3 percent thought that 75 to 100 percent would be helped.

The NCS-R also assessed respondents' willingness to seek treatment if needed and their level of comfort with doing so (Mojtabai, 2007). Of NCS-R respondents, 83.1 percent said that they probably or definitely would go for professional help for a serious emotional problem. However, only 78.5 percent of respondents said that they would be comfortable talking to a professional about personal problems. Twenty-eight percent of respondents indicated that they would feel embarrassed if friends knew about their getting professional help, and 7.5 percent said that they would be very embarrassed. These responses support the idea that stigma is perceived as a barrier to needed treatment, as do data from the 2010 NSDUH (SAMHSA, 2012). In NSDUH, approximately 10 percent of respondents reported not seeking needed mental health treatment because they did not want others to find out or because neighbors or other members of the community would have a negative opinion. About 8 percent feared negative repercussions for their jobs or being committed or having to take medicine. We examined the societal costs of stigma on service members. To calculate the costs associated with mental health stigma, we construct a microsimulation model that assesses the societal costs within the United States of service members who do not seek appropriate mental health care because of public or self-stigma. These costs include treatment and rehabilitation costs for service members with PTSD and depression, medical costs associated with suicide attempts and completions, value of lives lost to suicide, and lost productivity stemming from PTSD and depression.

This appendix describes the regression analyses we used to derive key estimates for barriers to care used in the model, as well as the model's guiding assumptions and parameters and sensitivity analyses we conducted to determine the impact that each parameter has on cost outcomes.

Model Description

Microsimulation models are event-driven simulations that generate individual life histories that can vary by socioeconomic and health-related characteristics. Such models are appropriate when needing to capture the complex set of behavioral responses that exist for unique individuals. An advantage of the microsimulation approach is that it can treat MHDs as recurring conditions, allowing for both remission and relapse over time. The microsimulation model that we use for this task is a revised version of the one published by Kilmer et al. (2011) that was developed as part of the RAND Invisible Wounds project (Tanielian and Jaycox, 2008).

In this section, we provide a brief summary of the model framework (details can be found in Kilmer et al., 2011). The model takes a representative cohort of service personnel and models their life course over two years, taking into account probabilistic events that may occur as a result of an MHD. Health events include entering treatment for an MHD, remission, relapse, suicide attempts, and death by suicide. Other events included in the model are labor-market outcomes, such as retention within DoD, career progression within the military conditional on retention, employment in the civilian sector, and civilian earnings. Individuals in our model can take four MHD states: namely no MHD, PTSD, major depressive disorder (MDD), or comorbid PTSD/MDD. People who are in any one of the latter three states have MHDs and can be in any of the following three treatment states: no treatment (no Tx), usual care (UC), or evidence-based treatment (EBT). Therefore, there are ten population groups (i.e., $1 + 3 \times 3$) (see Table D.1). Someone's labor-market outcomes and his or her suicide-related probability depend on his or her specific population group. Change in someone's population group affects costs.

Cost Outcomes

We consider the following set of costs for each population group:

- total treatment cost (Ttreat): This is defined by the sum of the treatment costs over all PWMHDs' related treatment costs over the two-year period.
- total productivity lost (Tprod): Someone's productivity loss is defined by the differences between his or her MHD-free wage and his or her actual wage over the two-year period. If he or she remains MHD-free, his or her productivity loss is zero. The total productivity loss is found by summing the productivity losses over all people in our population cohort.
- total cost of a suicide (Tsui): This is defined by the total medical cost of fatal and nonfatal suicide events.
- total cost of suicide and value of life lost (TsuiV): This is defined by the sum of Tsui and the total cost or potency of life lost due death by suicide.

We also considered an aggregate cost by summing Ttreat, Tprod, and TsuiV.

The first step in the model is to initialize the population cohort. The original model used data of troops deployed as part of OEF or OIF on June 30, 2008. We have updated this data set to include all military members (active and reserve from all service branches) deployed in June 2012. The number of records in this data set is N = 183,527. These data contained information on service members' military status (active or reserve), branch, rank, time in service, sex, and age. There was missing data for the following variables; rank (n = 3, 0.002 percent), age (n = 86, 0.05 percent), and

Treatment		MHD						
Status	No MHD	MDD	PTSD	Comorbid PTSD/MDD				
No Tx	No MHD and therefore no Tx	MDD and no Tx	PTSD and no Tx	PTSD/MDD and no Tx				
UCC	N/A	MDD and on UC	PTSD and on UC	PTSD/MDD and on UC				
EBT	N/A	MDD and on EBT	PTSD and on EBT	PTSD/MDD and on EBT				

Table D.1 Population Groups in the Microsimulation Model

NOTE: N/A = not applicable.

time in service (n = 3,500, 2 percent). Hot-deck imputation was done to impute data for these missing observations. Out of this total number of records, n = 45,500 people are randomly selected and assigned an age, race/ethnicity, education based on the 2004 Congressional Budget Office report Education Attainment and Compensation of Enlisted Personnel, and civilian labor force status based on the 2012 Current Population Survey (Bureau of Labor Statistics, undated). Of these 45,500 records, a population of 40,686 troops (representing 22 percent of the 183,527) is assigned to develop MDD, PTSD, or both during the two years post-return. The remaining 4,814 troops (45,500 – 40,686) serve as the control group for the productivity calculations. Of the 40,686 people who develop MHDs, 68 percent are assigned PTSD. Two-thirds of these people are set to develop PTSD during the two-year period, and the remaining one-third are instead initialized to start with PTSD at the beginning of the simulation run. Half of those who have or who develop PTSD also have or develop comorbid MDD. The remaining individuals (i.e., 33 percent of 40,686 individuals) begin the simulation with MDD. The final step of the initialization process removes people from our sample population because of attrition from military service. This is done by applying a removal probability that depends on MHD status assigned to the person and comes from Hoge, Castro, et al. (2004).

Time Steps in the Model

We iterate the model in time steps representing one-quarter of a year. There are thus eight time steps considered in the model. In each time step, the model performs six steps:

- 1. The first step updates the population group for each individual. At the beginning of the quarter, each PWMHD may initiate treatment. This occurs with a probability given by the parameter *treatment probability*. Someone who initiates treatment may go into either UC or EBT. The probability that the person enters EBT is controlled by the parameter EBT Probability. The model assumes that (i) a person cannot switch between EBT and UC; (ii) the success rate of treatment (i.e., the remission rate) depends on the MHD and on the treatment type received by the person; and (iii) any beneficial effects of treatment (i.e., remissions) are not seen until the beginning of the next quarter. If treatment is successful, then the person is remitted and enters the no-MHD state. However, someone who has received treatment and who was remitted at some previous time step may relapse back to his or her previous MHD state. If instead treatment is not successful, the person is assumed to have an 80-percent chance of continuing the same course of treatment in the next quarter.
- 2. The second step in the model deals with suicide events. People who have PTSD or MDD may attempt suicide. The annual suicide attempt probabilities are taken from Gibbons et al. (2007). These probabilities depend on the person's

age group (0.01114 for those younger than 26, 0.00539 for those 26 to 45, and 0.00226 for those over 45) but not on his or her treatment status.

If someone attempts suicide, he or she will die from that attempt with a probability that depends on the military status: The probability is 8.4 percent if he or she is still actively serving in the military and 4.0 percent if he or she is a civilian (U.S. Army, 2007; Goldsmith et al., 2002).

- 3. The third step updates the labor status. Someone in the model can either stay as actively serving in the military or leave the military and enter one of three civilian labor-force status (CLFS) groups: full, part, or unemployed. The probabilities that determine which are based on a distribution of veterans in the 2012 Current Population Survey. The CLFS can change only when someone experiences a change in mental health status.
- 4. Wages are updated depending on military status. Military pay for full-time and reservists are pulled from the official 2012 military pay tables. These depend on rank and years of service. Civilian wages depend on the CLFS, demographic variables, and MHD status. For someone with a mental health condition, wages are decreased based on rates reported in Savoca and Rosenheck (2000).
- 5. The fifth step deals with active-duty transitions. Someone may leave full-time active duty at the end of the quarter. The probabilities for this transition come from Hoge, Castro, et al. (2004) and vary depending on mental health status and length of time since returning home. Service members may also leave active duty at the end of the quarter because of attrition. Depending on rank and branch, some service members who leave active duty enter the reserves.
- 6. The final step deals with promotions. The promotion probabilities come from the Defense Manpower Requirements Report (Office of the Under Secretary of Defense for Personnel and Readiness, 2012) and vary depending on rank and on branch. Individuals can be promoted only once during the two-year period that the model simulates.

Model Parameters

There are 16 key model parameters, which we list in this section. Baseline values are given in parentheses:

- treatment probability: the probability per quarter that a PWMHD initiates treatment (7.5 percent)
- PTSD probability: the proportion of veterans with MHDs who have PTSD or PTSD together with MDD (68 percent)
- EBT probability: the proportion of people who have an MHD and are initiating treatment who access EBT (30 percent)
- civilian PTSD wages: the decrement factor in wages received by a civilian affected by PTSD (15.75 percent)

- military PTSD wages: the decrement factor in wages received by an active service member affected by PTSD (7.88 percent)
- civilian MDD wages: the decrement factor in wages received by a civilian affected by MDD (45.23 percent)
- military MDD wages: the decrement factor in wages received by an active service member by MDD (22.62 percent)
- PTSD relapse: the probability per quarter that someone who was previously affected by PTSD will relapse (55 percent)
- MDD relapse: the probability per quarter that someone who was previously affected by MDD will relapse (54 percent)
- civilian suicide attempt: the probability that a suicide attempt by a civilian results in death (4.00 percent)
- military suicide attempt: the probability that a suicide attempt by a service member results in death (8.60 percent)
- remission PTSD UC: the probability per quarter that someone affected by PTSD and treated by UC is remitted (30 percent)
- remission PTSD EBT: the probability per quarter that someone affected by PTSD and treated by EBT is remitted (30 percent)
- remission MDD UC: the probability per quarter that someone affected by MDD and treated by UC is remitted (40 percent)
- remission MDD EBT: the probability per quarter that someone affected by MDD and treated by EBT is remitted (48 percent)
- suicide attempt age: a multiplicative factor that scales the suicide attempt rates based on age (1.00).

Sensitivity Analysis

Many of the parameters used in the model are the same as those used in the original version of the model by Kilmer et al. (2011). However, the model now uses an updated data set that includes military members (active and reserve, all four branches) deployed in June 2012 and updated wage data from the *Defense Manpower Requirements Report* for 2012 (Office of the Under Secretary of Defense for Personnel and Readiness, 2012). We updated all costs in the model to 2012 dollars using the consumer price index calculator. We have also updated the parameter value for treatment probability based on prevalence of treatment-seeking found in the regression analyses of the Invisible Wounds service member survey. Here, yearly prevalence at baseline was 26.8 percent. This corresponds to a quarterly treatment probability of 7.5 percent found by using the mathematical expression

$$q = (1-p)^{\frac{1}{4}},$$

relating the quarterly treatment probability q to the yearly treatment probability p. Updating the values for other model parameters, such as suicide rates and EBT Probability, proved to be a much harder task. One of the main aims of our meeting with the expert panel was to discuss the model parameter values used. To aid the discussion, we decided to run a sensitivity analysis on the model parameters to determine which parameters have the greatest impact on costs.

We carried out our sensitivity analysis as follows. For each of the 16 parameters, we ran our simulation model by first setting its value to a lower-bound value and then to an upper-bound value. We instead fixed the values of all other model parameters to their baseline values. Consequently, we ran a set of 32 simulation runs, a pair of runs for each model parameter. For most model parameters, we set the lower and upper values, respectively, to 90 percent and 110 percent of their baseline value. For other parameters, such as treatment probability, we set the lower and upper bounds as given by their 95-percent CIs. For each of our 32 simulation runs, we recorded the costs per observation (i.e., per number of PWMHDs). For a given pair of simulation runs, associated with a given parameter, we reordered the variation in the cost outputs. Large variations in the cost outputs gave an indication that the uncertainty in the parameter value has a big influence on costs. Table D.2 summarizes our findings.

Treatment probability with the four remission probabilities has the largest influence on Ttreat. However, other important parameters that have a large impact on costs include the parameter *suicide attempt age* that controls and modulates the probabilities of suicide attempts by age group given by Gibbons et al. (2007). Another important parameter is probability of EBT, which controls the proportion of individuals who begin treatment who go into EBT as opposed to UC.

	Five Most-Impactful Parameters								
Cost	1	2	3	4	5				
Tprod	Civilian MDD	Military MDD	Military PTSD	Treatment	Remission PTSD				
	wages	wages	wages	probability	UC				
Ttreat	Remission PTSD	Remission MDD	Remission PTSD	Remission MDD	Treatment				
	UC	EBT	EBT	UC	probability				
Tsui	Suicide attempt age	Treatment probability	Remission PTSD UC	Probability of EBT	PTSD probability				
TsuiV	Treatment	Remission PTSD	Suicide attempt	PTSD	Remission PTSD				
	probability	EBT	age	probability	UC				
Aggregated	Treatment	Remission PTSD	Probability of	Suicide attempt	Civilian MDD				
cost	probability	UC	EBT	age	wages				

Table D.2 Parameters Ranked According to Their Impact on Costs

Estimating the Effects of Stigma on Medical and Societal Costs

Our specific research task was to use the microsimulation model to estimate the medical and societal costs resulting from the mental health stigma. As we describe in Chapter Four, we intended to use evidence-based estimates to assess the added costs introduced by stigma. We approached this by running the model representing a world in which stigma was absent and then assessing the difference between this ideal condition and the actual conditions modeled in the simulation. This cost can be estimated by changing the relationship that links the magnitude of stigma to a service member's probability of seeking treatment. Thus, the parameter *treatment probability* depends on a measure of the level of stigma. Presumably, the baseline value used for treatment probability includes the effects of stigma in the status quo. In order to obtain the added costs introduced by stigma, we would need to run the model using a different value of treatment probability representing a world in which stigma was absent.

Literature Review

We conducted an extensive literature search that specifically focused on the association between mental health care utilization and stigma. Our literature search has focused on various settings, including military and nonmilitary, as well as U.S. and non-U.S. settings. Published studies used surveys to measure level of stigma. As described in Chapter Three, these surveys assessed public and personal stigma, opinions about MHDs, and attitudes toward counseling using a variety of question types and response options (e.g., "members of my unit might have less confidence in me," "my unit bosses might treat me differently").

Although one might expect that perceptions of greater levels of stigma would be clearly associated with lower probabilities of treatment-seeking propensity, the findings in the literature have been mixed. Some studies have suggested that stigma reduces the likelihood that someone will initiate mental health treatment. For example, Hoge, Castro, et al. (2004) focused on U.S. veterans returning from Iraq and Afghanistan and found that the perception of stigma is an important barrier to care especially among those suffering from MHDs. PWMHDs were found to be twice as likely to report concern about possible stigmatization than those with no MHDs. In this study, approximately 40 percent of PWMHDs reported that seeking treatment would be too embarrassing. Although this study is important because it shows that stigma is prevalent, particularly among those with MHDs, it did not measure a direct association between stigma and treatment utilization.

Pietrzak et al. (2009) conducted a study of the perceived stigma and barriers to mental health care utilization among U.S. veterans. They found a statistically significant, although weak, correlation between negative beliefs about mental health care and stigma and between negative beliefs and the likelihood of utilization of mental health counseling and medication services but did not assess the correlation between stigma and utilization. A much stronger relationship between stigma and likelihood of treatment-seeking was found in an Australian study (Barney, Griffiths, Jorm, et al., 2006). This study found that a person's internalization of stigma about help-seeking for depression reduces the likelihood of help-seeking from any professional source. Out of all the studies we explored, this study provides the strongest association between stigma and mental health care utilization.

Other studies suggest that stigma does not have any impact on mental health care utilization. Gould et al. (2010) recently found that service members who say that stigma would be a barrier to care are no more or less likely than other service members to be interested in care or to actually seek care. Other recent civilian research has also showed little association or, in some cases, no association of stigma with help-seeking (Schomerus and Angermeyer, 2008). Most studies on nonmilitary U.S. populations also found no significant relationship between perceived stigma and actual mental health treatment-seeking (S. Brown, 2010; Vogel, Wade, Wester, et al., 2007).

In summary, the majority of the studies support the finding that, although stigma is widely reported as a barrier to care, there is no evidence that it affects actual mental health care utilization. We were unable to find robust empirical published literature on the direct impact of stigma on mental health treatment-seeking. Therefore, we decided to derive an estimate using data from a RAND military survey that included questions on mental health utilization, as well as stigma and other barriers to care.

RAND Military Survey on Stigma and Other Barriers to Care

As part of the Invisible Wounds study, RAND researchers conducted a large population-based survey on people previously deployed as part of OEF or OIF (Schell and Marshall, 2008). This study offered the ability to assess the extent to which specific barriers to care are associated with subsequent mental health treatment in a sample of previously deployed U.S. service members who had been identified as having a need for mental health treatment. The researchers conducted baseline interviews between August 2007 and January 2008. A total of N = 2,120 respondents completed the interview.¹ To assess barriers to seeking health care for mental health concerns, respondents were asked, "If you wanted help for an emotional or personal problem, which of the following would make it difficult?" This question was followed by statements posed as potential barriers to treatment. Respondents endorsed each statement that they thought would make it difficult to get treatment by responding "yes." Potential barriers to care were drawn from three separate instruments: the NCS-R (see, e.g., Kessler, Chiu, et al., 2005); the Hoge, Castro, et al. (2004) study of barriers to care in the military; and our own instrument, which was developed for use among people with a range of traumatic experiences (e.g., Wong, Marshall, et al., 2006). From across

¹ This includes almost 200 individuals who were counted as nonrespondents in the original report (Schell and Marshall, 2008) but who completed the survey after analyses for the report were undertaken.

these instruments, we selected distinct barriers, maintaining all of the factors found by Hoge, Castro, et al. (2004) to be highly endorsed in a military sample.

The researchers then followed up with respondents approximately 18 months after baseline (N = 1,010). The majority of attrition was due to service members who had relocated since baseline and who lacked valid phone number information. They conducted all of the analyses using both poststratification weights and attrition weights. The poststratification weights matched the baseline sample to the characteristics of the previously deployed force on branch of services and, within each branch of service, on age, gender, rank, marital status, component (active versus reserve), and current duty status. Attrition weights resulted in a follow-up sample that matched the full baseline sample on all demographic, military, and mental health variables, including service utilization and probable mental health diagnoses. The current analyses were restricted to those who were service.² Finally, analyses were restricted to people who had a probable need for treatment at follow-up. This included anyone who experienced any one of the following:

- sought treatment in the past year
- thought that they should have gotten treatment but did not
- had a moderate level of PTSD symptoms in the past month but may not have a current diagnosis
- had a moderate level of depression symptoms in the past month but may not have a current diagnosis.

The final analytic sample included 279 active-duty service members who had a need for treatment at follow-up.

As part of this follow-up interview, the researchers assessed participants' mental health service utilization over the prior 12 months. This included the number and length of visits both to mental health specialty providers and to other medical providers for mental health symptoms or disorders. They assessed prescriptions for psychotropic drugs, as well as the duration of and adherence to pharmacotherapy. For the purposes of the current analyses, they used these mental health treatment items to derive a measure of minimally adequate care over the prior 12 months, using a definition adapted from P. Wang et al. (2005). This determination is designed to assess whether the participant has been engaged in at least a low level of mental health treatment during those 12 months (for precise definition, see Schell and Marshall, 2008). Similar to what the researchers found at baseline (Schell and Marshall, 2008), 27 percent of those service

² All analyses were replicated on the broader sample that includes people who have left the military. None of the conclusions or costs estimates differs substantively in this broader sample. However, the current service member subsample is being presented because it better represents the effects of stigma in the military.

members who had a need for treatment at follow-up received minimally adequate care during the prior 12 months.

RAND Regression Model: Stigma

To identify the possible effect of mental health stigma on service utilization, we constructed a logistic regression model in which a range of factors assessed at baseline predicted minimally adequate care at follow-up. These predictors included baseline mental health symptoms, mental health service utilization before baseline, age, gender, rank, branch of service, and component of service. In addition to these covariates, we included as predictors several variables related to endorsed barriers to mental health care from the baseline survey.

The analytic strategy was to combine the measures that were thought to be directly assessing the stigma of mental health symptoms or disorders or treatment into a measure of stigma. We identified five items as relatively direct measures of a concern that other people might evaluate or treat the respondent poorly because of their stereotypes or prejudice against those with mental health symptoms or disorders. We considered the following five barriers to be indicators of stigma:

- My friends and family would respect me less.
- My spouse or partner would not want me to get treatment.
- My co-workers would have less confidence in me if they found out.
- My commander or supervisor has asked us not to get treatment.
- My commander or supervisor might respect me less.

The number of these five items endorsed as a barrier to care was our measure for mental health stigma in the model predicting mental health treatment at followup. This count was standardized such that the mean score in our population was 1. Because the OR from the logistic regression model reflects a unit change of 1 in the predictor, this standardized measure reflects the odds of treatment, comparing the current level of stigma with one in which stigma was completely eliminated.

The other barriers to treatment assessed at baseline were also included in the model as covariates. We entered each individual barrier item as a separate predictor. These included the following:

- It could harm my career.
- I do not think my treatment would be kept confidential.
- I would not know where to get help or whom to see.
- It would be difficult to arrange transportation to treatment.
- It would be difficult to schedule an appointment.
- Mental health care would cost too much money.
- Even good mental health care is not very effective.
- The medications that might help have too many side effects.

- It would be difficult to get child care or time off of work.
- My family or friends would be more helpful than a mental health professional.
- Religious counseling would be more helpful than mental health treatment.
- I could lose contact or custody of my children.
- I could lose medical or disability benefits.
- I could be denied a security clearance in the future.
- I have received treatment before and it did not work.

In order to maximize the power to detect an effect of stigma, the model was pruned using a parsimony criterion. Specifically, covariates (i.e., predictors of treatment utilization other than stigma) were dropped from the model when doing so resulted in a more parsimonious model as judged by the Akaike information criterion (Akaike, 1987). This is a relatively conservative approach to model pruning that is approximately equivalent to removing unnecessary covariates sequentially until all remaining covariates are significant at the p < 0.20 level. The final (pruned) regression model includes stigma, age, officer status, branch or component, baseline PTSD symptoms, and the barriers listed below:

- I would think less of myself if I could not handle it on my own.
- I do not think my treatment would be kept confidential.
- It could harm my career.
- I could lose medical or disability benefits.
- My family or friends would be more helpful than a mental health professional.

Finally, we were worried that several of the barriers that were included as covariates may themselves be partially affected by concerns about stigma. In particular, "It could harm my career" could reflect a mix of stigma-related harms, as well as harms from military policies that are not themselves stigma (e.g., it could affect a security clearance or prevent someone from deploying with his or her unit). Similarly, "I do not think my treatment would be kept confidential" might indirectly reflect fears about stigma that would be the result of disclosure of treatment. To the extent that stigma causes these two barrier items, it may be inappropriate to include them as covariates in the model. Specifically, including them as covariates might result in a downward bias on the size of our estimate of stigma's effect on treatment utilization. To assess for this possibility, we conduct sensitivity tests in which the effect of stigma were estimated with and without those items in the model. Analysis of the data found that the yearly probability of treatment initiation by service members with a probable need for mental health treatment is 26.8 percent. Table D.3 shows the regression model result. Consistent with our review of the empirical published literature is our finding that stigma did not significantly predict subsequent treatment utilization.
Table D.3 Stigma Regression Model				
OR for Stigma	95% LL	95% UL		
0.997	0.745	1.334		

NOTE: LL = lower limit. UL = upper limit.

The regression model shows that, if we were to completely eliminate the effect of stigma, the yearly probability of seeking treatment would still remain 26.8 percent and that, with a 95-percent CI of the effect of stigma, we can be relatively confident that the true effect would result in rates of utilization between 21.5 percent and 32.9 percent. Table D.4 summarizes the results.

Some of the barriers used as covariates within the model may themselves be the causal effect of stigma. We therefore decided to rerun the model excluding particular barriers to see how this affected the estimate. For the first modification, we estimated the effect of stigma while not controlling for "I do not think my treatment would be kept confidential" (modification 1). Results are shown in Table D.5.

For the second modification, we estimated the effect of stigma while not controlling for either "I do not think my treatment would be kept confidential" or "It could harm my career" (modification 2). Table D.6 shows the results.

In summary, the effect of eliminating stigma in predicting mental health treatment utilization was never statistically significant at the p < 0.05 level, regardless of

P _o : Original Prevalence (%)	O: Original Odds	P _A : Adjusted Previous Estimate (%)	95% LL (%)	95% UL (%
26.8	0.366	26.8	32.9	21.5
Table D.5 Results of the F	Regression M	odel with Modificatio	on 1	
	Regression M	odel with Modificatio	on 1	
	Regression M O: Original Odds	P _A : Adjusted Previous	on 1 95% LL (%)	95% UL (%)

Table D.4

Table D.6 **Results of the Regression Model with Modification 2**

P₀: Original Prevalence (%)	O: Original Odds	P _A : Adjusted Previous Estimate (%)	95% LL (%)	95% UL (%)
26.8	0.366	30.5	36.3	25.2

whether we include as covariates the two barriers that may be indirect measures of stigma. In all cases, the effect size of stigma as a predictor of mental health treatment is descriptively small.

RAND Regression Model: Other Barriers to Care

Stigma, as defined in this study, was not found to affect treatment utilization. However, the data showed that utilization depends mostly on age, branch component, PTSD status, and two other barriers to care:

- I could lose medical or disability benefits.
- My family or friends would be more helpful than a mental health professional.

However, the first of these barriers was reported as a barrier only by those not in treatment; no one in treatment endorsed this barrier. Therefore, we were not able to estimate an OR for this first barrier, the fear that medical or disability benefits will be lost if one seeks treatment. However, we proceeded in constructing a regression model that predicts utilization based on the second barrier—namely, that support from family and friends provides a more helpful alternative to professional mental health treatment. Results of the regression model are shown in Table D.7. The OR for this barrier was significantly less than 1, showing a statistically significant effect of this barrier on treatment utilization; people who believed that mental health professionals were not particularly helpful were much less likely to subsequently seek mental health care.

Because 30 percent of our population sample reported believing that family or friends are more helpful than seeking professional mental health treatment, we used the estimate given in Table D.7 to predict utilization if the endorsement of this barrier could be brought down to 15 percent and 0 percent. Table D.8 shows results, which suggest an increase in mental health treatment of approximately 7 percent if this barrier could be eliminated.

Table D.7 Odds Ratio for the Barrier "My Family and Friends Would Be More Helpful Than a Mental Health Professional"

OR	95% LL	95% UL
0.295	0.121	0.718

Table D.8

Change in Treatment Utilization Based on a Reduction and Elimination of
the Barrier "My Family and Friends Would Be More Helpful Than a Mental
Health Professional"

Percentage Endorsing the Barrier (%)	P _A : Adjusted P ₀ : Original Previous Estima Prevalence (%) (%)		95% LL (%)	95% UL (%)
15	26.8	30.5	33.4	27.8
0	26.8	34.5	40.8	28.8

Expert Panel to Vet Model Assumptions and Parameters

In September 2013, we convened a panel of ten experts who were asked to provide feedback on

- model assumptions and parameters, particularly for those parameters that have the greatest impact on cost outcomes
- our approach to quantifying the effect of stigma on mental health treatment utilization
- our analysis of other barriers to care that may affect mental health treatment utilization.

This section summarizes the five issues we addressed and the feedback from experts.

Probability of Entering Evidence-Based Treatment

We asked expert panelists for feedback on an updated estimate for EBT probability and for feedback on the CI used for this parameter. We focused on this parameter because EBT probability has a strong impact on the aggregated cost and Tsui. In relation to the first two questions regarding parameter values used and their CIs, the expert panel did not have strong opinions opposing our parameter choices or the results of the sensitivity analysis. The experts viewed the choice of 30 percent for the EBT probability as reasonable. Following our meeting, Deniz Fikretoglu stated that, for the case of the Canadian Forces, "The concern has been that EBT may not be used as often or as effectively as we would like: Whether it's better than 30 percent or not would be just guesswork." This general view seemed to be shared by other experts in our panel and indeed in the previous expert panel that initially approved of the parameter values to be used in the model published by Kilmer et al. (2011). This indicates that our variability of ±10 percent in the CI seems to be a conservative choice.

Suicide Attempts

We also asked expert panelists for feedback on an updated estimate of suicide attempts among military service members and feedback on the CIs we used (i.e., ± 10 percent). Our sensitivity analysis also showed that the parameter value for Suicide Attempt Age is also very important in determining Tsui and aggregated cost. This is because the potency of life lost due to death by suicide is very high and this affects our outcome for the TsuiV, as well as the aggregated cost. Consequently, varying the probabilities that affect suicide attempt rates and deaths by suicide has a big impact on costs. The expert panel did not have any objections or concerns that we were using the same rates assumed by the previous version of the model based on Gibbons et al. (2007) findings. The expert panel did not know of any new studies or data source that we could use to extract suicide attempt rates in the military for PWMHDs and broken down by age group. An uncertainty CI of ± 10 percent over the rates reported by Gibbons et al. also did not find any particular objections.

Remission Rates

The model relies on four remission probabilities (i.e., remission PTSD UC, remission PTSD EBT, remission MDD UC, and remission MDD EBT), each of which has a significant impact on costs (in particular, Ttreat). We asked experts for feedback on updated remission rates and on the CIs used to generate those remission rates. We used the same baseline values for the remission probabilities as in the original model (Kilmer et al., 2011). The treatment success probabilities in the original model were based on remission rates reported in existing literature (Dimidjian et al., 2006; Keller et al., 2000; Kessler, Sonnega, et al., 1995; Kocsis et al., 1988; Ludman et al., 2007; Schnurr et al., 2007). The expert panel did not object to the baseline values used or the CI range assumed.

Stigma and Other Barriers to Care

As described previously, we found that changing the levels of stigma has little or no effect on the parameter treatment probability. Additionally, because we found that mental health stigma is not a significant barrier to treatment utilization, we decided to use RAND military survey data to identify other, more-significant barriers. We found that the perception that support from family and friends is more effective than professional mental health treatment is a significant barrier to care. We asked experts for feedback on these findings and for recommendations for any other studies that quantify the effects of stigma or other barriers to care on mental health utilization. The expert panel agreed with our findings from the literature search and from the analysis on the RAND military survey data that stigma does not influence utilization. However, the expert panel pointed out that we had focused only on the hypothesis that stigma of having an MHD affects the probability of seeking treatment. The expert panel suggested that stigma could influence other probabilities and processes not reflected in the

microsimulation model. Most importantly, it was argued that stigma could influence suicide attempt rates. The expert panel was not able to provide references on where such a link was found or by how much. Carole Roan Gresenz noted that the model assumes no effect of treatment on suicide rates. That is indeed true: All people in the model belonging to the same age group who currently have MHDs are assumed to have the same probability of a suicide attempt regardless of their treatment status or history. Therefore, there is no direct influence of treatment on suicide rates per individual. However, at the aggregate level, treatment does reduce the number of PWMHDs, which then indirectly affects the aggregated number of suicide attempts.

In summary, the expert panel indicated agreement with the model assumptions and parameters and made some recommendations about how model findings should be presented, which we have implemented in the next section.

Microsimulation Model Results

Following recommendations made by the expert panel, we first present model runs for the hypothetical case in which PWMHDs do not enter treatment. We then present model runs showing the model baseline, which represents suicide and cost outcomes without stigma or any other barriers to care removed. Finally, we present findings from the model run with stigma and with the barrier to care described above (i.e., perception that friends and family are more helpful than a mental health professional).

Simulation 1: No Service Members Receive the Needed Treatment

We model the hypothetical case in which PWMHDs do not enter treatment by setting the value of the probability of treatment to 0. We ran 100 independent model realizations, each using the same set of parameter values but with a different random seed. Therefore, each realization resulted in different dynamics due to chance effects only. Table D.9 shows the mean values over all 100 realizations of the total number of people who begin with or who develop MHDs during the two-year period. There are three rows in this table, each showing a population with one of the three MHD states who are not in treatment. We show also the mean number of suicide attempts and the mean number of deaths by suicide. Values given within the parentheses show the standard deviation obtained over our 100 realizations.

Table D.10 shows the total costs associated with each population group. It gives all costs in millions of dollars except for Tsui, which is given in thousands of dollars. Tsui includes all medical and procedural costs of a suicide attempt but excludes the value of life loss from death by suicide. However, we show the costs associated with suicides that include the value of life lost (TsuiV), and we give this in millions of dollars.

Group	Observations	Number of Suicide Attempts	Number of Deaths by Suicide
PTSD, no Tx	13,835	141	11
	(83)	(12)	(3)
PTSD/MDD, no Tx	13,820	141	11
	(82)	(13)	(3)
MDD, no Tx	13,022	200	15
	(33)	(16)	(4)

Table D.9 Suicide Attempts and Deaths by Suicide When the Probability of Treatment Is Zero

NOTE: Numbers in parentheses are standard deviations (SDs).

Table D.10
Mean Cost Output When the Probability of Treatment Is Zero

Group	Ttreat (\$)	Tprod (\$ millions)	Tsui (\$ thousands)	TsuiV (\$ millions)	Aggregated Cost (\$ millions)
PTSD, no Tx	0	111.97 (12.24)	744.36 (63.95)	93.90 (27.71)	204.87 (31.62)
PTSD/MDD, no Tx	0	263.69 (11.62)	745.75 (70.03)	92.10 (25.79)	355.78 (28.65)
MDD, no Tx	0	367.03 (10.62)	1,057.90 (85.76)	127.47 (34.71)	494.50 (36,35)

NOTE: Numbers in parentheses are SDs.

Simulation 2: Baseline Model Without Stigma or Other Barriers to Care

Tables D.11 and D.12 provide the corresponding tables for the probability of treatment set to its baseline value of 7.5 percent per quarter. All nine rows appear here for each of the MHD states and treatment states. According to our literature search and our regression analysis of the RAND military survey, stigma was found not to have a statistically significant effect on treatment utilization. Therefore, Tables D.11 and D.12 for the baseline case are unchanged by modifying the stigma level in our microsimulation model. Stochastic variability, driven by chance effects and shown by the SDs within the parentheses in the tables, dominates the variability that is observed if the probability of treatment is varied within the 95-percent CI bounds shown in Table D.6.

Simulation 3: Model with Stigma

Analysis of the regression data found that the yearly probability of treatment initiation by service members with MHDs is 26.8 percent. From this, we extracted that the quarterly probability of treatment initiation (i.e., the value of the treatment-probability parameter) would be 7.5 percent. The regression model shows that, if we were to com-

Group	Observations	Number of Suicide Attempts	Number of Deaths by Suicide
PTSD, no Tx	9,305	88	9
	(69)	(10)	(3)
PTSD, UC	3,162	28	1
	(101)	(6)	(1)
PTSD, EBT	1,363	12	0
	(101)	(3)	(1)
PTSD/MDD, no Tx	9,310	88	8
	(78)	(9)	(3)
PTSD/MDD, UC	3,163	29	1
	(109)	(6)	(1)
PTSD/MDD, EBT	1,358	11	0
	(98)	(4)	(1)
MDD, no Tx	6,976	110	12
	(55)	(11)	(3)
MDD, UC	4,219	42	1
	(126)	(7)	(1)
MDD, EBT	1,821	17	1
	(119)	(4)	(1)

Table D.11
Suicide Attempts and Deaths by Suicide at Baseline

NOTE: Numbers in parentheses are the SDs resulting from stochastic fluctuations.

pletely eliminate the effect of stigma, the yearly probability of seeking treatment would still remain 26.8 percent and that, with a 95-percent CI of the effect of stigma, we can be relatively confident that the true effect would result in rates of utilization between 21.5 percent and 32.9 percent. The medical and societal costs would therefore not change from the baseline costs, reported above. Therefore, we do not report these.

Simulation 4: Model with a Barrier to Care

Tables D.13 through D.16 present our results when exploring a reduction in the barrier to care relating to "my family and friends would be more helpful than a mental health professional." In Tables D.13 and D.14, we show results for the case in which the number of respondents who endorse this barrier is reduced by 50 percent and the probability of treatment increases to 30.5 percent. We ran 100 realizations of the model with this value for the probability of treatment. We further ran 100 realizations of the model with the probability of treatment set at the lower- and upper-bound values of the 95-percent CI, as given in Table D.10. This produced a range of outputs in observations, number of suicides attempted and deaths by suicide, and in costs. In the tables, we report the mean value given over the 100 realizations when the probability of treatment was set at 30.5 percent. In the parentheses, we report either (1) the SD due to the

Group	Ttreat	Tprod	Tsui	TsuiV	Aggregated Cost
	(\$ millions)	(\$ millions)	(\$ thousands)	(\$ millions)	(\$ millions)
PTSD, no Tx	0	66.56	460.15	70.43	136.99
	(0)	(7.71)	(54.14)	(27.18)	(27.5)
PTSD, UC	2.74	24.21	150.52	9.02	35.97
	(0.1)	(3.48)	(30.47)	(8.34)	(9.42)
PTSD, EBT	4.42	9.49	62.86	3.57	17.48
	(0.35)	(1.86)	(17.66)	(5.68)	(5.99)
PTSD/MDD, no Tx	0	160.76	460.64	67.71	228.47
	(0)	(6.65)	(48.26)	(22.47)	(23.73)
PTSD/MDD, UC	2.74	55.32	155.99	8.87	66.94
	(0.1)	(3.55)	(30.63)	(7.57)	(7.3)
PTSD/MDD, EBT	6.02	22.3	61.55	3.41	31.72
	(0.45)	(2.27)	(20.4)	(5.03)	(5.78)
MDD, no Tx	0	196.13	579.86	98.18	294.31
	(0)	(4.81)	(59.41)	(25.54)	(25.74)
MDD, UC	2.65	82.38	230.63	11.62	96.65
	(0.08)	(4.33)	(36.73)	(10.07)	(10.73)
MDD, EBT	3.09	33.82	93.65	4.84	41.75
	(0.2)	(2.94)	(22.27)	(6.05)	(6.77)

Table D.12		
Cost Output	at	Baseline

NOTE: Numbers in parentheses are the SDs resulting from stochastic fluctuations.

variability associated with stochastic chance-driven effects or (2) the range in values produced when the parameter value for the probability of treatment varies between its lower- and upper-bound values. We choose to report the bigger variability between the two.

Tables D.15 and D.16 show results for the case in which the number of respondents who endorse the friends-and-family barrier is reduced by 100 percent and the probability of treatment increases to 34.5 percent, as shown in Table D.10. Similarly to how we reported in the previous tables, we present the results over 100 independent realizations and varied the probability of treatment between the lower and upper bounds of the 95-percent CI given in Table D.10. We thus report in the parentheses the bigger of the two variabilities—namely, the stochastic variability and the variability resulting from a parameter-value change in the probability of treatment.

Summary of Medical and Societal Costs

Not surprisingly, as the probability of treatment increases (going down the rows in the tables in the previous section), total treatment costs increase. The elimination

Group	Observations	Number of Suicide Attempts	Number of Deaths by Suicide
PTSD, no Tx	8,746	81	8
	(8,316–9,168)	(78–88)	(3)
PTSD, UC	3,546	32	1
	(3,258–3,856)	(5)	(1)
PTSD, EBT	1,536	13	0
	(1,408–1,672)	(4)	(1)
PTSD/MDD, no Tx	8,747	82	8
	(8,301–9,158)	(9)	(3)
PTSD/MDD, UC	3,548	32	1
	(3,263–3,850)	(5)	(1)
PTSD/MDD, EBT	1,533	13	1
	(1,407–1,664)	(4)	(1)
MDD, no Tx	6,286	99	11
	(5,766–6,792)	(91–108)	(3)
MDD, UC	4,707	48	2
	(4,351–5,067)	(44–52)	(1)
MDD, EBT	2,028	20	1
	(1,873–2,186)	(4)	(1)

Table D.13 Suicide Attempts and Deaths by Suicide When the Barrier of Friends and Family Is Reduced by 50 Percent

NOTE: There are two sources of variability. The first is given by the regression model and provides a CI for the probability-of-treatment parameter. To explore this variability, we ran the microsimulation twice. The first run used the lower bound value for the probability-of-treatment parameter, and the second used the upper bound value for the treatment parameter.

In addition to this variability, there is stochastic variability. This is noise-induced or chance-driven variability. So, for each of the two runs, we ran 100 replications using the same settings and initial conditions. Although the settings are the same, each of these realizations can produce slightly different results because of chance. So what we report is the sum of the effects of these two variabilities that gives the widest range (i.e., difference) between the two.

In this table, the numbers in parentheses show the large value between (1) the range produced by varying the probability of treatment within the 95-percent CIs given by the regression model and (2) the SD resulting from stochastic fluctuations.

of the friends-and-family barrier does not produce a dramatic increase in treatment costs. Reducing this barrier by 50 percent would increase treatment costs by just under \$3 million. However, looking at Tprod and aggregated cost, we see that this may be a cost-effective intervention. Reducing this barrier by 50 percent would result in more than \$9 million in savings in lost productivity and aggregated cost. Not surprisingly, the benefit and cost-effectiveness is even greater if this barrier is completely eliminated (100-percent reduction). In this case, the aggregated cost would fall by more than

Group	Ttreat	Tprod	Tsui	TsuiV	Aggregated Cost
	(\$ millions)	(\$ millions)	(\$ thousands)	(\$ millions)	(\$ millions)
PTSD, no Tx	0	61.31	422.52	68.85	130.16
	(0)	(7.98)	(407.34–459.52)	(23.3)	(25)
PTSD, UC	3.09	26.96	175.22	11.91	41.97
	(2.83–3.36)	(24.68–28.75)	(26.54)	(9.48)	(10.47)
PTSD, EBT	4.98	10.98	71.73	4.14	20.1
	(4.54–5.44)	(2.42)	(22.32)	(5.46)	(6.26)
PTSD/MDD, no Tx	0	148.78	431.48	62.95	211.73
	(0)	(139.86–157.7)	(46.58)	(22.98)	(201.51–226.01)
PTSD/MDD, UC	3.09	61.76	172.93	10.29	75.15
	(2.83–3.37)	(57.26–66.72)	(26.94)	(9.25)	(71.59–82.66)
PTSD/MDD, EBT	6.84	25.15	71.15	4.32	36.31
	(6.26–7.45)	(23.32–26.89)	(19.94)	(6.07)	(6.73)
MDD, no Tx	0	177.64	517.47	91.18	268.82
	(0)	(162.88–190.98)	(478.57–566.94)	(27.26)	(251.18–289.96)
MDD, UC	2.98	91.44	257.7	16.05	110.46
	(2.74–3.23)	(85.22–98.69)	(237.27–281.31)	(11.37)	(100.92–118.91)
MDD, EBT	3.45	37.37	107.94	5.52	46.34
	(3.18–3.73)	(34.74–40.26)	(23.87)	(6.91)	(42.04–51.06)

Table D.14
Cost Output When the Barrier of Friends and Family Is Reduced by 50 Percent

NOTE: Like in Table D.13, numbers in parentheses are the large value between (1) the range produced by varying the probability of treatment within the 95-percent CIs given by the regression model and (2) the SD resulting from stochastic fluctuations.

\$32 million from the baseline. Notice that, if we eliminate this barrier, cost savings are more than double those produced by a 50-percent reduction (Table D.17). This is determined by both a larger cost saving in total productivity and a smaller number of suicide attempts and, although a marginal reduction, also a smaller number in deaths by suicide (Table D.18).

Literature Reviewed to Estimate Stigma

Alvidrez, Jennifer, Lonnie R. Snowden, and Dawn M. Kaiser, "The Experience of Stigma Among Black Mental Health Consumers," *Journal of Health Care for the Poor and Underserved*, Vol. 19, No. 3, 2008, pp. 874–893.

Aromaa, Esa, Asko Tolvanen, Jyrki Tuulari, and Kristian Wahlbeck, "Personal Stigma and Use of Mental Health Services Among People with Depression in a General Population in Finland," *BMC Psychiatry*, Vol. 11, No. 52, 2011.

Bambauer, K. Z., and H. G. Prigerson, "The Stigma Receptivity Scale and Its Association with Mental Health Service Use Among Bereaved Older Adults," *Journal of Nervous and Mental Disease*, Vol. 194, No. 2, 2006, pp. 139–141.

Group	Observations	Number of Suicide Attempts	Number of Deaths by Suicide
PTSD, no Tx	8,144	74	8
	(7,269–9,013)	(65–84)	(3)
PTSD, UC	3,964	35	1
	(3,371–4,586)	(31–40)	(1)
PTSD, EBT	1,716	14	1
	(1,451–1,972)	(4)	(1)
PTSD/MDD, no Tx	8,160	76	7
	(7,267–8,997)	(65–85)	(3)
PTSD/MDD, UC	3,964	35	1
	(3,376–4,587)	(30–40)	(1)
PTSD/MDD, EBT	1,707	15	1
	(1,453–1,978)	(4)	(1)
MDD, no Tx	5,580	87	10
	(4,563–6,606)	(74–104)	(3)
MDD, UC	5,193	53	2
	(4,479–5,902)	(46–58)	(2)
MDD, EBT	2,249	21	1
	(1,932–2,554)	(18–24)	(1)

Table D.15 Suicide Attempts and Deaths by Suicide When the Barrier of Friends and Family Is Reduced by 100 Percent

NOTE: Numbers in parentheses are the large value between (1) the range produced by varying the probability of treatment within the 95-percent CIs given by the regression model and (2) the SD resulting from stochastic fluctuations.

Barney, L. J., K. M. Griffiths, H. Christensen, and A. F. Jorm, "The Self-Stigma of Depression Scale (SSDS): Development and Psychometric Evaluation of a New Instrument," *International Journal of Methods in Psychiatric Research*, Vol. 19, No. 4, 2010, pp. 243–254.

Barney, L. J., K. M. Griffiths, A. F. Jorm, and H. Christensen, "Stigma About Depression and Its Impact on Help-Seeking Intentions," *Australian and New Zealand Journal of Psychiatry*, Vol. 40, No. 1, January 2006, pp. 51–54.

Blumenthal, R., and J. Endicott, "Barriers to Seeking Treatment for Major Depression," *Depression and Anxiety*, Vol. 4, 1996–1997, pp. 273–278.

Britt, Thomas W., Tiffany M. Greene-Shortridge, Sarah Brink, Quyen B. Nguyen, Jaclyn Rath, Anthony L. Cox, Charles W. Hoge, and Carl Andrew Castro, "Perceived Stigma and Barriers to Care for Psychological Treatment: Implications for Reactions to Stressors in Different Contexts," *Journal* of Social and Clinical Psychology, Vol. 27, No. 4, 2008, pp. 317–335.

Brown, C., K. O. Conner, V. C. Copeland, N. Grote, S. Beach, D. Battista, and C. F. Reynolds III, "Depression Stigma, Race, and Treatment Seeking Behavior and Attitudes," *Journal of Community Psychology*, Vol. 38, No. 3, April 2010, pp. 350–368.

Group	Ttreat	Tprod	Tsui	TsuiV	Aggregated Cost
	(\$ millions)	(\$ millions)	(\$ thousands)	(\$ millions)	(\$ millions)
PTSD, no Tx	0	55.18	387.47	65.76	120.95
	(0)	(47.52–64.26)	(340.57–441.35)	(22.6)	(102.47–134.51)
PTSD, UC	3.47	28.84	189.29	11.94	44.25
	(2.93–4.05)	(25.68–32.74)	(165.34–215.63)	(9.86)	(39.31–52.09)
PTSD, EBT	5.6	11.82	77.18	4.58	22
	(4.7–6.47)	(10.11–13.12)	(22.82)	(5.48)	(18.55–26.77)
PTSD/MDD, no Tx	0	135.8	395.85	61.13	196.93
	(0)	(116.88–155.54)	(342.64–445.35)	(22.28)	(174.55–223.84)
PTSD/MDD, UC	3.47	67.82	188.83	12.17	83.46
	(2.94–4.06)	(58.88–76.56)	(160.62–214.66)	(8.96)	(73.23–92.58)
PTSD/MDD, EBT	7.61	27.6	79.59	4.56	39.78
	(6.46–8.93)	(23.82–31.44)	(20.74)	(5.8)	(35.23–45.92)
MDD, no Tx	0	156.32	457.98	82.1	238.42
	(0)	(128.23–186.9)	(383.04–545.42)	(23.38)	(204.5–279.13)
MDD, UC	3.31	100.22	286.36	16.5	120.03
	(2.82–3.82)	(87.81–111.75)	(247.22–315.6)	(13.49)	(104.42–133.92)
MDD, EBT	3.85	41.11	115.11	7.16	52.12
	(3.28–4.41)	(36.11–45.91)	(98.91–129.03)	(7.8)	(46.12–58.9)

Table D.16	
Cost Output When the Barrier of Friends and Family Is Reduced by 100 Percent	

NOTE: Numbers in parentheses are the large value between (1) the range produced by varying the probability of treatment within the 95-percent CIs given by the regression model and (2) the SD resulting from stochastic fluctuations.

Table D.17 Total Aggregate Costs for Each Simulation (in millions of dollars)

Simulation	Ttreat	Tprod	Tsui	TsuiV	Aggregated Cost
No Tx	0	741.69	2.55	313.47	1,055.16
Baseline treatment	21.65	650.98	2.26	277.64	950.28
50% reduction in friends-and- family barrier	24.42	641.39	2.23	275.23	941.04
100% reduction in friends-and- family barrier	27.32	624.71	2.18	265.9	917.94

Brown, M. C., A. H. Creel, C. C. Engel, R. K. Herrell, and C. W. Hoge, "Factors Associated with Interest in Receiving Help for Mental Health Problems in Combat Veterans Returning from Deployment to Iraq," *Journal of Nervous and Mental Disease*, Vol. 199, No. 10, October 2011, pp. 797–801.

Simulation	Number of Suicide Attempts	Number of Deaths by Suicide
No treatment	482	37
Baseline treatment	425	33
50% reduction in friends-and- family barrier	420	33
100% reduction in friends-and- family barrier	410	32

Table D.18 Aggregate Suicide Attempts and Deaths by Suicide for Each Simulation

Bruffaerts, Ronny, Koen Demyttenaere, Irving Hwang, Wai-Tat Chiu, Nancy Sampson, Ronald C. Kessler, Jordi Alonso, Guilherme Borges, Giovanni de Girolamo, Ron de Graaf, Silvia Florescu, Oye Gureje, Chiyi Hu, Elie G. Karam, Norito Kawakami, Stanislav Kostyuchenko, Viviane Kovess-Masfety, Sing Lee, Daphna Levinson, Herber Matschinger, Jose Posada-Villa, Rajesh Sagar, Kate M. Scott, Dan J. Stein, Toma Tomov, Maria Carmen Viana, and Matthew K. Nock, "Treatment of Suicidal People Around the World," *British Journal of Psychiatry*, Vol. 199, 2011, pp. 64–70.

Cooper-Patrick, Lisa, Neil R. Powe, Mollie W. Jenckes, Junius J. Gonzales, David M. Levine, and Daniel E. Ford, "Identification of Patient Attitudes and Preferences Regarding Treatment of Depression," *Journal of General Internal Medicine*, Vol. 12, No. 7, 1997, pp. 431–438.

Corrigan, Patrick, "How Stigma Interferes with Mental Health Care," *American Psychologist*, Vol. 59, No. 7, October 2004, pp. 614–625.

Downs, Marilyn F., and Daniel Eisenberg, "Help Seeking and Treatment Use Among Suicidal College Students," *Journal of American College Health*, Vol. 60, No. 2, 2012, pp. 104–114.

Eaton, K. M., C. W. Hoge, S. C. Messer, A. A. Whitt, O. A. Cabrera, D. McGurk, A. Cox, and C. A. Castro, "Prevalence of Mental Health Problems, Treatment Need, and Barriers to Care Among Primary Care-Seeking Spouses of Military Service Members Involved in Iraq and Afghanistan Deployments," *Military Medicine*, Vol. 173, No. 11, November 2008, pp. 1051–1056.

Edlund, Mark J., John C. Fortney, Christina M. Reaves, Jeffrey M. Pyne, and Dinesh Mittal, "Beliefs About Depression and Depression Treatment Among Depressed Veterans," *Medical Care*, Vol. 46, No. 6, June 2008, pp. 581–589.

Eisenberg, Daniel, Marilyn F. Downs, Ezra Golberstein, and Kara Zivin, "Stigma and Help Seeking for Mental Health Among College Students," *Medical Care Research and Review*, Vol. 66, No. 5, October 2009, pp. 522–541.

Fikretoglu, D., S. Guay, D. Pedlar, and A. Brunet, "Twelve Month Use of Mental Health Services in a Nationally Representative, Active Military Sample," *Medical Care*, Vol. 46, No. 2, February 2008, pp. 217–223.

Gibbons, Robert D., C. Hendricks Brown, Kwan Hur, Sue M. Marcus, Dulal K. Bhaumik, and J. John Mann, "Relationship Between Antidepressants and Suicide Attempts: An Analysis of the Veterans Health Administration Data Sets," *American Journal of Psychiatry*, Vol. 164, No. 7, July 1, 2007, pp. 1044–1049.

Givens, J. L., I. R. Katz, S. Bellamy, and W. C. Holmes, "Stigma and the Acceptability of Depression Treatments Among African Americans and Whites," *Journal of General Internal Medicine*, Vol. 22, No. 9, September 2007, pp. 1292–1297.

Golberstein, Ezra, Daniel Eisenberg, and Sarah E. Gollust, "Perceived Stigma and Mental Health Care Seeking," *Psychiatric Services*, Vol. 59, No. 4, April 2008, pp. 392–399.

, "Perceived Stigma and Help-Seeking Behavior: Longitudinal Evidence from the Healthy Minds Study," *Psychiatric Services*, Vol. 60, No. 9, September 1, 2009, pp. 1254–1256.

Goldsmith, S., T. C. Pellmar, A. M. Kleinman, and W. E. Bunney, eds., *Reducing Suicide: A National Imperative*, Washington, D.C.: National Academies Press, 2002.

Gorman, Lisa A., Adrian J. Blow, Barbara D. Ames, and Philip L. Reed, "National Guard Families After Combat: Mental Health, Use of Mental Health Services, and Perceived Treatment Barriers," *Psychiatric Services*, Vol. 62, No. 1, January 1, 2011, pp. 28–34.

Gould, Matthew, Amy Adler, Mark Zamorski, Carl Castro, Natalie Hanily, Nicole Steele, Steve Kearney, and Neil Greenberg, "Do Stigma and Other Perceived Barriers to Mental Health Care Differ Across Armed Forces?" *Journal of the Royal Society of Medicine*, Vol. 103, No. 4, April 1, 2010, pp. 148–156.

Halter, M. J., "Stigma and Help Seeking Related to Depression: A Study of Nursing Students," *Journal of Psychosocial Nursing and Mental Health Services*, Vol. 42, No. 2, February 2004, pp. 42–51.

Hoerster, K. D., C. A. Malte, Z. E. Imel, Z. Ahmad, S. C. Hunt, and M. Jakupcak, "Association of Perceived Barriers with Prospective Use of VA Mental Health Care Among Iraq and Afghanistan Veterans," *Psychiatric Services*, Vol. 63, No. 4, April 2012, pp. 380–382.

Hoge, Charles W., Jennifer L. Auchterlonie, and Charles S. Milliken, "Mental Health Problems, Use of Mental Health Services, and Attrition from Military Service After Returning from Deployment to Iraq or Afghanistan," *Journal of the American Medical Association*, Vol. 295, No. 9, 2006, pp. 1023–1032.

Hoge, Charles W., Carl A. Castro, Stephen C. Messer, Dennis McGurk, Dave I. Cotting, and Robert L. Koffman, "Combat Duty in Iraq and Afghanistan, Mental Health Problems, and Barriers to Care," *New England Journal of Medicine*, Vol. 351, No. 1, July 1, 2004, pp. 13–22.

Iversen, Amy C., Lauren van Staden, Jamie Hacker Hughes, Neil Greenberg, Matthew Hotopf, Roberto J. Rona, Graham Thornicroft, Simon Wessely, and Nicola T. Fear, "The Stigma of Mental Health Problems and Other Barriers to Care in the UK Armed Forces," *BMC Health Services Research*, Vol. 11, No. 1, 2011, p. 31.

Jorm, Anthony F., Jo Medway, Helen Christensen, Alisa E. Korten, Patricia A. Jacomb, and Bryan Rodgers, "Attitudes Towards People with Depression: Effects on the Public's Help-Seeking and Outcome When Experiencing Common Psychiatric Symptoms," *Australian and New Zealand Journal of Psychiatry*, Vol. 34, No. 4, August 2000, pp. 612–618.

Kessler, Ronald C., Wai Tat Chiu, Olga Demler, and Ellen E. Walters, "Prevalence, Severity, and Comorbidity of 12-Month DSM-IV Disorders in the National Comorbidity Survey Replication," *Archives of General Psychiatry*, Vol. 62, No. 6, 2005, pp. 617–627.

Kilmer, Beau, Christine Eibner, Jeanne S. Ringel, and Rosalie Liccardo Pacula, "Invisible Wounds, Visible Savings? Using Microsimulation to Estimate the Costs and Savings Associated with Providing Evidence-Based Treatment for PTSD and Depression to Veterans of Operation Enduring Freedom and Operation Iraqi Freedom," *Psychological Trauma: Theory, Research, Practice, and Policy*, Vol. 3, No. 2, June 2011, pp. 201–211.

Kim, Paul Y., Thomas W. Britt, Robert P. Klocko, Lyndon A. Riviere, and Amy B. Adler, "Stigma, Negative Attitudes About Treatment, and Utilization of Mental Health Care Among Soldiers," *Military Psychology*, Vol. 23, No. 1, January 2011, pp. 65–81. Kim, Paul Y., Jeffrey L. Thomas, Joshua E. Wilk, Carl A. Castro, and Charles W. Hoge, "Stigma, Barriers to Care, and Use of Mental Health Services Among Active Duty and National Guard Soldiers After Combat," *Psychiatric Services*, Vol. 61, No. 6, 2010, pp. 582–588.

Leaf, Phillip J., Martha Livingston Bruce, and Gary L. Tischler, "The Differential Effect of Attitudes on the Use of Mental Health Services," *Social Psychiatry*, Vol. 21, 1986, pp. 187–192.

Lee, S., H. S. Juon, G. Martinez, C. E. Hsu, E. S. Robinson, J. Bawa, and G. X. Ma, "Model Minority at Risk: Expressed Needs of Mental Health by Asian American Young Adults," *Journal of Community Health*, Vol. 34, No. 2, April 2009, pp. 144–152.

Lee, S., A. Tsang, and K. Kwok, "Twelve-Month Prevalence, Correlates, and Treatment Preference of Adults with DSM-IV Major Depressive Episode in Hong Kong," *Journal of Affective Disorders*, Vol. 98, No. 1–2, February 2007, pp. 129–136.

Link, Bruce G., Elmer L. Struening, M. Rahav, Jo C. Phelan, and L. Nuttbrock, "On Stigma and Its Consequences: Evidence from a Longitudinal Study of Men with Dual Diagnoses of Mental Illness and Substance Abuse," *Journal of Health and Social Behavior*, Vol. 38, No. 2, June 1997, pp. 177–190.

Markowitz, Fred E., "The Effects of Stigma on the Psychological Well-Being and Life Satisfaction of Persons with Mental Illness," *Journal of Health and Social Behavior*, Vol. 39, No. 4, December 1998, pp. 335–347.

Mojtabai, R., M. Olfson, and D. Mechanic, "Perceived Need and Help-Seeking in Adults with Mood, Anxiety, or Substance Use Disorders," *Archives of General Psychiatry*, Vol. 59, No. 1, January 2002, pp. 77–84.

Momen, Nausheen, Chris P. Strychacz, and Erik Viirre, "Perceived Stigma and Barriers to Mental Health Care in Marines Attending the Combat Operational Stress Control Program," *Military Medicine*, Vol. 171, No. 10, October 2012, pp. 1143–1148.

Nam, Suk Kyung, Seong In Choi, Ji Hee Lee, Mi Kyoung Lee, A. Reum Kim, and Sang Min Lee, "Psychological Factors in College Students' Attitudes Toward Seeking Professional Psychological Help: A Meta-Analysis," *Professional Psychology: Research and Practice*, Vol. 44, No. 1, February 2013, pp. 37–45.

Ng, T. P., A. Z. Jin, R. Ho, H. C. Chua, C. S. Fones, and L. Lim, "Health Beliefs and Help Seeking for Depressive and Anxiety Disorders Among Urban Singaporean Adults," *Psychiatric Services*, Vol. 59, No. 1, January 2008, pp. 105–108.

Okumura, Y., and S. Sakamoto, "Depression Treatment Preferences Among Japanese Undergraduates: Using Conjoint Analysis," *International Journal of Social Psychiatry*, Vol. 58, No. 2, March 2012, pp. 195–203.

O'Mahony, Joyce Maureen, Tam Truong Donnelly, Shelley Raffin Bouchal, and David Este, "Cultural Background and Socioeconomic Influence of Immigrant and Refugee Women Coping with Postpartum Depression," *Journal of Immigrant and Minority Health*, Vol. 8, No. 1, January 2006.

Ouimette, Paige, Dawne Vogt, Michael Wade, Vanessa Tirone, Mark A. Greenbaum, Rachel Kimerling, Charlene Laffaye, Julie E. Fitt, and Craig S. Rosen, "Perceived Barriers to Care Among Veterans Health Administration Patients with Posttraumatic Stress Disorder," *Psychological Services*, Vol. 8, No. 3, August 2011, pp. 212–223.

Park, S., M. J. Cho, J. N. Bae, S. M. Chang, H. J. Jeon, B. J. Hahm, J. W. Son, S. G. Kim, A. Bae, and J. P. Hong, "Comparison of Treated and Untreated Major Depressive Disorder in a Nationwide Sample of Korean Adults," *Community Mental Health Journal*, Vol. 48, No. 3, June 2012, pp. 363–371.

Perlick, D. A., R. A. Rosenheck, J. F. Clarkin, J. A. Sirey, J. Salahi, E. L. Struening, and B. G. Link, "Stigma as a Barrier to Recovery: Adverse Effects of Perceived Stigma on Social Adaptation of Persons Diagnosed with Bipolar Affective Disorder," *Psychiatric Services*, Vol. 52, No. 12, December 2001, pp. 1627–1632.

Pietrzak, R. H., D. C. Johnson, M. B. Goldstein, J. C. Malley, and S. M. Southwick, "Perceived Stigma and Barriers to Mental Health Care Utilization Among OEF-OIF Veterans," *Psychiatric Services*, Vol. 60, No. 8, August 2009, pp. 1118–1122.

Rae Olmsted, Kristine L., Janice M. Brown, J. Russ Vandermaas-Peeler, Stephen J. Tueller, Ruby E. Johnson, and Deborah A. Gibbs, "Mental Health and Substance Use Treatment Stigma Among Soldiers," *Military Psychology*, Vol. 23, January 2011, pp. 52–64.

Rickwood, Debra, Frank P. Deane, Coralie J. Wilson, and Joseph Ciarrochi, "Young People's Help-Seeking for Mental Health Problems," *Australian e-Journal for the Advancement of Mental Health*, Vol. 4, No. 3, 2005.

Rochlen, Aaron B., Jonathan J. Mohr, and Byron K. Hargrove, "Development of the Attitudes Toward Career Counseling Scale," *Journal of Counseling Psychology*, Vol. 46, No. 2, March 1999, pp. 196–206.

Sansone, R. A., V. G. Matheson, G. A. Gaither, and N. Logan, "Concerns About Career Stigma by Military Parents of Children with Psychiatric Illness," *Military Medicine*, Vol. 173, No. 2, February 2008, pp. 134–137.

Savoca, E., and R. Rosenheck, "The Civilian Labor Market Experiences of Vietnam-Era Veterans: The Influence of Psychiatric Disorders," *Journal of Mental Health Policy and Economics*, Vol. 3, No. 4, December 1, 2000, pp. 199–207.

Schell, Terry L., and Grant N. Marshall, "Survey of Individuals Previously Deployed for OEF/ OIF," in Terri Tanielian and Lisa H. Jaycox, eds., *Invisible Wounds of War: Psychological and Cognitive Injuries, Their Consequences, and Services to Assist Recovery*, Santa Monica, Calif.: RAND Corporation, MG-720-CCF, 2008, pp. 87–116.

Schomerus, Georg, and M. C. Angermeyer, "Stigma and Its Impact on Help-Seeking for Mental Disorders: What Do We Know?" *Epidemiologia e psichiatria sociale*, Vol. 17, No. 1, January–March 2008, pp. 31–37.

Schomerus, Georg, Charlotte Auer, Dieter Rhode, Melanie Luppa, Harald J. Freyberger, and Silke Schmidt, "Personal Stigma, Problem Appraisal and Perceived Need for Professional Help in Currently Untreated Depressed Persons," *Journal of Affective Disorders*, Vol. 139, No. 1, June 2012, pp. 94–97.

Schomerus, Georg, H. Matschinger, and M. C. Angermeyer, "The Stigma of Psychiatric Treatment and Help-Seeking Intentions for Depression," *European Archives of Psychiatry and Clinical Neuroscience*, Vol. 259, No. 5, August 2009, pp. 298–306.

Shea, Munyi, and Christine J. Yeh, "Asian American Students' Cultural Values, Stigma, and Relational Self-Construal: Correlates of Attitudes Toward Professional Help Seeking," *Journal of Mental Health Counseling*, Vol. 30, No. 2, April 2008, pp. 157–172.

Sirey, Jo Anne, Martha L. Bruce, George S. Alexopoulos, Deborah A. Perlick, Steven J. Friedman, and Barnett S. Meyers, "Stigma as a Barrier to Recovery: Perceived Stigma and Patient-Rated Severity of Illness as Predictors of Antidepressant Drug Adherence," *Psychiatric Services*, Vol. 52, No. 12, December 2001, pp. 1615–1620.

Sirey, Jo Anne, Martha L. Bruce, George S. Alexopoulos, Deborah A. Perlick, Patrick Raue, Steven J. Friedman, and Barnett S. Meyers, "Perceived Stigma as a Predictor of Treatment Discontinuation in Young and Older Outpatients with Depression," *American Journal of Psychiatry*, Vol. 158, No. 3, March 1, 2001, pp. 479–481.

Stewart, Victoria, Perceived Social Support, Suicidal Ideation, Stigma, and Help Seeking Behaviors Among College Students, Scranton, Pa.: Marywood University, doctoral thesis, 3357205, 2009.

Stone, Fred P., *Attitudinal and Perceptual Barriers to Accessing Mental Health Services Among Members of the U.S. Air Force*, Salt Lake City, Utah: University of Utah, doctoral thesis, 98-015D, 1998.

Sudom, Kerry, Mark Zamorski, and Bryan Garber, "Stigma and Barriers to Mental Health Care in Deployed Canadian Forces Personnel," *Military Psychology*, Vol. 24, No. 4, January 2012, pp. 414–431.

Thornicroft, Graham, "No Time to Lose: Onset and Treatment Delay for Mental Disorders," *Epidemiology and Psychiatric Sciences*, Vol. 21, No. 1, March 2012, pp. 59–61.

Turner, Erlanger A., *Parental Attitudes Toward Child Mental Health Services: The Influence of Ethnicity and Child Characteristics on Help-Seeking Intentions*, College Station, Tex.: Texas A&M University, doctoral thesis, 2009.

U.S. Army, Suicide Risk Management and Surveillance Office, Army Behavioral Health Technology Office, *Army Suicide Event Report: Calendar Year 2006*, Tacoma, Wash.: Madigan Army Medical Center, 2007.

Vogel, David L., Nathaniel G. Wade, and Ashley H. Hackler, "Perceived Public Stigma and the Willingness to Seek Counseling: The Mediating Roles of Self-Stigma and Attitudes Toward Counseling," *Journal of Counseling Psychology*, Vol. 54, No. 1, January 2007, pp. 40–50.

Vogel, David L., Nathaniel G. Wade, S. R. Wester, L. Larson, and Ashley H. Hackler, "Seeking Help from a Mental Health Professional: The Influence of One's Social Network," *Journal of Clinical Psychology*, Vol. 63, No. 3, March 2007, pp. 233–245.

Vogel, David L., Stephen R. Wester, Meifen Wei, and Guy A. Boysen, "The Role of Outcome Expectations and Attitudes on Decisions to Seek Professional Help," *Journal of Counseling Psychology*, Vol. 52, No. 4, 2005, pp. 459–470.

Vogt, Dawne, "Mental Health–Related Beliefs as a Barrier to Service Use for Military Personnel and Veterans: A Review," *Psychiatric Services*, Vol. 62, No. 2, 2011, pp. 135–142.

Wang, J. L., "Perceived Work Stress, Imbalance Between Work and Family/Personal Lives, and Mental Disorders," *Social Psychiatry and Psychiatric Epidemiology*, Vol. 41, No. 7, July 2006, pp. 541–548.

Wang, Min-Ling, *The Cultural Beliefs and Mental Health Help-Seeking Behaviors of People with Schizophrenia in Taiwan*, Washington, D.C.: Catholic University of America, doctoral thesis, 2011.

Wong, E. C., G. N. Marshall, T. L. Schell, M. N. Elliott, K. Hambarsoomians, C. Chun, and S. M. Berthold, "Barriers to Mental Health Care Utilization for U.S. Cambodian Refugees," *Journal of Consulting and Clinical Psychology*, Vol. 74, No. 6, December 2006, pp. 1116–1120.

Wong E. C., T. L. Schell, L. H. Jaycox, G. N. Marshall, T. Tanielian, and J. N. Miles, "Mental Health Treatment Experiences of U.S. Service Members Previously Deployed to Iraq and Afghanistan," *Psychiatric Services*, Vol. 64, No. 3, March 1, 2013, pp. 277–279.

Wrigley, S., H. Jackson, F. Judd, and A. Komiti, "Role of Stigma and Attitudes Toward Help-Seeking from a General Practitioner for Mental Health Problems in a Rural Town," *Australian and New Zealand Journal of Psychiatry*, Vol. 39, No. 6, June 2005, pp. 514–521.

Identifying programs funded by DoD that focus on stigma reduction was not a straightforward task. As a result, we used a multifaceted approach to identifying programs for inclusion in this report. Our general approach was to identify as many potential programs as possible in order to ensure that we did not omit any and to apply the exclusion criteria (see next section) only after we had adequate information about each potential program, generally obtained through an interview with a program representative. The methods we used to identify programs were web and other media searching, scanning program materials, reviewing relevant documents that were available to the general public, consulting with military personnel, obtaining DoD lists of programs, consulting with topic-area experts within and outside of RAND, and snowball sampling.

Initial Program Identification

We identified 26 potential programs, five of which met our inclusion criteria and are included in this report. Determinations about inclusion and rationale for exclusion, as well as detailed descriptions of each program, are included at the end of this appendix.

Our criteria for inclusion were as follows:

- The activity focuses on stigma reduction. As mentioned in Chapter Six, programs that create a culture of help-seeking by universally discussing mental health concerns and encouraging treatment-seeking certainly contribute to stigma-reduction efforts in the military. However, it is difficult to quantify these efforts because service members are being educated about mental health concerns and encouraged to seek help as needed from a vast array of DoD programs and services. Our decision to exclude these programs as out of scope is not meant to suggest that these efforts are not important or not effective. Rather, we are taking these efforts as a given and focusing our resources on better understanding programs that have taken a more specific or targeted approach to stigma reduction.
- The program was focused on mental health and not treatment for drug or alcohol problems.

- The activity is sponsored or funded by DoD, including through
 - any DoD office, activity, agency, service, or command
 - the VA/DoD Joint Incentive Fund
 - any DoD memorandum of understanding or memorandum of agreement
 - funding by one of the branches of service.
- The activity has a target audience that includes, at a minimum, active-duty, National Guard, or Reserve component service members or their family members.
- The activity conducts its efforts either in theater or out of theater and was in operation at some point between January 2013 and July 2013.

Recruiting Program Representatives for Interviews

RAND staff made at least four attempts via phone or email to contact staff responsible for each potential program. For a small number of programs whose staff did not respond (n = 3), we were able to develop descriptions of the programs from publicly available documentation (primarily via the Internet) and make a determination of inclusion.

Program Interviews

We conducted 30- to 60-minute interviews with someone from each potential program that appeared to meet our inclusion criteria; some were subsequently excluded because of the information obtained during the interview. We conducted interviews between January 2012 and July 2013, and the information included in the program descriptions was correct as of the date of the interview. As an extra assurance, we sent a report including the program description prepared by RAND staff to each interviewee, who was asked to review the report and ensure its accuracy. RAND staff have not independently verified information reported to us by the program representatives.

The topics addressed during the interviews are as follows:

- branch of service and installations served
- organization responsible for administering the program
- start and end dates of the program in its current form
- mission, goal, or objectives of the program
- alignment with a specific theoretical model or definition of stigma
- stigma-reduction activities or services provided by the program (e.g., objectives, audience, mode, frequency, timing in relation to deployment)
- target for stigma reduction (individual or self, public, institution or organization)
- evidence base for the content of the program
- program funding and staffing.

Determination of Program Inclusion or Exclusion

We reviewed the activities and mission or goal of all programs to first determine whether the program met our inclusion criteria. Table E.1 lists all programs from which we sought information, whether the program was included in this report, and, if not, reason for its exclusion. For those meeting inclusion criteria, we used information from the interviews to categorize these programs into activities that address stigma in the institutional context, those that address it in the public context, and that that support

Program	Included or Excluded and Justification
Afterdeployment.org	Included
Ask, Care, Escort	Excluded: does not directly or primarily target stigma; focuses on culture shift
Army confidential alcohol treatment and education pilot	Excluded: focuses on alcohol and drugs
Army Suicide Prevention Program	Excluded: does not directly or primarily target stigma; focuses on culture shift
America's Heroes at Work	Excluded: not DoD funded
Battlemind	Excluded: no longer a stand-alone program; now part of Comprehensive Soldier and Family Fitness
Breaking the Stigma	Included
Buddy to Buddy	Excluded: not DoD funded
Comprehensive Soldier and Family Fitness	Excluded: does not directly or primarily target stigma; focuses on culture shift
EBH	Included
Healing Heroes	Excluded: no longer in existence
Marine Corps COSC	Excluded: does not directly or primarily target stigma; focuses on culture shift
Marine Corps Suicide Prevention Program	Excluded: does not directly or primarily target stigma; focuses on culture shift
Military and Family Life Counseling	Excluded: does not directly or primarily target stigma; focuses on culture shift
Military OneSource	Excluded: does not directly or primarily target stigma; focuses on culture shift
Military Pathways	Included
Navy Operational Stress Control	Excluded: does not directly or primarily target stigma; focuses on culture shift

Table E.1 Programs Contacted and Decisions About Inclusion

Table	E.1—	Continu	ed
-------	------	---------	----

Program	Included or Excluded and Justification
OSCAR	Excluded: does not directly or primarily target stigma; focuses on culture shift
PTSD Awareness Day	Excluded: not DoD funded
Real Warriors Campaign	Included
Soldier Evaluation for Life Fitness	Excluded: no longer in existence
Stigma Reduction Communication Campaign Research—Army	Excluded: not yet in operation
Suicide Awareness Voices of Education	Excluded: not DoD funded
Theater of War	Excluded: not DoD funded
Virtual Behavioral Telehealth and Technology at Tripler	Excluded: does not directly or primarily target stigma; focuses on culture shift
Yellow Ribbon	Excluded: does not directly or primarily target stigma; focuses on culture shift

NOTE: OSCAR = Operational Stress Control and Readiness.

people with mental health concerns. A program may contribute to stigma reduction in more than one context.

Descriptions of Programs Included

Program descriptions of those included in our analysis are listed in Tables E.2 through E.6.

Table E.2 Afterdeployment.org

Aspect	Description	
Program description		
Brief description of program and objectives		
Target population or branch of service	Service members, veterans, family, and providers; DoD-wide, although, for stigma, the focus is on the individual.	
Target outcomes	For stigma, the objective is to change attitudes toward treatment-seeking and treatment-seeking behaviors.	
Approach to stigma redu	ction	
Definition of stigma (or model)	The definition of stigma used was developed in consultation with several leading stigma researchers and extends beyond help-seeking to include self, family or peer, and institutional stigma. Materials also note, "The modern concept of stigma has four necessary parts: 1. Labeling someone with a condition (e.g., saying someone is 'depressed') 2. Stereotyping people with that condition (e.g., thinking that every depressed person is 'weak') 3. Creating a division—a superior 'us' group and a devalued 'them' group resulting in loss of status in the community 4. Discriminating against someone on the basis of their label."	
Target for stigma reduction	Increase awareness of one's own attitude toward mental health treatment- seeking by, for instance, addressing common myths.	
Stigma-reduction activities	Stigma is an important concern when considering the effectiveness of approaches to and recommendations for treatment for other psychological health-related topics addressed by the website. In the future, stigma will be folded into clinical topics on the website so that it has a more pervasive presence. For instance, a service member may use the website to assess sleep difficulties and receive the recommendation to seek further assistance but choose not to because of his or her misconceptions regarding stigma. Rather than maintaining a separate topic page on the website regarding stigma going forward, the redesigned website will address stigma within the sleep topic area itself. Within the current topic module, there is a self-administered assessment to identify the perceived comfort level for seeking treatment. It also contains eight videos of individuals sharing their personal stories about stigma toward help-seeking and links to additional resources. An interactive resource booklet addresses stigma myths, gives suggestions for changing stigma in oneself and others. A link is provided to allow interested visitors to enroll in an ongoing research project with T2's research group to further validate the stigma assessment that appears on Afterdeployment.org.	

Table E.2—Continued

Aspect	Description	
Status of implementation	on and evaluation	
Setting	Web-based program	
Implementation history	The website was started in 2008 and, as of March 2013, had been recently revised.	
Evaluation design and outcomes	The stigma assessment used in Afterdeployment.org was developed and validated by T2 researchers. Currently, the website provides an opportunity for users to enroll in an ongoing study to further validate the assessment fo a wider population.	
Managing office	T2 at DCoE	
Funder	DCoE	
Published information or website	http://afterdeployment.t2.health.mil/	

SOURCE: Interviews with program representatives. NOTE: T2 = National Center for Telehealth and Technology.

Table E.3 Breaking the Stigma

Aspect	Description	
Program description		
 Brief description of program and objectives Breaking the Stigma is a USASOC training program that aims to build resiliency and optimize performance by reinforcing the importance of maintaining psychological fitness and seeking behavioral health care when needed. The training is built around a 24-minute video in which series of senior-ranking, respected, and proven SOF operators share the stories of dealing with combat- and deployment-related issues and the consequences (to their units, careers, and families) of seeking or not see help. The training consists of viewing the video, a presentation by two the video participants, an overview of available resources, a briefing frunit providers, and a briefing from commanders (who share their own connections and experiences). Rather than being a remedial medical program targeting high-risk individuals, Breaking the Stigma is designed to be a command-driven seeking to achieving peak performance and resiliency. 		
Target population or branch of service	Breaking the Stigma focuses on reaching the soldiers and leaders in the SOF community.	
Target outcomes	Increased behavioral health care utilization; decreased critical-incident reports (e.g., violence, accidents, injuries); increased normalization and encouragement of treatment-seeking; increased awareness of the costs and benefits to self and unit of seeking help	

Table E.3—Continued

Aspect	Description	
Approach to stigma redu	ction	
Definition of stigma (or model)	No specific definition is used. Stigma is identified as a barrier to treatment- seeking. Additionally, stigma is associated with the negative perceptions about engaging in a particular type of activity or behavior.	
Target for stigma reduction	Stigma-reduction targets include dispelling myths related to negative consequences of seeking treatment, embedding providers within units to facilitate rapport and build trust, and emphasizing psychological performance as a component of peak performance and true leadership.	
Stigma-reduction activities	Stigma-reduction activities include presenting the Breaking the Stigma training to units and connecting soldiers and leaders with resources and materials (website, video, and downloadable guides).	
Status of implementation	and evaluation	
Setting	Primarily, Breaking the Stigma is presented to SOF units as a training, but resources are also available on its website, including a link to the video on YouTube.	
Implementation history	In April 2012, LTG John F. Mulholland, USASOC commanding general, began to focus on addressing stigma, enjoining commanders to address myths and set an example of supporting help-seeking behavior. The Breaking the Stigma video was created later that year and was incorporated into a training tasker. By December 2012, the training was being presented to SOF units. In April 2013, the video was posted to YouTube and has promoted widespread circulation.	
Evaluation design and outcomes	Breaking the Stigma is tracking YouTube hits, monitoring the percentage of help-seeking people who are referencing the video and training as a reason for seeking care, monitoring data reported from U.S. Army Public Health Command on the number of new cases of certain mental health conditions (higher rates may be an indication of higher utilization), and tracking critical incident reports (e.g., violence, accidents, injuries).	
Managing office	USASOC	
Funder	Internally funded by USASOC	
Published information or website	http://www.soc.mil/Stigma/Index.html	

SOURCE: Interviews with program representatives.

NOTE: A tasker is a formal request from one military division or department to another to perform a specific task.

Aspect Description	
Program description	
Brief description of program and objectives	The EBH program is an early intervention and treatment program that provides multidisciplinary behavioral health care to soldiers in close proximit to their unit areas and in close coordination with unit leaders. In doing so, the program streamlines the number of behavioral health providers involved in the treatment process and enables strong relationships to form between operational leaders and behavioral health providers.
Target population or branch of service	Army-wide
Target outcomes	The target outcomes are (1) early behavioral health identification and intervention, (2) increased soldier readiness (pre-, during, and postdeployment), (3) enhanced continuity of behavioral health care, (4) enhanced quality of behavioral health care, and (5) erosion of the stigma commonly associated with behavioral health care in the military setting.
Approach to stigma reduc	ction
Definition of stigma (or model)	No specific definition is used. The program utilizes a public health (preventive) approach to behavioral health.
Target for stigma reduction	Stigma-reduction targets include (1) unit and peer climate regarding behavioral health issues and treatment-seeking and (2) leadership support for behavioral health issues and treatment-seeking.
Stigma-reduction activities	The program establishes working relationships between behavioral health providers and key battalion personnel. Behavioral health providers meet with battalion commanders at least twice a month and work more frequentl with company commanders and first sergeants. Program staff can therefore observe trends in attitudes toward behavioral health within a given unit and can prompt an intervention if stigmatizing attitudes or behaviors are found to exist.
	Through its primary activities, the program also incidentally targets stigma as a barrier to care by encouraging soldiers to talk with their peers about behavioral health issues. Similarly, the program prompts commanders to become proponents of behavioral health.
Status of implementation	and evaluation

Table E.4 Embedded Behavioral Health

Setting Army installations The program began in January 2009 at Fort Carson in response to a series Implementation of homicides allegedly perpetrated by soldiers from units at Fort Carson. In history July 2009, a report titled Investigation of Homicides at Fort Carson, Colorado, November 2008–May 2009 (U.S. Army Center for Health Promotion and Preventive Medicine, 2009) was issued. In response to issues highlighted in the report, the program was expanded. It became an Army-wide program in 2012. The program staffs 13 behavioral health and support personnel per brigade combat team. Teams typically consist of one psychiatrist or psychiatric nurse practitioner, three social workers, three psychologists, two psychiatric assistants, one licensed practical nurse, one case manager, and two administrative staff members. One or two behavioral health officers, who belong to the brigade rather than medical team, work with the team when the unit is not deployed.

pect	Description	
Evaluation design and outcomes	 Evaluation efforts include the following: leader surveys, which are conducted among leadership at various levels within six months of implementation and 90 days before deplot ment. The surveys solicit general feedback on the program. soldier satisfaction surveys, which solicit general feedback on the program Army Public Health Command assessments (published in April 2010 ar September 2011), which outline some outcomes, including soldier satisfaction; trust of behavioral health providers; unit leader satisfaction with behavioral health; improvement of behavioral health no post; providers' timeliness, knowledge of soldiers' behavioral health histor and respect for soldiers; and location hours and convenience data on the number of soldiers sent off post for individual therapy, psychotropic medication, medical management, or inpatient hospitalization. 	
Managing office	MEDCOM/Behavioral Health System of Clinical Care	
Funder	MEDCOM public health funds	
Published information or website	None	

Table E.4—Continued

SOURCE: Interviews with program representatives. NOTE: MEDCOM = U.S. Army Medical Command.

Table E.5 Military Pathways

Aspect	Description
Program description	
Brief description of program and objectives	 The mission of Military Pathways (in conjunction with its parent program, Screening for Mental Health) is to promote the improvement of mental health by providing service members and their families with education, screening, and treatment resources; reduce stigma and empower participants to take the first step in addressing mental health concerns for themselves or a loved one; and educate military families and service members about treatment services available to them. Program activities include the following: online mental health screenings for depression, alcoholism, generalized anxiety disorder, bipolar disorder, PTSD, and adolescent depression. The web portal also includes a learning section, a military mental health blog, and a postassessment resource and referral section. Video Doctor, an interactive video program designed to walk users through stages of change to assist them in seeking treatment or in caring for themselves educational and promotional materials that are disseminated at a range of events, including the NASD, PTSD Screening Day, and NDSD SOS, an educational program for military middle and high school children (within the DoDEA system) a family resilience program, which disseminates educational and promotional materials to family readiness groups and family readiness centers.

Table E.5—Continued

Aspect Description		
Target population or branch of service		
Target outcomes	Target outcomes include (1) increases in the utilization of mental health screening tools and referral resources, (2) increases in help-seeking or self- care behaviors, and, ultimately, (3) improvements in mental health.	
Approach to stigma reduc	ction	
Definition of stigma (or model)	No specific definition is used. The program utilizes Prochaska and DiClemente's transtheoretical model of behavior change (2005) as the basis for the program.	
Target for stigma reduction	Stigma-reduction targets include (1) individual knowledge, attitudes, and beliefs about mental health, treatment-seeking, and recovery and (2) self-efficacy as it relates to treatment-seeking.	
Stigma-reduction activities	The program offers a video, "A Different Kind of Courage," in conjunction with a two-workbook curriculum. The video features current and former service members and spouses who share their personal stories of seeking treatment for mental health issues. The video can be accessed on the Military Pathways website. More generally, the program strives to normalize treatment-seeking through its range of educational activities.	
Status of implementation	and evaluation	
Setting	Online and in person at events on military installations	
Implementation history	In 2006, Screening for Mental Health was approached by the Deputy Assistant Secretary of Defense for Force Health Protection and Readiness to provide a mental health education program that reduces stigma, educates service members and military families about mental health and alcohol problems, and encourages appropriate help-seeking and treatment. Military Pathways has been offered in both English and Spanish to better address the cultural diversity of service members and families affected by deployment. The program's mental health self-assessments are currently administered to roughly 300,000 participants per year.	
Evaluation design and outcomes	 Various evaluation methods (both internal and external) have been used to ensure the success of the program: Internal: Weekly, monthly, and quarterly progress reports are provided to T2. These reports include data on numbers of screening, kit production, and quantity and quality of promotional material. In addition, online surveys that measure customer satisfaction are available, as are processes and tools that track customer comments and suggestions. Focus groups and personal interviews have been implemented to ensure cultural appropriateness of the interventions. External: Two program evaluation projects have been conducted: (1) The University of Connecticut Health Center completed an evaluation of the SOS suicide prevention program and its impact on military children enrolled in DoDEA and civilian high-impact schools, and (2) the University of New Hampshire completed an evaluation on the effectiveness of the online screening program and of NASD and NDSD in-person events. In addition, Westat evaluated the online screening module in 2006 and 2007. 	
Managing office	Screening for Mental Health (via government contract)	
Funder	Government contract	

Aspect	Description
Published information or website	http://www.militarymentalhealth.org/

Table E.5—Continued

SOURCE: Interviews with program representatives.

NOTE: NASD = National Alcohol Screening Day. NDSD = National Depression Screening Day.

Table E.6		
Real Warriors Campaign		

Aspect	Description	
Program description		
Brief description of program and objectives	 The Real Warriors Campaign is a multimedia public awareness campaign designed to encourage help-seeking behavior among service members, veterans, and military families coping with invisible wounds. Sponsored by DCoE, the campaign is an integral part of DoD's overall effort to encourage warriors and their families to seek appropriate care and support for psychological health concerns. The campaign is designed to do the following: Create awareness about the resources available for psychological health care and support among service members, their families, their commanders, and the public at large. Create understanding regarding the challenges service members may feel prevent them from seeking care or support for psychological health concerns. Create awareness about the concepts of resilience and early intervention, as well as the roles they play in successful care, recovery, and reintegration for returning service members and overall force readiness. 	
Target population or branch of service	 The Real Warriors Campaign provides information that is tailored to meet the needs of the military community, including the following: active-duty military members of the National Guard and reserve families veterans health care professionals. 	
Target outcomes	Increase the likelihood that a service member or veteran experiencing an invisible wound will reach out for appropriate psychological health care or support.	

Table E.6—Continued

Aspect	Description	
Approach to stigma reduction		
Definition of stigma (or model)	 According to the J-MHAT 7 survey in 2011 and previous Army MHAT surveys, warriors are concerned that seeking care could negatively affect their military careers; make them look weak to their peers and commands; or cause them to lose their security clearances. The Real Warriors Campaign is an integral part of DoD's overall efforts to encourage help-seeking behavior for invisible wounds and combat the stigma associated with seeking care that was identified by the DoD Task Force on Mental Health in its 2007 report. The campaign's framework is based on the health belief model (HBM), a social marketing theory that uses the desire to avoid negative health consequences as a motivator for behavior change and the belief in positive outcomes as a reinforcing principle. The HBM is built on four constructs: perceived susceptibility: a person's assessment of his or her risk of getting the condition perceived severity: a person's assessment of the seriousness of the condition and its potential consequences perceived barriers: a person's assessment of the positive consequences that facilitate or discourage adoption of the promoted behavior perceived benefits: a person's assessment of the positive consequences of adopting the behavior. 	
Target for stigma reduction	 The campaign is designed to do the following: Create awareness about the resources available for psychological health care and support among service members, their families, their commanders, and the public at large. Create understanding regarding the challenges service members may feel prevent them from seeking care or support for psychological health concerns. Create awareness of the concepts of resilience and early intervention, as well as the roles they play in successful care, recovery, and reintegration for returning service members and overall force readiness. The campaign uses a targeted messaging strategy and a variety of communication tactics to encourage help-seeking behavior for invisible wounds among service members, veterans, and military families. The campaign's core messages are specifically designed to address the psychosocial barriers and motivators to care for psychological health concerns that exist within the military community. 	

spect	Description
Stigma-reduction activities	 The campaign's core messages are specifically designed to address the barriers and motivators to seeking care for psychological health concerns that exist within the military community. The core messages let service members, veterans, and their families know the following: Reaching out for help is a sign of strength that benefits the person, his or her family, and the entire military community. It is possible to seek care for psychological health concerns and maintain a successful military or civilian career. Warriors are not alone in coping with psychological health concerns, and every service member, veteran, and his or her family members should feel comfortable reaching out to their units, chains of command, fellow warriors, and community resources for support. Experiencing psychological stress as a result of deployment is common and successful care and positive outcomes are greatly assisted by early intervention. To disseminate these key messages, the campaign uses multiple communication channels, including traditional and social media, outreach and partnership activities, conferences and events, multimedia, and web. All campaign messages, imagery, and resources are tailored to each branch and target audience (primary audience: active-duty service members, members of the National Guard and Reserve, veterans; secondary audience: families; tertiary audience: health care providers) to provide the most-relevant information and resources to each audience. The campaign features video profiles of real service members of varying ranks and services candidly sharing their stories of coping with and successfully sought care for psychological health concerns. The videos were developed as a result of primary research and literature and studies on stigma, mental health, and health behavior change indicating that service members wanted to see people like them who have coped with and successfully sought care for psychological health concerns. These

Table E.6—Continued

Status of implementation and evaluation

Setting

The touchstone of the campaign is its website, which includes more than 115 articles that provide audiences with practical tools, tips, and resources for coping with invisible wounds and building resilience; a live chat feature that offers 24/7 access to trained resource consultants at the DCoE Outreach Center; and video profiles of real service members of varying ranks and services who candidly share their stories of coping with and successfully seeking care for psychological health concerns. To disseminate its key messages, the campaign uses multiple communication

channels, including traditional and social media, outreach and partnership activities, conferences and events, multimedia, and web.

	Table	e E.6–	–Contii	nued
--	-------	--------	---------	------

Aspect	Description
Implementation history	 Description The Real Warriors Campaign was developed in response to the 2007 DoD Task Force on Mental Health recommendation 5.1.1.1, which stated that DoD "should implement an anti-stigma public education campaign using evidence based techniques to provide factual information about mental disorders." The goals and objectives of the campaign are further supported by numerous studies and reports, including those from J-MHAT 7, the U.S. Army, and the 2010 report of the DoD Task Force on the Prevention of Suicide by Members of the Armed Forces, which indicate that stigma remains a critical barrier to service members accessing needed psychological health care or support. For example, among recommendations from the 2010 DoD Task Force on the Prevention of Suicide by Members of the Armed Forces was an instruction to "develop messages focusing on help seeking behaviors, positive leadership, command involvement and promoting the well-being, connectedness and psychological well-being of service members." Additionally, one of the primary goals of the VA/DoD IMHS is to "advance [psychological health] care through community partnership, education, and successful public communication." The Real Warriors Campaign was developed in direct response to Public Law 109-163, the NDAA for FY 2006, § 723, which called for the Secretary of Defense to "establish within the DoD a task force to examine matters relating to mental health and the Armed Forces." The campaign also responds to DoD Directive 6200.04, "Force Health Protection (FHP)," which mandates that all service members, and the Military Health System shall promote, improve, conserve, and restore physical and mental well-being of members of the armed forces across the full range of military activities and operations. Subsequent mandates further support the campaign's continued execution, including the following: Public Law 110-417, the NDAA for FY 2009, § 733, called for a DoD task f

pect	Description
Evaluation design and outcomes	To continue to identify best practices, chronicle campaign achievements, and make necessary course corrections, the campaign team conducts weekly, monthly, and quarterly reviews and process evaluations of campaign activities. Additionally, the Real Warriors Campaign has conducted usability tests and heuristic evaluations of its website and certain interactive and mobile elements to assess design elements and information flow and make adjustments as needed to enhance user experiences with campaign online elements and incorporate visual cues to action to further motivate users to access available psychological health care tools and resources. Extensive market research, including focus groups, key-informant interviews a review of literature, and situational analysis that served as a basis for developing campaign messages and tactics that would effectively reach audiences and stakeholders were conducted prior to the launch of the campaign. In the spring of 2012, the Real Warriors Campaign team conducte updated market research to build on its formative research. Updated market research promotes greater understanding of commonly perceived barriers to care and patterns among military personal regarding help-seeking behavior and helps refinement of stigma-reduction approaches that promote the effective use of psychological health resources to guide future campaign outreach. As of June 2013, the campaign was also in the process of executing two concurrent tasks (focus groups and an extended outcome survey of the issues service members and veterans feel are important, specifically as they relate to help-seeking behavior for psychological health concerns). Findings from these tasks will be used to make strategic recommendations to adjust campaign strategies, as well as provide insight into any shifts in knowledge, attitudes, beliefs, or opinions applicable to the campaign, with the intent to drive down the stigma associated with seeking psychological health care, encourage help-seeking behavior, and promote psychological
Managing office	DCoE
Funder	DCoE
Published information or website	http://www.realwarriors.net

SOURCE: Interviews with program representatives.

NOTE: NDAA = National Defense Authorization Act. FY = fiscal year. IMHS = Integrated Mental Health Strategy.

We identified and analyzed policies in DoD that could contribute to the stigmatization of those who access mental health care. This process involved three steps. First, DCoE supplied lists of policies that could potentially have implications for mental health stigma or discrimination. Then, we created a decision tree and used it to systematically determine whether each policy was likely to reduce stigma and discrimination, contribute to it, or have no implications. We then analyzed the content of each policy deemed relevant to stigma and discrimination and summarized the implications. Each step is detailed in this appendix.

Obtaining Policies and Determining Mental Health Relevance

To begin identifying policies that might have implications for stigmatization of those accessing mental health care, DCoE's subcontractor, Dynamics Research Corporation (DRC), conducted a search of policies of the following organizations: Air Force, Army, Coast Guard, DoD, Marine Corps, National Guard (Air and Army), Navy, and BUMED. DRC searched these policies for the following terms related to mental health or stigma:

- suicid-
- mental-
- psych-
- emotion
- counseling
- behavior
- stigma
- access
- barriers
- to care
- help-

- discriminat- (searched only if the policy contained at least one of the first six search terms)
- prejud- (searched only if the policy contained at least one of the first six search terms)
- stereotyp- (searched only if the policy contained at least one of the first six search terms).

DRC then purged policies that did not contain at least one search term from above, then provided spreadsheets, one for each organization, containing the following information about each of the remaining policies:

- service
- date of issuance
- DoDI or policy number
- DoDI or policy name
- a short description of the policy
- results of searches for each of the terms listed above
- the web address for accessing the policy online
- any additional notes on the policy.

Data on the number of policies DRC searched, the number of policies containing the search terms (i.e., hits), and the number of policies containing the search terms but that were not accessible or searchable are contained in Table F.1.

Identifying Policies That Could Reduce or Contribute to Stigma

Although the identification of policies based on mental health-related and stigmarelated search terms was a useful first step in identifying potential stigma-related policies, not all policies were directly related to mental health or stigma. For example, a policy containing *mental* might not be related to mental health but rather contain such words as *environmental* or *fundamental*. In order to further narrow the scope of the policies to only those that have implications for the stigmatization of service members with mental health challenges or seeking mental health care, we conducted a secondary review of the policies.

We created a two-step decision tree to guide our review (see Figure F.1). In the first step, a member of our team reviewed the policy to determine whether it explicitly addressed mental health or illness or mental health care. Policies that did not were excluded from review. Policies identified as mental health–relevant were also condensed as much as possible, combining policies that were revised versions of previous policies or that represented different chapters of the same manual.

Service	Policies Searched	Hits	Policies That Were Inaccessible or Not Searchable
Air Force	4,603	1,336	379
Army	2,533	1,002	429
National Guard	281	49	80
Navy	1,368	273	23
BUMED	360	93	244
Coast Guard	611	202	12
Marine Corps	1,096	267	279
DoD	1,372	336	112
Total	12,224	3,558 (29% of total searched)	1,558 (13% of total searched)

Table F.1Summary of Dynamics Research Corporation Policy Search Results

SOURCE: DCoE.

Figure F.1 Policy-Analysis Decision Tree



RAND RR426-F.1

In the second step, the remaining policies were coded for ways in which they might reduce or contribute to mental health stigma or discrimination. The items in this list were developed from our literature review and conceptual model for stigma (Chapter Two) and were added to during the course of our policy review. A policy could reduce mental health stigma or discrimination if any of these is true about it:
- It mandates a *stigma-reduction intervention*, such as education or training or a media campaign (see the literature review for a description of stigma-reduction interventions).
- It mandates that a mental health professional provide *cognitive restructuring* to all service members seeking treatment to improve their ability to cope with stigma.
- It ascribes *consequences for behavior that is prejudicial or discriminatory* against PWMHDs. These could include personnel actions or legal sanctions.
- It mandates that *service members with mental illness be treated no differently* from those without mental illness. Some examples of this would be policies that mandate that service members with mental illness stay with their units and are not prohibited from deploying.
- It makes the *process of accessing mental health care* less stigmatizing than it otherwise would be. This could include offering confidential care or placing mental health providers in less distinct locations.
- It *protects the privacy or confidentiality* of a service member's mental health information.

A policy could contribute to mental health stigma or discrimination if any of these is true of it:

- It uses *terminology* that is outside the range of standard psychiatric or psychological practice and may be considered to negatively portray PWMHDs (e.g., "bizarre," "defective," "temper tantrum," "childish outbursts").
- It implies *incompetence* in PWMHDs.
- It prohibits certain actions of PWMHDs (e.g., use of firearms, promotion).
- It mandates or implies a *lack of privacy or confidentiality* of one's mental health information.
- It involves *non-mental health professionals in determining mental fitness* or interpreting the implications of mental health symptoms or disorders.
- It separates those who are at risk for MHDs from those who are not (e.g., mandates *mental health screening* for specific groups).

Policies that did not relate to stigma or discrimination were excluded from review. See Table F.2 for the number of policies falling into each category.

Organization	Relevant Policies	Policies That May Contribute to Stigma	Policies That May Reduce Stigma
Air Force	88	46	19
Army	104	46	24
National Guard	13	10	2
Navy	54	24	10
BUMED	17	6	5
Coast Guard	48	17	22
Marine Corps	54	32	14
DoD	66	28	25
Total	444	209 (47% of all relevant policies)	121 (27% of all relevant policies)

Table F.2 Number of Policies That May Contribute to or Reduce Stigma

NOTE: The number of policies that may contribute to or reduce stigma are not discrete. Some policies may include provisions that could both contribute to and reduce stigma.

This appendix contains the DoD and service-specific policies we identified as having potential implications for stigma.

Table G.1	
U.S. Department of Defense Policies That May Either Contribute to or Reduce Stigma	

Document Number	Description	Has Content That May Contribute to Stigma	Has Content That May Reduce Stigma
AI 8	Updates established [policy], guidance, and procedures for taking disciplinary and adverse actions	х	
AI 9	Provides procedures for processing complaints of discrimination in employment on the basis of race, color, religion, sex, age, national origin, physical and/or mental disability, and/or retaliation		х
AI 31	Assigns responsibilities and procedures for developing, implementing, and [evaluating] EEO, [affirmative employment program], and Diversity Programs		х
AR 12-15/ OPNAVINST 4950.1H/AFR 50- 29/MCO 4950.2	This joint regulation is a consolidation of several regulations that cover the education and training of foreign personnel. It prescribes policies, responsibilities, procedures, and administration for the education and training of foreign military trainees by the Departments of the Army, Navy, and Air Force as authorized by U.S. security assistance legislation.	x	
AR 190-5/OPNAV 11200.5D/ AFI 31-218(I)/ MCO 5110.1D/ DLAR 5720	[C]overs motor vehicle traffic supervision. It outlines policy on vehicle registration; implements the 0.08 blood alcohol content as the standard for adverse administrative actions prescribed in this regulation; permits involuntary extraction of blood under revised Military Rules of Evidence in cases [in which] intoxicated driving is suspected; provides policy on towing, storing, and impounding vehicles; adopts the National Highway Traffic Safety Administration technical standards for breathalyzer equipment; establishes traffic points for seat belt and child restraint device violations; and requires that new safety requirements be included in the installation traffic code.	x	

Document Number	Description	Has Content That May Contribute to Stigma	Has Content That May Reduce Stigma
AR 190-8/ OPNAVINST 3461.6/AFJI 31- 304/MCO 3461.1	This regulation implements [DoDD 2310.1] and establishes policies and planning guidance for the treatment, care, accountability, legal status, and administrative procedures for Enemy Prisoners of War, Civilian Internees, Retained Persons, and Other Detainees. This regulation is a consolidation of [AR 190-8] and [AR] 190-57 and incorporates [SECNAVINST] 3461.3 and [AFJI] 31-304 Policy and procedures established herein apply to the services and their capabilities to the extent that they are resourced and organized for enemy prisoner of war operations.	x	
AR 601-270/ OPNAVINST 1100.4C CH-2/ AFI 36-2003_IP/ MCO 1100.75F/ COMDTINST M1100.2E	This regulation covers military entrance processing station operational policies, programs, and procedures. It implements [DoDD 1145.2] governing personnel enlisting in the military and the processing of Selective Service registrants.		x
DoD 5200.2-R	Acceptance and retention of DoD military, civilian, consultant and contractor personnel and of granting such persons access to classified information or assignment to a sensitive position	x	
DoD 5210.42-R	[Implements] the PRP to select and maintain only the most reliable people to perform duties associated with nuclear weapons	x	
DoD 5400.11-R	Uniform procedures for implementation of the DoD Privacy Program	х	
DoD 6025.13-R	Various components [making up DoD's] efforts to ensure that beneficiaries receive [good] care	х	
DoD 6025.18-R	Uses and disclosures of protected health information.	х	х
DoD 6055.05-M	[Identifies] the known health risks associated with specific jobs, processes, and exposures	х	
DoDD 1020.1	This Directive implements [references] to prohibit discrimination based on handicap in programs and activities receiving Federal financial assistance disbursed by [DoD] and in programs and activities conducted by [DoD].		x

Document Number	Description	Has Content That May Contribute to Stigma F	Has Content That May Reduce Stigma
DoDD 1350.2	This Directive: 1.1. Reissues [an earlier directive]. 1.2. Regulates the Department of Defense MEO Program and assigns responsibilities for ensuring DoD-wide compliance with the broad program objectives outlined in [other references]. 1.3. Provides for education and training in [equal opportunity] and human relations. 1.4. Prescribes the functions of the [DEOC] and [the Defense Equal Opportunity Management Institute]. 1.5. Establishes Department- wide standards for discrimination complaint processing and resolution as set forth in the DEOC Task Force on Discrimination and Sexual Harassment Report [1995]. The recommendations contained in the DEOC Task Force Report are set out in [a later section in this directive]; implementation of this Directive shall be consistent with [other sections in this directive]. 1.6. Provides standard terms and definitions pertaining to the MEO program.		x
DoDD 1440.1	This Directive: 1.1. Implements [other references] by establishing the Civilian [EEO] Program, to include affirmative action programs, consistent with guidance from [EEOC, OPM], and the DoD Human Goals Charter 1.2. Consolidates in a single document provisions of [other references] and, therefore, cancels these references. 1.3. Authorize, as an integral part of the Civilian EEO Program, the establishment of [SEPs] entitled the [Federal Women's Program], the [Hispanic Employment Program], and the [Program for People with Disabilities], the Asian/Pacific Islander Employment Program , the [AIEP], and the [Black Employment Program]. 1.4. Establishes the [DEOC], the Civilian EEO Review Board, and SEP Boards. 1.5. Authorizes the issuance of DoD Instructions and Manuals to implement this Directive and guidance from standard-setting Agencies such as EEOC and OPM, consistent with DoD 5025.1-M		X
DoDD 5200.2	1.1. Reissues [an earlier directive] to update the policy and responsibilities for the DoD Personnel Security Program under [other references]. 1.2. Continues to authorize the publication of DoD 5200.2-R in accordance with DoD 5025.1-M		x
DoDD 5220.6	This Directive reissues [an earlier directive] to update policy, responsibilities, and procedures of the Defense Industrial Personnel Security Clearance Review Program implementing enclosure 1.	x	

Document Number	Description	Has Content That May Contribute to Stigma	Has Content That May Reduce Stigma
DoD GEN-36A/ DA Pam 600- 2(Rev. 1988)/ NAVEDTRA 46905-A/Navy Stock No. 0503- LP-001-1760/ AFP 190-13(Rev. 1988)/NAVMC 2563(Rev. 1988)	[Book on serving as an officer in the U.S. armed forces]	х	
DoDI 1300.18	Assigns responsibilities and establishes uniform personnel policies and procedures for reporting, recording, notifying, and assisting the [next of kin] whenever DoD casualties are sustained	х	
DoDI 1325.07	Implements policy, assigns responsibilities, and prescribes procedures to carry out the administration and operation of military correctional programs and facilities and the administration and operation of military clemency and parole programs	х	
DoDI 1342.12	Implements policy, assigns responsibilities, and prescribes procedures under 20 U.S.C. Chapter 33 [Education of Individuals with Disabilities]	x	
DoDI 1400.25- V810	This Volume of this Instruction implements DoD policy, prescribes procedures, and delegates authority on implementing the DoD injury compensation program under the "Federal Employees' Compensation Act" (FECA), section 8101 of title 5, United States Code (U.S.C.) which provides benefits to civilian employees of the Federal Government for disability due to personal injury, disease, or death arising from or within the scope of their employment.		Х
DoDI 2310.4	Implements policy, assigns responsibilities, and prescribes procedures for repatriating U.S. military personnel, DoD civilian employees, and DoD contractor employees who have been [prisoners of war], held hostage by terrorists (inside or outside [CONUS], detained in peacetime by a hostile foreign government evading enemy capture, or were otherwise missing under hostile conditions		x
DoDI 3020.41	Establishes policy, assigns responsibilities, and provides procedures for OCS, including OCS program management, contract support integration, and integration of defense contractor personnel into contingency operations outside the United States	x	
DoDI 3025.19	Establishes policy, assigns responsibilities, and provides procedures for employment of DoD capabilities and information sharing in support of the [U.S. Secret Service]	x	

Document Number	Description	Has Content That May Contribute to Stigma	Has Content That May Reduce Stigma
DoDI 5154.30	Implements policy, assigns responsibilities, and prescribes procedures for the administration and management of [the Armed Forces Institute of Pathology]		x
DoDI 5210.65	Reissues policy and responsibilities for the management of the DoD Chemical Agent Security Program	х	
DoDI 5210.87	Implements policy, assigns responsibilities, and prescribes procedures for the screening, nomination, selection, approval and continued evaluation of DoD military and civilian personnel and contractor employees assigned to, or utilized in, [presidential support activities]	x	
DoDI 5210.89	Establishes minimum standards for securing and safeguarding [biological select agents or toxins] in [DoD] custody or possession	х	
DoDI 5525.14	Establishes policy, assigns responsibilities, and prescribes procedures for DoD [law enforcement officers flying armed] aboard commercial aircraft	х	
DoDI 6000.14	Establishes policy, assigns responsibilities, and provides procedures for implementation of the Consumer Bill of Rights and Responsibilities		х
DoDI 6040.45	Establishes policy and assigns responsibilities for [service treatment record] and [nonservice treatment record] life cycle management	х	х
DoDI 6060.2	Updates policy, responsibilities, and procedures for [child-development programs] for eligible minor children of DoD military and civilian personnel		х
DoDI 6060.4	Implements policy, assigns responsibilities, and prescribes procedures for youth programs (YPs) for children and youth, kindergarten through grade 12, of DoD military members, civilian employees, and other eligible patrons of [morale, welfare, and recreation] programs		х
DoDI 6130.03	a. Reissues [DoDD] 6130.3 as a [DoDI] in accordance with the authority in DoDD 5124.02 to establish policy, assign responsibilities, and prescribe procedures for physical and medical standards for appointment, enlistment, or induction in the Military Services. b. Establishes medical standards, which, if not met, are grounds for rejection for military service. Other standards may be prescribed for a mobilization for a national emergency. c. Incorporates and cancels DoDI 6130.4	x	

Document Number	Description	Has Content That May Contribute to Stigma	Has Content That May Reduce Stigma
DoDI 6200.03	Ensures mission assurance and readiness by protecting installations, facilities, personnel, and other assets in managing the impact of public health emergencies caused by all-hazards incidents		х
DoDI 6400.06	Establishes, implements, and updates domestic abuse policies; identifies and assigns responsibilities for preventing and responding to domestic abuse		х
DoDI 6490.03	Assigns responsibilities for deployment health activities		x
DoDI 6490.04	Implements DoD policy; assigns responsibility; and prescribes procedures for the referral, evaluation, treatment and administrative management of service members who may require mental health evaluation, psychiatric hospitalization and/or assessment of risk for potentially dangerous behavior	x	x
DoDI 6490.05	Establishes policy and assigns responsibilities for developing [COSC] programs within the Military Departments, the Combatant Commands, and joint service operations		Х
DoDI 6490.06	Establishes and implements counseling policies and identifies and assigns responsibilities for providing counseling support	х	x
DoDI 6490.07	Establishes policy, assigns responsibilities, and provides procedures for ensuring that service members and DoD civilian employees, including Coast Guard service members and civilian employees at all times, including when the Coast Guard is a service in the Department of Homeland Security deployed and deploying on contingency deployments are medically able to accomplish their duties in deployed environments	х	
DoDI 6490.08	Establishes policy, assigns responsibilities, and prescribes procedures for health care providers for determining command notification requirements as applied to: (1) service members' involvement in mental health care and (2) service members voluntarily seeking drug and alcohol abuse education (as distinguished from substance abuse treatment), requiring DoD personnel to receive education pertaining to drug and alcohol abuse	х	x
DoDI 6490.09	Establishes policy, assigns responsibilities, and prescribes procedures to ensure visible leadership and advocacy for the psychological health and mental health disease and injury protection of the Military Service members		x

Document Number	Description	Has Content That May Contribute to Stigma	Has Content That May Reduce Stigma
DoDI 6490.10	Establishes policy for the Military Departments, assigns responsibilities, and prescribes guidelines for establishment of Military Department policy and procedures to ensure continuity of behavioral health care at the losing and gaining installations when service members transition from one health care provider to another when transferring to a new duty station or transitioning out of the service	x	x
DoDI 6495.02	Implements policy, assigns responsibilities, provides guidance and procedures, and establishes the [SAAC] for the DoD [SAPR] Program		x
DoDM 5105.21- V3	Assigns responsibilities and prescribes procedures for the implementation of [the Director of Central Intelligence] and Director of National Intelligence	x	
DoDM 7730.47- M-V1	Assign[s] responsibilities and prescribe procedures for managing the [Defense Incident-Based Reporting System]	х	
DTM-11-011	Establishes the policy for person-to-person mental health assessments for each Service member deployed in connection with a contingency operation in accordance with section 1074m of title 10, United States Code	х	x
DTM-11-015	In accordance with the authority in [DoDD 5124.02], establishes policy, assigns responsibilities, and prescribes procedures for the [Integrated Disability Evaluation System], which is superseding the legacy [Disability Evaluation System].	х	
Manual for Courts-Martial	The Manual for Courts-Martial , United States (2012 Edition) updates the [manual] (2008 Edition). It is a complete reprinting and incorporates the [Manual for Courts-Martial] (2008 Edition), including all amendments to the Rules for Courts-Martial, Military Rules of Evidence (Mil. R. Evid.), and Punitive Articles made by the President in [EOs] from 1984 to present, and specifically including EO 13468 (24 July 2008); EO 13552 (31 August 2010); and EO 13593 (13 December 2011). See [appendix].	x	x

NOTE: AI = administrative instruction. EEO = equal employment opportunity. OPNAVINST = Office of the Chief of Naval Operations instruction. AFR = Air Force Reserve. DLAR = Defense Logistics Agency regulation. AFJI = Air Force joint instruction. SECNAVINST = Secretary of the Navy instruction. COMDTINST = commandant instruction. PRP = Personnel Reliability Program. DEOC = Defense Equal Opportunity Council. EEOC = Equal Employment Opportunity Commission. OPM = Office of Personnel Management. SEP = special-emphasis program. AIEP = American Indian/Alaskan Native Employment Program. NAVEDTRA = Naval Education and Training. NAVMC = Navy and Marine Corps form. DA = Department of the Army. CONUS= continental United States. OCS = operational contract support. SAAC = Sexual Assault Advisory Council. SAPR = Sexual Assault Prevention and Response. DTM = directive-type memorandum. EO = executive order.

Document Number	Description	Has Content That May Contribute to Stigma	Has Content That May Reduce Stigma
ADRP 6-22	ADRP 6-22 establishes and describes the leader attributes and core leader competencies that facilitate focused feedback, education, training, and development across all leadership levels.	х	х
AR 1-75	This regulation implements [DoDD] 5132.3 and supplements DOD 5105.38-M. This revision updates the regulation due to legislative changes to the Foreign Assistance Act of 1961, the Arms Export Control Act, and internal DOD policy. The revision also incorporates changes involving the activities supporting security assistance organizations. This regulation assigns responsibilities and provides guidance for assignment of personnel to security assistance organizations; morale, welfare, and recreational activities of security assistance organization personnel; development of the security assistance organization budget and fiscal procedures; preparation of Joint Tables of Allowances; and settlement of Foreign Military Sales claims of foreign governments against security assistance organization personnel.	x	
AR 20-1	This regulation prescribes the responsibility and policy for the selection and duties of [IGs] throughout the Army. It describes [IG] functions, including teaching and training, inspections, assistance, and investigations. Although the [fundamental IG] role and functions have not changed, this revision incorporates numerous policy and mandated procedural changes affecting [IG] activities. This regulation implements DODD 5505.6.		x
AR 25-6	This regulation implements [DoDD] 4650.2, establishes policy and management responsibilities for the Army Military Affiliate Radio System, and incorporates policy on amateur radio operations. It describes a program sponsored by the DOD and supported by the Department of the Army in which military installations, military units, clubs, and volunteer-licensed amateur radio operators and stations provide emergency communications for emergency management agencies on a local, national, and international basis as an adjunct to existing Department of the Army communications.		x
AR 25-55	This regulation updates the Freedom of Information Act Program in accordance with the Freedom of Information Reform Act of 1986. The Reform Act required agency promulgation of regulations specifying a uniform schedule of fees and guidelines for determining waiver or reduction of such fees.		x

Table G.2 Army Policies That May Either Contribute to or Reduce Stigma

Document Number	Description	Has Content That May Has Content Contribute to That May Stigma Reduce Stigma
AR 40-3	This regulation addresses specific programs (Aviation Medicine, Army Blood, Army Transplant and Organ/ Tissue Donation) as well as auditory evaluations, hearing aids, oral health services, veterinary medical care, and orthopedic footwear. It provides operational policy for nutrition care management, pharmacy management, and medical laboratory management. It sets policy and procedures for implementing advance directives, do-not-resuscitate, and withhold/withdraw orders; medical libraries; psychological test materials; emergency medical services; and air ambulances. Pertinent Federal statutes, regulations, and other standards governing these programs/services are cited throughout.	x
AR 40-8	This regulation provides implementation guidance for NATO STANAG 3474 AMD (edition 4).	x
AR 40-66	This regulation prescribes policies for preparing and using medical reports and records in accordance with [NATO STANAGs] 2348 ED.3(1) and 2132ED.2 and American–British–Canadian–Australian Quadripartite Standardization Agreement 470 ED.1.	Х
AR 40-400	This consolidated regulation prescribes policies and mandated tasks governing the management and administration of patients. It includes [DoD] and statutory policies regarding medical care entitlements and managed care practices.	Х
AR 40-501	This publication implements DODD 6130.3 and [DoDI] 6130.4. It provides information on medical fitness standards for induction, enlistment, appointment, retention, and related policies and procedures.	Х
AR 50-1	This regulation prescribes policies, procedures, and responsibilities for the Army Biological Surety Program in accordance with DODD 5210.88.	Х
AR 50-5	This regulation prescribes policies, procedures, and responsibilities for the Army Nuclear Surety Program and implements DOD Directive 5210.42, <i>Nuclear Weapons [PRP</i>].	х
AR 135-178	This regulation implements DODI 1332.14 and DODI 1332.30. It establishes policies, standards, and procedures governing the administrative separation of certain enlisted Soldiers of the [ARNG] and the United States Army Reserve.	X

Document Number	Description	Has Content That May Contribute to Stigma	Has Content That May Reduce Stigma
AR 165-1	This regulation prescribes policies on Total Army religious support activities, religious ministries, Chaplain and Chaplain Assistant personnel, Chaplain recruitment, the Chaplain Candidate Program, policy development, mobilization and readiness, training, moral leadership, management of information, logistics, and resources.		x
AR 190-30	This regulation establishes policies and procedures for selection of [MP] investigators and Department of the Army civilian detectives and investigators, issuance and control of [MP] investigators' credentials, operational procedures, types and categories of offenses investigated by [MP] investigators, investigator reports and case folders, and the uniform for [MP] investigators.	x	
AR 190-47	This regulation covers policies governing the Army Corrections System and implements [DoDD] 1325.4.	х	х
AR 190-58	This regulation establishes Army policy on personal security.	х	
AR 195-3	This revision covers acceptance, accreditation, withdrawal of acceptance/accreditation, and warrant officer application criteria for criminal investigative personnel.	x	
AR 195-5	This regulation establishes [policies] and procedures on criminal investigation evidence procedures, including the collection, accounting, preservation, and disposition of evidence. It also delineates responsibility between [MP] and the U.S. Army Criminal Investigation Command as they apply to evidence procedures.	х	
AR 210-25	This regulation incorporates existing direction and guidance from the Secretary of the Army for the general governance and operating policies of the United States Military Academy	x	
AR 210-26	This regulation incorporates existing direction and guidance from the Secretary of the Army for the general governance and operating policies of the United States Military Academy.	x	
AR 210-35	This regulation provides guidance for establishing and managing civilian inmate labor programs on Army installations. It provides guidance on establishing prison camps on Army installations. It addresses recordkeeping and reporting incidents related to the Civilian Inmate Labor Program and/or prison camp administration.	х	

Document Number	Description	Has Content That May Contribute to Stigma	Has Content That May Reduce Stigma
AR 215-1	This regulation implements [DoD] and congressional policies. It contains administration, operation, and management policies governing the Army's morale, welfare, and recreation programs and general policies on [NAF] instrumentalities.	x	
AR 215-3	This regulation provides uniform policies governing personnel management and administration for non- appropriated fund instrumentalities (NAFI) employees of the Department of the Army (DA).		х
AR 350-1	This regulation consolidates policy and guidance for Army training and leader development and supports a full-spectrum, force protection, expeditionary Army.	x	
AR 350-51	This new regulation prescribes eligibility requirements and procedures for processing applications of Active Army personnel for Officer Candidate School and for Army Band Officer Candidate School.	х	
AR 360-1	This major revision provides guidelines for command and public information, including information released to the media, and community relations programs intended for internal and external audiences with interest in the U.S. Army. This revision also defines the staff relationship of the public affairs officer; [and] identifies public affairs as an inherently governmental function		X
AR 380-67	This regulation implements the DOD and Department of the Army Personnel Security Program and takes precedence over all other departmental issuances affecting these programs.	x	х
AR 600-8-19	This regulation prescribes policies and procedures governing promotion and reduction of Army enlisted personnel.	х	
AR 600-8-22	This regulation provides Department of the Army policy, criteria, and administrative instructions concerning individual military decorations, Good Conduct Medal, service medals and ribbons, combat and special skill badges and tabs, unit decorations, and trophies, and similar devices awarded in recognition of accomplishments.	х	
AR 600-20	It prescribes the policy and responsibility of command, which includes well-being of the force, military and personal discipline and conduct, the Army Equal Opportunity Program, Prevention of Sexual Harassment, and the Army Sexual Assault Prevention and Response Program.	x	x
AR 600-63	This publication prescribes policy and sets forth responsibilities for all aspects of the Army Health Promotion Program.		х

Document Number	Description	Has Content That May Contribute to Stigma	Has Content That May Reduce Stigma
AR 601-10	This regulation covers the management and mobilization of retired Soldiers of the Army during war, national emergency, or when otherwise authorized by law.	x	
AR 601-210	This regulation governs eligibility criteria, policies, and procedures for enlistment and processing of persons into the Regular Army, the Army Reserve, and [ARNG] for enlistment on or after the effective date of this regulation.	x	
AR 601-280	This regulation prescribes the criteria for the Army Retention Program. For those soldiers serving in the active Army, it outlines procedures for immediate reenlistment or extension of enlistment.	x	
AR 608-75	This regulation outlines the policies and procedures for the Exceptional Family Member Program. It implements DODD 1342.17 and portions [of] DODD 1020.1. It also implements DODI 1315.19 and DODI 1342.12.		x
AR 614-30	This publication complies with and implements [DoD] policies in DODD 1315.7 and DODI 1315.18.	х	
AR 614-200	This regulation provides guidance on the selection of enlisted Soldiers for assignment, utilization, reclassification, detail, transfer, and training as implemented by DODI 1315.18.	x	
AR 621-202	This regulation establishes a reference for educational incentives and entitlements authorized by public law. It provides Army-unique policies, responsibilities, and procedures governing these educational benefits for Soldiers and former Soldiers of the Active Army.		x
AR 623-3	This regulation prescribes the policy and tasks for the Army's Evaluation Reporting System, including officer, noncommissioned officer, and academic evaluation reports focused on the assessment of performance and potential. It includes policy statements, operating tasks, and rules in support of operating tasks. It has been revised to update policy on the use of new grade plate officer evaluation forms, integration of a rater's profile for rated officers in the grades of O-5 and below, strengthening of rating chain accountability, integration of Army leadership doctrine on officer evaluation reports, use of the officer evaluation support form, and requirements for counseling and assessing rated Soldiers on fostering a climate of dignity and respect and adhering to the Sexual Harassment/Assault Response and Prevention Program.	x	X

Document Number	Description	Has Content That May Contribute to Stigma	Has Content That May Reduce Stigma
AR 633-30	This is a change to AR 633-30, 6 November 1964. This change implements policies and procedures required by [DoDD] 1325.4, the decision of the U.S. Court of Military Appeals in the case of <i>U.S. v. Allen</i> , 17 MJ 126 (CMA 1984) and the Manual for Courts-Martial, United States, 1984.	x	
AR 635-200	This regulation implements DODI 1332.14 and DODI 1332.30. Statutory authority for this regulation is established under Sections 1169, 12313(a), and 12681, Title 10, United States Code.	x	х
AR 690-600	This regulation establishes policies and mandated procedures for counseling, filing, processing, investigating, settling, deciding, and taking action on equal employment opportunity discrimination complaints.		х
ATTP 3-21.9	ATTP 3-21.9 describes how the [SBCT] Infantry rifle platoons and squads fight. ATTP 3-21.9 discusses how they operate as Infantry, mounted in their Stryker [ICVs], with the ICV in support, and as part of the Stryker rifle company. The content includes principles, tactics, techniques, procedures, terms, and symbols that apply to small-unit operations. ATTP 3-21.9 is intentionally designed as a companion manual to FM 3-21.8.	x	
ATTP 3-21.71	This manual provides a doctrinal framework for the Bradley Fighting Vehicle (BFV) equipped Infantry rifle platoon and squads. It addresses the BFV and all variations, rifle platoon and squad combat and noncombat operations, across the spectrum of conflict.		х
ATTP 3-39.10	[This] is aligned with [FM] 3-39 and the [MP] Corps Regiment's keystone operational doctrine. It provides guidance for commanders and staffs on [MP L&O] operations. L&O operations support all elements of full spectrum operations (offensive, defensive, and stability or civil support operations).		х
ATTP 4-02	As the Army's capstone medical doctrine statement, this publication identifies medical functions and procedures that are essential for operations covered in other Army Medical Department (AMEDD) proponent manuals. This publication depicts Army Health System operations from the point of injury, illness, or wounding through successive roles of care within the theater and evacuation to the [CONUS] support base.	x	
FM 3-19.4	This field manual addresses [MP] maneuver and mobility support (MMS), area security (AS), internment and resettlement (I/R), [L&O], and police intelligence operations (PIO) across the full spectrum of Army operations.	x	

Table	G.2-	-Continue	ed
-------	------	-----------	----

Document Number	Description	Has Content That May Contribute to Stigma	Has Content That May Reduce Stigma
FM 3-34.468	This manual is a guide for engineer personnel responsible for planning, designing, and conducting blasting operations in pits and quarries. This manual should be used in conjunction with current Army blasting doctrine.	х	
FM 3-50.1	[This] is the Army's doctrinal publication for personnel recovery. It presents doctrinal guidance and direction for Army personnel recovery operations and is the foundation for developing tactics and techniques, handbooks, and unit standard operating procedures.	x	x
FM 4-02.6	This field manual provides information on the employment, functions, and operations of divisional and nondivisional medical companies of Army of Excellence (AOE) and Army XXI divisions to include separate brigades, the [SBCT], and the armored cavalry regiment. It is intended to serve as doctrine and a primary reference publication for medical planners and the medical commander and his [or her] staff.	x	
FM 4-02.51	This publication outlines the functions and operations of each [COSC] element within an area of operations (AO). This field manual establishes Army doctrine and provides guidance for conducting COSC support for combat, stability, and reconstruction operations from brigade to theater level.		х
FM 4-25.11 C1	This publication outlines both self-aid and aid to other service members (buddy aid). More importantly, it emphasizes prompt and effective action in sustaining life and preventing or minimizing further suffering and disability.	x	x
FM 6-22	FM 6-22 is the Army's keystone field manual on leadership. It established leadership doctrine and fundamental principles for all officers, noncommissioned officers and Army civilians across all components.	x	x
FM 6-22.5	The focus of this publication is to inform leaders and Soldiers of the stressors of combat (offense and defense), stability, and civil support operations and to provide information on [COSC]. It provides guidance on how to prevent, reduce, identify, and manage combat and operational stress reactions (COSRs) in the Soldier's own unit to the maximum extent possible.		x

NOTE: ADRP = Army doctrine reference publication. IG = inspector general. NATO = North Atlantic Treaty Organization. STANAG = standardization agreement. ARNG = Army National Guard. MP = military police. NAF = nonappropriated fund. ICV = infantry carrier vehicle. FM = field manual. L&O = law and order.

Document Number	Description	Has Content That May Contribute to Stigma	Has Content That May Reduce Stigma
Navy policies			
OPNAVINST 1220.1D	Purpose: To provide guidance for removing or changing Navy Enlisted Classification (NEC) Codes used to designate enlisted personnel trained as nuclear propulsion plant operators.	х	
OPNAVINST 1300.14D	Purpose: To publish revised Navy policy and guidance for determining suitability of Navy servicemembers and family members for overseas or remote duty assignments per [references] and implemented by [references]. This instruction is a complete revision and should be reviewed in its entirety.	х	
OPNAVINST 1500.75B	Purpose: This instruction establishes policy and procedures to abate or minimize mishaps during high-risk training. Due to recent changes to the Navy's training organization, this instruction has been extensively revised and should be read in its entirety.	x	
OPNAVINST 1700.9E	Purpose: This instruction implements policy, assigns responsibilities, incorporates changes in organizational responsibility, and prescribes procedures under [references] for the operation of Child and Youth Programs (CYPs) on naval installations and in government housing to ensure a healthy, safe environment, and promote quality programs	x	x
OPNAVINST 1720.4A	Purpose: To provide policy, procedures and assign responsibilities for the Navy's Suicide Prevention Program per [references].		x
OPNAVINST 1754.7	Purpose: To establish policy, responsibilities, and authority to implement and execute a dynamic Returning Warrior Workshop (RWW) program, in accordance with [the] authority and requirements of [references].		x
OPNAVINST 3591.1F	Purpose: To establish Navy policy and prescribe minimum requirements for individual small arms training and qualification per [references]. This instruction is a complete revision and shall be reviewed in its entirety.	x	
OPNAVINST 3710.7U	Purpose: To issue policy and procedural guidance applicable to a broad spectrum of users and complements individual Naval Air Training and Operating Procedures Standardizations (NATOPS) manuals. This is a complete revision and should be reviewed in its entirety.	x	

Table G.3 Navy and U.S. Navy Bureau of Medicine and Surgery Policies That May Either Contribute to or Reduce Stigma

ocument umber	Description	Has Content That May Contribute to Stigma	Has Content That May Reduce Stigma
OPNAVINST 3750.6R	Purpose: To add the use of [Office of the Chief of Naval Operations] 3750/16 to the list of mandated forms and to update information in [two chapters].		х
OPNAVINST 5500.11F	Purpose: To prescribe regulations, procedures, and criteria governing issuance of authorizations to enter certain defense areas, the entry of which is prohibited except as authorized by the Secretary of the Navy (SECNAV)	x	
OPNAVINST 5510.60M	Purpose: To update security policy and procedural guidance for the protection of classified information and materials in the custody of the Chief of Naval Operations (CNO), the Office of the Chief of Naval Operations (OPNAV) staff or other Metro Washington, DC, [DON] offices for which the CNO has cognizance for security	x	
OPNAVINST 5530.16A	Purpose: To prescribe policies, procedures, and responsibilities for the Navy Biological Surety Program per [references].	x	
OPNAVINST 6100.2A	Purpose: To update the comprehensive Navy Health and Wellness Promotion Program that improves and sustains military readiness by focusing on health, fitness, and quality of life for Sailors, [DoD] personnel, and other beneficiaries as required by [references]	x	
OPNAVINST 6420.1A	Purpose: To provide physical requirements for non- submarine personnel embarked on submarines, including non-submarine military personnel, civilians, governmental and contractor employees, and military dependents.	x	
SECNAV M-5510.30	This manual establishes specific policy set forth in SECNAVINST 5510.30B, "Department of Navy (DON) Personnel Security Program (PSP) Instruction." It is intended to provide maximum uniformity and effectiveness in the application of PSP policies throughout DON.	x	
SECNAVINST 1640.9c	Purpose: To issue standardized policies and procedures for the operation of Navy and Marine Corps confinement facilities. This instruction is a complete revision and should be reviewed in its entirety.		x
SECNAVINST 1730.9	Purpose: This instruction provides policy on confidential communications to Navy chaplains. The unconstrained ability to discuss personal matters in complete privacy encourages full and complete disclosure by personnel and family members seeking chaplain assistance		х

Document Number	Description	Has Content That May Has Content Contribute to That May Stigma Reduce Stigma
SECNAVINST 1850.4E	Purpose: To revise and simplify policies and procedures for evaluation of physical fitness for duty and disposition of physical disability in the [DON] in compliance with Chapter 61 and Section 1554 of [Title 10, U.S. Code] and with [DoDD 1332.18, DoDI 1332.38, DoDI 1332.39, and SECNAVINST 5300.30C]. This instruction is a complete revision and should be reviewed in its entirety.	X
SECNAVINST 1920.6C CH-4	Purpose: To transmit changes per the repeal of section 654 of title 10, United States Code, "Don't Ask, Don't Tell." A complete revision to this instruction will be issued at a later date.	х
SECNAVINST 5312.12	Purpose: To assign responsibility and prescribe procedures for implementation of [reference]. This instruction is a complete revision and should be reviewed in its entirety.	Х
SECNAVINST 5510.35B	Purpose: a. To establish and maintain the [DON] Nuclear Weapons [PRP] per [DoDI 5210.42, DoD 5210.42-R, and SECNAVINST 8120.1].	X
SECNAVINST 5815.3J	Purpose: To publish regulations consistent with [references] for implementation of systems of clemency, parole, retention and enlistment of selected court-martialed offenders who were subject to the authority of the Secretary of the Navy at the time of their offense.	х
SECNAVINST 6320.24A	Purpose: To issue [DON] policy, assign responsibility, and prescribe procedures per [reference] for the referral, evaluation, treatment and administrative management of services members who are directed by their commands for mental health evaluation and/or assessment of risk for potentially dangerous behavior.	х
BUMED policies		
Chief, BUMED, 2008	This change completely revises Chapter 6, Dental Corps. This revision takes into account the changes in authority and responsibilities of BUMED (Echelon II) post integration of the Dental Division as well as the latest clinical guidelines driven by recent instructions.	x
Chief, BUMED, 2012	Purposes of medical examinations [and] Interpretation and application of physical standards	x
BUMEDINST 1755.1	This instruction supplements [reference] by assigning responsibilities and providing guidance and procedures specific to the provision of early intervention services (EIS) by Navy educational and developmental intervention services (EDIS) programs.	x

Table	G.3—	Conti	nued

Document Number	Description	Has Content That May Contribute to Stigma	Has Content That May Reduce Stigma
BUMEDINST 5041.6	To implement the provisions of [references] throughout Budget Submitting Office (BSO)-18 and clarify responsibilities for the operation of the Navy Medicine (NAVMED) Hotline Program. [Enclosure] is a list of references used in this interaction.	Х	x
BUMEDINST 6300.17	To establish guidance in the delivery of case management (CM) services within Navy Medicine		х
BUMEDINST 6400.3A	To identify a class of specialized psychological materials, and provide policy in the use, security, and supervision of psychological testing procedures in medical treatment facilities and other medical and operational platforms		x
BUMEDINST 6520.2	To provide policy guidance for the evaluation and disposition of patients presenting with suicidal ideation or behavior	х	
BUMEDNOTE 6100	To provide preliminary implementation guidance for the newly required NAVMED 6100/8, <i>Mental Health</i> <i>Assessment</i> , to be completed by specified deployers in four time frames and to announce the additional training requirement for providers who conduct the assessment	x	

NOTE: CYP = Child and Youth Program. DON = Department of the Navy. BUMEDINST = BUMED instruction. BUMEDNOTE = BUMED note.

Table G.4

Marine Corps Policies That May Either Contribute to or Reduce Stigma

Document Number	Description	Has Content That May Contribute to Stigma	Has Content That May Reduce Stigma
MCO 1001R.1K	The [reserve component] is an integral part of the Total Force Marine Corps and provides augmentation and reinforcement in times of war or national emergency. This Order establishes the policies and responsibilities for the administration and personnel management of the Marine Corps Reserve as outlined in [references].	х	
MCO 1040.31	This Order revises the policies, procedures, and standards for the operation of the Marine Corps Enlisted Retention and Career Development Program.	х	
MCO 1200.17D	The publication of this Order synchronizes the Marine Corps Human Resources Development Process.	х	
MCO 1306.16E	To provide current policy and procedures concerning conscientious objection applicable to all Marines	х	

Document Number	Description	Has Content That May Contribute to Stigma	Has Content That May Reduce Stigma
MCO 1326.7E	To establish assignment criteria and issue instructions relative to the selection and screening process for personnel assigned to duty with [Marine Helicopter Squadron 1]	х	
MCO 1553.2B	Establish management policies and procedures for the establishment and operations of Marine Corps Formal Learning Centers	х	
MCO 1560R.30B	To publish implementing guidance for [Montgomery GI Bill—Reserve] education benefits authorized by [statute]		x
MCO 1720.2	To provide policy and procedural guidance for the Marine Corps Suicide Prevention Program	х	
MCO 1730.6E	This Order implements [SECNAVINST 1730.7D]. This is a complete revision and should be reviewed in its entirety.		х
MCO 1754.11	To provide policy and procedural guidance for the effective execution and use of the [FAP] in order to support the commander's responsibility to prevent and respond to child abuse and domestic abuse, and support and treat eligible beneficiaries with counseling services in accordance with the references	х	
MCO 3000.19A	This Order establishes Service policy and tasks that support [references] and outlines procedures to conduct Marine Corps Reserve Component (Re) activation, integration, and deactivation as described in [references]. This Order is designed to support contingency planning, crisis action planning, and sustained operations leading to rapid augmentation and/or reinforcement of the [active component] of the U.S. Marine Corps. As such, this Order is directive in nature.		x
MCO 3120.11	Individual service parachuting programs are guided by policy directing their conduct and administration by [DoD EAs] in accordance with [references]. [DoDD 5100.1] assigns the U.S. Army as the EA for all airborne matters common to both the Army and the Marine Corps. This Order establishes policy and procedural guidance for the administration of Marine Corps parachuting programs and takes precedence where Marine Corps equipment and doctrine are not supported by EA policy, procedure and/or doctrine. All Marine Corps parachuting programs will be administered in compliance with this Order.	x	

Table	G.4-	-Continu	Jed
-------	------	----------	-----

Document Number	Description	Has Content That May Has Content Contribute to That May Stigma Reduce Stigma
MCO 3150.4	[DoD EAs] establish policy that guides the conduct and administration of individual Service diving programs, per [references]. [DoDI 3224.04] assigns the Assistant Secretary of Defense for Special Operations and Low- Intensity Conflict, Department of the Navy as the DOD Diving Proponent.	x
MCO 5100.29B	Force preservation is a vital element of our combat readiness; death, serious injury, and the loss of materiel assets due to mishaps directly and negatively impacts the warfighting capability of the entire Marine Corps. Engaged leadership at all levels is the key to ensuring a command climate that demands the preservation of Marine Corps assets through risk management. This Order and [Marine Corps Safety Program] establishes the minimum requirements of the Marine Corps Safety Program based on the references and policies required by resources listed in [the enclosure].	x
MCO 5300.17	To provide policy and procedural guidance to commanders, substance abuse personnel, and Marines in order to effectively utilize and execute the Marine Corps Substance Abuse Program, per [references]. This Order provides both policy and guidance to commanders so that they may improve their capability to treat and prevent alcohol and drug abuse problems that detract from unit performance and mission readiness.	x
MCO 5530.16	To promulgate policy and guidance for the establishment, management, training, and employment of a Security Augmentation Force at all Marine Corps installations per [references]	x
MCO 5580.2B	This Order is a complete revision of the last Marine Corps [law enforcement manual] and should be reviewed in its entirety. It provides updated policy that commanding officers, provost marshals and police chiefs require, and establishes guidelines and procedures for MPs, police officers and security personnel in the performance of their duties per [references].	x
MCO 8400.6	To establish standard licensing procedures for qualifying, testing, and licensing ordnance vehicle operators. Ordnance vehicles are defined as any wheeled or tracked vehicle configured to conduct a combat mission or support an ordnance/maintenance operation. Implementation of the provisions contained herein will enhance selection and certification of qualified ordnance vehicle operators, an essential element for safe and efficient operations. A listing of ordnance vehicles requiring licensed operators is provided in [earlier versions of the order].	x

Document		Has Content That May Contribute to	Has Content That May
Number	Description	Stigma	Reduce Stigma
MCO 12410.24	To establish the framework for [civilian leadership development] for the U.S. Marine Corps		x
MCO 12630.3	To provide policy and procedures on implementing [Family and Medical Leave Act of 1993 (Pub. L. 103- 3)] for civilian employee of [HQMC]; Headquarters Battalion, Henderson Hall; Marine Barracks, 8th and I Streets, Washington, DC; and the Marine Corps Institute		x
MCO 12713.8	To implement the [Handicapped Individuals Program].	х	х
MCO 12771.2	To establish the administrative grievance procedure for civilian employees at [HQMC] and serviced activities.	х	х
MCO P1100.72C	This manual establishes the criteria for enlistment, procedures governing the processing of applicants and summarizes recruiting support programs to be used in accomplishing the enlisted recruiting mission.	x	
MCO P1300.8R	To implement DoD policy and provide definitive guidance on the assignment and [PCS] of Marines	х	
MCO P1326.6D	To establish criteria and instructions relative to selecting, screening and preparing enlisted Marines for assignment to Special Duties (Drill Instructor, Recruiting, Marine Security Guard, Marine Corps Security Forces) and Independent Duties	x	
MCO P1400.32D	To publish regulations and policies concerning enlisted promotions	х	x
MCO P1610.7F	The recent addition of the Marine Corps Combat Fitness Test and modification to the Body Fat Percentage reporting criteria necessitate changes to the [PES]		x
MCO P3500.44A	To establish training standards, regulations and policies regarding the training of Navy Chaplains and [RPs] assigned to the Marine Corps. Per Volume II of the Manual of Navy Enlisted Manpower and Personnel Classifications [NAVPERS 18068F], this order identifies core entry-level training requirements to award RPs with the NEC 2401.		x
MCO P3500.75	Per the references, this manual establishes training standards, regulations and policies regarding the training of assigned Navy personnel in Health Services. This order identifies core-entry level training requirements to award Hospital men with [NECs] 8404, 8427, and 8403 and Dental Technicians with the NEC 8707.	х	

Document Number	Description	Has Content That May Contribute to Stigma	Has Content
MCO P3550.10	In order to achieve the highest levels of readiness, organizations must maintain rigorous and realistic training programs based on approved training standards. Training resources must enhance safe and challenging live-fire training, enabling Marine units to train as they fight.	x	
MCO P12000.11A	To transmit new page inserts into the Manual		х
NAVMC 2660	To promulgate NAVMC 2660, <i>Discussion Leaders'</i> Course	х	
NAVMC 2768	This handbook is designed to heighten the awareness of commanders and functional managers to the potential for fraud and waste within their activities.	x	
NAVMC 3500.18B (PRELIM) Part 1	Contains non-[military occupational specialty-]specific individual events for all Marines that are introduced during the [entry-level training] pipeline.	x	
NAVMC 3500.59A Part 2	Same as NAVMC 3500.59A Part 1.		x
NAVMC DIR 1500.58	This Directive elaborates upon procedures and methods and provides guidance for implementing the [Marine Corps Mentoring Program].	x	x

NOTE: HQMC = Headquarters, U.S. Marine Corps. PCS = permanent change of station. PES = Performance Evaluation System. RP = religious program specialist.

Table G.5 Air Force Policies That May Either Contribute to or Reduce Stigma

Document Number	Description	Has Content That May Contribute to Stigma	Has Content That May Reduce Stigma
5AFI 36-102	This instruction implements AFPD 36-1, <i>General Civilian Personnel Provisions and Authorities</i> . It provides managers and supervisors with instructions for administration of [master labor contract] and [IHA] employees.	х	
AETC Instruction 36-2205, Vols. 2, 3, 5, 6, 8, and 12	This instruction implements AFPD 36-26, <i>Total</i> Force Development, establishes policy for student administration, conduct and documentation, and provides management guidelines for all AETC fighter pilot training programs.	x	

Document Number	Description	Has Content That May Contribute to Stigma	Has Content That May Reduce Stigma
AFGM 48-04	This memorandum replaces previous guidance in AFI48- 120_AFGM3 issued 8 March 2011, which instituted Congressionally mandated mental health assessments (referred to in AFI48-120_AFGM3 as Deployment Resiliency Assessments) for members of the Armed Forces deployed in connection with a contingency operation		X
AFGSCGM 10-02	This is an [AFGSCGM] immediately implementing policy for [AFMAN 10-3902], the management of the [PRP].	х	
AFI 10-203	This instruction implements AFPD 10-2, <i>Readiness</i> , October 30, 2006; Title 10, United States Code Sections 136(d) and 671. This Instruction describes how to communicate to commanders['] individual member restrictions due to medical reasons. The application of restrictions is a commander's program with medical recommendations. It also describes the disposition and management of members who have duty limitations and reporting requirements. It interfaces with AFPD 44-1, <i>Medical Operations</i> , AFPD 48-1, <i>Aerospace</i> <i>Medical Program</i> .		x
AFI 10-403	This instruction implements AFPD 10-4, Operations Planning: Air and Space Expeditionary Force (AEF) and AFI 10-401, Air Force Operations Planning and Execution. It provides the basic requirements for Air Force deployment planning and execution at all levels of command to support contingency and exercise deployment operations. This instruction applies to all [MAJCOMs, ANG, field operating agencies, direct reporting units], and Air Force civilian personnel.		х
AFI 11-402	It sets the procedures for initiating aviation/parachutist service, awarding Air Force aeronautical ratings and aviation badges, and gives guidance that applies to administering initiation and termination of aviation/ parachutist service and award of ratings/badges.	x	
AFI 13-204V3_ AFSOCSUP_I	It sets policy and describes unit, [MAJCOM] and [Air Force] roles for managing airfield operations at [Air Force] locations.	x	
AFI 31-204_ USAFESUP_ INCIRLIKABSUP_I	This instruction implements [AFPD] 31-2, <i>Law</i> <i>Enforcement</i> . It assigns responsibilities and establishes procedures for motor vehicle traffic supervision on Air Force installations in [CONUS] and overseas ([outside CONUS]) areas.	x	
AFI 31-205	It describes how to administer corrections, rehabilitation and parole and clemency programs; how to secure confinement/corrections facilities; and how to control and transfer military inmates.	x	

Document Number	Description	Has Content That May Contribute to Stigma	Has Content That May Reduce Stigma
AFI 31-206	This instruction implements AFPD 31-2, <i>Air Provost</i> <i>Operations</i> . It provides guidance on general [special forces] duties and law enforcement operations.	X	
AFI 31-207	Gives war and peacetime requirements for arming [Air Force] personnel and the use of deadly force	x	х
AFI 31-501	This instruction implements [AFPD] 31-5, <i>Personnel Security Program Policy</i> . It provides guidance for personnel security investigations and clearance needs.	х	
AFI 31-501_ AFGM1.1	By Order of the Secretary of the Air Force, effective as of the date of this Memorandum, [Air Force] Guidance Memorandum 1 dated 24 Aug 2011 to immediately change AFI 31-501 is reissued.	x	
AFI 32-3001_ AFDWSUP_I	[l]dentifies Air Force [explosive ordnance disposal] program requirements	х	
AFI 36-704	This instruction implements AFPD 36-7, <i>Employee and Labor-Management Relations</i> . It provides instructions for maintaining discipline and for taking disciplinary and adverse actions against certain civilian employees paid with appropriated funds only.	x	x
AFI 36-2002	This instruction implements [AFPD] 36-20, Accession of Air Force Military Personnel. It tells how to enlist qualified individuals into the Regular Air Force and the [delayed-entry program] and how to: access [ARC] members onto specified [EAD] tours; access ARC members who apply for [regular Air Force] enlistment while on EAD; and reenlist officers with a statutory entitlement. Reenlist airmen removed from [TDRL].	x	
AFI 36-2110_ AFDWSUP_I	[E]stablishes criteria for assignment of military personnel to satisfy operational, rotational, and training (including formal education and professional military education/development) requirements include [temporary duty] and [PCS]	x	
AFI 36-2113	This instruction sets the policy for management of first sergeants and implements AFPD 36-21, <i>Utilization</i> <i>and Classification of Air Force Military Personnel</i> . The procedures implemented in this instruction are not [a] basis for change in numbers or type and kind of manpower requirements or authorizations. It applies to all Air Force personnel.	x	
AFI 36-2501_ AFGM1	By Order of the Secretary of the Air Force, this [Air Force] Guidance Memorandum immediately changes AFI 36-2501, Officer Promotions and Selective Continuation. Compliance with this publication is mandatory. To the extent its directions are inconsistent with other Air Force publications, the information herein prevails, in accordance with AFI 33-360, Publications and Forms Management.	x	

Document Number	Description	Has Content That May Contribute to Stigma	Has Content That May Reduce Stigma
AFI 36-2706	This instruction implements [AFPD] 36-27, Equal Opportunity, 22 May 2009. It prohibits unlawful discrimination and harassment, and reprisal. It establishes the requirements for the Air Force [MEO] Program and the Civilian [EEO] Program. This publication applies to all military and civilian Air Force personnel, including [Air Force Reserve Command] Units.		X
AFI 36-2910	This publication sets guidelines for Line of Duty and Misconduct Determinations It applies to all active duty members as well as members and units of the [ARC]—the [ANG] and the [Air Force] Reserve It implements [AFPD] 36-29, <i>Military Standards</i> .	x	x
AFI 36-3002	This instruction implements [AFPD] 36-30, <i>Military</i> <i>Entitlements</i> , and [DoDI] 1300.18, <i>Personnel Casualty</i> <i>Matters, Policies, and Procedures</i> . It describes procedures for the Casualty Services Program for all levels of command and all Air Force organizations.	x	
AFI 36-3206	This [AFI] explains how to discharge active duty Air Force officers for substandard performance of duty, misconduct, moral or professional dereliction, homosexual conduct, or in the interest of national security. It prescribes procedures for disposing of cases involving officers and for processing cases approved under AFI 31-501, <i>Personnel Security Program</i> <i>Management</i> .	x	
AFI 36-3208	This instruction states how [the Air Force] administratively separate[s] enlisted members for all reasons except physical disability or court-martial. It contains standards and procedures for implementing Air Force Policy Directive (AFPD) 36-32, <i>Military Retirements and Separations</i> , concerning voluntary and involuntary separations.	x	
AFI 36-3209	This instruction applies to all officer and enlisted members not serving on active duty with the Regular Air Force. Chapter 5 applies to members assigned to the Retired Reserve not receiving retired pay. Recalled members are subject to this instruction if they are removed from retired status and returned to an active status. This instruction implements AFPD 36-32, <i>Military Retirements and Separations</i> , 28 May 1993, and establishes procedures for administrative separation or discharge of [ANG] and Air Force Reserve members under those conditions and reasons outlined herein.	x	

Document Number	Description	Has Content That May Contribute to Stigma	Has Content That May Reduce Stigma
AFI 36-3212	This instruction describes how to retire or discharge Air Force members who are unfit to do their military duties because of physical disability. It outlines procedures for examining, and discharging or retiring members from the [TDRL]. It also provides disposition instructions for unfit members who remain on active duty in [limited assignment status]. Chapter 8 applies to certain [ANG] and United States Air Force Reserve members not on [EAD].	x	
AFI 36-6001	This instruction implements [AFPD] 36-60, 28 March 2008, Sexual Assault Prevention and Response (SAPR) Program, and [DoDI] 6495.02, 23 June 2006, Sexual Assault Prevention and Response (SAPR) Program Procedures. It assigns responsibility for the prevention of and response to sexual assault and establishes command relationships, authorities and responsibilities in support of the policy.	x	x
AFI40-301_ USAFASUP_I	[D]escribes the responsibilities of [FAP] agencies, FAP staff, and other [Air Force] personnel who are instrumental to the implementation and operation of the Air Force FAP.	x	
AFI 40-402_ USAFASUP_I	[P]rovides guidance and procedures for conducting research investigation at [MTFs, clinical investigation facilities], and other medical support centers and for using human subjects in [research, development, test, and evaluation] conducted or funded by the Air Force.	x	
AFI 40-404	How the program is managed, responsibilities, information processing procedures, data processing, and data analysis	x	x
AFI 41-210	[H]ow to manage TRICARE Operations and Patient Administration functions including determining eligibility for care, protecting medical information, managing health records, the preparation and disposition of medical documentation and managing other administrative activities to support patients	x	х
AFI 41-307	[I]mplements nursing considerations and standards of care in the [aeromedical evacuation] system	х	Х
AFI 44-109	[D]escribes rights for members who are directed by their commanders to undergo a mental health evaluation	х	x
AFI 44-153	[G]uidance for [TSR] teams at all active duty Air Force installations, integrating resources and efforts of the [AFR] and [ANG]	x	
AFI 44-165	[D]efines the ASF mission and scope of care; [explains] how to manage and transport patients; and specifies support responsibilities of the [MTF]	х	

Document Number	Description	Has Content That May Contribute to Stigma	Has Content That May Reduce Stigma
AFI 44-172	[G]uidance for the operation of the [mental health] services and the assessment and treatment of [Air Force] personnel and beneficiaries with [mental health] problems	x	
AFI 48-123	It establishes procedures, requirements, recording and medical standards for medical examinations given by the Air Force. It prescribes procedures and references the authority for retiring, discharging, or retaining members who, because of physical disability, are unfit to perform their duties	x	
AFI 51-102V2	[D]irects the structure and programs for Chaplain Assistant professional development and describes the process for vectoring and accessing Chaplain Assistants.	x	
AFI 51-201_ AMCSUP_I	This instruction implements the [UCMJ], the Manual for Courts-Martial , United States, 2005. It provides guidance and procedures for administering military justice. Users of this instruction must familiarize themselves with the UCMJ, [Manual for Courts- Martial], and applicable [DoD] Directives.	x	
AFI 90-505	[R]equirements to conduct education and training to prevent acts of harm to self and raise awareness to prevent suicide and suicidal behavior in Air Force communities		х
AFMAN 31- 201V4	This manual implements AFPD 31-2, <i>Air Provost</i> <i>Operations</i> . This Manual sets forth guidance regarding [SF] standards and procedures of Air Force civilian and military personnel, including the Air Force Reserve and [ANG] serving in SF roles.	x	
AFMAN 48-146	[G]uidance on overall [OEH] program management and incorporates [risk management] principles into the OEH program	x	х
AFPAM 36-2241	This pamphlet implements AFPD 36-22, Air Force Military Training, and AFI 36-2642, Professional Advancement and Continuous Education. Information in this guide is taken primarily from Air Force publications and based on knowledge requirements from the [MKTS] as determined by the MKTS Advisory Council of the [Chief Master Sergeant of the Air Force] and [MAJCOM command chief master sergeants].		х
AFPAM 44-160	[E]ducation, preventative services, stress management	x	х
AFRCI 36-2001	The purpose of this instruction is to provide necessary procedural guidance for Air Force Reserve Recruiting personnel to recruit qualified persons to meet [AFR] manning requirements without regard to race, color, religion, sex, or national origin.	x	

Document Number	Description	Has Content That May Contribute to Stigma	Has Content That May Reduce Stigma
CFETP 3S1X1	This [CFETP] is a comprehensive education and training document that identifies life-cycle education/ training requirements, training support resources, and minimum core task requirements for this specialty		x
CFETP 4C0X1	This CFETP provides the information necessary for [Air Force career-field managers, MAJCOM functional managers], commanders, training managers, supervisors and trainers, to plan, develop, manage, and conduct an effective career field training program.	х	
CFETP 5R0X1C1	[Chaplain assistant CFETP]	x	
PACAFDIR 90- 254	This Directory implements AFPD 90-2, Inspector General—The Inspection System, and AFI 90-201, Inspector General Activities. Attachment 1 of this directory is a Corrections Program [mission performance checklist] and supports guidance in AFI 31-205, Air Force Corrections System.	х	

NOTE: 5AFI = Fifth Air Force instruction. AFPD = Air Force policy directive. IHA = indirect hire agreement. AETC = Air Education and Training Command. AFGM = Air Force guidance memorandum. AFGSCGM = Air Force Global Strike Command guidance memorandum. AFMAN = Air Force manual. MAJCOM = major command. ANG = Air National Guard. USAFESUP = U.S. Air Forces in Europe supplement. INCIRLIKABSUP = Incirlik Air Base supplement. AFDWSUP = Air Force District of Washington supplement. ARC = Air Reserve Component. EAD = extended active duty. TDRL = temporary disability retired list. FAP = Family Advocacy and General Counseling Programs. ASF = aeromedical staging facility. MTF = medical treatment facility. SF = security force. OEH = occupational and environmental health. AFPAM = Air Force pamphlet. MKTS = Military Knowledge and Testing System. AFRCI = Air Force Reserve Command instruction. CFETP = career-field education and training plan. PACAFDIR = Pacific Air Forces directory.

Table G.6 National Guard Policies That May Either Contribute to or Reduce Stigma

Document Number	Description	Has Content That May Contribute to Stigma	Has Content That May Reduce Stigma
AFI10-404_ ANGSUP_I	The [ESSP] is a subset of the overall expeditionary site planning process and serves as the foundation for Air Force expeditionary operations. This AFI is the governing document for ESSP and provides detailed information for use by planners at the strategic, operational, and tactical levels.		x
AFI 13-204, Vol. 3, ANG Supplement	This instruction implements AFPD 13-2, Air Traffic Control, Airspace, Airfield, and Range Management It directs the administration of facilities, the use of equipment, the operations, and the training of airfield operations. It outlines duties and responsibilities of AOF members assigned to the unit level.	х	

Document Number	Description	Has Content That May Contribute to Stigma	Has Content That May Reduce Stigma
ANGDIR 90- 266	This new directory implements [AFPD] 90-2, Inspector General—The Inspection System, and is applicable to all [ANG] Security Forces Squadrons. Units may supplement this directory to add internal compliance items. This directory may be used in whole or in part by [HHQ] during visits or exercises. [HHQ/IG] will use this directory in whole or in part during evaluations and exercises.	х	
ANGI 36- 2001	DoDI 1215.13 policy requires establishment of training programs that provide a minimum number of [inactive- duty training periods], and [active-duty training] required for attaining the prescribed unit readiness status and maintaining individual proficiency. Due to the high visibility of the training program, state headquarters, commanders and unit training managers must ensure [that] proper internal controls, resource management, and documentation prescribed by this instruction [are] accomplished.	х	
ANGI 36- 2002	This instruction prescribes the eligibility requirements and procedures for enlisting and reenlisting in the [ANG] and as a Reserve of the Air Force.	x	
CNGBI 1800.01	This instruction outlines the requirement to assess, on an individual basis, whether a particular child or youth with a special need may participate in the National Guard [CYP] with or without reasonable accommodation.	х	
NGR 385-10	This regulation explains the need for safety in the ARNG; outlines responsibilities; and provides policy for the development, implementation, and management of the ARNG Safety Program.	x	
NGR 500-3/ ANGI 10- 2503	This regulation instruction prescribes policies, procedures, and responsibilities governing the employment of National Guard [weapons-of-mass-destruction civil support teams] in support of the National Guard homeland security, homeland defense, contingency operations, special events, incident of national significance, and [defense support of civil authorities] mission.	x	
NGR 500-4/ ANGI 10- 2504	This regulation/instruction prescribes policies, procedures, and responsibilities governing the deployment and employment of National Guard [Chemical, Biological, Radiological, Nuclear, and High-Yield Explosive] Enhanced Response Force Packages in support of the National Guard Homeland Security mission.	х	
NGR 600-200	This regulation prescribes the criteria, policies, processes, procedures and responsibilities to classify; assign; utilize; transfer within and between states; provides Special Duty Assignment Pay; separate, and appoint to and from Command Sergeant Major, [ARNG] and [ARNGUS] enlisted Soldiers.	х	

Document Number	Description	Has Content That May Contribute to Stigma	Has Content That May Reduce Stigma
NGR 614-1	This regulation prescribes rules for use of the ING, enlistment into the ING as part of the [Recruit Force Pool], transfer between active status and the ING, accounting for personnel in the ING, restrictions on Soldiers while assigned to the ING, reporting, annual muster requirements, entitlements, [and] responsibilities, and describes the effect of a mobilization, call or order to active duty on Soldiers in the ING.	х	
TPR 630	This regulation prescribes the Human Resources Absence and Leave Program for National Guard technicians and replaces the leave regulations that were used to supplement the Federal Personnel Manual.		x

NOTE: ESSP = Expeditionary Site Survey Process. HHQ = higher headquarters. ANGI = ANG instruction. CNGBI = Chief, National Guard Bureau instruction. NGR = National Guard regulation. ARNGUS = ARNG of the United States. ING = Inactive Army National Guard. AOF = airfield operations flight.

Table G.7 Coast Guard Policies That May Either Contribute to or Reduce Stigma

Document Number	Description	Has Content That May Contribute to Stigma	Has Content That May Reduce Stigma
COMDTINST 1730.4C	To promulgate policy, assign responsibilities and implement this Instruction for religious ministries within the Coast Guard.		х
COMDTINST 1734.1A	To provide policy and procedures, and assign responsibilities for the Coast Guard's Suicide Prevention Program.	x	х
COMDTINST 1740.7B	This Instruction establishes the Coast Guard's [employee assistance program] and prescribes associated policy, procedures and responsibilities.		x
COMDTINST 1750.7C	The [CGFAP] was developed and implemented in 1982 as the result of a congressional mandate to reduce and prevent family violence through the case management process. The CGFAP shall intervene in all suspicions, suspected and reported incidents of family violence within the Coast Guard and establish family violence preventive resources and services. The CGFAP's intent is to prevent and reduce the incidence of family violence and create an environment of intolerance for such behavior. Education and proper intervention and rehabilitation services result in prevention of the escalation of violence, serious injury and homicide. Successful resolution of family violence cases results in the retention of productive Coast Guard members and successful accomplishment of the Coast Guard mission.	x	

Document Number	Description	Has Content That May Contribute to Stigma	Has Content That May Reduce Stigma
COMDTINST 1754.3A	To update guidance for providing services intended to minimize the potential for psychological injury to employees, Coast Guard members, and their family members who have been involved in or affected by a critical incident. [Critical-incident stress management] includes pre-incident training, critical-incident interventions, and post-incident follow-up.	х	
COMDTINST 1754.7A	The purpose of this instruction is to provide policy and guidance concerning the Coast Guard Special Needs Program.		х
COMDTINST 1755.1	This instruction promulgates applicable DoD/[DoD Dependents Schools] directives pertaining to minor dependents' schooling in overseas areas.		x
COMDTINST 5230.60B	[T]his Instruction establishes U.S. Coast Guard policy, procedures, roles, and responsibilities for implementing the 1998 amendments to Section 508 of the Rehabilitation Act Section 508 requires Federal departments and agencies that develop, procure, maintain, or use [electronic and information technology] to ensure that Federal employees and members of the public with disabilities have access to and use of information and data, comparable to that of Federal employees and members of the public without disabilities. Moreover, the [Coast Guard] desires to meet the demands of citizen-centric electronic government, attract and retain the best talent, and maintain a productive workforce. To accomplish that goal [the Coast Guard] must create an environment that enables the abilities of persons with disabilities.		х
COMDTINST 6400.1B	This Instruction establishes policy, procedure and standards for Student Externship Programs in Coast Guard health care facilities.		х
COMDTINST 12300.9	This Instruction establishes a Coast Guard program under which Headquarters Program Managers may compete for centrally funded billets allocated for a 2 to 3-year period to develop candidates to meet future workforce needs.		x
COMDTINST 12335.4A	This Instruction establishes the Coast Guard's policy for merit promotion and the approved plan for merit promotion for Coast Guard–wide use.		x
COMDTINST 16798.2	This Instruction establishes the air crew qualification and promulgates the qualification requirements, training syllabus, and instructor guide for the Auxiliary air crew program.	x	
COMDTINST M1000.3	This Manual establishes Coast Guard policy and procedures concerning the accessions, evaluations, personnel boards, and promotions for the Coast Guard officer corps.	x	х

Document Number	Description	Has Content That May Contribute to Stigma	Has Content That May Reduce Stigma
COMDTINST M1000.4	This Manual establishes Coast Guard policy and procedures concerning separations and retirements for all military personnel.	х	х
COMDTINST M1000.8	This Manual establishes Coast Guard policy and procedures concerning military personnel assignments and authorized absences.	x	
COMDTINST M1600.2	This Manual establishes Coast Guard policy and procedures concerning the Code of Conduct, Civil arrest and convictions, absentee and deserters, dissident and protest activities, and military corrections/confinement for all personnel.		x
COMDTINST M1700.1	This Manual establishes Coast Guard policy on civil matters and dependent welfare and special benefits for military personnel.	х	
COMDTINST M1754.10D	This instruction establishes policy and prescribes procedures for the Coast Guard SAPR Program. The ultimate purpose of this program is to eliminate sexual assault within the Coast Guard by providing a culture of prevention, education and training, response capability, victim support, reporting procedures, and accountability that enhances the safety and well-being of all its members.	х	x
COMDTINST M3150.1C	This Manual establishes policy and procedures for the administration, application and execution of diving operations within the Coast Guard.	х	
COMDTINST M3710.1F	This manual promulgates a revision of the Coast Guard Air Operations Manual. It prescribes policy, standards, instructions and capabilities pertinent to all phases of Coast Guard flight operations and is intended for use by operational commanders, unit commanding officers, aircrews tasked with air operations, as well as customers of Coast Guard aviation	х	
COMDTINST M5100.47	The purpose of this Manual is to promulgate safety and environmental health policies, standards and guidelines and define safety and environmental health responsibilities.	x	
COMDTINST M5110.1B	This Notice promulgates Change One to [the postal manual].	х	

Document Number	Description	Has Content That May Contribute to Stigma	Has Content That May Reduce Stigma
COMDTINST M5350.4C	This Manual provides policy and guidance for Coast Guard [MEO] and Civilian [EEO] Programs. It outlines responsibilities and procedures aimed at equipping Coast Guard employees with confidential access to timely and expert advice on EEO/[MEO] matters, and provides tools that will assist unit leaders in building a model EEO workplace. The model EEO program prescribed by the [EEOC] ensures that all employees and applicants for employment enjoy equality of opportunity in the Coast Guard workplace regardless of race, sex, national origin, color, religion, disability or reprisal for engaging in prior protected activity. It is the policy of the Coast Guard to extend to members of the military workforce as much as is practical, the same protections afforded the civilian workforce.		x
COMDTINST M5400.17	To issue the [organization manual] and [SOPs] for Coast Guard [port security units].		x
COMDTINST M5810.1E	This Manual prescribes the Judge Advocate General of the Coast Guard policies, regulations, and procedures applicable to the administration of military justice in the Coast Guard pursuant to, and in support of [the UCMJ and the Manual for Courts-Martial].		х
COMDTINST M5830.1A	This manual establishes procedures for the appointment, conduct and review of Administrative Investigations.	х	x
COMDTINST M6000.1E	This Manual establishes policies, procedures, and health care standards for all active duty and reserve Coast Guard members and other Service Members assigned to duty with the Coast Guard.	x	x
COMDTINST M6410.3A	This Manual establishes policy, assigns responsibilities, and provides guidelines regarding the Coast Guard Aviation Medicine Program.	x	
COMDTINST M12271.1A	The purpose of this Notice is to provide changes to the subject Manual. The contents are intended for all units with [NAF] employees.		x
COMDTINST M12750.4	This Manual provides guidance and procedures for the processing of disciplinary, performance, and adverse actions to correct situations of unacceptable employee conduct and performance. It also provides advice and guidance for the processing of adverse action appeals and administrative grievances. All actions covered by this Manual shall be processed in accordance with applicable laws, controlling rules and regulations, and the guidance and procedures set forth in this Manual.		х
Table G.7—Continued

Document Number	Description	Has Content That May Contribute to Stigma	Has Content That May Reduce Stigma
COMDTINST M12792.3A	This Manual defines the health care services and occupational health program that is available to civilian personnel as required under [references]. In accordance with [federal statute], civilian employees paid from [NAF] are eligible for the same level and availability of care as [Coast Guard] "appropriated funds" personnel, subject to the regulations of the [OPM]. Civilian employees of other Federal agencies and contractor employees (e.g., security guards) are not covered by the provisions of this Instruction unless an exception is specifically identified in this Instruction.	х	
COMDTINST M16000.10A	This Manual established policies and procedures for the activities associated with the investigation of marine casualties, investigation into recreational boating deaths, detecting violations of criminal and other statutes for law enforcement purposes, taking remedial law enforcement action such as civil penalties and suspension and revocation, investigating pollution, and enforcing international treaties.	х	
COMDTINST M16790.1G	This Manual establishes policies and procedures for all Coast Guard members who are involved with the administration of the Coast Guard Auxiliary, including Auxiliarists, military, and civilian personnel.	х	x

NOTE: CGFAP = Coast Guard Family Advocacy Program. SOP = standard operating procedure.

APPENDIX H

Policies That Contain Negative Terminology with Implications for Stigma

Table H.1Policies That Contain Negative Terminology with Implications for Stigma

Organization	Document Number	Date of Issuance	Document Title	Implications for Stigma
Air Force	AFI 31-205	August 31, 2010	The Air Force Corrections System	"The following characteristics have been found less desirable in successfully completing the Return to Duty Program: patterns of poor judgment, patterns of impulsivity, history of violent behavior or serious psychological problems (bipolar disorder, recurrent major depression, any psychotic disorder, diagnosed personality disorder), untreated sexual offense, conflict avoidance and lack of initiative." Embedding MHDs in the list of undesirable characteristics communicates that an MHD renders someone as undesirable as someone who has a history of violence or sexual offense.
Air Force	AFI 36-2002	April 7, 1999	Regular Air Force and Special Category Accessions	When listing ineligibility criteria, uses the following language: "Has questionable moral character, history of antisocial behavior (including psychosis), frequent difficulties with law enforcement agencies, transsexualism and other gender identity disorders, exhibitionism transvestitism voyeurism, and other paraphilias." By linking DSM-recognized MHDs with "questionable moral character" and "frequent difficulties with law enforcement agencies," promotes a negative connotation of all.
Air Force	AFI 40-402, U.S. Air Force Academy Supplement	February 14, 2011	Protection of Human Subjects in Biomedical and Behavioral Research	When defining <i>institutionalized mentally infirm person</i> , the definition includes reference to "the mentally ill, the retarded, the emotionally disturbed, the psychotic, or the senile." This language identifies people by their disorders rather than specifying them as PWMHDs or other disorders (e.g., person with psychosis versus "the psychotic").
Air Force	AFI 44-172	March 14, 2011	Mental Health	Policy includes the text "Those with a potential personality disorder or other unsuitable condition will be referred" The term <i>unsuitable</i> has a negative connotation for PWMHDs.
Air Force	AFI 48-123	July 19, 2012	Medical Examinations and Standards	PWMHDs may be rendered "unsuitable." The term <i>unsuitable</i> has a negative connotation for PWMHDs.

Table H.1—Continued

Organization	Document Number	Date of Issuance	Document Title	Implications for Stigma
Air Force	AFI31-207_ AFDWSUP_I	May 18, 2012	Arming and Use of Force by Air Force Personnel	The policy includes a list of "personality and behavioral factors that may affect suitability to bear firearms" that includes descriptions of behaviors (some of which may be mental health symptoms) that use language with negative connotations (e.g., "Temper Tantrums. Childish outbursts of anger, sulking, or pouting over minor disappointments"; irritability, also described as being "overly sensitive/defensive," "arrogant/argumentative," and "insubordinate/ hostile"; and "desperate acting out/getting attention"). This language can affirm negative stereotypes about PWMHDs being childish.
Air Force	AFMAN 31- 201V4	November 17, 2011	High-Risk Response	When describing hostage and nonhostage crisis situations, nonhostage situations are described with the following language: "Subjects often act in an emotional, senseless, and self-destructive manner and have no clear goals. Unable to control their emotions in response to life's many stressors, they are motivated by anger, rage, frustration, hurt, confusion, or <i>depression</i> , and often exhibit purposeless, self-defeating behavior. They have no substantive or escape demands or totally unrealistic demands that they have no reasonable expectation of fulfilling. Disgruntled employees, jilted lovers, rejected spouses, aggrieved individuals, idealistic fanatics, <i>individuals with mental illness</i> , and others with unfulfilled aspirations who feel they have been wronged by others or events fall into this broad category. Non-Hostage situations often result in Domestic Violence, Workplace Violence, or an Active Shooter scenario." This description uses value-laden language and, by using the terms <i>depression</i> and <i>mental illness</i> , associates various MHDs with violence and unpredictability.
Air Force	AFRCI 36- 2001	June 1, 2009	Air Force Reserve Recruiting Procedures	When listing ineligibility criteria, uses the following language: "Questionable moral character, history of antisocial behavior, alcoholism, frequent difficulties with law enforcement agencies, history of psychotic disorders, transsexualism and other gender identity disorders, exhibitionism, transvestitism, voyeurism and other paraphilias." By linking DSM-recognized MHDs with "questionable moral character" and "frequent difficulties with law enforcement agencies," promotes a negative connotation of all.
Army	AR 190-58	March 22, 1989	Personal Security	Includes text stating that Special Response Teams may respond to situations involving "barricaded criminals and mentally disturbed persons with or without hostages, sniper incidents, threatened suicides, combatting terrorism operations, drug raids, warrant apprehension of dangerous individuals, and protective service missions." The term <i>mentally disturbed</i> is used to refer to PWMHDs and associates them with violence and dangerousness.

Table H.1—Continued

Organization	Document Number	Date of Issuance	Document Title	Implications for Stigma
Army	AR 623-3	June 5, 2012	Evaluation Reporting System	"Behavioral health issues include a variety of <i>unusual or inappropriate</i> behaviors that may be associated with post-traumatic stress disorder, mild traumatic brain injuries, combat stress or other stress, and/or suicidal thoughts or tendencies." MHDs are described in a value-laden manner (i.e., <i>unusual</i> , <i>inappropriate</i>).
Army	AR 635-200	June 6, 2005 (rapid action revision September 6, 2011)	Active Duty Enlisted Administrative Separations	Uses the phrase "mentally defective, deranged, or abnormal." Although "mentally defective" has a specific meaning in a legal context, "deranged" and "abnormal" do not.
Army	ATTP 3-39.10	June 20, 2011	Law and Order Operations	Policy includes the text: "Criminals, persons with psychological issues, insurgent elements, and many other categories of undesirable personnel may attempt to gain employment within the police force." The placement of the phrase "persons with psychological issues" amid the list reinforces a stereotype that people with mental health symptoms or disorders are untrustworthy or dangerous.
Marine Corps	NAVMC DIR 1500.58	February 13, 2006	Marine Corps Mentoring Program (MCMP) Guidebook	Includes the phrase "no one trusts or respects a leader who is not in emotional control at all times." Stating that a service member will not be trusted or respected if not in emotional control at all times may implicitly communicate that hiding emotional problems or symptoms of MHDs is necessary to avoid negative judgment from peers.
USD(P&R)	DoDI 1342.12	April 11, 2005	Provision of Early Intervention and Special Education Services to Eligible DoD Dependents	This policy uses the phrase "children who are schizophrenic," which uses mental health status as a descriptor for the children. Using alternative phrasing, such as "children with schizophrenia," would more clearly promote the idea that schizophrenia is a medical diagnosis.

NOTE: DSM = Diagnostic and Statistical Manual of Mental Disorders. Emphasis added in some cases.

APPENDIX I

Methods Used to Conduct the Expert Panel to Refine and Vet Priorities for Mental Health Stigma Reduction in the U.S. Department of Defense

This appendix describes how we convened the expert panel and who the expert panelists were, as well the process used by the expert panelists to develop the final set of recommended priorities included in this report. The final list of priorities is shown as Table I.1 of this appendix.

Expert Panel Methods

Experts participated in a modified Delphi process to develop a set of recommended priorities for mental health stigma reduction in DoD.

Convening the Expert Panel

We convened a panel consisting of 12 experts in two key areas:

- mental health stigma
- mental health in the military (PTSD, deployment psychology).

The experts who served on the panel in each of these areas are presented in the rest of this section.

Area 1: Experts on Mental Health Stigma

Rebecca L. Collins, Ph.D., heads the Health Promotion and Disease Prevention program in RAND Health. She currently leads an evaluation of California statewide efforts to reduce mental health stigma and discrimination. Collins' other research focuses on media's effects on adolescent development, and she leads an ongoing national longitudinal study testing associations between television viewing and adolescent sexual attitudes and behavior.

Patrick Corrigan, Psy.D., is a distinguished professor and associate dean for research in the Institute of Psychology at the Illinois Institute of Technology. He has been principal investigator of federally funded studies on rehabilitation, the stigma of

mental illness, and consumer-operated services and has contributed to the development and evaluation of antistigma programs in DoD and the VA.

Tracy Stecker, Ph.D., is an assistant professor at the Dartmouth Psychiatric Research Center at Dartmouth Medical School's Department of Community and Family Medicine. Stecker researches help-seeking behaviors in individuals with mental illness and has focused on the treatment of service members returning from the Iraq and Afghanistan wars with PTSD and substance abuse issues.

Bernice A. Pescosolido, Ph.D., is a distinguished professor of sociology at Indiana University and the director of the Indiana Consortium for Mental Health Services Research. Pescosolido's research focuses on social issues in health, illness, and healing, and she has been principal investigator for several major National Institutes of Health– funded studies on stigma.

David L. Vogel, Ph.D., is a professor in the Department of Psychology at Iowa State University. His research has focused on the role of stereotypes on people's interactions, measuring the stereotypes of stigmatized groups and their effects on relationships, help-seeking, and counselor training.

Nathaniel G. Wade, Ph.D., is a professor in the Department of Psychology and the director of Network Community Counseling Services at Iowa State University. His research interests include the psychology of forgiveness and religion, as well as counseling process and outcome.

Area 2: Experts on Mental Health in the Military (PTSD, deployment psychology)

Thomas Britt, Ph.D., is a professor of social and organizational psychology at Clemson University. Britt's research focuses on military psychology, including investigating how barriers to care (such as stigma) influence mental health seeking among military personnel and what factors promote resilience among service members returning from deployments.

Kristie Gore, Ph.D., is a senior behavioral social scientist at the RAND Corporation and the associate director for military health in the Forces and Resources Policy Center in the RAND National Security Research Division. Her research focuses on the evaluation of treatment strategies, care-seeking behavior, and care models, and she was recently involved in a large-scale Army effort to treat PTSD and depression in service members seeking primary care.

Charles W. Hoge, M.D., colonel, U.S. Army (retired), led the U.S. military's research program on the mental health effects of the Iraq and Afghanistan wars at the Walter Reed Army Institute of Research. A national expert, Hoge is frequently cited and called upon to provide testimony for his expertise in war-related mental health issues, including PTSD and TBI.

Nancy Skopp, Ph.D., is a research psychologist and program manager in the Research, Outcomes, Surveillance, and Evaluation Division of T2. Her research

focuses on suicidal behavior among service members and veterans and technologybased psychological health interventions for military personnel.

Christopher Warner, M.D., lieutenant colonel, U.S. Army, is the Army Surgeon General's psychiatric consultant and deputy commander for clinical services, U.S. Army Medical Department Activity–Alaska. He has published several studies that focus on the psychological effects of deployments and on the effectiveness of mental health screening.

John Roberts served in the U.S. Marine Corps from 1982 to 1996 and is currently the warrior relations executive vice president at the Wounded Warrior Project, an organization that helps meet the needs of service members with physical and mental injuries and raises public awareness about the needs of these warriors.

Developing a List of Priorities for Mental Health Stigma Reduction in the U.S. Department of Defense

We utilized a modified version of the RAND/UCLA Appropriateness Method (Fitch et al., 2001) to develop a list of recommended priorities. First, we sent experts a summary of the findings from the report and a set of proposed priorities, based on these findings. We presented proposed priorities as affirmative statements about what DoD should do to reduce mental health stigma among service members. We presented a short rationale for each priority and possible short- and long-term steps to achieve that priority. During the meeting, panelists discussed the strengths and weaknesses of each proposed priority and brainstormed on any priorities that were missing from the list. Panelists then rated each proposed priority on its validity and its importance to DoD. We used a nine-point Likert scale for each rating.

We defined a priority to be valid if the following was true:

• Adequate scientific evidence or professional consensus exists to support a link between the proposed priority and reducing stigma or improving service members' help-seeking.

We defined a priority to be *important* if both of the following were true:

- Addressing or undertaking the priority has a critical influence on reducing stigma or improving service members' help-seeking.
- There are serious adverse consequences from not addressing or undertaking the priority.

Expert-panel members were also given the opportunity to provide comments on each of the proposed priorities to help clarify how items should be modified or improved and to propose new priorities. After the panel had completed all ratings and submitted comments, the ratings were calculated, and we assembled the final list of priorities. An item was included in the final list of priorities if it received no more than one rating below a 6 (out of a possible 9) on both validity and importance.

Table I.1 shows the final list of priorities. The list contains 13 of the priorities the RAND team originally proposed, nine of which we have edited for clarity, and four new priorities proposed by experts.

Table I.1 Expert Ratings of Stigma-Reduction Priorities

Prior	ity	Importance	Validity				
Inter	Intervention						
1	Explore interventions that directly increase treatment-seeking.	8.3, 6–9	7.4, 3–9				
2	Consider evidence-based approaches to empowering service members with MHDs to support their peers.	7.8, 4–9	6.8, 4–9				
3	Design new or adapt existing intervention-delivery mechanisms to minimize operational barriers for service members seeking treatment.	7.6, 5–9	6.7, 4–9				
4	Embed stigma-reduction interventions in clinical treatment.	6.4, 2–9	6.2, 2–9				
5	Implement and evaluate stigma-reduction programs that target service members who have not yet developed symptoms of MHDs.	6.4, 2–9	5.9, 2–9				
Polic	у						
1	Provide better guidance for policies in which an MHD or treatment prohibits job opportunities or actions.	7.1, 1–9	6.9, 3–9				
2	Review the stigmatizing language identified in policies to determine whether it should be removed.	6.8, 5–9	6.4, 5–9				
3	Offer incentives for positive mental health behaviors.	6.1, 1–9	5.3, 1–8				
Rese	Research and evaluation						
1	Improve measures of prevalence to improve tracking of stigma and other barriers to care.	6.6, 2–9	6.3, 2–9				
2	Continue to improve and evaluate the modifications made to existing programs that begin to address stigma and other barriers to care.	7.4, 5–9	6.3, 4–9				
3	Examine the dynamic nature of stigma and how it interacts with interna and external conditions over time.	1 7.4, 2–9	6.6, 2–9				
4	Review classified DoD and service-specific policies to determine potential implications for mental health stigma and discrimination.	6.4, 1–9	5.2, 1–9				
Over	arching (i.e., cuts across programs, policies, and research)						
1	Convene a task force to explore the tensions between a command's need to know a service member's mental health status and treatment history and the service member's need for privacy	6.7, 1–9	6.4, 3–9				

NOTE: The first value in each importance or validity cell is the mean value assigned by members of the expert panel. The range in each cell is the range of values assigned by the members.

We found that the expert ratings of the proposed priorities varied widely, as evidenced by the ranges in the table above. Expert ratings most strongly converged on ratings regarding intervention priority 1 (explore interventions that directly increase treatment-seeking). Experts' ratings of importance for this priority ranged from 6 to 9. The validity of this priority was rated 6 or above except for two expert ratings. The mean scores for this priority were the highest of all priorities in both importance and validity. Experts also strongly agreed on the importance of intervention priorities 2 and 3 and research and evaluation priority 3; all but one rating of importance was below 6 for these priorities. Additionally, only two experts rated the importance of overarching priority 1, policy priority 1, and research and evaluation priority 2 lower than 6. Intervention priority 5, policy priority 3, and research and evaluation priority 4 average ratings were below 6 for importance or validity or both. The validity of research and evaluation priority 4 had disagreement among experts: Six out of ten rankings fell below 6. Research and evaluation priority 4 also had the lowest mean rating for validity of any other priority. Overall, the mean validity ratings were lower than importance ratings for the majority of priorities. The range of validity ratings was 5.2 to 7.4, compared with 4.4 to 8.3 for importance. Intervention priority 1 and policy priority 3 were the only validity ratings that received less two or less rankings below 6.

The mean rankings for an additional policy priority suggested disagreement among experts. This priority (which did not make the final list above) received four rankings below 4 on validity and four rankings below 3 on importance. Because of this significant disagreement, this policy priority was not included in the list of final priorities, but we describe it next.

Excluded Policy Priority: Develop a Policy That Dictates Consequences for Service Members Who Display Prejudice or Acts of Discrimination Against Other Service Members Based on Mental Health Status

We were unable to locate any policies that specified consequences for service members who discriminate against or harass other service members who have mental health conditions. In order to deter prejudice and discrimination, DoD should institute a DoD-wide policy that specifies clear consequences for prejudicial or discriminatory behaviors. This policy could be modeled after a similar policy instituted by the Coast Guard (COMDTNOTE 12271). This policy states that "making disparaging references, expressing a stereotyped view, or associating undesirable characteristics with a person's . . . disability" is an offense that results in punishment varying from a ten-day suspension to removal.

- Short-term steps: Institute a policy that specifies consequences for prejudicial or discriminatory behaviors related to mental health status.
- Long-term steps: Monitor enforcement of the policy.

5AFI 36-102—See Commander, Fifth Air Force, 2012 (2012).

Ackerman, Sandra R., *Attitudes Toward Help-Seeking and Mental Health Among College Athletes: Impact of a Psycho-Educational Workshop*, South Orange, N.J.: Seton Hall University, doctoral thesis, 2011. As of January 29, 2014: http://scholarship.shu.edu/dissertations/37/

Adler, Amy B., Paul D. Bliese, Dennis McGurk, Charles W. Hoge, and Carl Andrew Castro, "Battlemind Debriefing and Battlemind Training as Early Interventions with Soldiers Returning from Iraq: Randomization by Platoon," *Journal of Consulting and Clinical Psychology*, Vol. 77, No. 5, October 2009, pp. 928–940.

, "Battlemind Debriefing and Battlemind Training as Early Interventions with Soldiers Returning from Iraq," *Sport, Exercise, and Performance Psychology*, Vol. 1, Supp. 1, August 2011, pp. 66–83.

ADTP 6-22—See Headquarters, Department of the Army, 2012c.

Ægisdóttir, Stefanía, Michael P. O'Heron, Joel M. Hartong, Sarah A. Haynes, and Miranda K. Linville, "Enhancing Attitudes and Reducing Fears About Mental Health Counseling: An Analogue Study," *Journal of Mental Health Counseling*, Vol. 33, No. 4, October 2011, pp. 327–346.

AETC Instruction 36-2205, Vol. 2—*See* Commander, Air Education and Training Command, 2012a.

AETC Instruction 36-2205, Vol. 3—*See* Commander, Air Education and Training Command, 2012c.

AETC Instruction 36-2205, Vol. 5—*See* Commander, Air Education and Training Command, 2013b.

AETC Instruction 36-2205, Vol. 6—*See* Commander, Air Education and Training Command, 2012d.

AETC Instruction 36-2205, Vol. 8—*See* Commander, Air Education and Training Command, 2013a.

AETC Instruction 36-2205, Vol. 12—*See* Commander, Air Education and Training Command, 2012b.

AFGM 48-04—See Headquarters, U.S. Air Force, Surgeon General, 2012.

AFGSCGM 10-02—See Headquarters, Air Force Global Strike Command, 2012.

AFI 10-203—See Secretary of the Air Force, 2013.

AFI 10-403—*See* Secretary of the Air Force, 2012d (2013).

- AFI 10-404—See Secretary of the Air Force, 2011c.
- AFI 11-402-See Secretary of the Air Force, 2010c (2012).
- AFI 13-204V3_AFSOCSUP_I-See Secretary of the Air Force, 2010a (2012).
- AFI 31-204_USAFESUP_INCIRLIKABSUP_I—See Secretary of the Air Force, 2000b (2011).
- AFI 31-205-See Secretary of the Air Force, 2004a (2010).
- AFI 31-206—See Secretary of the Air Force, 2002c (2008).
- AFI 31-207—See Secretary of the Air Force, 2009a (2012).
- AFI 31-501—See Secretary of the Air Force, 2005a (2012).
- AFI 32-3001_AFDWSUP_I—See Secretary of the Air Force, 2007a (2009).
- AFI 36-704—See Secretary of the Air Force, 1994.
- AFI 36-2002—See Secretary of the Air Force, 1999 (2011).
- AFI 36-2110_AFDWSUP_I—See Secretary of the Air Force, 2005c (2007).
- AFI 36-2113—See Secretary of the Air Force, 2007c (2010).
- AFI 36-2706—See Secretary of the Air Force, 2010b.
- AFI 36-2910—See Secretary of the Air Force, 2002b (2010).
- AFI 36-3206—See Secretary of the Air Force, 2004b (2013).
- AFI 36-3208—See Secretary of the Air Force, 2004c (2013).
- AFI 36-3209—See Secretary of the Air Force, 2005b (2011).
- AFI 36-3212—See Secretary of the Air Force, 2006a (2009).
- AFI 36-6001-See Secretary of the Air Force, 2008b (2010).
- AFI 40-301—See Air Force Medical Operations Agency, 2009.
- AFI 40-402_USAFASUP_I—See Secretary of the Air Force, 2005d.
- AFI 40-404—See Secretary of the Air Force, 1997.
- AFI 41-210-See Secretary of the Air Force, 2012a.
- AFI 41-307—See Secretary of the Air Force, 2003 (2014).
- AFI 44-109—See Secretary of the Air Force, 2000a (2010).
- AFI 44-153—See Secretary of the Air Force, 2011b.
- AFI 44-165—See Secretary of the Air Force, 2007b.
- AFI 44-172—See Secretary of the Air Force, 2011a.
- AFI 48-123—See Secretary of the Air Force, 2013c.
- AFI 52-102V2—See U.S. Air Force, 2012.
- AFI 90-505—See Secretary of the Air Force, 2012c.
- AFMAN 10-3902—See Secretary of the Air Force, 2006b (2012).
- AFMAN 31-201V4—See Secretary of the Air Force, 2002a.
- AFMAN 48-146—See Secretary of the Air Force, 2012e (2012).

AFPAM 36-2241—See Department of the Air Force, 2013.

AFPAM 44-160—See U.S. Air Force, 2001.

AFPD 10-2—See Secretary of the Air Force, 2012f.

AFRCI 36-2001—See Commander, Air Force Reserve Command, 2009 (2011).

Agerbo, Esben, Majella Byrne, William W. Eaton, and Preben B. Mortensen, "Marital and Labor Market Status in the Long Run in Schizophrenia," *Archives of General Psychiatry*, Vol. 61, No. 1, 2004, pp. 28–33.

Aho-Mustonen, Kati, Jari Tiihonen, Eila Repo-Tiihonen, Olli-Pekka Ryynänen, Raili Miettinen, and Hannu Räty, "Group Psychoeducation for Long-Term Offender Patients with Schizophrenia: An Exploratory Randomised Controlled Trial," *Criminal Behaviour and Mental Health*, Vol. 21, 2011, pp. 163–176.

AI 8—See Director of Administration and Management, 2008.

AI 9—See Office of Equal Opportunity and Diversity, 2013b.

AI 31—See Office of Equal Opportunity and Diversity, 2013a.

Air Force Medical Operations Agency, Mental Health Division, *Family Advocacy*, Air Force Instruction 40-301, November 30, 2009. As of February 4, 2014: http://static.e-publishing.af.mil/production/1/af_sg/publication/afi40-301/afi40-301.pdf

Ajzen, Icek, "The Theory of Planned Behavior," *Organizational Behavior and Human Decision Processes*, Vol. 50, 1991, pp. 179–211.

Ajzen, Icek, Robert Louis Heilbroner, Martin Fishbein, and Lester C. Thurow, *Understanding Attitudes and Predicting Social Behavior*, Englewood Cliffs, N.J.: Prentice-Hall, 1980.

Akaike, Hirotugu, "Factor Analysis and AIC," *Psychometrika*, Vol. 52, No. 3, September 1987, pp. 317–332.

Allport, Gordon W., The Nature of Prejudice, Reading, Mass.: Addison-Wesley Pub. Co., 1979.

Alrutz, Anna Stowe, *The Role of Communication, Prior Experience and Beliefs as Factors Influencing Combat Stress Recommendations from Military Spouses*, Florida State University, master's thesis, 2006.

Altindag, A., M. Yanik, A. Ucok, K. Alptekin, and M. Ozkan, "Effects of an Antistigma Program on Medical Students' Attitudes Towards People with Schizophrenia," *Psychiatry and Clinical Neurosciences*, Vol. 60, No. 3, June 2006, pp. 283–288.

Alvidrez, Jennifer, Lonnie R. Snowden, and Dawn M. Kaiser, "The Experience of Stigma Among Black Mental Health Consumers," *Journal of Health Care for the Poor and Underserved*, Vol. 19, No. 3, 2008, pp. 874–893.

Alvidrez, Jennifer, Lonnie R. Snowden, Stephen M. Rao, and Alicia Boccellari, "Psychoeducation to Address Stigma in Black Adults Referred for Mental Health Treatment: A Randomized Pilot Study," *Community Mental Health Journal*, Vol. 45, 2009, pp. 127–136.

American Forces Information Service, U.S. Department of Defense, *The Armed Forces Officer*, U.S. Department of Defense GEN-36A, Department of the Army Pamphlet 600-2 (Rev. 1988), Naval Education and Training 46905-A, Air Force Pamphlet 190-13 (Rev. 1988), Navy and Marine Corps Form 2563 (Rev. 1988), c. 1988. As of February 7, 2014:

http://www.marines.mil/Portals/59/Publications/NAVMC%202563.pdf

American Psychological Association, Presidential Task Force on Preventing Discrimination and Promoting Diversity, *Dual Pathways to a Better America: Preventing Discrimination and Promoting Diversity*, Washington, D.C., 2012. As of January 29, 2014: http://www.apa.org/pubs/info/reports/promoting-diversity.aspx

Amnesty International, "Mental Health," undated; referenced September 13, 2013. As of January 29, 2014:

http://www.amnesty.ie/mentalhealth

Anderson, Kelly, and Jehannine C. Austin, "Effects of a Documentary Film on Public Stigma Related to Mental Illness Among Genetic Counselors," *Journal of Genetic Counseling*, Vol. 21, No. 4, August 2012, pp. 573–581.

Anderson, Rosemary J., and David Pierce, "Assumptions Associated with Mental Health Literacy Training: Insights from Initiatives in Rural Australia," *Advances in Mental Health*, Vol. 10, No. 3, June 2012, pp. 258–267.

ANGDIR 90-266-See Chief, National Guard Bureau, 2010 (2012).

Angermeyer, Matthias C., and S. Dietrich, "Public Beliefs About and Attitudes Towards People with Mental Illness: A Review of Population Studies," *Acta Psychiatrica Scandinavica*, Vol. 113, No. 3, March 2006, pp. 163–179.

Angermeyer, Matthias C., A. Holzinger, and Herbert Matschinger, "Mental Health Literacy and Attitude Towards People with Mental Illness: A Trend Analysis Based on Population Surveys in the Eastern Part of Germany," *European Psychiatry*, Vol. 24, No. 4, May 2009, pp. 225–232.

Angermeyer, Matthias C., and Herbert Matschinger, "The Effect of Personal Experience with Mental Illness on the Attitude Towards Individuals Suffering from Mental Disorders," *Social Psychiatry and Psychiatric Epidemiology*, Vol. 31, No. 6, 1996, pp. 321–326.

——, "Causal Beliefs and Attitudes to People with Schizophrenia: Trend Analysis Based on Data From Two Population Surveys in Germany," *British Journal of Psychiatry*, Vol. 186, 2005, pp. 331–334.

Angermeyer, Matthias C., Herbert Matschinger, and S. G. Riedel-Heller, "Whom to Ask for Help in Case of a Mental Disorder? Preferences of the Lay Public," *Social Psychiatry and Psychiatric Epidemiology*, Vol. 34, No. 4, 1999, pp. 202–210.

ANGI 10-2503—See National Guard Bureau, 2011.

ANGI 10-2504—See Department of the Army and Department of the Air Force, 2009b.

ANGI 36-2001—See Chief, National Guard Bureau, 2009.

ANGI 36-2002—See Chief, National Guard Bureau, 2012b.

AR 1-75—*See* Headquarters, Department of the Army; Headquarters, Department of the Navy; and Headquarters, Department of the Air Force, 2000.

AR 20-1—See Headquarters, Department of the Army, 2010f (2012).

AR 25-6—See Headquarters, Department of the Army, 2014a.

AR 25-55—See Headquarters, Department of the Army, 1997.

AR 40-3—See Headquarters, Department of the Army, 2013b.

AR 40-8—See Headquarters, Department of the Army, 2007c.

AR 40-66—See Headquarters, Department of the Army, 2008b (2010).

AR 40-400—See Headquarters, Department of the Army, 2010a (2011).

AR 40-501—See Headquarters, Department of the Army, 2007f (2011).

AR 50-1—See Headquarters, Department of the Army, 2008c.

AR 50-5—See Headquarters, Department of the Army, 2000a.

AR 135-178—See Headquarters, Department of the Army, 2007a (2011).

AR 165-1—See Headquarters, Department of the Army, 2009d.

AR 190-5—*See* Headquarters, Department of the Army; Headquarters, Department of the Navy; Headquarters, Department of the Air Force; Headquarters, Department of the Marine Corps; and Defense Logistics Agency, 2006.

AR 190-8—*See* Headquarters, Department of the Army; Headquarters, Department of the Navy; Headquarters, Department of the Air Force; and Headquarters, Department of the Marine Corps, 1997.

AR 190-30—See Headquarters, Department of the Army, 2005c.

AR 190-47—See Headquarters, Department of the Army, 2006b.

AR 190-58—See Headquarters, Department of the Army, 1989.

AR 195-3—See Headquarters, Department of the Army, 1987.

AR 195-5—See Headquarters, Department of the Army, 2013a.

AR 210-25—See Headquarters, Department of the Army, 2004b.

AR 210-26—See Headquarters, Department of the Army, 2009e (2011).

AR 210-35—See Headquarters, Department of the Army, 2005a.

AR 215-1—See Headquarters, Department of the Army, 2010d.

AR 215-3—See Headquarters, Department of the Army, 2003b.

AR 350-1—See Headquarters, Department of the Army, 2009f (2011).

AR 350-51—See Headquarters, Department of the Army, 2001.

AR 360-1—See Headquarters, Department of the Army, 2000b.

AR 380-67—See Headquarters, Department of the Army, 2014b.

AR 600-8-4—See Headquarters, Department of the Army, 2008d.

AR 600-8-19—See Headquarters, Department of the Army, 2010c (2011).

AR 600-8-22—See Headquarters, Department of the Army, 2006f (2013).

AR 600-8-101—See Headquarters, Department of the Army, 2003a.

AR 600-20—See Headquarters, Department of the Army, 2008a.

AR 600-63—See Headquarters, Department of the Army, 2007b (2010).

AR 601-1-See Headquarters, Department of the Army, 2007d (2011).

AR 601-10—See Headquarters, Department of the Army, 2009c.

AR 601-210—See Headquarters, Department of the Army, 2011a (2013).

AR 601-280—See Headquarters, Department of the Army, 2006a (2011).

AR 608-18-See Headquarters, Department of the Army, 2007e (2011).

AR 608-75—See Headquarters, Department of the Army, 2006e (2011).

AR 614-30—See Headquarters, Department of the Army, 2010b.

AR 614-200—See Headquarters, Department of the Army, 2009a (2011).

AR 621-202—See Headquarters, Department of the Army, 1992 (2011).

AR 623-3—See Headquarters, Department of the Army, 2012b.

AR 633-30—*See* Headquarters, Department of the Army; and Headquarters, Department of the Air Force, 1989.

AR 635-200—See Headquarters, Department of the Army, 2005b (2011).

AR 690-600—See Headquarters, Department of the Army, 2004a.

Arboleda-Flórez, Julio, and Heather Stuart, "From Sin to Science: Fighting the Stigmatization of Mental Illnesses," *Canadian Journal of Psychiatry*, Vol. 57, No. 8, August 2012, pp. 457–463.

Aromaa, Esa, Asko Tolvanen, Jyrki Tuulari, and Kristian Wahlbeck, "Personal Stigma and Use of Mental Health Services Among People with Depression in a General Population in Finland," *BMC Psychiatry*, Vol. 11, No. 52, March 31, 2011a.

———, "Predictors of Stigmatizing Attitudes Towards People with Mental Disorders in a General Population in Finland," *Nordic Journal of Psychiatry*, Vol. 65, No. 2, April 2011b, pp. 125–132.

Assistant Secretary of Defense for Command, Control, Communications, and Intelligence, "DoD Personnel Security Program," Department of Defense Directive 5200.2, April 9, 1999.

Assistant Secretary of Defense for Force Management Policy, "The DoD Civilian Equal Employment Opportunity (EEO) Program," Department of Defense Directive 1440.1, May 21, 1987; incorporating through change 3 April 17, 1992; certified current as of November 21, 2003. As of February 7, 2014:

http://www.dtic.mil/whs/directives/corres/pdf/144001p.pdf

———, Child Development Programs (CDPs), Department of Defense Instruction 6060.2, January 19, 1993; administrative reissuance incorporating through change 2 August 25, 1998. As of February 7, 2014:

http://www.dtic.mil/whs/directives/corres/pdf/606002p.pdf

Assistant Secretary of Defense for Health Affairs, "Rehabilitation and Referral Services for Alcohol and Drug Abusers," Department of Defense Instruction 1010.6, March 13, 1985.

—_____, DoD Health Information Privacy Regulation, Department of Defense Regulation 6025.18-R, January 2003a. As of February 7, 2014: http://www.dtic.mil/whs/directives/corres/pdf/602518r.pdf

, "Armed Forces Institute of Pathology Operations," Department of Defense Instruction 5154.30, March 18, 2003b. As of February 7, 2014: http://www.dtic.mil/whs/directives/corres/pdf/515430p.pdf

———, *Military Health System (MHS) Clinical Quality Assurance (CQA) Program Regulation*, Department of Defense Regulation 6025.13-R, June 11, 2004. As of June 10, 2014: https://www.qmo.amedd.army.mil/ptsafety/p602513r.pdf

Assistant Secretary of Defense for International Security Affairs, *Repatriation of Prisoners of War (POW), Hostages, Peacetime Government Detainees and Other Missing or Isolated Personnel,* Department of Defense Instruction 2310.4, November 21, 2000. As of February 7, 2014: http://www.dtic.mil/dpmo/laws_directives/documents/dodi_2310_4_repatriation.pdf

Assistant Secretary of Defense for Manpower, Reserve Affairs, and Logistics, *Nondiscrimination on the Basis of Handicap in Programs and Activities Assisted or Conducted by the Department of Defense*, Department of Defense Directive 1020.1, March 31, 1982; incorporating change 1 November 16, 1994; certified current as of November 21, 2003. As of February 7, 2014: http://www.dtic.mil/whs/directives/corres/pdf/102001p.pdf

Atilola, Olayinka, and Funmilayo Olayiwola, "Stigmatization of Mental Illness in Nigerian Home Videos," *Psychiatric Services*, Vol. 62, No. 6, June 1, 2011, pp. 683–684.

ATTP 3-21.9—See Headquarters, Department of the Army, 2010g.

ATTP 3-21.71—See Headquarters, Department of the Army, 2010e.

ATTP 3-39.10—See Headquarters, Department of the Army, 2011b.

ATTP 4-02—See Headquarters, Department of the Army, 2013.

Back Each Other Up, undated home page; referenced September 13, 2013. As of January 29, 2014: http://backeachotherup.org/

Bahora, Masuma, Sonya Hanafi, Victoria H. Chien, and Michael T. Compton, "Preliminary Evidence of Effects of Crisis Intervention Team Training on Self-Efficacy and Social Distance," *Administration and Policy in Mental Health*, Vol. 35, No. 3, May 2008, pp. 159–167.

Baldwin, M. L., and S. C. Marcus, "Stigma, Discrimination, and Employment Outcomes among Persons with Mental Health Disabilities," in Izabela Z. Schultz and E. Sally Rogers, eds., *Work Accommodation and Retention in Mental Health*, New York: Springer, 2011, pp. 53–69.

Bambauer, K. Z., and H. G. Prigerson, "The Stigma Receptivity Scale and Its Association with Mental Health Service Use Among Bereaved Older Adults," *Journal of Nervous and Mental Disease*, Vol. 194, No. 2, 2006, pp. 139–141.

Barke, Antonia, Seth Nyarko, and Dorothee Klecha, "The Stigma of Mental Illness in Southern Ghana: Attitudes of the Urban Population and Patients' Views," *Social Psychiatry and Psychiatric Epidemiology*, Vol. 46, No. 11, November 2011, pp. 1191–1202.

Barlas, Frances M., William Bryan Higgins, Jacqueline C. Pflieger, and Kelly Diecker, 2011 Department of Defense Health Related Behaviors Survey of Active Duty Military Personnel, Washington, D.C.: U.S. Department of Defense, February 2013. As of February 3, 2014: http://www.murray.senate.gov/public/_cache/files/889efd07-2475-40ee-b3b0-508947957a0f/ final-2011-hrb-active-duty-survey-report.pdf

Barnes, J. B., A. Nickerson, A. B. Adler, and B. T. Litz, "Perceived Military Organizational Support and Peacekeeper Distress: A Longitudinal Investigation," *Psychological Services*, Vol. 10, No. 2, May 2013, pp. 177–182.

Barney, L. J., K. M. Griffiths, H. Christensen, and A. F. Jorm, "The Self-Stigma of Depression Scale (SSDS): Development and Psychometric Evaluation of a New Instrument," *International Journal of Methods in Psychiatric Research*, Vol. 19, No. 4, 2010, pp. 243–254.

Barney, L. J., K. M. Griffiths, A. F. Jorm, and H. Christensen, "Stigma About Depression and Its Impact on Help-Seeking Intentions," *Australian and New Zealand Journal of Psychiatry*, Vol. 40, No. 1, January 2006, pp. 51–54.

Barrett, Clark C., "Unarmed and Dangerous: The Holistic Preparation of Soldiers for Combat," *Ethical Human Psychology and Psychiatry*, Vol. 13, No. 2, 2011, pp. 95–114.

Barry, Colleen L., Emma E. McGinty, Jon S. Vernick, and Daniel W. Webster, "After Newtown: Public Opinion on Gun Policy and Mental Illness," *New England Journal of Medicine*, Vol. 368, March 21, 2013, pp. 1077–1081.

Bathje, Geoff J., and John B. Pryor, "The Relationships of Public and Self-Stigma to Seeking Mental Health Services," *Journal of Mental Health Counseling*, Vol. 33, No. 2, April 2011, pp. 161–176.

Beldie, Alina, Johan A. den Boer, Cecilia Brain, Eric Constant, Maria Luisa Figueira, Igor Filipcic, Benoît Gillain, Miro Jakovljevic, Marek Jarema, Daniela Jelenova, Oguz Karamustafalioglu, Blanka Kores Plesnicar, Andrea Kovacsova, Klara Latalova, Josef Marksteiner, Filipa Palha, Jan Pecenak, Jan Prasko, Dan Prelipceanu, Petter Andreas Ringen, Norman Sartorius, Erich Seifritz, Jaromir Svestka, Magdalena Tyszkowska, and Johannes Wancata, "Fighting Stigma of Mental Illness in Midsize European Countries," *Social Psychiatry and Psychiatric Epidemiology*, Vol. 47, No. 1, April 2012, pp. 1–38.

Beltran, Ruth O., Justin Newton Scanlan, Nicola Hancock, and Tim Luckett, "The Effect of First Year Mental Health Fieldwork on Attitudes of Occupational Therapy Students Towards People with Mental Illness," *Australian Occupational Therapy Journal*, Vol. 54, No. 1, March 2007, pp. 42–48.

Ben-Zeev, D., P. W. Corrigan, T. W. Britt, and L. Langford, "Stigma of Mental Illness and Service Use in the Military," *Journal of Mental Health*, Vol. 21, No. 3, June 2012, pp. 264–273.

Blumenthal, R., and J. Endicott, "Barriers to Seeking Treatment for Major Depression," *Depression and Anxiety*, Vol. 4, 1996–1997, pp. 273–278.

Boardman, Felicity, Frances Griffiths, Renata Kokanovic, Maria Potiriadis, Christopher Dowrick, and Jane Gunn, "Resilience as a Response to the Stigma of Depression: A Mixed Methods Analysis," *Journal of Affective Disorders*, Vol. 135, No. 1–3, December 2011, pp. 267–276.

Borinstein, A. B., "Public Attitudes Toward Persons with Mental Illness," *Health Affairs*, Vol. 11, No. 3, August 1992, pp. 186–196.

Bormann, Jill E., Lin Liu, Steven R. Thorp, and Ariel J. Lang, "Spiritual Wellbeing Mediates PTSD Change in Veterans with Military-Related PTSD," *International Journal of Behavioral Medicine*, Vol. 19, No. 4, December 2012, pp. 496–502.

Borum, R., M. W. Deane, H. J. Steadman, and J. Morrissey, "Police Perspectives on Responding to Mentally Ill People in Crisis: Perceptions of Program Effectiveness," *Behavioral Sciences and the Law*, Vol. 16, No. 4, Autumn 1998, pp. 393–405.

Bower, Deborah L., and W. Gene Pettit, "Albuquerque Police Department's Crisis Intervention Team: A Report Card," *FBI Law Enforcement Bulletin*, Vol. 70, No. 2, February 2001, pp. 1–6.

Boysen, Guy A., and David L. Vogel, "The Relationship Between Level of Training, Implicit Bias, and Multicultural Competency Among Counselor Trainees," *Training and Education in Professional Psychology*, Vol. 2, No. 2, May 2008, pp. 103–110.

Brancu, Mira, Kristy Straits-Tröster, and Harold Kudler, "Behavioral Health Conditions Among Military Personnel and Veterans: Prevalence and Best Practices for Treatment," *North Carolina Medical Journal*, Vol. 72, No. 1, 2011, pp. 54–60.

Briggs, Cynthia A., and Stephanie Reneson, "Counseling Addicted Veterans: What to Know and How to Help," *VISTAS*, 2010. As of June 8, 2014: http://counselingoutfitters.com/vistas/vistas10/Article_88.pdf

Britt, Thomas W., Elizabeth A. Bennett, Michael Crabtree, Christine Haugh, Kalifa Oliver, Anna McFadden, and Cynthia L. S. Pury, "The Theory of Planned Behavior and Reserve Component Veteran Treatment Seeking," *Military Psychology*, Vol. 23, No. 1, January 2011, pp. 82–96.

Britt, Thomas W., Tiffany M. Greene-Shortridge, Sarah Brink, Quyen B. Nguyen, Jaclyn Rath, Anthony L. Cox, Charles W. Hoge, and Carl Andrew Castro, "Perceived Stigma and Barriers to Care for Psychological Treatment: Implications for Reactions to Stressors in Different Contexts," *Journal* of Social and Clinical Psychology, Vol. 27, No. 4, 2008, pp. 317–335. Britt, Thomas W., Kathleen M. Wright, and DeWayne Moore, "Leadership as a Predictor of Stigma and Practical Barriers Toward Receiving Mental Health Treatment: A Multilevel Approach," Psychological Services, Vol. 9, No. 1, February 2012, pp. 26-37.

Brohan, Elaine, Dolores Gauci, Norman Sartorius, and Graham Thornicroft, "Self-Stigma, Empowerment and Perceived Discrimination Among People with Bipolar Disorder or Depression in 13 European Countries: The GAMIAN-Europe Study," Journal of Affective Disorders, Vol. 129, No. 1–3, March 2011, pp. 56–63.

Bronfenbrenner, Urie, The Ecology of Human Development: Experiments by Nature and Design, Cambridge, Mass.: 1979.

Brown, C., K. O. Conner, V. C. Copeland, N. Grote, S. Beach, D. Battista, and C. F. Reynolds III, "Depression Stigma, Race, and Treatment Seeking Behavior and Attitudes," Journal of Community Psychology, Vol. 38, No. 3, April 2010, pp. 350-368.

Brown, Kaylene, and Loretta J. Bradley, "Reducing the Stigma of Mental Illness," Journal of Mental Health Counseling, Vol. 24, No. 1, January 2002, pp. 81-87.

Brown, M. C., A. H. Creel, C. C. Engel, R. K. Herrell, and C. W. Hoge, "Factors Associated with Interest in Receiving Help for Mental Health Problems in Combat Veterans Returning from Deployment to Iraq," Journal of Nervous and Mental Disease, Vol. 199, No. 10, October 2011, pp. 797-801.

Brown, Seth A., "Implementing a Brief Hallucination Simulation as a Mental Illness Stigma Reduction Strategy," Community Mental Health Journal, Vol. 46, No. 5, October 2010, pp. 500-504.

Brown, Seth A., Yolanda Evans, Kelly Espenschade, and Maureen O'Connor, "An Examination of Two Brief Stigma Reduction Strategies: Filmed Personal Contact and Hallucination Simulations," Community Mental Health Journal, Vol. 46, No. 5, October 2010, pp. 494-499.

Bruffaerts, Ronny, Koen Demyttenaere, Irving Hwang, Wai-Tat Chiu, Nancy Sampson, Ronald C. Kessler, Jordi Alonso, Guilherme Borges, Giovanni de Girolamo, Ron de Graaf, Silvia Florescu, Oye Gureje, Chiyi Hu, Elie G. Karam, Norito Kawakami, Stanislav Kostyuchenko, Viviane Kovess-Masfety, Sing Lee, Daphna Levinson, Herber Matschinger, Jose Posada-Villa, Rajesh Sagar, Kate M. Scott, Dan J. Stein, Toma Tomov, Maria Carmen Viana, and Matthew K. Nock, "Treatment of Suicidal People Around the World," British Journal of Psychiatry, Vol. 199, 2011, pp. 64-70.

Bryan, Craig J., and Chad E. Morrow, "Circumventing Mental Health Stigma by Embracing the Warrior Culture: Lessons Learned from the Defender's Edge Program," Professional Psychology: Research and Practice, Vol. 42, No. 1, February 2011, pp. 16-23.

Buchanan, Avril, and Michael Murray, "Using Participatory Video to Challenge the Stigma of Mental Illness: A Case Study," International Journal of Mental Health Promotion, Vol. 14, No. 1, 2012, pp. 35–43.

BUMEDINST 1755.1—See Bureau of Medicine and Surgery, 2005.

BUMEDINST 5041.6—See Bureau of Medicine and Surgery, 2011b.

BUMEDINST 6300.17-See Bureau of Medicine and Surgery, 2009.

BUMEDINST 6400.3A—See Bureau of Medicine and Surgery, 2012.

BUMEDINST 6520.2-See Bureau of Medicine and Surgery, 2011a.

BUMEDNOTE 6100—See Bureau of Medicine and Surgery, 2010.

Bureau of Labor Statistics, "Labor Force Statistics from the Current Population Survey," undated. As of February 6, 2014:

http://www.bls.gov/cps/

Bureau of Medicine and Surgery, *Educational and Developmental Intervention Services (EDIS) Early Intervention Services*, Bureau of Medicine and Surgery Instruction 1755.1, October 17, 2005. As of February 9, 2014:

http://www.med.navy.mil/directives/externaldirectives/1755.1.pdf

———, "Navy Medicine Clinical Case Management," Bureau of Medicine and Surgery Instruction 6300.17, November 23, 2009. As of February 9, 2014: http://www.med.navy.mil/directives/externaldirectives/6300.17.pdf

, "Spring 2010 Physical Fitness Assessment," Bureau of Medicine and Surgery Notice 6100, February 17, 2010. As of February 9, 2014: http://www.med.navy.mil/directives/Internal%20Notes/ Note%206100%20(17%20Feb%202010).pdf

———, "Evaluation and Disposition of Patients Presenting with Suicidal Ideation or Behavior," Bureau of Medicine and Surgery Instruction 6520.2, March 7, 2011a. As of February 9, 2014: http://www.med.navy.mil/directives/ExternalDirectives/6520.2.pdf

, "Navy Medicine Hotline Program," Bureau of Medicine and Surgery Instruction 5041.6, July 25, 2011b. As of February 5, 2014: http://www.med.navy.mil/directives/ExternalDirectives/5041.6.pdf

, "Ships and Stations Having Medical Department Personnel," Bureau of Medicine and Surgery Instruction 6400.3A, August 24, 2012. As of February 9, 2014: http://www.med.navy.mil/directives/ExternalDirectives/6400.3A.pdf

Burnam, M. Audrey, Lisa S. Meredith, Terri Tanielian, and Lisa H. Jaycox, "Mental Health Care for Iraq and Afghanistan War Veterans," *Health Affairs*, Vol. 28, No. 3, May–June 2009, pp. 771–782.

Burns, Shaun Michael, and James R. Mahalik, "Suicide and Dominant Masculinity Norms Among Current and Former United States Military Servicemen," *Professional Psychology: Research and Practice*, Vol. 42, No. 5, October 2011, pp. 347–353.

Byrne, Peter, "Stigma of Mental Illness: Changing Minds, Changing Behavior," *British Journal of Psychiatry*, Vol. 174, January 1999, pp. 1–2.

Calear, Alison L., Kathleen M. Griffiths, and Helen Christensen, "Personal and Perceived Depression Stigma in Australian Adolescents: Magnitude and Predictors," *Journal of Affective Disorders*, Vol. 129, No. 1–3, March 2011, pp. 104–108.

Castro, Carl Andrew, Amy B. Adler, Dennis McGurk, and Paul D. Bliese, "Mental Health Training with Soldiers Four Months After Returning from Iraq: Randomization by Platoon," *Journal of Traumatic Stress*, Vol. 25, No. 4, August 2012, pp. 376–383.

Castro, Carl Andrew; Charles C. Engel, Jr.; and Amy B. Adler, "The Challenge of Providing Mental Health Prevention and Early Intervention in the U.S. Military," in Brett T. Litz, ed., *Early Intervention for Trauma and Traumatic Loss*, New York: Guilford Press, 2004, pp. 301–318.

Centers for Disease Control and Prevention, Substance Abuse and Mental Health Services Administration, National Association of County Behavioral Health and Developmental Disability Directors, National Institute of Mental Health, and Carter Center Mental Health Program, *Attitudes Toward Mental Illness: Results from the Behavioral Risk Factor Surveillance System*, Atlanta, Ga.: Centers for Disease Control and Prevention, 2012. As of January 30, 2014: http://www.cdc.gov/hrqol/mental_health_reports/mental_health_reports.html

Cerit, Cem, Arzu Filizer, Ümit Tural, and Ali Evren Tufan, "Stigma: A Core Factor on Predicting Functionality in Bipolar Disorder," *Comprehensive Psychiatry*, Vol. 53, No. 5, July 2012, pp. 484–489.

Cersovsky, Steven B., "Fighting the War Within: Suicide as an Individual and Public Health Challenge in the U.S. Army," *Psychiatry: Interpersonal and Biological Processes*, Vol. 74, No. 2, 2011, pp. 110–114.

CFETP 3S1X1—See Department of the Air Force, 2011.

CFETP 4C0X1—See Department of the Air Force, 2007.

CFETP 5R0X1C1—See Secretary of the Air Force, 2014.

Chamberlin, Judi, "A Working Definition of Empowerment," *Journal of Psychiatric Rehabilitation*, Vol. 20, No. 4, Spring 1997, pp. 43–46.

Chan, Jenny Y. N., Winnie W. S. Mak, and Lawrence S. C. Law, "Combining Education and Video-Based Contact to Reduce Stigma of Mental Illness: '*The Same or Not the Same*' Anti-Stigma Program for Secondary Schools in Hong Kong," *Social Science and Medicine*, Vol. 68, No. 8, April 2009, pp. 1521–1526.

Chief, Bureau of Medicine and Surgery, *Change 132 Manual of the Medical Department U.S. Navy, Naval Medical Publication 117*, November 17, 2008. As of June 10, 2014: http://www.med.navy.mil/directives/Documents/NAVMED%20P-117%20%28MANMED%29/ MANMED%20Change%20132.pdf

———, Change 140 Manual of the Medical Department U.S. Navy, Naval Medical Publication 117, May 3, 2012. As of June 10, 2014:

http://www.med.navy.mil/directives/Documents/NAVMED%20P-117%20%28MANMED%29/ Chapter%2015,%20Medical%20Examinations%20%28incorporates%20Changes%20128,%20 130,%20135-140%20below%29.pdf

Chief, National Guard Bureau, *Management of Training and Operational Support Within the Air National Guard*, Air National Guard Instruction 36-2001, October 19, 2009. As of February 24, 2014:

http://static.e-publishing.af.mil/production/1/ang/publication/angi36-2001/angi36-2001.pdf

———, "Compliance and Standardization Requirements List (C&SRL) Arming and Use of Force," Air National Guard Directory 90-266, January 26, 2010; certified current January 23, 2012. As of February 20, 2014:

http://static.e-publishing.af.mil/production/1/ang/publication/angdir90-266/angdir90-266.pdf

———, "Guidance on the Participation of Children with Special Needs in the National Guard Child and Youth Program (CYP)," Chief National Guard Bureau Instruction 1800.01, April 19, 2012a.

—_____, Enlistment and Reenlistment in the Air National Guard and as a Reserve of the Air Force, Air National Guard Instruction 36-2002, October 1, 2012b. As of February 24, 2014: http://static.e-publishing.af.mil/production/1/ang/publication/angi36-2002/angi36-2002.pdf

Christensen, Brandon N. and Joanne Yaffe, "Factors Affecting Mental Health Service Utilization Among Deployed Military Personnel," *Military Medicine*, Vol. 177, No. 3, March 2012, pp. 278–283.

Cifu, David X., and Cory Blake, "Re-Establishing Normalcy," in David X. Cifu and Cory Blake, eds., *Overcoming Post-Deployment Syndrome: A Six-Step Mission to Health*, New York: DemosHealth, 2011.

Clark-Hitt, Rose, Sandi W. Smith, and Jordan S. Broderick, "Help a Buddy Take a Knee: Creating Persuasive Messages for Military Service Members to Encourage Others to Seek Mental Health Help," *Health Communication*, Vol. 27, No. 5, 2012, pp. 429–438.

Clement, Sarah, O. Schauman, T. Graham, F. Maggioni, S. Evans-Lacko, N. Bezborodovs, C. Morgan, N. Rüsch, J. S. L. Brown, and G. Thornicroft, "What Is the Impact of Mental Health–Related Stigma on Help-Seeking? A Systematic Review of Quantitative and Qualitative Studies," *Psychological Medicine*, January 2014, pp. 1–17.

Clement, Sarah, and Graham Thornicroft, "SAPPHIRE Programme on Stigma and Discrimination in Mental Health: Research Findings," *Psychiatrische Praxis*, 2011, p. 38.

Clement, Sarah, Adrienne van Nieuwenhuizen, Aliya Kassam, Clare Flach, Anisha Lazarus, Melanie de Castro, Paul McCrone, Ian Norman, and Graham Thornicroft, "Filmed v. Live Social Contact Interventions to Reduce Stigma: Randomised Controlled Trial," *British Journal of Psychiatry*, Vol. 201, No. 1, 2012, pp. 57–64.

Clinton, Michael, "Collaborative Education and Social Stereotypes," *Australian and New Zealand Journal of Mental Health Nursing*, Vol. 8, No. 3, 1999, pp. 100–103.

CNGBI 1800.01-See Chief, National Guard Bureau, 2012a.

Code of Federal Regulations, Title 7, Agriculture, Subtitle B, Regulations of the Department of Agriculture, Chapter III, Animal and Plant Health Inspection Service, Department of Agriculture, Part 331, Possession, Use, and Transfer of Select Agents and Toxins. As of February 9, 2014: http://www.ecfr.gov/cgi-bin/text-idx?c=ecfr&tpl=/ecfrbrowse/Title07/7cfr331_main_02.tpl

———, Title 9, Animals and Animal Products. As of February 9, 2014: http://www.ecfr.gov/cgi-bin/text-idx?tpl=/ecfrbrowse/Title09/9tab_02.tpl

, Title 42, Public Health. As of February 9, 2014: http://www.ecfr.gov/cgi-bin/text-idx?c=ecfr;sid=9a0f708ce16687214b2fd14aa7e4dcea;rgn=div5;view=text;node=42%3A1.0.1.1.2;idno=42;cc=ecfr

Collins, Rebecca L., Eunice C. Wong, Jennifer L. Cerully, Dana Schultz, and Nicole K. Eberhart, Interventions to Reduce Mental Health Stigma and Discrimination: A Literature Review to Guide Evaluation of California's Mental Health Prevention and Early Intervention Initiative, Santa Monica, Calif.: RAND Corporation, TR-1318-CMHSA, 2012. As of January 30, 2014: http://www.rand.org/pubs/technical_reports/TR1318.html

COMDTINST 1730.4C—See Commandant, U.S. Coast Guard, 2012a.

COMDTINST 1734.1A—See Commandant, U.S. Coast Guard, 2009c.

COMDTINST 1740.7B—See Commandant, U.S. Coast Guard, 2011j.

COMDTINST 1750.7C-See Commandant, U.S. Coast Guard, 2006.

COMDTINST 1754.3A—See Commandant, U.S. Coast Guard, 2011e.

COMDTINST 1754.7A—See Commandant, U.S. Coast Guard, 1996.

COMDTINST 1755.1-See Commandant, U.S. Coast Guard, undated.

COMDTINST 5230.60B—See Commandant, U.S. Coast Guard, 2011d.

COMDTINST 6400.1B—See Commandant, U.S. Coast Guard, 2012c.

COMDTINST 12300.9-See Commandant, U.S. Coast Guard, 2005.

COMDTINST 12335.4A—See Commandant, U.S. Coast Guard, 2003.

COMDTINST 16798.2-See Commandant, U.S. Coast Guard, 1999.

COMDTINST M1000.4—See Commandant, U.S. Coast Guard, 2011i.

COMDTINST M1600.2—See Commandant, U.S. Coast Guard, 2011g.

COMDTINST M1700.1-See Commandant, U.S. Coast Guard, 2011h.

COMDTINST M1754.10D—See Commandant, U.S. Coast Guard, 2012d.

COMDTINST M3150.1C—See Commandant, U.S. Coast Guard, 2009a.

COMDTINST M5100.47—See Commandant, U.S. Coast Guard, 1990.

COMDTINST M5110.1B—See Commandant, U.S. Coast Guard, 2008b.

COMDTINST M5350.4C-See Commandant, U.S. Coast Guard, 2010.

COMDTINST M5400.17—See Commandant, U.S. Coast Guard, 2004.

COMDTINST M5810.1E—See Commandant, U.S. Coast Guard, 2011c.

COMDTINST M5830.1A—See Commandant, U.S. Coast Guard, 2007.

COMDTINST M6000.1E-See Commandant, U.S. Coast Guard, 2011a.

COMDTINST M6410.3A—See Commandant, U.S. Coast Guard, 2012b.

COMDTINST M12750.4—See Commandant, U.S. Coast Guard, 1987.

COMDTINST M12792.3A—See Commandant, U.S. Coast Guard, 2011b.

COMDTINST M16000.1A-See Commandant, U.S. Coast Guard, 2008a.

COMDTINST M16790.1G-See Commandant, U.S. Coast Guard, 2011f.

COMDTNOTE 12271-See Commandant, U.S. Coast Guard, 2001.

Commandant of the Marine Corps, *Conscientious Objectors*, Marine Corps Order 1306.16E, November 21, 1986. As of June 10, 2014: http://www.marines.mil/Portals/59/Publications/MCO%201306.16E.pdf

, "Handicapped Individuals Program (HIP)," Marine Corps Order 12713.8, April 25, 1990a.

, "Grievance Procedure," Marine Corps Order 12771.2, October 16, 1990b. As of February 19, 2014:

http://www.marines.mil/Portals/59/Publications/MCO%2012771.2.pdf

Montgomery GI Bill—*Reserve (MGIB-R)*, Marine Corps Order 1560R.30B, August 23, 1993a. As of February 19, 2014:

http://www.marines.mil/Portals/59/Publications/MCO%201560R.30B.pdf

——, "Family and Medical Leave," Marine Corps Order 12630.3, October 14, 1993b. As of February 19, 2014:

http://www.marines.mil/Portals/59/Publications/MCO%2012630.3.pdf

— Marine Corps Personnel Assignment Policy, Marine Corps Order P1300.8R, October 4, 1994. As of February 17, 2014:

http://www.marines.mil/Portals/59/Publications/MCO%20P1300.8R%20W%20CH%201-8.pdf

———, "Civilian Leadership Development," Marine Corps Order 12410.24, June 26, 1997. As of February 19, 2014:

http://www.marines.mil/Portals/59/Publications/MCO%2012410.24.pdf

———, Selection and Screening Criteria for Marines Assigned to Marine Helicopter Squadron One (HMX-1), Marine Corps Order 1326.7E, November 30, 2004.

—_____, Health Services Training and Readiness Manual, Marine Corps Order P3500.75, May 6, 2005a. As of February 19, 2014:

http://www.marines.mil/Portals/59/Publications/MCO%20P3500.75.pdf

——, *Policies and Procedures for Range and Training Area (RTA) Management*, Marine Corps Order P3550.10 Ch 1, July 7, 2005b.

——, *Religious Ministry Team (RMT) Training and Readiness Manual*, Marine Corps Order P3500.44A, July 20, 2005c. As of February 19, 2014: http://www.marines.mil/Portals/59/Publications/MCO%20P3500.44A.pdf

—_____, Marine Corps Promotion Manual, Volume 2, Enlisted Promotions, Marine Corps Order P1400.32D, May 11, 2006. As of February 17, 2014: http://www.newriver.marines.mil/Portals/17/Documents/MCO%20P1400.32D.pdf

_____, Law Enforcement Manual, Marine Corps Order 5580.2B, August 27, 2008.

—_____, Marine Corps Reserve Administrative Management Manual, Marine Corps Order 1001R.1K, March 22, 2009a. As of February 17, 2014: http://www.marines.mil/Portals/59/Publications/MCO%201001R.1K.pdf

—____, Marine Corps Parachuting Policy and Program Administration, Marine Corps Order 3120.11, May 4, 2009b.

—_____, Marine Corps Diving Policy and Program Administration, Marine Corps Order 3150.4, May 4, 2009c. As of February 17, 2014: http://www.marines.mil/Portals/59/Publications/MCO%203150.4.pdf

——, U.S. Marine Corps Total Force Mobilization, Activation, Integration, and Deactivation Plan, Marine Corps Order 3000.19A, August 29, 2010a. As of February 19, 2014: http://www.marines.mil/Portals/59/Publications/MCO%203000%2019A_1%20(2).pdf

—_____, Enlisted Retention and Career Development Program, Marine Corps Order 1040.31, September 8, 2010b. As of February 17, 2014: http://www.marines.mil/Portals/59/Publications/MCO%201040.31.pdf

—_____, Performance Evaluation System, Marine Corps Order P1610.7F Ch 2, November 19, 2010c. As of February 19, 2014:

http://www.hqmc.marines.mil/Portals/133/Docs/MCO%20P1610_7F%20W%20CH%201-2.pdf

—_____, Management of Marine Corps Formal Schools and Training Detachments, Marine Corps Order 1553.2B, April 1, 2011a. As of February 17, 2014: http://www.t3s.marines.mil/Portals/64/MCO_1553.2B.pdf

—____, Marine Corps Substance Abuse Program, Marine Corps Order 5300.17, April 11, 2011b. As of February 17, 2014:

http://www.med.navy.mil/sites/nmcphc/Documents/policy-and-instruction/mco-530017.pdf

— Marine Corps Safety Program, Marine Corps Order 5100.29B, July 28, 2011c. As of February 17, 2014:

http://www.marines.mil/Portals/59/Publications/MCO%205100.29B.pdf

———, Security Augmentation Force Program, Marine Corps Order 5530.16, August 26, 2011d. As of February 17, 2014:

http://www.marines.mil/Portals/59/Publications/MCO%205530.16.pdf

———, Security Cooperation (SC) Training and Readiness (T&R) Manual, Navy and Marine Corps 3500.59A, Part 1, October 3, 2011e. As of February 19, 2014: http://www.marines.mil/Portals/59/Publications/NAVMC%203500.59A_1.pdf

———, Security Cooperation (SC) Training and Readiness (T&R) Manual, Navy and Marine Corps 3500.59A, Part 2, October 3, 2011f. As of February 19, 2014: http://www.marines.mil/Portals/59/Publications/NAVMC%203500.59A_2.pdf

—_____, Marine Corps Nonappropriated Fund Personnel Policy Manual, Marine Corps Order P12000.11A Ch 4, December 27, 2011g. As of February 19, 2014: http://www.marines.mil/Portals/59/Publications/MCO%20P12000.11A%20W%20CH1-4.pdf

—_____, Marine Corps Family Advocacy and General Counseling Program, Marine Corps Order 1754.11, March 26, 2012a. As of February 17, 2014: http://www.marines.mil/Portals/59/Publications/MCO%201754_11.pdf

— Marine Corps Suicide Prevention Program (MCSPP), Marine Corps Order 1720.2, April 10, 2012b. As of February 17, 2014: http://www.marines.mil/Portals/59/Publications/MCO%201720.2.pdf

———, Marine Corps Common Skills (MCCS) Training and Readiness (T&R) Manual, Navy and Marine Corps 3500.18B, preliminary, Part 1, May 10, 2012c. As of February 19, 2014: http://www.marines.mil/Portals/59/Publications/NAVMC%203500_18B%20PT%201.pdf

, "Religious Ministry in the Marine Corps," Marine Corps Order 1730.6E, June 4, 2012d. As of February 17, 2014:

http://www.marines.mil/Portals/59/Publications/MCO%201730_6E.pdf

, Military Occupational Specialties Manual, Marine Corps Order 1200.17D, June 5, 2012e.

_____, Licensing Procedures for Ordnance Vehicle Operators, Marine Corps Order 8400.6 CH 3, December 17, 2012f. As of February 17, 2014:

http://www.marines.mil/Portals/59/Publications/MCO%208400.6%20W%20CH%201-3.pdf

Commandant, U.S. Coast Guard, *DoD/DoD Dependent Schools (DoDDS) Directives Pertaining to Minor Dependents' Schooling in Overseas Areas*, Commandant Instruction 1755.1, undated. As of February 24, 2014:

http://www.uscg.mil/directives/ci/1000-1999/CI_1755_1.pdf

——, Civilian Personnel Actions: Discipline, Performance, Adverse Actions, Appeals, and Grievances, Commandant Instruction M12750.4, December 15, 1987. As of February 24, 2014: http://www.uscg.mil/directives/cim/12000-12999/CIM_12750_4.pdf

—____, Safety and Environmental Health Manual, Commandant Instruction M5100.47, April 1990. As of February 24, 2014:

http://www.uscg.mil/directives/cim/5000-5999/CIM_5100_47.pdf

——, "Coast Guard Special Needs Program," Commandant Instruction 1754.7A, January 30, 1996. As of February 24, 2014: http://www.uscg.mil/baseAlameda/WLS/CI_1754_7A.pdf

———, Auxiliary Air Crew Qualification Program, Commandant Instruction 16798.2, August 16, 1999. As of February 24, 2014: http://www.uscg.mil/auxiliary/publications/comdtinst/M16798.2.pdf

—____, Ch-1 to Coast Guard Nonappropriated Fund (NAF) Personnel Manual, COMDINST M12271.1A, Commandant Notice 12271, March 30, 2001.

———, "Merit Promotion Program and National Merit Promotion Plan," Commandant Instruction 12335.4A, January 24, 2003. As of February 24, 2014: http://www.uscg.mil/directives/ci/12000-12999/CI_12335_4A.pdf

——, Port Security Unit (PSU) Organization Manual (ORGMAN), Commandant Instruction M5400.17, September 3, 2004. As of February 24, 2014: http://www.uscg.mil/directives/cim/5000-5999/CIM_5400_17.pdf

——, "Career Entry-Level Opportunity (CEO) Program," Commandant Instruction 12300.9, May 19, 2005. As of February 24, 2014: http://www.uscg.mil/directives/ci/12000-12999/CI 12300 9.pdf

—_____, Coast Guard Family Advocacy Program, Commandant Instruction 1750.7C, June 27, 2006. As of February 24, 2014: http://www.uscg.mil/directives/ci/1000-1999/CI_1750_7C.pdf

—_____, Administrative Investigations Manual, Commandant Instruction Manual 5830.1A, September 7, 2007a. As of February 24, 2014: http://www.uscg.mil/directives/cim/5000-5999/CIM_5830_1A.pdf

——, "Coast Guard Operations Manual," Commandant Instruction Manual 3710.1F, October 22, 2007b.

———, USCG Marine Safety Manual, Vol. V: Investigations and Enforcement, Commandant Instruction Manual 16000.10A, April 2008a. As of February 24, 2014: http://www.uscg.mil/directives/cim/16000-16999/CIM_16000_10A.pdf

——, U.S. Coast Guard Postal Manual, Commandant Instruction Manual 5110.1B, June 2008b. As of February 24, 2014: http://www.uscg.mil/directives/cim/5000-5999/CIM_5110_1B.pdf

———, *Coast Guard Diving Policies and Procedures Manual*, Vol. I, Commandant Instruction Manual 3150.1C, June 2009a. As of February 24, 2014: http://www.uscg.mil/directives/cim/3000-3999/CIM_3150_1C.pdf

, "Change 3 to Coast Guard Nonappropriated Fund Personnel Manual," Commandant Instruction Manual 12271.1A, June 4, 2009.

———, Suicide Prevention Program, Commandant Instruction 1734.1A, December 7, 2009c. As of February 24, 2014:

http://www.uscg.mil/directives/ci/1000-1999/CI_1734_1A.pdf

—_____, Coast Guard Civil Rights Manual, Commandant Instruction M5350.4C, May 2010. As of February 24, 2014:

http://www.uscg.mil/directives/cim/5000-5999/CIM_5350_4c.pdf

—, Coast Guard Medical Manual, Commandant Instruction M6000.1E, April 29, 2011a.

———, Civilian Employee Health Care and Occupational Health Program, Commandant Instruction M12792.3A, May 2011b. As of February 24, 2014: http://www.uscg.mil/directives/cim/12000-12999/CIM_12792_3A.pdf

—____, Military Justice Manual, Commandant Instruction M5810.1E, May 2011c. As of February 24, 2014:

http://www.uscg.mil/directives/cim/5000-5999/CIM_5810_1e.pdf

———, *Coast Guard Implementation of the Rehabilitation Act, Section 508*, Commandant Instruction 5230.60B, June 21, 2011d. As of February 24, 2014: http://www.uscg.mil/directives/ci/5000-5999/CI_5230_60B.pdf

———, Critical Incident Stress Management (CISM), Commandant Instruction 1754.3A, July 18, 2011e. As of February 24, 2014: http://www.uscg.mil/directives/ci/1000-1999/CI_1754_3A.pdf

——, Auxiliary Manual, Commandant Instruction M16790.1G, August 17, 2011f. As of February 24, 2014:

http://www.uscg.mil/directives/cim/16000-16999/CIM_16790_1G.pdf

—____, Discipline and Conduct, Commandant Instruction M1600.2, September 2011g. As of February 24, 2014:

http://www.uscg.mil/directives/cim/1000-1999/CIM_1600_2.pdf

—______, Military Civil and Dependent Affairs, Commandant Instruction M1700.1, September 2011h. As of February 24, 2014: http://www.uscg.mil/directives/cim/1000-1999/CIM_1700_1.pdf

—____, Military Separations, Commandant Instruction M1000.4, September 2011i. As of February 24, 2014: http://www.uscg.mil/directives/cim/1000-1999/CIM_1000_4.pdf

, "Employee Assistance Program (EAP)," Commandant Instruction 1740.7B, September 2, 2011j. As of February 24, 2014:

http://www.uscg.mil/directives/ci/1000-1999/CI_1740_7B.pdf

——, "Officer Accessions, Evaluations, and Promotions," Commandant Instruction 1000.3, September 29, 2011k.

——, "Military Assignments and Authorized Absences," Commandant Instruction 1000.8, September 29, 2011l.

—_______, Religious Ministries Within the Coast Guard, Commandant Instruction 1730.4C, March 9, 2012a. As of February 24, 2014: http://www.uscg.mil/directives/ci/1000-1999/CI_1730_4C.pdf

_____, *Coast Guard Aviation Medicine Manual*, Commandant Instruction M6410.3A, April 4,

2012b. As of February 24, 2014: http://www.uscg.mil/directives/cim/6000-6999/CIM_6410_3A.pdf

——, Student Externship Programs (SEP), Commandant Instruction 6400.1B, April 4, 2012c. As of February 24, 2014:

http://www.uscg.mil/directives/ci/6000-6999/CI_6400_1B.pdf

——, Sexual Assault Prevention and Response (SAPR) Program, Commandant Instruction M1754.10D, April 19, 2012d. As of February 24, 2014: http://www.uscg.mil/directives/cim/1000-1999/CIM_1754_10D.pdf

Commander, Air Education and Training Command, *Formal Flying Training Administration and Management*, Vol. 2: *Airmanship and Airmanship Support Training*, Air Education and Training Command Instruction 36-2205, February 28, 2012a. As of February 20, 2014: http://static.e-publishing.af.mil/production/1/aetc/publication/aetci36-2205v2/aetc36-2205v2.pdf

——, Formal Flying Training Administration and Management, Vol. 12: Aviation Leadership Program—Introductory Flight Training (ALP-IFT), Air Education and Training Command Instruction 36-2205, August 8, 2012b. As of February 20, 2014: http://static.e-publishing.af.mil/production/1/aetc/publication/aetci36-2205v12/aetci36-2205v12.pdf

———, Formal Flying Training Administration and Management, Vol. 3: *Initial Flight Screening (IFS)*, Air Education and Training Command Instruction 36-2205, December 20, 2012c. As of February 20, 2014:

http://static.e-publishing.af.mil/production/1/aetc/publication/aetci36-2205v3/aetci36-2205v3.pdf

—______, Formal Flying Training Administration and Management, Vol. 6: *Fighter and Introduction to Fighter Fundamentals (IFF)*, Air Education and Training Command Instruction 36-2205, December 21, 2012d. As of February 20, 2014:

http://static.e-publishing.af.mil/production/1/aetc/publication/aetci36-2205v6/aetci36-2205v6.pdf

—_____, Formal Flying Training Administration and Management, Vol. 8: *Special Operations/Personnel Recovery*, Air Education and Training Command Instruction 36-2205, April 30, 2013a; certified current on December 5, 2013. As of February 20, 2014:

http://static.e-publishing.af.mil/production/1/aetc/publication/aetci36-2205v8/aetci36-2205v8.pdf

—_____, Formal Flying Training Administration and Management, Vol. 5: *Combat Systems Officer (CSO)*, Air Education and Training Command Instruction 36-2205, September 4, 2013b. As of February 20, 2014:

http://static.e-publishing.af.mil/production/1/aetc/publication/aetci36-2205v5/aetci36-2205v5.pdf

Commander, Air Force Reserve Command, *Air Force Reserve Recruiting Procedures*, Air Force Reserve Command Instruction 36-2001, June 1, 2009; incorporating change 1, October 19, 2011. As of February 20, 2014:

http://static.e-publishing.af.mil/production/1/afrc/publication/afrci36-2001/afrci36-2001.pdf

Commander, Fifth Air Force, *Personnel Administration of Master Labor Contract (MLC) and Indirect Hire Agreement (IHA) Employees (PA)*, Fifth Air Force Instruction 36-102, April 16, 2012; certified current August 3, 2012. As of February 20, 2014:

http://static.e-publishing.af.mil/production/1/5_af/publication/5afi36-102/5afi36-102.pdf

Commander, Pacific Air Forces, *Security Forces Mission Performance Checklist: PACAF Corrections Program*, Pacific Air Forces Directory 90-254, June 29, 2005; certified current March 16, 2009. As of February 21, 2014:

http://static.e-publishing.af.mil/production/1/pacaf/publication/pacafdir90-254/pacafdir90-254.pdf

Compton, Michael T., Michelle L. Esterberg, Robin McGee, Raymond J. Kotwicki, and Janet R. Oliva, "Brief Reports: Crisis Intervention Team Training—Changes in Knowledge, Attitudes, and Stigma Related to Schizophrenia," *Psychiatric Services*, Vol. 57, No. 8, 2006, pp. 1199–1202.

Congressional Budget Office, *Educational Attainment and Compensation of Enlisted Personnel*, February 2004. As of February 6, 2014:

http://www.cbo.gov/sites/default/files/cbofiles/ftpdocs/51xx/doc5108/02-23-enlistedpersonnel.pdf

Cook, Trevor M., and JianLi Wang, "Descriptive Epidemiology of Stigma Against Depression in a General Population Sample in Alberta," *BMC Psychiatry*, Vol. 10, April 2010.

Cooper-Patrick, Lisa, Neil R. Powe, Mollie W. Jenckes, Junius J. Gonzales, David M. Levine, and Daniel E. Ford, "Identification of Patient Attitudes and Preferences Regarding Treatment of Depression," *Journal of General Internal Medicine*, Vol. 12, No. 7, 1997, pp. 431–438.

Corker, E., S. Hamilton, C. Henderson, C. Weeks, V. Pinfold, D. Rose, P. Williams, C. Flach, V. Gill, E. Lewis-Holmes, and G. Thornicroft, "Experiences of Discrimination Among People Using Mental Health Services in England 2008–2011," *British Journal of Psychiatry*, Vol. 202, 2013, pp. s58–s63.

Corrigan, Patrick W., "Target-Specific Stigma Change: A Strategy for Impacting Mental Illness Stigma," *Psychiatric Rehabilitation Journal*, Vol. 28, No. 2, 2004a, pp. 113–121. As of July 25, 2014: http://www.ncbi.nlm.nih.gov/pubmed/15605746

——, "How Stigma Interferes with Mental Health Care," *American Psychologist*, Vol. 59, No. 7, October 2004b, pp. 614–625.

———, On the Stigma of Mental Illness: Practical Strategies for Research and Social Change, Washington, D.C.: American Psychological Association, 2005.

———, "Best Practices: Strategic Stigma Change (SSC)—Five Principles for Social Marketing Campaigns to Reduce Stigma," *Psychiatric Services*, Vol. 62, No. 8, August 1, 2011, pp. 824–826.

, "Research and the Elimination of the Stigma of Mental Illness," *British Journal of Psychiatry*, Vol. 201, No. 1, 2012a, pp. 7–8.

, "Where Is the Evidence Supporting Public Service Announcements to Eliminate Mental Illness Stigma?" *Psychiatric Services*, Vol. 63, No. 1, January 2012b, pp. 79–82.

Corrigan, Patrick W., and Betsy Gelb, "Three Programs That Use Mass Approaches to Challenge the Stigma of Mental Illness," *Psychiatric Services*, Vol. 57, No. 3, March 1, 2006, pp. 393–398.

Corrigan, Patrick W., Amy Kerr, and Lissa Knudsen, "The Stigma of Mental Illness: Explanatory Models and Methods for Change," *Applied and Preventive Psychology*, Vol. 11, No. 3, September 2005, pp. 179–190.

Corrigan, Patrick W., Jonathon Larson, Molly Sells, Nathaniel Niessen, and Amy C. Watson, "Will Filmed Presentations of Education and Contact Diminish Mental Illness Stigma?" *Community Mental Health Journal*, Vol. 43, No. 2, April 2007, pp. 171–181.

Corrigan, Patrick W., Fred E. Markowitz, and Amy C. Watson, "Structural Levels of Mental Illness Stigma and Discrimination," *Schizophrenia Bulletin*, Vol. 30, No. 3, 2004, pp. 481–491.

Corrigan, Patrick W., Fred E. Markowitz, Amy Watson, David Rowan, and Mary Ann Kubiak, "An Attribution Model of Public Discrimination Towards Persons with Mental Illness," *Journal of Health and Social Behavior*, Vol. 44, No. 2, June 2003, pp. 162–179.

Corrigan, Patrick W., Scott B. Morris, Patrick J. Michaels, Jennifer D. Rafacz, and Nicolas Rüsch, "Challenging the Public Stigma of Mental Illness: A Meta-Analysis of Outcome Studies," *Psychiatric Services*, Vol. 63, No. 10, October 2012, pp. 963–973.

Corrigan, Patrick W., and John R. O'Shaughnessy, "Changing Mental Illness Stigma as It Exists in the Real World," *Australian Psychologist*, Vol. 42, No. 2, June 2007, pp. 90–97.

Corrigan, Patrick W., and David L. Penn, "Lessons from Social Psychology on Discrediting Psychiatric Stigma," *American Psychologist*, Vol. 54, No. 9, September 1999, pp. 765–776.

Corrigan, Patrick W., Karina J. Powell, and Nicolas Rüsch, "How Does Stigma Affect Work in People with Serious Mental Illnesses?" *Psychiatric Rehabilitation Journal*, Vol. 35, No. 5, September 2012, pp. 381–384.

Corrigan, Patrick W., Jennifer D. Rafacz, Julie Hautamaki, Jessica Walton, Nicolas Rüsch, Deepa Rao, Patricia Doyle, Sarah O'Brien, John Pryor, and Glenn Reeder, "Changing Stigmatizing Perceptions and Recollections About Mental Illness: The Effects of NAMI's In Our Own Voice," *Community Mental Health Journal*, Vol. 46, No. 5, October 2010, pp. 517–522.

Corrigan, Patrick W., Jennifer Rafacz, and Nicolas Rüsch, "Examining a Progressive Model of Self-Stigma and Its Impact on People with Serious Mental Illness," *Psychiatry Research*, Vol. 189, No. 3, October 30, 2011, pp. 339–343.

Corrigan, Patrick W., and Deepa Rao, "On the Self-Stigma of Mental Illness: Stages, Disclosure, and Strategies for Change," *Canadian Journal of Psychiatry*, Vol. 57, No. 8, August 2012, pp. 464–469.

Corrigan, Patrick W., L. Philip River, Robert K. Lundin, David L. Penn, Kyle Uphoff-Wasowski, John Campion, James Mathisen, Christine Gagnon, Maria Bergman, Hillel Goldstein, and Mary Anne Kubiak, "Three Strategies for Changing Attributions About Severe Mental Illness," *Schizophrenia Bulletin*, Vol. 27, No. 2, 2001, pp. 187–195.

Corrigan, Patrick W., David Roe, and Hector W. H. Tsang, *Challenging the Stigma of Mental Illness: Lessons for Therapists and Advocates*, Chichester, West Sussex, UK: John Wiley and Sons, 2011.

Corrigan, Patrick W., David Rowan, Amy Green, Robert Lundin, Philip River, Kyle Uphoff-Wasowski, Kurt White, and Mary Anne Kubiak, "Challenging Two Mental Illness Stigmas: Personal Responsibility and Dangerousness," *Schizophrenia Bulletin*, Vol. 28, No. 2, 2002, pp. 293–309.

Corrigan, Patrick W., and Nicolas Rüsch, "Mental Illness Stereotypes and Clinical Care: Do People Avoid Treatment Because of Stigma?" *Psychiatric Rehabilitation Skills*, Vol. 6, No. 3, 2002, pp. 312–334.

Corrigan, Patrick W., and Jenessa R. Shapiro, "Measuring the Impact of Programs That Challenge the Public Stigma of Mental Illness," *Clinical Psychology Review*, Vol. 30, No. 8, December 2010, pp. 907–922.

Corrigan, Patrick W., V. Thompson, D. Lambert, Y. Sangster, J. G. Noel, and J. Campbell, "Perceptions of Discrimination Among Persons with Serious Mental Illness," *Psychiatric Services*, Vol. 54, No. 8, August 2003, pp. 1105–1110.

Corrigan, Patrick W., and A. Wassel, "Understanding and Influencing the Stigma of Mental Illness," *Journal of Psychosocial Nursing and Mental Health Services*, Vol. 46, No. 1, January 2008, pp. 42–48.

Corrigan, Patrick W., and Amy C. Watson, "Understanding the Impact of Stigma on People with Mental Illness," *World Psychiatry*, Vol. 1, No. 1, February 2002, pp. 16–20.

Corrigan, Patrick W., Amy C. Watson, and Victor Ottati, "From Whence Comes Mental Illness Stigma?" *International Journal of Social Psychiatry*, Vol. 49, No. 2, 2003, pp. 142–157.

Costello, E. J., D. L. Foley, and A. Angold, "10-Year Research Update Review: The Epidemiology of Child and Adolescent Psychiatric Disorders: II. Developmental Epidemiology," *Journal of the American Academy of Child and Adolescent Psychiatry*, Vol. 45, No. 1, January 2006, pp. 8–25.

Creel, A. H., R. N. Rimal, G. Mkandawire, K. Böse, and J. W. Brown, "Effects of a Mass Media Intervention on HIV-Related Stigma: 'Radio Diaries' Program in Malawi," *Health Education Research*, Vol. 26, No. 3, June 2011, pp. 456–465.

Crisp, A., M. Gelder, E. Goddard, and H. Meltzer, "Stigmatization of People with Mental Illnesses: A Follow-Up Study Within the Changing Minds Campaign of the Royal College of Psychiatrists," *World Psychiatry*, Vol. 4, No. 2, June 2005, pp. 106–113.

Crocker, Jennifer, Brenda Major, and Claude Steele, "Social Stigma," in Daniel Todd Gilbert, Susan T. Fiske, and Gardner Lindzey, eds., *The Handbook of Social Psychology*, Boston, Mass.: McGraw-Hill, 1998, pp. 504–553.

Crowley, Kathleen, *The Power of Procovery in Healing Mental Illness: Just Start Anywhere*, San Francisco, Calif.: Kennedy Carlisle, 2000.

Dalack, G. W., A. J. Blow, M. Valenstein, L. Gorman, J. Spinner, S. Marcus, M. Kees, S. McDonough, J. F. Greden, B. Ames, B. Francisco, J. R. Anderson, J. Bartolacci, and R. Lagrou, "Public-Academic Partnerships: Working Together to Meet the Needs of Army National Guard Soldiers: An Academic-Military Partnership," *Psychiatric Services*, Vol. 61, No. 11, November 2010, pp. 1069–1071.

Dalky, Heyam F., "Mental Illness Stigma Reduction Interventions: Review of Intervention Trials," *Western Journal of Nursing Research*, Vol. 34, No. 4, June 2012, pp. 520–547.

Davis, James Allan, and Tom W. Smith, General Social Survey, Chicago, Ill.: National Opinion Research Center, 2006.

DCoE—See Defense Centers of Excellence for Psychological Health and Traumatic Brain Injury.

Dear, Michael, and Glenda Laws, "Anatomy of a Decision: Recent Land Use Zoning Appeals and Their Effect on Group Home Locations in Ontario," *Canadian Journal of Community Mental Health*, Vol. 5, No. 1, 1986, pp. 5–17.

Defense Centers of Excellence for Psychological Health and Traumatic Brain Injury, "Behavioral Health Stigma and Access to Care," unpublished, 2012.

Delaney, Kathleen R., "Psychiatric Mental Health Nurses: Stigma Issues We Fail to See," Archives of Psychiatric Nursing, Vol. 26, No. 4, August 2012, pp. 333–335.

DeMartino, Robert E., Alexander E. Crosby, Marlene EchoHawka, David A. Litts, Jane Pearson, Gerald A. Reed, and Margaret West, "A Call to Collaboration: The Federal Commitment to Suicide Prevention," *Suicide and Life-Threatening Behavior*, Vol. 33, No. 2, Summer 2003, pp. 101–110.

Department of the Air Force, *AFSC 4C0X1 Mental Health Service Specialty*, Career Field Education and Training Plan 4C0X1, Parts I and II, January 1, 2007. As of June 10, 2014: http://static.e-publishing.af.mil/production/1/af_sg/publication/cfetp4c0x1/cfetp4c0x1.pdf

AFSC 3S1X1 Equal Opportunity, Career Field Education and Training Plan 3S1X1, Parts I and II, March 15, 2011. As of June 10, 2014: http://static.e-publishing.af.mil/production/1/af_a1/publication/cfetp3s1x1/cfetp_3s1x1.pdf

——, *Professional Development Guide, 40th Anniversary Edition*, Air Force Pamphlet 36-2241, October 1, 2013. As of February 21, 2014: http://static.e-publishing.af.mil/production/1/af_a1/publication/afpam36-2241/afpam36-2241.pdf

Department of the Army and Department of the Air Force, *Army National Guard Safety Program*, National Guard Regulation 385-10, September 12, 2008. As of February 24, 2014: http://www.ngbpdc.ngb.army.mil/pubs/385/ngr385_10.pdf

———, Enlisted Personnel Management, National Guard Regulation 600-200, July 31, 2009a. As of February 24, 2014:

http://www.ngbpdc.ngb.army.mil/pubs/600/ngr600_200.pdf

— National Guard CBRNE Enhanced Response Force Package Management, National Guard Regulation 500-4, Air National Guard Instruction 10-2504, October 16, 2009b.

— Inactive Army National Guard, National Guard Regulation 614-1, March 18, 2010. As of February 24, 2014:

http://www.ngbpdc.ngb.army.mil/pubs/614/ngr614_1.pdf

Department of Defense Task Force on Mental Health, *An Achievable Vision: Report of the Department of Defense Task Force on Mental Health*, Falls Church, Va.: Defense Health Board, June 2007. As of January 30, 2014:

http://www.health.mil/dhb/downloads/MHTF-Report-Final.pdf

Department of Defense Task Force on the Prevention of Suicide by Members of the Armed Forces, *The Challenge and the Promise: Strengthening the Force, Preventing Suicide and Saving Lives—Final Report of the Department of Defense Task Force on the Prevention of Suicide by Members of the Armed Forces*, Falls Church, Va.: Defense Health Board, August 2010.

Devine, Patricia G., "Stereotypes and Prejudice: Their Automatic and Controlled Components," *Journal of Personality and Social Psychology*, Vol. 56, No. 1, January 1989, pp. 5–18.

Devine, Patricia G., and Lindsay B. Sharp, "Automaticity and Control in Stereotyping and Prejudice," in Todd D. Nelson, ed., *Handbook of Prejudice, Stereotyping, and Discrimination*, New York: Psychology Press, 2009, pp. 61–87.

Diala, C., C. Muntaner, C. Walrath, K. J. Nickerson, T. A. LaVeist, and P. J. Leaf, "Racial Differences in Attitudes Toward Professional Mental Health Care and in the Use of Services," *American Journal of Orthopsychiatry*, Vol. 70, No. 4, October 2000, pp. 455–464.

Dickstein, Benjamin D., Dawne S. Vogt, Sonia Handa, and Brett T. Litz, "Targeting Self-Stigma in Returning Military Personnel and Veterans: A Review of Intervention Strategies," *Military Psychology*, Vol. 22, No. 2, April 2010, pp. 224–236.

Dietrich, Sandra, Roland Mergl, Philine Freudenberg, David Althaus, and Ulrich Hegerl, "Impact of a Campaign on the Public's Attitudes Towards Depression," *Health Education Research*, Vol. 25, No. 1, 2010, pp. 135–150.

Dimidjian, Sona, Steven D. Hollon, Keith S. Dobson, Karen B. Schmaling, Robert J. Kohlenberg, Michael E. Addis, Robert Gallop, Joseph B. McGlinchey, David K. Markley, Jackie K. Gollan, David C. Atkins, David L. Dunner, and Neil S. Jacobson, "Randomized Trial of Behavioral Activation, Cognitive Therapy, and Antidepressant Medication in the Acute Treatment of Adults with Major Depression," *Journal of Consulting and Clinical Psychology*, Vol. 74, No. 4, August 2006, pp. 658–670.

Dingfelder, Sadie F., "The Military's War on Stigma," *Monitor on Psychology*, Vol. 40, No. 6, June 2009, p. 52.

Director of Administration and Management, Office of the Secretary of Defense, *Disciplinary and Adverse Actions*, Administrative Instruction 8, May 7, 2008. As of February 7, 2014: http://www.dtic.mil/whs/directives/corres/pdf/a008p.pdf

DoD 5200.2-R-See Office of the Deputy Under Secretary of Defense for Policy, 1987 (1996).

DoD 5210.42-R—See Office of the Under Secretary of Defense for Acquisition, Technology, and Logistics, 2006 (2009).

DoD 5400.11-R-See Office of the Director for Administration and Management, 2007.

DoD 6025.13-R-See Assistant Secretary of Defense for Health Affairs, 2004.

DoD 6025.18-R-See Assistant Secretary of Defense for Health Affairs, 2003a.

DoD 6055.05-M—See Under Secretary of Defense for Acquisition, Technology, and Logistics, 2007 (2008).

DoDD 1020.1—See Assistant Secretary of Defense for Manpower, Reserve Affairs, and Logistics, 1982 (2003).

DoDD 1350.2—See Under Secretary of Defense for Personnel and Readiness, 1995 (2003).

DoDD 1440.1-See Assistant Secretary of Defense for Force Management Policy, 1987 (2003).

DoDD 5200.2—See Assistant Secretary of Defense for Command, Control, Communications, and Intelligence, 1999.

DoDD 5220.6-See General Counsel, 1992 (1999).

DoDD 6200.04—See Under Secretary of Defense for Personnel and Readiness, 2004 (2007).

DoDI 1010.6-See Assistant Secretary of Defense for Health Affairs, 1985.

DoDI 1300.18-See Under Secretary of Defense for Personnel and Readiness, 2008 (2009).

DoDI 1342.12—See Under Secretary of Defense for Personnel and Readiness, 2005a.

DoDI 1400.25-V810—See Under Secretary of Defense for Personnel and Readiness, 2005b (2009).

DoDI 2040.2—See Under Secretary of Defense for Policy, 2008.

DoDI 2310.4—See Assistant Secretary of Defense for International Security Affairs, 2000. DoDI 3020.41—See Under Secretary of Defense for Acquisition, Technology, and Logistics, 2011. DoDI 3025.19-See Under Secretary of Defense for Policy, 2011. DoDI 5154.30-See Assistant Secretary of Defense for Health Affairs, 2003b. DoDI 5210.42—See Under Secretary of Defense for Acquisition, Technology, and Logistics, 2012. DoDI 5210.65-See Under Secretary of Defense for Intelligence, 2007. DoDI 5210.87—See Executive Secretary, 1998. DoDI 5210.89-See Under Secretary of Defense for Intelligence, 2006. DoDI 5525.14-See Under Secretary of Defense for Personnel and Readiness, 2011a. DoDI 6000.14—See Under Secretary of Defense for Personnel and Readiness, 2011d (2013). DoDI 6040.45-See Under Secretary of Defense for Personnel and Readiness, 2010d. DoDI 6060.2-See Assistant Secretary of Defense for Force Management Policy, 1993 (1998). DoDI 6130.03-See Under Secretary of Defense for Personnel and Readiness, 2010c (2011). DoDI 6200.03—See Under Secretary of Defense for Personnel and Readiness, 2010b (2013). DoDI 6400.06-See Under Secretary of Defense for Personnel and Readiness, 2007 (2011). DoDI 6490.03-See Under Secretary of Defense for Personnel and Readiness, 2006. DoDI 6490.04—See Under Secretary of Defense for Personnel and Readiness, 2013a. DoDI 6490.05—See Under Secretary of Defense for Personnel and Readiness, 2011e (2013).

- DoDI 6490.06—See Under Secretary of Defense for Personnel and Readiness, 2009 (2011).
- DoDI 6490.07-See Under Secretary of Defense for Personnel and Readiness, 2010a.
- DoDI 6490.08—See Under Secretary of Defense for Personnel and Readiness, 2011c.

DoDI 6490.09-See Under Secretary of Defense for Personnel and Readiness, 2012a (2013).

DoDI 6490.10-See Under Secretary of Defense for Personnel and Readiness, 2012b.

DoDI 6495.02-See Under Secretary of Defense for Personnel and Readiness, 2013c.

DoDM 5105.21-V1—See Under Secretary of Defense for Intelligence, 2012a.

DoDM 5105.21-V3-See Under Secretary of Defense for Intelligence, 2012b.

DoDM 7730.47-M-V1—See Under Secretary of Defense for Personnel and Readiness, 2010e.

"DoD Recognizes Stigma and Changes Security Clearance Question on Mental Health," *Military Medicine*, July 2008, p. 2.

DoD Task Force on Mental Health-See Department of Defense Task Force on Mental Health.

DoD Task Force on the Prevention of Suicide by Members of the Armed Forces—*See* Department of Defense Task Force on the Prevention of Suicide by Members of the Armed Forces.

Downs, Marilyn F., "Changing the Meaning of Help: Clinical Approaches to Reducing Stigma Among Suicidal Young Adults," *Families in Society*, Vol. 93, No. 1, January–March 2012, pp. 22–28.

Downs, Marilyn F., and Daniel Eisenberg, "Help Seeking and Treatment Use Among Suicidal College Students," *Journal of American College Health*, Vol. 60, No. 2, 2012, pp. 104–114.

Doyle, Kerry Esme, and Mary Cruickshank, "Stereotyping Stigma: Undergraduate Health Students' Perceptions at Handover," *Journal of Nursing Education*, Vol. 51, No. 5, May 2012, pp. 255–261.

Druss, Benjamin G., and Robert A. Rosenheck, "Mental Disorders and Access to Medical Care in the United States," *American Journal of Psychiatry*, Vol. 155, No. 12, December 1, 1998, pp. 1775–1777.

DTM-11-011—See Under Secretary of Defense for Personnel and Readiness, 2011b.

DTM-11-015-See Under Secretary of Defense for Personnel and Readiness, 2011f.

Eaton, K. M., C. W. Hoge, S. C. Messer, A. A. Whitt, O. A. Cabrera, D. McGurk, A. Cox, and C. A. Castro, "Prevalence of Mental Health Problems, Treatment Need, and Barriers to Care Among Primary Care-Seeking Spouses of Military Service Members Involved in Iraq and Afghanistan Deployments," *Military Medicine*, Vol. 173, No. 11, November 2008, pp. 1051–1056.

Economou, M., E. Louki, L. E. Peppou, C. Gramandani, L. Yotis, and C. N. Stefanis, "Fighting Psychiatric Stigma in the Classroom: The Impact of an Educational Intervention on Secondary School Students' Attitudes to Schizophrenia," *International Journal of Social Psychiatry*, Vol. 58, No. 5, 2012, pp. 544–551.

Edlund, Mark J., John C. Fortney, Christina M. Reaves, Jeffrey M. Pyne, and Dinesh Mittal, "Beliefs About Depression and Depression Treatment Among Depressed Veterans," *Medical Care*, Vol. 46, No. 6, June 2008, pp. 581–589.

Eisenberg, Daniel, Marilyn F. Downs, and Ezra Golberstein, "Effects of Contact with Treatment Users on Mental Illness Stigma: Evidence from University Roommate Assignments," *Social Science and Medicine*, Vol. 75, No. 6, September 2012, pp. 1122–1127.

Eisenberg, Daniel, Marilyn F. Downs, Ezra Golberstein, and Kara Zivin, "Stigma and Help Seeking for Mental Health Among College Students," *Medical Care Research and Review*, Vol. 66, No. 5, October 2009, pp. 522–541.

Engel, Charles C.; Thomas Oxman; Christopher Yamamoto; Darin Gould; Sheila Barry; Patrice Stewart; John W. Williams, Jr.; and Allen J. Dietrich, "RESPECT-Mil: Feasibility of a Systems-Level Collaborative Care Approach to Depression and Post-Traumatic Stress Disorder in Military Primary Care," *Military Medicine*, Vol. 173, No. 10, October 2008, pp. 935–940.

Esposito-Smythers, Christianne, Jennifer Wolff, Keith M. Lemmon, Mary Bodzy, Rebecca R. Swenson, and Anthony Spirito, "Military Youth and the Deployment Cycle: Emotional Health Consequences and Recommendations for Intervention," *Journal of Family Psychology*, Vol. 25, No. 4, August 2011, pp. 497–507.

Essler, Vicky, Anthony Arthur, and Theodore Stickley, "Using a School-Based Intervention to Challenge Stigmatizing Attitudes and Promote Mental Health in Teenagers," *Journal of Mental Health*, Vol. 15, No. 2, 2006, pp. 243–250.

Evans-Lacko, Sara, E. Brohan, R. Mojtabai, and Graham Thornicroft, "Association Between Public Views of Mental Illness and Self-Stigma Among Individuals with Mental Illness in 14 European Countries," *Psychological Medicine*, Vol. 42, No. 8, August 2012, pp. 1741–1752.

Evans-Lacko, Sara, Claire Henderson, and Graham Thornicroft, "Public Knowledge, Attitudes and Behaviour Regarding People with Mental Illness in England 2009–2012," *British Journal of Psychiatry*, Vol. 202, 2013, pp. s51–s57.

Evans-Lacko, Sara, Jillian London, Sarah Japhet, Nicholas Rüsch, Clare Flach, Elizabeth Corker, Claire Henderson, and Graham Thornicroft, "Mass Social Contact Interventions and Their Effect on Mental Health Related Stigma and Intended Discrimination," *BMC Public Health*, Vol. 12, 2012, p. 489. Evans-Lacko, Sara, Estelle Malcolm, Keon West, Diana Rose, Jillian London, Nicolas Rüsch, Kirsty Little, Claire Henderson, and Graham Thornicroft, "Influence of Time to Change's Social Marketing Interventions on Stigma in England 2009–2011," *British Journal of Psychiatry*, Vol. 202, 2013, pp. 77–88.

Executive Secretary, Office of the Secretary of Defense, *Selection of DoD Military and Civilian Personnel and Contractor Employees for Assignment to Presidential Support Activities (PSAs)*, Department of Defense Instruction 5210.87, November 30, 1998. As of February 7, 2014: http://www.dtic.mil/whs/directives/corres/pdf/521087p.pdf

Farina, Amerigo, and Robert D. Feliner, "Employment Interviewer Reactions to Former Mental Patients," *Journal of Abnormal Psychology*, Vol. 82, No. 2, October 1973, pp. 268–272.

Fass, Daniel, *Increasing the Accuracy of the Military's Post-Deployment Mental Health Screening Strategies*, Indiana State University, dissertation, 2010.

Fenton, Wayne S., Crystal R. Blyler, and Robert K. Heinssen, "Determinants of Medication Compliance in Schizophrenia: Empirical and Clinical Findings," *Schizophrenia Bulletin*, Vol. 23, No. 4, 1997, pp. 637–651.

Fikretoglu, D., S. Guay, D. Pedlar, and A. Brunet, "Twelve Month Use of Mental Health Services in a Nationally Representative, Active Military Sample," *Medical Care*, Vol. 46, No. 2, February 2008, pp. 217–223.

Finkelstein, Joseph, and Oleg Lapshin, "Reducing Depression Stigma Using a Web-Based Program," *International Journal of Medical Informatics*, Vol. 76, No. 10, October 2007, pp. 726–734.

Finnegan, Alan, Sara Finnegan, Paula McGee, Mike Srinivasan, and Robin Simpson, "Predisposing Factors Leading to Depression in the British Army," *British Journal of Nursing*, Vol. 19, No. 21, 2010, pp. 1355–1362.

Fishbein, Martin, and Icek Ajzen, Belief, Attitude, Intention, and Behavior: An Introduction to Theory and Research, Reading, Mass.: Addison-Wesley, 1975.

Fishbein, Martin, Joseph Cappella, Robert Hornik, Sarah Sayeed, Marco Yzer, and R. Kirkland Ahern, "The Role of Theory in Developing Effective Antidrug Public Service Announcements," in William D. Crano and Michael Burgoon, eds., *Mass Media and Drug Prevention: Classic and Contemporary Theories and Research*, Mahwah, N.J.: Lawrence Erlbaum, 2002, pp. 89–117.

Fitch, Kathryn, Steven J. Bernstein, Mary S. Aguilar, Bernard Burnand, Juan Ramon LaCalle, Pablo Lazaro, Mirjam van het Loo, Joseph McDonnell, Janneke Vader, and James P. Kahan, *The RAND/UCLA Appropriateness Method User's Manual*, Santa Monica, Calif.: RAND Corporation, MR-1269-DG-XII/RE, 2001. As of January 31, 2014: http://www.rand.org/pubs/monograph_reports/MR1269.html

http://www.rand.org/pubs/monograph_teports/witt20/.html

FM 3-34.468—See Headquarters, Department of the Army, 2003c.

FM 3-50.1—See Headquarters, Department of the Army, 2011c.

FM 4-02—See Headquarters, Department of the Army, 2013c.

FM 4-02.51—See Headquarters, Department of the Army, 2006c.

FM 4-02.6—See Headquarters, Department of the Army, 2002a.

FM 4-25.11 C1—*See* Headquarters, Department of the Army; Headquarters, Department of the Navy; Headquarters, Department of the Air Force; and Commandant of the Marine Corps, 2002.

FM 6-22—See Headquarters, Department of the Army, 2006d.

FM 6-22.5—See Headquarters, Department of the Army, 2009b.
Freeman, Sharon Morgillo, Leslie Lundt, Edward J. Swanton, and Bret A. Moore, "Myths and Realities of Pharmacotherapy in the Military," in Sharon Morgillo Freeman, Bret A. Moore, and Arthur Freeman, eds., *Living and Surviving in Harm's Way: A Psychological Treatment Handbook for Pre- and Post-Deployment of Military Personnel*, New York: Routledge, 2009, pp. 329–346.

Friedrich, B., S. Evans-Lacko, C. Henderson, and G. Thornicroft, "What Purpose Does a Website Linked to a Mental Health Anti-Stigma Campaign Serve? Evaluation of the Open UP Website Using Surveys and Google Analytics," *Psychiatrische Praxis*, Vol. 38, 2011.

Fung, Kelvin M. T., Hector W. H. Tsang, and Wai-ming Cheung, "Randomized Controlled Trial of the Self-Stigma Reduction Program Among Individuals with Schizophrenia," *Psychiatry Research*, Vol. 189, No. 2, September 2011, pp. 208–214.

Fung, Kelvin M. T., Hector W. H. Tsang, and Patrick W. Corrigan, "Self-Stigma of People with Schizophrenia as Predictor of Their Adherence to Psychosocial Treatment," *Psychiatric Rehabilitation Journal*, Vol. 32, No. 2, Fall 2008, pp. 95–104.

Fung, Kelvin M. T., Hector W. H. Tsang, Patrick W. Corrigan, Chow S. Lam, and Wai-ming Cheng, "Measuring Self-Stigma of Mental Illness in China and Its Implications for Recovery," *International Journal of Social Psychiatry*, Vol. 53, No. 5, 2007, pp. 408–418.

Fung, Mang-tak, Stages of Change, Self-Stigma, and Treatment Compliance Among Chinese Adults with Severe Mental Illness, Hong Kong Polytechnic University, doctoral thesis, 2010.

Gaebel, Wolfgang, Harald Zäske, Anja E. Baumann, Joachim Klosterkötter, Wolfgang Maier, Petra Decker, and Hans-Jürgen Möller, "Evaluation of the German WPA 'Program Against Stigma and Discrimination Because of Schizophrenia—Open the Doors': Results from Representative Telephone Surveys Before and After Three Years of Antistigma Interventions," *Schizophrenia Research*, Vol. 98, No. 1–3, January 2008, pp. 184–193.

Gates, Robert M., "Policy Implementation: Mental Health Question, Standard Form (SF) 86, Questionnaire for National Security Positions," memorandum from the Secretary of Defense to secretaries of the military departments, chair of the Joint Chiefs of Staff, undersecretaries of defense, assistant secretaries of defense, general counsel of the U.S. Department of Defense, inspector general of the U.S. Department of Defense, director for administration and management, directors of defense agencies, and directors of U.S. Department of Defense field activities, Washington, D.C., April 18, 2008. As of February 3, 2014:

http://www.army.mil/docs/OSD_Guidance_on_Revised_Q21.pdf

General Counsel, U.S. Department of Defense, *Defense Industrial Personnel Security Clearance Review Program*, Department of Defense Directive 5220.6, January 2, 1992; administrative reissuance incorporating through change 4 April 20, 1999. As of February 7, 2014: http://www.dtic.mil/whs/directives/corres/pdf/522006p.pdf

Ghaffari, Azadeh S., *Stigma and Psychological Help-Seeking of Veterans*, Purdue University, doctoral thesis, 2011.

Gibbons, Robert D., C. Hendricks Brown, Kwan Hur, Sue M. Marcus, Dulal K. Bhaumik, and J. John Mann, "Relationship Between Antidepressants and Suicide Attempts: An Analysis of the Veterans Health Administration Data Sets," *American Journal of Psychiatry*, Vol. 164, No. 7, July 1, 2007, pp. 1044–1049.

Gibbs, Deborah A., Kristine L. Rae Olmsted, Janice M. Brown, and A. Monique Clinton-Sherrod, "Dynamics of Stigma for Alcohol and Mental Health Treatment Among Army Soldiers," *Military Psychology*, Vol. 23, No. 1, January 2011, pp. 36–51. Givens, J. L., I. R. Katz, S. Bellamy, and W. C. Holmes, "Stigma and the Acceptability of Depression Treatments Among African Americans and Whites," *Journal of General Internal Medicine*, Vol. 22, No. 9, September 2007, pp. 1292–1297.

Goff, Meghan, Acupuncture for the Treatment of Post Traumatic Stress Disorder, Pacific University School of Physician Assistant Studies, Paper 262, 2011.

Goffman, Erving, *Stigma: Notes on the Management of Spoiled Identity*, Englewood Cliffs, N.J.: Prentice-Hall, 1963.

, Stigma: Notes on the Management of Spoiled Identity, New York: Simon and Schuster, 1986.

Golberstein, Ezra, Daniel Eisenberg, and Sarah E. Gollust, "Perceived Stigma and Mental Health Care Seeking," *Psychiatric Services*, Vol. 59, No. 4, April 2008, pp. 392–399.

, "Perceived Stigma and Help-Seeking Behavior: Longitudinal Evidence from the Healthy Minds Study," *Psychiatric Services*, Vol. 60, No. 9, September 1, 2009, pp. 1254–1256.

Goldsmith, S., T. C. Pellmar, A. M. Kleinman, and W. E. Bunney, eds., *Reducing Suicide: A National Imperative*, Washington, D.C.: National Academies Press, 2002.

Gorman, Lisa A., Adrian J. Blow, Barbara D. Ames, and Philip L. Reed, "National Guard Families After Combat: Mental Health, Use of Mental Health Services, and Perceived Treatment Barriers," *Psychiatric Services*, Vol. 62, No. 1, January 1, 2011, pp. 28–34.

Gould, Matthew, Amy Adler, Mark Zamorski, Carl Castro, Natalie Hanily, Nicole Steele, Steve Kearney, and Neil Greenberg, "Do Stigma and Other Perceived Barriers to Mental Health Care Differ Across Armed Forces?" *Journal of the Royal Society of Medicine*, Vol. 103, No. 4, April 1, 2010, pp. 148–156.

Gould, Matthew, Neil Greenberg, and J. Hetherton, "Stigma and the Military: Evaluation of a PTSD Psychoeducational Program," *Journal of Traumatic Stress*, Vol. 20, No. 4, August 2007, pp. 505–515.

Greenberg, Neil, Victoria Langston, Brian Everitt, Amy Iversen, Nicola T. Fear, Norman Jones, and Simon Wessely, "A Cluster Randomized Controlled Trial to Determine the Efficacy of Trauma Risk Management (TRiM) in a Military Population," *Journal of Traumatic Stress*, Vol. 23, No. 4, August 2010, pp. 430–436.

Greenberg, Neil, Victoria Langston, Amy C. Iversen, and Simon Wessely, "The Acceptability of 'Trauma Risk Management' Within the UK Armed Forces," *Occupational Medicine*, Vol. 61, No. 3, May 2011, pp. 184–189.

Greene-Shortridge, Tiffany M., Thomas W. Britt, and Carl Andrew Castro, "The Stigma of Mental Health Problems in the Military," *Military Medicine*, Vol. 172, No. 2, February 2007, pp. 157–161.

Griffiths, Kathleen M., Helen Christensen, Anthony F. Jorm, Kimberley Evans, and Chloe Groves, "Effect of Web-Based Depression Literacy and Cognitive–Behavioural Therapy Interventions on Stigmatising Attitudes to Depression: Randomised Controlled Trial," *British Journal of Psychiatry*, Vol. 185, No. 4, 2004, pp. 342–349.

Gsell, Margaret Ray, *Individual Contributions to Stigma and Attitudes Toward Help-Seeking Among Rural Emerging Adult College Students*, Richmond, Va.: Virginia Commonwealth University, doctoral thesis, December 20, 2010.

Hackler, Ashley Hill, *Contact and Stigma Toward Mental Illness: Measuring the Effectiveness of Two Video Interventions*, Iowa State University, doctoral thesis, 2011.

Halter, M. J., "Stigma and Help Seeking Related to Depression: A Study of Nursing Students," *Journal of Psychosocial Nursing and Mental Health Services*, Vol. 42, No. 2, February 2004, pp. 42–51.

Hamilton, David L., and Jeffrey W. Sherman, "Stereotypes," in Robert S. Wyer and Thomas K. Srull, eds., *Handbook of Social Cognition*, Vol. 2: *Applications*, Hillsdale, N.J.: L. Erlbaum Associates, 1994, pp. 44–91.

Hammen, Constance L., and Stefanie D. Peters, "Interpersonal Consequences of Depression: Responses to Men and Women Enacting a Depressed Role," *Journal of Abnormal Psychology*, Vol. 87, No. 3, June 1978, pp. 322–332.

Hammer, Joseph H., and David L. Vogel, "Men's Help Seeking for Depression: The Efficacy of a Male-Sensitive Brochure About Counseling," *Counseling Psychologist*, Vol. 38, No. 2, February 2010, pp. 296–313.

, "Assessing the Utility of the Willingness/Prototype Model in Predicting Help-Seeking Decisions," *Journal of Counseling Psychology*, Vol. 60, No. 1, January 2013, pp. 83–97.

Hasson-Ohayon, Ilanit, Itamar Levy, Shlomo Kravetz, Adi Vollanski-Narkis, and David Roe, "Insight into Mental Illness, Self-Stigma, and the Family Burden of Parents of Persons with a Severe Mental Illness," *Comprehensive Psychiatry*, Vol. 52, No. 1, January–February 2011, pp. 75–80.

Hayes, Steven C., Jason B. Luoma, Frank W. Bond, Akihiko Masuda, and Jason Lillis, "Acceptance and Commitment Therapy: Model, Processes and Outcomes," *Behaviour Research and Therapy*, Vol. 44, No. 1, January 2006, pp. 1–25.

Hayward, Peter, and Jenifer A. Bright, "Stigma and Mental Illness: A Review and Critique," *Journal of Mental Health*, Vol. 6, No. 4, 1997, pp. 345–354.

Headquarters, Air Force Global Strike Command, *AFGSC Personnel Reliability Program* (*PRP*) *Procedural Guidance Memorandum*, Air Force Global Strike Command Guidance Memorandum 10-02, August 2, 2012. As of February 20, 2014: http://static.e-publishing.af.mil/production/1/afgsc/publication/afgscgm10-02/afgscgm10-02.pdf

Headquarters, Department of the Army, *Acceptance, Accreditation, and Release of United States Army Criminal Investigation Command Personnel*, Army Regulation 195-3, April 22, 1987. As of February 8, 2014:

http://www.apd.army.mil/pdffiles/r195_3.pdf

———, Personal Security, Army Regulation 190-58, March 22, 1989. As of February 9, 2014: http://www.apd.army.mil/pdffiles/r190_58.pdf

———, Army Educational Incentives and Entitlements, Army Regulation 621-202, February 3, 1992; rapid action revision September 6, 2011. As of February 9, 2014: http://www.apd.army.mil/pdffiles/r621_202.pdf

—_____, The Department of the Army Freedom of Information Act Program, Army Regulation 25-55, November 1, 1997. As of February 9, 2014: http://www.asamra.army.mil/docs/R25_55.pdf

—____, The Army Public Affairs Program, Army Regulation 360-1, September 15, 2000b. As of February 9, 2014:

http://www.asaie.army.mil/Public/IE/Toolbox/documents/r360_1.pdf

——, United States Army Officer Candidates School, Army Regulation 350-51, June 11, 2001. As of February 9, 2014:

http://www.apd.army.mil/pdffiles/r350_51.pdf

———, The Medical Company: Tactics, Techniques, and Procedures, Field Manual 4-02.6, August 2002a. As of February 7, 2014:

http://www.globalsecurity.org/military/library/policy/army/fm/4-02-6/fm4-02-6_c1_2004.pdf

———, Military Police Leaders' Handbook, Field Manual 3-19.4, August 2, 2002b. As of February 7, 2014:

http://armypubs.army.mil/doctrine/DR_pubs/dr_a/pdf/fm3_19x4.pdf

——, Personnel Processing (In-, Out-, Soldier Readiness Mobilization, and Deployment Processing), Army Regulation 600-8-101, July 18, 2003a. As of February 5, 2014: http://www.apd.army.mil/pdffiles/r600_8_101.pdf

——, *Nonappropriated Funds Personnel Policy*, Army Regulation 215-3, August 29, 2003b. As of February 9, 2014:

http://www.apd.army.mil/pdffiles/r215_3.pdf

———, Seabee Quarry Blasting Operations and Safety Manual, Field Manual 3-34.468, December 2003c.

—_____, Equal Employment Opportunity Discrimination Complaints, Army Regulation 690-600, February 9, 2004a. As of February 9, 2014: http://armypubs.army.mil/epubs/pdf/r690_600.pdf

———, "Vending Facility Program for the Blind on Federal Property," Army Regulation 210-25, June 30, 2004b. As of February 8, 2014: http://armypubs.army.mil/epubs/pdf/r210_25.pdf

———, Civilian Inmate Labor Program, Army Regulation 210-35, January 14, 2005a. As of February 8, 2014: http://armypubs.army.mil/epubs/pdf/r210_35.pdf

—______, Active Duty Enlisted Administrative Separations, Army Regulation 635-200, June 6, 2005b; rapid action revision September 6, 2011. As of February 8, 2014: http://www.apd.army.mil/pdffiles/r635_200.pdf

—_____, Military Police Investigations, Army Regulation 190-30, November 1, 2005c. As of February 8, 2014:

http://www.apd.army.mil/pdffiles/r190_30.pdf

— Army Retention Program, Army Regulation 601-280, January 31, 2006a; rapid action revision September 15, 2011. As of February 8, 2014: http://www.apd.army.mil/pdffiles/r601_280.pdf

The Army Corrections System, Army Regulation 190-47, June 15, 2006b. As of February 8, 2014:

http://www.apd.army.mil/pdffiles/r190_47.pdf

Combat and Operational Stress Control, Field Manual 4-02.51, July 2006c. As of February 8, 2014:

https://www.fas.org/irp/doddir/army/fm4-02-51.pdf

, Army Leadership: Competent, Confident, and Agile, Field Manual 6-22, October 2006d. As of February 7, 2014:

http://www.fas.org/irp/doddir/army/fm6-22.pdf

———, Exceptional Family Member Program, Army Regulation 608-75, November 22, 2006e; rapid action revision February 24, 2011. As of February 9, 2014: http://www.apd.army.mil/pdffiles/r608_75.pdf

—_____, Military Awards, Army Regulation 600-8-22, December 11, 2006f; rapid action revision June 24, 2013. As of February 8, 2014: http://www.apd.army.mil/pdffiles/r600_8_22.pdf

—_____, Enlisted Administrative Separations, Army Regulation 135-178, March 13, 2007a; rapid action revision September 13, 2011. As of February 7, 2014: http://www.apd.army.mil/pdffiles/r135_178.pdf

—_____, Army Health Promotion, Army Regulation 600-63, May 7, 2007b; rapid action revision September 7, 2010. As of February 3, 2014: http://www.apd.army.mil/pdffiles/r600_63.pdf

———, "Temporary Flying Restrictions Due to Exogenous Factors Affecting Aircrew Efficiency," Army Regulation 40-8, May 16, 2007c. As of February 7, 2014: http://www.apd.army.mil/pdffiles/r40_8.pdf

———, Assignment of Enlisted Personnel to the U.S. Army Recruiting Command, Army Regulation 601-1, October 12, 2007d; rapid action revision September 6, 2011.

—_____, The Army Family Advocacy Program, Army Regulation 608-18, October 30, 2007e; rapid action revision September 13, 2011. As of February 5, 2014: http://www.apd.army.mil/pdffiles/r608_18.pdf

—_____, Standards of Medical Fitness, Army Regulation 40-501, December 14, 2007f; rapid action revision August 4, 2011. As of February 7, 2014: http://www.apd.army.mil/pdffiles/r40_501.pdf

———, Army Command Policy, Army Regulation 600-20, March 18, 2008a; rapid action revision September 20, 2012. As of February 8, 2014: http://www.apd.army.mil/pdffiles/r600_20.pdf

—______, Medical Record Administration and Healthcare Documentation, Army Regulation 40-66, June 17, 2008b; rapid action revision January 4, 2010. As of February 7, 2014: http://www.apd.army.mil/pdffiles/r40_66.pdf

———, Biological Surety, Army Regulation 50-1, July 28, 2008c. As of February 8, 2014: http://www.fas.org/irp/doddir/army/ar50-1.pdf

—_____, Line of Duty Policy, Procedures, and Investigations, Army Regulation 600-8-4, September 4, 2008d. As of February 5, 2014: http://www.apd.army.mil/pdffiles/r600_8_4.pdf

—_____, Enlisted Assignments and Utilization Management, Army Regulation 614-200, February 26, 2009a; rapid action revision October 11, 2011. As of February 4, 2014: http://www.apd.army.mil/pdffiles/r614_200.pdf

———, Combat and Operational Stress Control Manual for Leaders and Soldiers, Field Manual 6-22.5, March 2009b. As of February 7, 2014: http://armypubs.army.mil/doctrine/DR_pubs/dr_a/pdf/fm6_22x5.pdf

——, Management and Recall to Active Duty of Retired Soldiers of the Army in Support of Mobilization and Peacetime Operations, Army Regulation 601-10, March 13, 2009c. As of February 8, 2014:

http://www.apd.army.mil/pdffiles/r601_10.pdf

, Army Chaplain Corps Activities, Army Regulation 165-1, December 3, 2009d. As of February 8, 2014:

http://www.usachcs.army.mil/pdf/activities.pdf

— United States Military Academy, Army Regulation 210-26, December 9, 2009e; rapid action revision September 6, 2011. As of February 8, 2014: http://www.apd.army.mil/pdffiles/r210_26.pdf

—_____, Army Training and Leader Development, Army Regulation 350-1, December 18, 2009f; rapid action revision August 4, 2011. As of February 7, 2014: http://www.apd.army.mil/pdffiles/r350_1.pdf

—_______, Patient Administration, Army Regulation 40-400, January 27, 2010a; rapid action revision September 15, 2011. As of February 8, 2014: http://www.apd.army.mil/pdffiles/r40_400.pdf

— Overseas Service, Army Regulation 614-30, March 3, 2010b. As of February 8, 2014: http://www.apd.army.mil/pdffiles/r614_30.pdf

—_____, Enlisted Promotions and Reductions, Army Regulation 600-8-19, April 30, 2010c; rapid action revision December 27, 2011. As of February 7, 2014: http://www.apd.army.mil/pdffiles/r600_8_19.pdf

—_____, Military Morale, Welfare, and Recreation Programs and Nonappropriated Fund Instrumentalities, Army Regulation 215-1, September 24, 2010d. As of February 8, 2014: http://armypubs.army.mil/epubs/pdf/r215_1.pdf

—_____, Mechanized Infantry Platoon and Squad (Bradley), Army Tactics, Techniques, and Procedures 3-21.71, November 2010e. As of February 9, 2014: http://armypubs.army.mil/doctrine/DR_pubs/dr_a/pdf/attp3_21x71.pdf

—_____, Inspector General Activities and Procedures, Army Regulation 20-1, November 29, 2010f; rapid action revision July 3, 2012. As of February 8, 2014: http://www.apd.army.mil/pdffiles/r20_1.pdf

—_____, SBCT Infantry Rifle Platoon and Squad, Army Tactics, Techniques, and Procedures 3-21.9, December 8, 2010g. As of February 4, 2014: http://armypubs.army.mil/doctrine/DR pubs/dr a/pdf/attp3 21x9.pdf

———, Active and Reserve Components Enlistment Program, Army Regulation 601-210, February 8, 2011a; rapid action revision March 12, 2013. As of February 8, 2014: http://www.apd.army.mil/pdffiles/r601_210.pdf

Law and Order Operations, Army Tactics, Techniques, and Procedures 3-39.10, June 2011b. As of February 8, 2014:

http://armypubs.army.mil/doctrine/DR_pubs/dr_a/pdf/attp3_39x10.pdf

———, Army Personnel Recovery, Field Manual 3-50.1, November 2011c. As of February 7, 2014: https://www.fas.org/irp/doddir/army/fm3-50-1.pdf

——, Army 2020: Generating Health and Discipline in the Force Ahead of the Strategic Reset, rev. 2, Washington, D.C., 2012a. As of January 31, 2014: http://www.armyg1.army.mil/hr/suicide/docs/ Army_2020_Generating_Health_and_Discipline_in_the_Force_Report_2012_GOLD_BOOK.pdf

Evaluation Reporting System, Army Regulation 623-3, June 5, 2012b. As of February 9, 2014:

http://www.apd.army.mil/pdffiles/r623_3.pdf

—_____, Army Leadership, Army Doctrine and Training Publication 6-22, August 2012c. As of February 7, 2014:

http://www.chapnet.army.mil/pdf/ADRP%206_22_new.pdf

———, Evidence Procedures, Army Regulation 195-5, February 22, 2013a. As of February 9, 2014: http://www.apd.army.mil/pdffiles/r195_5.pdf

—____, Medical, Dental, and Veterinary Care, Army Regulation 40-3, April 23, 2013b. As of February 8, 2014:

http://armypubs.army.mil/epubs/pdf/r40_3.pdf

— Army Health System, Field Manual 4-02, August 2013c. As of February 7, 2014: http://armypubs.army.mil/doctrine/DR_pubs/dr_a/pdf/fm4_02.pdf

—_____, Military Auxiliary Radio System and Amateur Radio Program, Army Regulation 25-6, January 3, 2014a. As of February 9, 2014: http://armypubs.army.mil/epubs/pdf/r25_6.pdf

Personnel Security Program, Army Regulation 380-67, January 24, 2014b. As of February 7, 2014:

http://www.apd.army.mil/pdffiles/r380_67.pdf

Headquarters, Department of the Army; and Headquarters, Department of the Air Force, *Military Sentences to Confinement*, Army Regulation 633-30, Air Force Regulation 125-30, February 28, 1989. As of February 8, 2014:

http://www.apd.army.mil/pdffiles/r633_30.pdf

Headquarters, Department of the Army; Headquarters, Department of the Navy; and Headquarters, Department of the Air Force, *Security Assistance: Joint Security Assistance Training (JSAT) Regulation*, Army Regulation 12-15, Office of the Chief of Naval Operations Instruction 4950.1H, Air Force Regulation 50-29, Marine Corps Order 4950.2, October 1, 1985.

———, Administrative and Logistical Support of Overseas Security Assistance Organizations (SAOs), Army Regulation 1-75, Secretary of the Navy Instruction 4900.49, Air Force Joint Instruction 16-104, March 27, 2000. As of February 8, 2014: http://www.apd.army.mil/pdffiles/r1_75.pdf

Headquarters, Department of the Army; Headquarters, Department of the Navy; Headquarters, Department of the Air Force; and Commandant of the Marine Corps, *First Aid*, Field Manual 4-25.11 C1, Naval Tactical Reference Publication 4-02.1.1, Air Force Manual 44-163(I), Marine Corps Reference Publication 3-02G, December 2002. As of February 7, 2014: http://armypubs.army.mil/doctrine/DR_pubs/dr_a/pdf/fm4_25x11.pdf

Headquarters, Department of the Army; Headquarters, Department of the Navy; Headquarters, Department of the Air Force; and Headquarters, Department of the Marine Corps, *Enemy Prisoners of War, Retained Personnel, Civilian Internees and Other Detainees*, Army Regulation 190-8, Office of the Chief of Naval Operations Instruction 3461.6, Air Force Joint Instruction 31-304, Marine Corps Order 3461.1, October 1, 1997. As of February 7, 2014: http://www.marines.mil/Portals/59/Publications/MCO%203461.1.pdf

Headquarters, Department of the Army; Headquarters, Department of the Navy; Headquarters, Department of the Air Force; Headquarters, Department of the Marine Corps; and Defense Logistics Agency, *Motor Vehicle Traffic Supervision*, Army Regulation 190-5, Office of the Chief of Naval Operations Instruction 11200.5D, Air Force Instruction 31-218(I), Marine Corps Order 5110.1D, Defense Logistics Agency Regulation 5720.1, May 22, 2006. As of February 7, 2014: http://www.apd.army.mil/pdffiles/r190_5.pdf

Headquarters, Department of the Army; Headquarters, Department of the Navy; Headquarters, Department of the Air Force; Headquarters, Department of the Marine Corps; and Headquarters, Department of the Coast Guard, *Military Entrance Processing Station (MEPS)*, Army Regulation 601-270, Office of the Chief of Naval Operations Instruction 1100.4C CH-2, Air Force Instruction 36-2003 IP, Marine Corps Order 1100.75F, Commandant Instruction Manual 1100.2E, March 23, 2007; rapid action revision September 13, 2011. As of February 7, 2014: http://www.apd.army.mil/pdffiles/r601_270.pdf

Headquarters, Department of the Army; Headquarters, Department of the Navy; Headquarters, Department of the Marine Corps; and Headquarters, Department of the Coast Guard, *Military Entrance Processing Station (MEPS)*, Army Regulation 601-270, Office of the Chief of Naval Operations Instruction 1100.4C CH-1, Marine Corps Order 1100.75F, Commandant Instruction M 1100.2E, March 23, 2007; rapid action revision September 13, 2011. As of February 19, 2014: http://www.marines.mil/Portals/59/Publications/MCO%201100.75F%20AR601_270.pdf

Headquarters, U.S. Marine Corps, *USMC Leadership Program Discussion Leaders' Course*, Navy and Marine Corps 2660, January 21, 1977.

———, Fraud Indicators Handbook, Navy and Marine Corps 2768, June 17, 1991. As of February 19, 2014: http://www.marines.mil/Portals/59/Publications/NAVMC%202768.pdf

———, Selecting, Screening, and Preparing Enlisted Marines for Special Duty Assignments and Independent Duties, Marine Corps Order P1326.6D, September 12, 1999. As of February 17, 2014: http://www.marines.mil/Portals/59/Publications/MCO%20P1326.6D%20W%20CH%201.pdf

—_____, Military Personnel Procurement Manual, Vol. 2: *Enlisted Procurement*, Marine Corps Order P1100.72C, June 18, 2004. As of February 17, 2014: http://www.marines.mil/Portals/59/Publications/MCO%20P1100.72C%20W%20ERRATUM.pdf

———, Marine Corps Mentoring Program (MCMP) Guidebook, Navy and Marine Corps Directive 1500.58, February 13, 2006. As of February 25, 2014: http://www.marines.mil/Portals/59/Publications/NAVMC%20DIR%201500.58.pdf

Headquarters, U.S. Air Force, Surgeon General, "Implementation of Revised Department of Defense Forms 2795, 2796, and 2900: Deployment Health Assessments," Air Force Guidance Memorandum 48-04, November 7, 2012. As of February 21, 2014: http://static.e-publishing.af.mil/production/1/af_ja/publication/afgm48-04/afgm48_04.pdf

Hegerl, Ulrich, David Althaus, Armin Schmidtke, and Günter Niklewski, "The Alliance Against Depression: 2-Year Evaluation of a Community-Based Intervention to Reduce Suicidality," *Psychological Medicine*, Vol. 36, No. 9, September 2006, pp. 1225–1233.

Hegerl, Ulrich, Roland Mergl, Inga Havers, Armin Schmidtke, Hartmut Lehfeld, Günter Niklewski, and David Althaus, "Sustainable Effects on Suicidality Were Found for the Nuremberg Alliance Against Depression," *European Archives of Psychiatry and Clinical Neuroscience*, Vol. 260, No. 5, 2010, pp. 401–406.

Hegerl, Ulrich, and Lisa Wittenburg, "Focus on Mental Health Care Reforms in Europe: The European Alliance Against Depression—A Multilevel Approach to the Prevention of Suicidal Behavior," *Psychiatric Services*, Vol. 60, No. 5, May 1, 2009, pp. 596–599.

Heijnders, Miriam, and Suzanne Van Der Meij, "The Fight Against Stigma: An Overview of Stigma-Reduction Strategies and Interventions," *Psychology, Health and Medicine*, Vol. 11, No. 3, 2006, pp. 353–363. Hemmens, Craig; Milo Miller; Velmer S. Burton, Jr.; and Susan Milner, "The Consequences of Official Labels: An Examination of the Rights Lost by the Mentally III and Mentally Incompetent Ten Years Later," *Community Mental Health Journal*, Vol. 38, No. 2, April 2002, pp. 129–140.

Henderson, Claire, Elizabeth Corker, Elanor Lewis-Holmes, Sarah Hamilton, Clare Flach, Diana Rose, Paul Williams, Vanessa Pinfold, and Graham Thornicroft, "England's Time to Change Antistigma Campaign: One-Year Outcomes of Service User–Rated Experiences of Discrimination," *Psychiatric Services*, Vol. 63, No. 5, May 1, 2012, pp. 451–457.

Henderson, Claire, Sara Evans-Lacko, Clare Flach, and Graham Thornicroft, "Responses to Mental Health Stigma Questions: The Importance of Social Desirability and Data Collection Method," *Canadian Journal of Psychiatry*, Vol. 57, No. 3, March 2012, pp. 152–160.

Hermann, R. C., C. K. Rollins, and J. A. Chan, "Risk-Adjusting Outcomes of Mental Health and Substance-Related Care: A Review of the Literature," *Harvard Review of Psychiatry*, Vol. 15, No. 2, March–April 2007, pp. 52–69.

Hickie, I., "Can We Reduce the Burden of Depression? The Australian Experience with BeyondBlue—The National Depression Initiative," *Australasian Psychiatry*, Vol. 12, Supp., 2004, pp. S38–S46.

Hill, Robert B., "Structural Discrimination: The Unintended Consequences of Institutional Processes," in Hubert J. O'Gorman, ed., *Surveying Social Life: Papers in Honor of Herbert H. Hyman*, Middletown, Conn.: Wesleyan University Press, 1988, pp. 142–177.

Hilton, James L., and William von Hippel, "Stereotypes," *Annual Review of Psychology*, Vol. 47, No. 1, 1996, pp. 237–271.

Hipes, Crosby, "The Stigma of Mental Health Treatment in the Military: An Experimental Approach," *Current Research in Social Psychology*, Vol. 18, No. 5, 2012.

Hoerster, K. D., C. A. Malte, Z. E. Imel, Z. Ahmad, S. C. Hunt, and M. Jakupcak, "Association of Perceived Barriers with Prospective Use of VA Mental Health Care Among Iraq and Afghanistan Veterans," *Psychiatric Services*, Vol. 63, No. 4, April 2012, pp. 380–382.

Hoge, Charles W., "Interventions for War-Related Posttraumatic Stress Disorder: Meeting Veterans Where They Are," *Journal of the American Medical Association*, Vol. 306, No. 5, August 3, 2011, pp. 549–551.

Hoge, Charles W., Jennifer L. Auchterlonie, and Charles S. Milliken, "Mental Health Problems, Use of Mental Health Services, and Attrition from Military Service After Returning from Deployment to Iraq or Afghanistan," *Journal of the American Medical Association*, Vol. 295, No. 9, 2006, pp. 1023–1032.

Hoge, Charles W., Carl A. Castro, Stephen C. Messer, Dennis McGurk, Dave I. Cotting, and Robert L. Koffman, "Combat Duty in Iraq and Afghanistan, Mental Health Problems, and Barriers to Care," *New England Journal of Medicine*, Vol. 351, No. 1, July 1, 2004, pp. 13–22.

Hornik, Robert C., ed., *Public Health Communication: Evidence for Behavior Change*, Mahwah, N.J.: Lawrence Erlbaum Associates, 2002.

Hoyt, Gary B., "Integrated Mental Health Within Operational Units: Opportunities and Challenges," *Military Psychology*, Vol. 18, No. 4, 2006, pp. 309–320.

IACP—See International Association of Chiefs of Police.

Idemudia, Erhabor S., and Nyambeni A. Matamela, "The Role of Stigmas in Mental Health: A Comparative Study," *Curationis*, Vol. 35, No. 1, 2012, pp. 1–8.

Ilic, Marie, Jost Reinecke, Gerd Bohner, Röttgers Hans-Onno, Thomas Beblo, Martin Driessen, Ulrich Frommberger, and Patrick William Corrigan, "Protecting Self-Esteem from Stigma: A Test of Different Strategies for Coping with the Stigma of Mental Illness," *International Journal of Social Psychiatry*, Vol. 58, No. 3, 2012, pp. 246–257.

Institute of Medicine, *Reducing Risks for Mental Disorders: Frontiers for Preventive Intervention Research*, Washington, D.C.: National Academies Press, 1994. As of June 19, 2014: http://www.nap.edu/catalog.php?record_id=2139

"Interactive Simulation Launched to Provide Information on Post-Traumatic Stress Disorder: T2 Virtual PTSD Experience," *Exceptional Parent*, Vol. 41, No. 2, February 2011, p. 40.

International Association of Chiefs of Police, "International Policing Division Steering Committee (IPDSC)," undated (a). As of February 5, 2014: http://www.theiacp.org/International-Policing-Division-Steering-Committee

, "National Law Enforcement Policy Center," undated (b). As of June 14, 2014:

http://www.iacp.org/Model-Policies-for-Policing

Iversen, Amy C., Lauren van Staden, Jamie Hacker Hughes, Neil Greenberg, Matthew Hotopf, Roberto J. Rona, Graham Thornicroft, Simon Wessely, and Nicola T. Fear, "The Stigma of Mental Health Problems and Other Barriers to Care in the UK Armed Forces," *BMC Health Services Research*, Vol. 11, No. 1, 2011, p. 31.

Jones, Edward Ellsworth, *Social Stigma: The Psychology of Marked Relationships*, New York: W. H. Freeman, 1984.

Jones, Simon, Louise Howard, and Graham Thornicroft, "Diagnostic Overshadowing': Worse Physical Health Care for People with Mental Illness," *Acta Psychiatrica Scandinavica*, Vol. 118, No. 3, September 2008, pp. 169–171.

Jones, Simon, K. Sinha, M. Swinton, C. Millar, D. Rayment, and M. Simmons, "OpenMinds: Creating a Mental Health Workshop for Teenagers to Tackle Stigma and Raise Awareness," *Psychiatria Danubina*, Vol. 23, Suppl. 1, September 2011, pp. 69–72.

Jorm, Anthony F., Helen Christensen, and Kathleen M. Griffiths, "The Impact of Beyondblue: The National Depression Initiative on the Australian Public's Recognition of Depression and Beliefs About Treatments," *Australian and New Zealand Journal of Psychiatry*, Vol. 39, No. 4, May 2005, pp. 248–254.

——, "Changes in Depression Awareness and Attitudes in Australia: The Impact of Beyondblue: The National Depression Initiative," *Australian and New Zealand Journal of Psychiatry*, Vol. 40, No. 1, January 2006, pp. 42–46.

Jorm, Anthony F., Betty A. Kitchener, J. A. Fischer, and S. Cvetkovski, "Mental Health First Aid Training by E-Learning: A Randomized Controlled Trial," *Australian and New Zealand Journal of Psychiatry*, Vol. 44, No. 12, December 2010, pp. 1072–1081.

Jorm, Anthony F., Betty A. Kitchener, Richard O'Kearney, and Keith B. G. Dear, "Mental Health First Aid Training of the Public in a Rural Area: A Cluster Randomized Trial," *BMC Psychiatry*, Vol. 4, 2004, p. 33.

Jorm, Anthony F., Jo Medway, Helen Christensen, Alisa E. Korten, Patricia A. Jacomb, and Bryan Rodgers, "Attitudes Towards People with Depression: Effects on the Public's Help-Seeking and Outcome When Experiencing Common Psychiatric Symptoms," *Australian and New Zealand Journal of Psychiatry*, Vol. 34, No. 4, August 2000, pp. 612–618. Kanter, J. W., L. C. Rusch, and M. J. Brondino, "Depression Self-Stigma: A New Measure and Preliminary Findings," *Journal of Nervous and Mental Disease*, Vol. 196, No. 9, September 2008, pp. 663–670.

Kassam, Aliya, Nick Glozier, Morven Leese, Joanne Loughran, and Graham Thornicroft, "A Controlled Trial of Mental Illness Related Stigma Training for Medical Students," *BMC Medical Education*, Vol. 11, July 2011.

Kassam, Aliya, Jeanne Williams, and Scott Patten, "Perceived Discrimination Among People with Self-Reported Emotional, Psychological, or Psychiatric Conditions in a Population-Based Sample of Canadians Reporting a Disability," *Canadian Journal of Psychiatry*, Vol. 57, No. 2, February 2012, pp. 102–110.

Keller, Martin B., James P. McCullough, Daniel N. Klein, Bruce Arnow, David L. Dunner, Alan J. Gelenberg, John C. Markowitz, Charles B. Nemeroff, James M. Russell, Michael E. Thase, Madhukar H. Trivedi, Janice A. Blalock, Frances E. Borian, Darlene N. Jody, Charles DeBattista, Lorrin M. Koran, Alan F. Schatzberg, Jan Fawcett, Robert M. A. Hirschfeld, Gabor Keitner, Ivan Miller, James H. Kocsis, Susan G. Kornstein, Rachel Manber, Philip T. Ninan, Barbara Rothbaum, A. John Rush, Dina Vivian, and John Zajecka, "A Comparison of Nefazodone, the Cognitive Behavioral-Analysis System of Psychotherapy, and Their Combination for the Treatment of Chronic Depression," *New England Journal of Medicine*, Vol. 342, May 18, 2000, pp. 1462–1470.

Kelley, C. L., T. W. Britt, A. B. Adler, and P. D. Bliese, "Perceived Organizational Support, Posttraumatic Stress Disorder Symptoms, and Stigma in Soldiers Returning from Combat," *Psychological Services*, Vol. 11, No. 2, May 2014, pp. 229–234.

Kelly, Brendan D., "The Power Gap: Freedom, Power and Mental Illness," *Social Science and Medicine*, Vol. 63, No. 8, October 2006, pp. 2118–2128.

Kessler, Ronald C., *National Comorbidity Survey: Baseline (NCS-1), 1990–1992*, Ann Arbor, Mich.: Inter-University Consortium for Political and Social Research, 2002.

Kessler, Ronald C., P. A. Berglund, M. L. Bruce, J. R. Koch, E. M. Laska, P. J. Leaf, R. W. Manderscheid, R. A. Rosenheck, E. E. Walters, and P. S. Wang, "The Prevalence and Correlates of Untreated Serious Mental Illness," *Health Services Research*, Vol. 36, No. 6, December 2001, pp. 987–1007.

Kessler, Ronald C., Wai Tat Chiu, Olga Demler, and Ellen E. Walters, "Prevalence, Severity, and Comorbidity of 12-Month DSM-IV Disorders in the National Comorbidity Survey Replication," *Archives of General Psychiatry*, Vol. 62, No. 6, 2005, pp. 617–627.

Kessler, Ronald C., Amanda Sonnega, Evelyn Bromet, Michael Hughes, and Christopher B. Nelson, "Posttraumatic Stress Disorder in the National Comorbidity Survey," *Archives of General Psychiatry*, Vol. 52, No. 12, 1995, pp. 1048–1060.

Khan, T. M., M. A. Hassali, H. Tahir, and A. Khan, "A Pilot Study Evaluating the Stigma and Public Perception About the Causes of Depression and Schizophrenia," *Iranian Journal of Public Health*, Vol. 40, No. 1, 2011, pp. 50–56.

Kilmer, Beau, Christine Eibner, Jeanne S. Ringel, and Rosalie Liccardo Pacula, "Invisible Wounds, Visible Savings? Using Microsimulation to Estimate the Costs and Savings Associated with Providing Evidence-Based Treatment for PTSD and Depression to Veterans of Operation Enduring Freedom and Operation Iraqi Freedom," *Psychological Trauma: Theory, Research, Practice, and Policy*, Vol. 3, No. 2, June 2011, pp. 201–211.

Kim, Paul Y., Thomas W. Britt, Robert P. Klocko, Lyndon A. Riviere, and Amy B. Adler, "Stigma, Negative Attitudes About Treatment, and Utilization of Mental Health Care Among Soldiers," *Military Psychology*, Vol. 23, No. 1, January 2011, pp. 65–81. Kim, Paul Y., Jeffrey L. Thomas, Joshua E. Wilk, Carl A. Castro, and Charles W. Hoge, "Stigma, Barriers to Care, and Use of Mental Health Services Among Active Duty and National Guard Soldiers After Combat," *Psychiatric Services*, Vol. 61, No. 6, 2010, pp. 582–588.

Kitchener, Betty A., and Anthony F. Jorm, "Mental Health First Aid Training for the Public: Evaluation of Effects on Knowledge, Attitudes and Helping Behavior," *BMC Psychiatry*, Vol. 2, 2002, p. 10.

———, "Mental Health First Aid Training in a Workplace Setting: A Randomized Controlled Trial," *BMC Psychiatry*, Vol. 4, 2004, p. 23.

———, "Mental Health First Aid Training: Review of Evaluation Studies," *Australian and New Zealand Journal of Psychiatry*, Vol. 40, No. 1, 2006, pp. 6–8.

Klose, Arian Lorene, *The Effects of Internalized Stigma of Mental Illness on Self-Efficacy*, North Dakota State University, doctoral thesis, 2010.

Knight, Matthew T. D., Til Wykes, and Peter Hayward, "Group Treatment of Perceived Stigma and Self-Esteem in Schizophrenia: A Waiting List Trial of Efficacy," *Behavioural and Cognitive Psychotherapy*, Vol. 34, No. 3, July 2006, pp. 305–318.

Know the Signs, "About Us," undated; referenced September 13, 2013. As of January 30, 2014: http://www.suicideispreventable.org/about.php/

Kocsis, J. H., A. J. Frances, C. Voss, J. J. Mann, B. J. Mason, and J. Sweeney, "Imipramine Treatment for Chronic Depression," *Archives of General Psychiatry*, Vol.45, No. 3, March 1988, pp. 253–257.

Kolodziej, Monika E., and Blair T. Johnson, "Interpersonal Contact and Acceptance of Persons with Psychiatric Disorders: A Research Synthesis," *Journal of Consulting and Clinical Psychology*, Vol. 64, No. 6, December 1996, pp. 1387–1396.

Kondrat, David C., and Theresa J. Early, "Battling in the Trenches: Case Managers' Ability to Combat the Effects of Mental Illness Stigma on Consumers' Perceived Quality of Life," *Community Mental Health Journal*, Vol. 47, No. 4, August 2011, pp. 390–398.

Kranke, Derrick A., Jerry Floersch, Bridget O. Kranke, and Michelle R. Munson, "A Qualitative Investigation of Self-Stigma Among Adolescents Taking Psychiatric Medication," *Psychiatric Services*, Vol. 62, No. 8, 2011, pp. 893–899.

Kranke, Derrick A., Joseph Guada, Bridget Kranke, and Jerry Floersch, "What Do African American Youth with a Mental Illness Think About Help-Seeking and Psychiatric Medication? Origins of Stigmatizing Attitudes," *Social Work in Mental Health*, Vol. 10, No. 1, 2012, pp. 53–71.

Kroska, Amy, and Sarah K. Harkness, "Coping with the Stigma of Mental Illness: Empirically-Grounded Hypotheses from Computer Simulations," *Social Forces*, Vol. 89, No. 4, 2011, pp. 1315–1339.

Kumar, Asha, Perceptions of Mental Disorders in South Indians: An Examination of Factors Influencing the Stigma Process, St. John's University, doctoral thesis, 2011.

Kumkale, G. Tarcan, and Dolores Albarracín, "The Sleeper Effect in Persuasion: A Meta-Analytic Review," *Psychological Bulletin*, Vol. 130, No. 1, January 2004, pp. 143–172.

Lake, James, "Nonconventional and Integrative Treatments of Alcohol and Substance Abuse," *Psychiatric Times*, Vol. 24, No. 6, May 1, 2007.

Lakeman, R., P. McGowan, L. MacGabhann, M. Parkinson, M. Redmond, I. Sibitz, C. Stevenson, and J. Walsh, "A Qualitative Study Exploring Experiences of Discrimination Associated with Mental-Health Problems in Ireland," *Epidemiology and Psychiatric Sciences*, Vol. 21, No. 3, September 2012, pp. 271–279.

Laraway, Lori A., *Stigma: "What Is It and Why Does the Operational Commander Need to Be Concerned?"* Newport, R.I.: Naval War College, Joint Military Operations Department, 2007. As of January 31, 2014:

http://oai.dtic.mil/oai/oai?&verb=getRecord&metadataPrefix=html&identifier=ADA476490

Lau, Richard R., Robert L. Kane, Sandra H. Berry, John E. Ware, and Douglas Roy, "Channeling Health: A Review of the Evaluation of Televised Health Campaigns," *Health Education Quarterly*, Vol. 7, No. 1, Spring 1980, pp. 56–89.

Lauber, Christoph, and Norman Sartorius, "At Issue: Anti-Stigma-Endeavours," *International Review of Psychiatry*, Vol. 19, No. 2, 2007, pp. 103–106.

Lavis, John N., Andrew D. Oxman, Simon Lewin, and Atle Fretheim, "SUPPORT Tools for Evidence-Informed Health Policymaking (STP)," *Health Research Policy and Systems*, Vol. 7, Suppl. 1, 2009.

Leaf, Phillip J., Martha Livingston Bruce, and Gary L. Tischler, "The Differential Effect of Attitudes on the Use of Mental Health Services," *Social Psychiatry*, Vol. 21, 1986, pp. 187–192.

Lebowitz, Matthew S., and Woo-kyoung Ahn, "Combining Biomedical Accounts of Mental Disorders with Treatability Information to Reduce Mental Illness Stigma," *Psychiatric Services*, Vol. 63, No. 5, May 1, 2012, pp. 496–499.

Lee, S., H. S. Juon, G. Martinez, C. E. Hsu, E. S. Robinson, J. Bawa, and G. X. Ma, "Model Minority at Risk: Expressed Needs of Mental Health by Asian American Young Adults," *Journal of Community Health*, Vol. 34, No. 2, April 2009, pp. 144–152.

Lee, S., A. Tsang, and K. Kwok, "Twelve-Month Prevalence, Correlates, and Treatment Preference of Adults with DSM-IV Major Depressive Episode in Hong Kong," *Journal of Affective Disorders*, Vol. 98, No. 1–2, February 2007, pp. 129–136.

Lincoln, Tania M., Elisabeth Arens, Cornelia Berger, and Winfried Rief, "Can Antistigma Campaigns Be Improved? A Test of the Impact of Biogenetic vs. Psychosocial Causal Explanations on Implicit and Explicit Attitudes to Schizophrenia," *Schizophrenia Bulletin*, Vol. 34, No. 5, September 2008, pp. 984–994.

Lindley, E., "More to It Than Stigma: The Multiple Ways Young People Make Sense of Mental Illness," *Mental Health Today*, May 2011, pp. 28–33.

Link, Bruce G., Francis T. Cullen, James Frank, and John F. Wozniak, "The Social Rejection of Former Mental Patients: Why Labels Matter," *American Journal of Sociology*, Vol. 92, No. 6, May 1987, pp. 1461–1500.

Link, Bruce G., Francis T. Cullen, Elmer Struening, Patrick E. Shrout, and Bruce P. Dohrenwend, "A Modified Labeling Theory Approach to Mental Disorders and Empirical Assessment," *American Sociological Review*, Vol. 54, No. 3, June 1989, pp. 400–423.

Link, Bruce G., Jerrold Mirotznik, and Francis T. Cullen, "The Effectiveness of Stigma Coping Orientations: Can Negative Consequences of Mental Illness Labeling Be Avoided?" *Journal of Health and Social Behavior*, Vol. 32, No. 3, September 1991, pp. 302–320.

Link, Bruce G., and Jo C. Phelan, "Conceptualizing Stigma," *Annual Review of Sociology*, Vol. 27, 2001, pp. 363–385.

Link, Bruce G., Elmer L. Struening, Sheree Neese-todd, Sara Asmussen, and Jo C. Phelan, "On Describing and Seeking to Change the Experience of Stigma," *Psychiatric Rehabilitation Skills*, Vol. 6, No. 2, 2002, pp. 201–231.

Link, Bruce G., Elmer L. Struening, M. Rahav, Jo C. Phelan, and L. Nuttbrock, "On Stigma and Its Consequences: Evidence from a Longitudinal Study of Men with Dual Diagnoses of Mental Illness and Substance Abuse," *Journal of Health and Social Behavior*, Vol. 38, No. 2, June 1997, pp. 177–190.

Link, Bruce G., Lawrence H. Yang, Jo C. Phelan, and Pamela Y. Collins, "Measuring Mental Illness Stigma," *Schizophrenia Bulletin*, Vol. 30, No. 3, 2004, pp. 511–541.

Livingston, J. D., and J. E. Boyd, "Correlates and Consequences of Internalized Stigma for People Living with Mental Illness: A Systematic Review and Meta-Analysis," *Social Science and Medicine*, Vol. 71, No. 12, December 2010, pp. 2150–2161.

Lucksted, Alicia, Amy Drapalski, Christine Calmes, Courtney Forbes, Bruce DeForge, and Jennifer Boyd, "Ending Self-Stigma: Pilot Evaluation of a New Intervention to Reduce Internalized Stigma Among People with Mental Illnesses," *Psychiatric Rehabilitation Journal*, Vol. 35, No. 1, Summer 2011, pp. 51–54.

Ludman, E. J., G. E. Simon, S. Tutty, and M. Von Korff, "A Randomized Trial of Telephone Psychotherapy and Pharmacotherapy for Depression: Continuation and Durability of Effects," *Journal of Consulting and Clinical Psychology*, Vol. 75, No. 2, April 2007, pp. 257–266.

Ludwikowski, Wyndolyn M. A., David Vogel, and Patrick Ian Armstrong, "Attitudes Toward Career Counseling: The Role of Public and Self-Stigma," *Journal of Counseling Psychology*, Vol. 56, No. 3, July 2009, pp. 408–416.

Luoma, J. B., B. S. Kohlenberg, S. C. Hayes, K. Bunting, and A. K. Rye, "Reducing Self-Stigma in Substance Abuse Through Acceptance and Commitment Therapy: Model, Manual Development, and Pilot Outcomes," *Addiction Research and Theory*, Vol. 16, No. 2, 2008, pp. 149–165.

Luty, Jason, Joby Maducolil Easow, and Vania Mendes, "Stigmatised Attitudes Towards the 'Stressed' or 'Ill' Models of Mental Illness," *Psychiatric Bulletin*, Vol. 35, No. 10, 2011, pp. 370–373.

Luty, Jason, Okon Umoh, Mohammed Sessay, and Arghya Sarkhel, "Effectiveness of Changing Minds Campaign Factsheets in Reducing Stigmatised Attitudes Towards Mental Illness," *Psychiatric Bulletin*, Vol. 31, No. 10, 2007, pp. 377–381.

MacInnes, D. L., and M. Lewis, "The Evaluation of a Short Group Programme to Reduce Self-Stigma in People with Serious and Enduring Mental Health Problems," *Journal of Psychiatric and Mental Health Nursing*, Vol. 15, No. 1, January 2008, pp. 59–65.

Macrae, C. Neil, Galen V. Bodenhausen, Alan B. Milne, and Jolanda Jetten, "Out of Mind but Back in Sight: Stereotypes on the Rebound," *Journal of Personality and Social Psychology*, Vol. 67, No. 5, November 1994, pp. 808–817.

Magaña, S. M., J. I. Ramírez García, M. G. Hernández, and R. Cortez, "Psychological Distress Among Latino Family Caregivers of Adults with Schizophrenia: The Roles of Burden and Stigma," *Psychiatric Services*, Vol. 58, No. 3, March 2007, pp. 378–384.

Major, Brenda, and Laurie T. O'Brien, "The Social Psychology of Stigma," *Annual Review of Psychology*, Vol. 56, February 2005, pp. 393–421.

Mann, Caroline E., and Melissa J. Himelein, "Putting the Person Back into Psychopathology: An Intervention to Reduce Mental Illness Stigma in the Classroom," *Social Psychiatry and Psychiatric Epidemiology*, Vol. 43, No. 7, 2008, pp. 545–551.

Manos, Rachel C., Laura C. Rusch, Jonathan W. Kanter, and Lisa M. Clifford, "Depression Self-Stigma as a Mediator of the Relationship Between Depression Severity and Avoidance," *Journal of Social and Clinical Psychology*, Vol. 28, No. 9, 2009, pp. 1128–1143.

Manzo, John F., "On the Sociology and Social Organization of Stigma: Some Ethnomethodological Insights," *Human Studies*, Vol. 27, No. 4, 2004, pp. 401–416.

Markowitz, Fred E., "The Effects of Stigma on the Psychological Well-Being and Life Satisfaction of Persons with Mental Illness," *Journal of Health and Social Behavior*, Vol. 39, No. 4, December 1998, pp. 335–347.

Marmar, Charles R., Mental Health Impact of Afghanistan and Iraq Deployment: Meeting the Challenge of a New Generation of Veterans," *Depression and Anxiety*, Vol. 26, No. 6, June 2009, pp. 493–497.

Martens, L., and J. Addington, The Psychological Well-Being of Family Members of Individuals with Schizophrenia," *Social Psychiatry and Psychiatric Epidemiology*, Vol. 36, No. 3, 2001, pp. 128–133.

Martin, Jack K., Annie Lang, and Sigrun Olafsdottir, "Rethinking Theoretical Approaches to Stigma: A Framework Integrating Normative Influences on Stigma (FINIS)," *Social Science and Medicine*, Vol. 67, No. 3, August 2008, pp. 431–440.

Martin, Jack K., Bernice A. Pescosolido, and Steven A. Tuch, "Of Fear and Loathing: The Role of 'Disturbing Behavior,' Labels, and Causal Attributions in Shaping Public Attitudes Toward People with Mental Illness," *Journal of Health and Social Behavior*, Vol. 41, No. 2, June 2000, pp. 208–223.

Martinez, Andres G., Paul K. Piff, Rodolfo Mendoza-Denton, and Stephen P. Hinshaw, "The Power of a Label: Mental Illness Diagnoses, Ascribed Humanity, and Social Rejection," *Journal of Social and Clinical Psychology*, Vol. 30, No. 1, 2011, pp. 1–23.

Masuda, Akihiko, Page L. Anderson, and Joshua Edmonds, "Help-Seeking Attitudes, Mental Health Stigma, and Self-Concealment Among African American College Students," *Journal of Black Studies*, Vol. 43, No. 7, October 2012, pp. 773–786.

Masuda, Akihiko, Steven C. Hayes, Lindsay B. Fletcher, Paul J. Seignourel, Kara Bunting, Scott A. Herbst, Michael P. Twohig, and Jason Lillis, "Impact of Acceptance and Commitment Therapy Versus Education on Stigma Toward People with Psychological Disorders," *Behaviour Research and Therapy*, Vol. 45, No. 11, November 2007, pp. 2764–2772.

Masuda, Akihiko, and Robert D. Latzman, "Examining Associations Among Factor-Analytically Derived Components of Mental Health Stigma, Distress, and Psychological Flexibility," *Personality and Individual Differences*, Vol. 51, No. 4, September 2011, pp. 435–438.

McCay, Elizabeth, Heather Beanlands, Robert Zipursky, Paul Roy, Molyn Leszcz, Janet Landeen, Kathy Ryan, Gretchen Conrad, Donna Romano, Daphene Francis, Jennifer Hunt, Lucia Costantini, and Eugene Chan, "A Randomised Controlled Trial of a Group Intervention to Reduce Engulfment and Self-Stigmatisation in First Episode Schizophrenia," *Advances in Mental Health*, Vol. 6, No. 3, 2007, pp. 212–220.

McFarling, L., M. D'Angelo, M. Drain, D. A. Gibbs, and K. L. Rae Olmsted, "Stigma as a Barrier to Substance Abuse and Mental Health Treatment," *Military Psychology*, Vol. 23, No. 1, 2011, pp. 1–5.

MCO 1001R.1K—See Commandant of the Marine Corps, 2009a.

MCO 1040.31—See Commandant of the Marine Corps, 2010b.

MCO 1100.75F—*See* Headquarters, Department of the Army; Headquarters, Department of the Navy; Headquarters, Department of the Marine Corps; and Headquarters, Department of the Coast Guard, 2007 (2011).

MCO 1200.17D—See Commandant of the Marine Corps, 2012.

MCO 1306.16E—See Commandant of the Marine Corps, 1986.

MCO 1326.7E—See Commandant of the Marine Corps, 2004.

MCO 1553.2B—See Commandant of the Marine Corps, 2011a.

MCO 1560R.30B—See Commandant of the Marine Corps, 1993a.

MCO 1720.2-See Commandant of the Marine Corps, 2012b.

MCO 1730.6E—See Commandant of the Marine Corps, 2012d.

MCO 1754.11—See Commandant of the Marine Corps, 2012a.

MCO 3120.11—See Commandant of the Marine Corps, 2009b.

MCO 3150.4—See Commandant of the Marine Corps, 2009c.

MCO 3461.1—*See* Headquarters, Department of the Army; Headquarters, Department of the Navy; Headquarters, Department of the Air Force; and Headquarters, Department of the Marine Corps, 1997.

MCO 4950.2—*See* Headquarters, Department of the Army; Headquarters, Department of the Navy; and Headquarters, Department of the Air Force, 1985.

MCO 5100.29B—See Commandant of the Marine Corps, 2011c.

MCO 5110.1D—*See* Headquarters, Department of the Army; Headquarters, Department of the Navy; Headquarters, Department of the Air Force; Headquarters, Department of the Marine Corps; and Defense Logistics Agency, 2006.

MCO 5300.17-See Commandant of the Marine Corps, 2011b.

MCO 5530.16—See Commandant of the Marine Corps, 2011d.

MCO 5580.2B—See Commandant of the Marine Corps, 2008.

MCO 8400.6—See Commandant of the Marine Corps, 2012f.

MCO 12410.24—See Commandant of the Marine Corps, 1997.

MCO 12630.3—See Commandant of the Marine Corps, 1993b.

MCO 12713.8—See Commandant of the Marine Corps, 1990a.

MCO 12771.2—See Commandant of the Marine Corps, 1990b.

MCO P1610.7F—See Commandant of the Marine Corps, 2010b.

MCO P1100.72C—See Headquarters, U.S. Marine Corps, 2004.

MCO P1300.8R—See Commandant of the Marine Corps, 1994.

MCO P1326.6D—See Headquarters, U.S. Marine Corps, 1999.

MCO P1400.32D-See Commandant of the Marine Corps, 2006.

MCO P1610.7F—See Commandant of the Marine Corps, 2010c.

MCO P3500.44A—See Commandant of the Marine Corps, 2005c.

MCO P3500.75—See Commandant of the Marine Corps, 2005a.

MCO P3550.10-See Commandant of the Marine Corps, 2005b.

MCO P12000.11A—See Commandant of the Marine Corps, 2011g.

Mehta, Nisha, Aliya Kassam, Morven Leese, Georgia Butler, and Graham Thornicroft, "Public Attitudes Towards People with Mental Illness in England and Scotland, 1994–2003," *British Journal of Psychiatry*, Vol. 194, No. 3, 2009, pp. 278–284.

Mental Health Advisory Team III, *Operation Iraqi Freedom 04–06*, Office of the Surgeon General, Multinational Force–Iraq, and Office of the Surgeon General, U.S. Army Medical Command, May 29, 2006.

Mental Health Advisory Team IV, *Operation Iraqi Freedom 05–07 Final Report*, Office of the Surgeon General, Multinational Force—Iraq, and Office of the Surgeon General, U.S. Army Medical Command, November 17, 2006.

Mental Health Advisory Team V, *Operation Iraqi Freedom 06–08*, Office of the Surgeon General, Multi-National Force–Iraq, and Office of the Surgeon General, U.S. Army Medical Command, February 14, 2008. As of June 9, 2014: http://www.dtic.mil/dtic/tr/fulltext/u2/a519676.pdf

Mental Health Advisory Team VI, *Operation Iraqi Freedom 07–09*, Office of the Surgeon General, Multinational Force–Iraq, and Office of the Surgeon General, U.S. Army Medical Command, May 8, 2009.

Mesidor, M., V. Gidugu, E. S. Rogers, V. M. Kash-Macdonald, and J. B. Boardman, "A Qualitative Study: Barriers and Facilitators to Health Care Access for Individuals with Psychiatric Disabilities," *Psychiatric Rehabilitation Journal*, Vol. 34, No. 4, Spring 2011, pp. 285–294.

Mickelson, Kristen D., "Perceived Stigma, Social Support, and Depression," *Personality and Social Psychology Bulletin*, Vol. 27, No. 8, August 2001, pp. 1046–1056.

Military Rules of Evidence, Privilege Concerning Mental Examination of the Accused, Rule 302. As of February 4, 2014:

http://www.loc.gov/rr/frd/Military_Law/pdf/MCM-2012.pdf

——, Psychotherapist–Patient Privilege, Rule 513.

Milliken, Charles S., Jennifer L. Auchterlonie, and Charles W. Hoge, "Longitudinal Assessment of Mental Health Problems Among Active and Reserve Component Soldiers Returning from the Iraq War," *Journal of the American Medical Association*, Vol. 298, No. 18, November 14, 2007, pp. 2141–2148.

Mino, Yoshio, Nobufumi Yasuda, Toshihide Tsuda, and Shinji Shimodera, "Effects of a One-Hour Educational Program on Medical Students' Attitudes to Mental Illness," *Psychiatry and Clinical Neurosciences*, Vol. 55, No. 5, 2001, pp. 501–507.

Mirabi, Mohsen, Maxine L. Weinman, Sandra M. Magnetti, and Kathleen N. Keppler, "Professional Attitudes Toward the Chronic Mentally Ill," *Psychiatric Studies*, Vol. 36, No. 4, April 1, 1985, p. 404.

Mittal, Dinesh, Greer Sullivan, Lakshminarayana Chekuri, Elise Allee, and Patrick W. Corrigan, "Empirical Studies of Self-Stigma Reduction Strategies: A Critical Review of the Literature," *Psychiatric Services*, Vol. 63, No. 10, October 2012, pp. 974–981.

Mojtabai, Ramin, "Americans' Attitudes Toward Mental Health Treatment Seeking: 1990–2003," *Psychiatric Services*, Vol. 58, No. 5, May 1, 2007, pp. 642–651.

Mojtabai, Ramin, William W. Eaton, and Pallab K. Maulik, "Pathways to Care: Need, Attitudes, Barriers," in William W. Eaton, ed., *Public Mental Health*, New York: Oxford University Press, 2012, pp. 415–439.

Mojtabai, R., M. Olfson, and D. Mechanic, "Perceived Need and Help-Seeking in Adults with Mood, Anxiety, or Substance Use Disorders," *Archives of General Psychiatry*, Vol. 59, No. 1, January 2002, pp. 77–84.

Momen, Nausheen, Chris P. Strychacz, and Erik Viirre, "Perceived Stigma and Barriers to Mental Health Care in Marines Attending the Combat Operational Stress Control Program," *Military Medicine*, Vol. 171, No. 10, October 2012, pp. 1143–1148.

Monteith, Lindsey L., and Jeremy W. Pettit, "Implicit and Explicit Stigmatizing Attitudes and Stereotypes About Depression," *Journal of Social and Clinical Psychology*, Vol. 30, No. 5, 2011, pp. 484–505.

Morrison, James K., Robert E. Becker, and Christine A. Bourgeois, "Decreasing Adolescents' Fear of Mental Patients by Means of Demythologizing," *Psychological Reports*, Vol. 44, No. 3, June 1979, pp. 855–859.

Mueller, B., C. Nordt, C. Lauber, P. Rueesch, P. C. Meyer, and W. Roessler, "Social Support Modifies Perceived Stigmatization in the First Years of Mental Illness: A Longitudinal Approach," *Social Science and Medicine*, Vol. 62, No. 1, January 2006, pp. 39–49.

Mullen, Michael, "On Total Force Fitness in War and Peace," *Military Medicine*, Vol. 175, No. 8S, August 2010, pp. 1–2.

Muñoz, Manuel, María Sanz, Eloísa Pérez-Santos, and María de los Ángeles Quiroga, "Proposal of a Socio–Cognitive–Behavioral Structural Equation Model of Internalized Stigma in People with Severe and Persistent Mental Illness," *Psychiatry Research*, Vol. 186, No. 2–3, April 30, 2011, pp. 402–408.

Nam, Suk Kyung, Seong In Choi, Ji Hee Lee, Mi Kyoung Lee, A. Reum Kim, and Sang Min Lee, "Psychological Factors in College Students' Attitudes Toward Seeking Professional Psychological Help: A Meta-Analysis," *Professional Psychology: Research and Practice*, Vol. 44, No. 1, February 2013, pp. 37–45.

Nathan, Matthew L., *Defense Health Program*, statement before the Subcommittee on Defense of the Senate Committee on Appropriations, Washington, D.C., March 28, 2012.

National Center for Complementary and Alternative Medicine, "Complementary, Alternative, or Integrative Health: What's in a Name?" last updated May 2013; referenced December 20, 2011.

National Comorbidity Survey, "NCS-R Interview Schedule and Respondent Booklet," undated. As of June 9, 2014:

http://www.hcp.med.harvard.edu/ncs/replication.php

National Guard Bureau, *Absence and Leave Program*, Technician Personnel Regulation 630, August 27, 2010. As of February 24, 2014: http://www.ngbpdc.ngb.army.mil/pubs/TPR/tpr630.pdf

———, *Weapons of Mass Destruction Civil Support Team Management*, National Guard Regulation 500-3, Air National Guard Instruction 10-2503, May 9, 2011. As of February 24, 2014: http://www.ngbpdc.ngb.army.mil/pubs/10/angi10_2503.pdf

National Institute of Mental Health, "If You or Someone You Know Is in Crisis and Needs Immediate Help," undated; referenced September 13, 2013. As of January 31, 2014: http://www.nimh.nih.gov/site-info/

if-you-or-someone-you-know-is-in-crisis-and-needs-immediate-help.shtml

NAVMC 2563—See American Forces Information Service, 1988.

NAVMC 2660—See Headquarters, U.S. Marine Corps, 1977.

NAVMC 2768—See Headquarters, U.S. Marine Corps, 1991.

NAVMC 3500.18B (Prelim) Part 1-See Commandant of the Marine Corps, 2012c.

NAVMC 3500.59A Part 1-See Commandant of the Marine Corps, 2011f.

NAVMC 3500.59A Part 2-See Commandant of the Marine Corps, 2011f.

NAVMC DIR 1500.58—See Headquarters, U.S. Marine Corps, 2006.

NAVMED P-117, change 132-See Chief, Bureau of Medicine and Surgery, 2008.

NAVMED P-117, change 140—See Chief, Bureau of Medicine and Surgery, 2012.

Naylor, Paul B., Helen A. Cowie, Stephen J. Walters, Lorenzo Talamelli, and Judith Dawkins, "Impact of a Mental Health Teaching Programme on Adolescents," *British Journal of Psychiatry*, Vol. 194, No. 4, 2009, pp. 365–370.

Newell, Carol, Kimberly Whittam, and Zannette Uriell, "2010 Behavioral Health Quick Poll," Millington, Tenn.: Navy Personnel Research, Studies, and Technology, Bureau of Naval Personnel, June 2010. As of June 14, 2014:

https://quickpolling.nprst.navy.mil/execsum_2010BehavioralHealth_Jun10.pdf

———, "2011 Behavioral Health Quick Poll," Millington, Tenn.: Navy Personnel Research, Studies, and Technology, Bureau of Naval Personnel, June 2011. As of June 14, 2014: https://quickpolling.nprst.navy.mil/execsum_2011BHQP_Jun11.pdf

Ng, T. P., A. Z. Jin, R. Ho, H. C. Chua, C. S. Fones, and L. Lim, "Health Beliefs and Help Seeking for Depressive and Anxiety Disorders Among Urban Singaporean Adults," *Psychiatric Services*, Vol. 59, No. 1, January 2008, pp. 105–108.

NGR 385-10—See Department of the Army and Department of the Air Force, 2008.

NGR 500-3—See National Guard Bureau, 2011.

NGR 500-4—See Department of the Army and Department of the Air Force, 2009b.

NGR 600-200-See Department of the Army and Department of the Air Force, 2009a.

NGR 614-1—See Department of the Army and Department of the Air Force, 2010.

Noar, S. M., "A 10-Year Retrospective of Research in Health Mass Media Campaigns: Where Do We Go from Here?" *Journal of Health Communication*, Vol. 11, No. 1, 2006, pp. 21–42.

Office of the Chief of Naval Operations, *Health and Wellness Promotion Program*, Office of the Chief of Naval Operations Instruction 6100.2A, March 15, 2007a. As of February 9, 2014: http://www.med.navy.mil/sites/nmcphc/Documents/policy-and-instruction/Governance-and-National-Health-Initiatives/OPNAVINST_6100_2A.pdf

, "Suitability Screening for Overseas and Remote Duty Assignment," Office of the Chief of Naval Operations Instruction 1300.14D, April 9, 2007b. As of February 9, 2014: http://doni.daps.dla.mil/Directives/01000%20Military%20Personnel%20Support/ 01-300%20Assignment%20and%20Distribution%20Services/1300.14D.pdf

———, Navy Physical Security and Law Enforcement Program, Office of the Chief of Naval Operations Instruction 5530.14E, January 28, 2009a.

———, Security Regulations for Offices Under the Cognizance of the Chief of Naval Operations, Office of the Chief of Naval Operations Instruction 5510.60M, March 23, 2009b.

——, Naval Aviation Safety Program, Office of the Chief of Naval Operations Instruction 3750.6R CH-4, April 8, 2009c. As of February 9, 2014: http://doni.daps.dla.mil/Directives/03000%20Naval%20Operations%20and%20Readiness/ 03-700%20Flight%20and%20Air%20Space%20Support%20Services/3750.6R.pdf ———, "Changing or Removing Primary Navy Enlisted Classification Codes for Nuclear Propulsion Plant Operators," Office of the Chief of Naval Operations Instruction 1220.1D, June 3, 2009d. As of February 9, 2014:

http://doni.daps.dla.mil/Directives/01000%20Military%20Personnel%20Support/01-200%20Personnel%20Classification%20and%20Designation/1220.1D.pdf

, "Returning Warrior Workshop Program," Office of the Chief of Naval Operations Instruction 1754.7, June 19, 2009e. As of February 9, 2014: http://doni.daps.dla.mil/Directives/01000%20Military%20Personnel%20Support/ 01-700%20Morale,%20Community%20and%20Religious%20Services/1754.7.pdf

———, "Suicide Prevention Program," Office of the Chief of Naval Operations Instruction 1720.4A, August 4, 2009f. As of February 9, 2014: http://doni.daps.dla.mil/Directives/01000%20Military%20Personnel%20Support/ 01-700%20Morale,%20Community%20and%20Religious%20Services/1720.4A.pdf

, Small Arms Training and Qualification, Office of the Chief of Naval Operations Instruction 3591.1F, August 12, 2009g. As of February 9, 2014: http://doni.daps.dla.mil/Directives/03000%20Naval%20Operations%20and%20Readiness/ 03-500%20Training%20and%20Readiness%20Services/3591.1F.pdf

———, Naval Air Training and Operating Procedures Standardization General Flight and Operating Instructions, Office of the Chief of Naval Operations Instruction 3710.7U, November 23, 2009h. As of February 9, 2014:

http://doni.daps.dla.mil/Directives/03000%20Naval%20Operations%20and%20Readiness/ 03-700%20Flight%20and%20Air%20Space%20Support%20Services/3710.7U.pdf

———, *Policy and Procedures for Conducting High-Risk Training*, Office of the Chief of Naval Operations Instruction 1500.75B, March 4, 2010. As of February 9, 2014: http://doni.daps.dla.mil/Directives/01000%20Military%20Personnel%20Support/ 01-500%20Military%20Training%20and%20Education%20Services/1500.75B.pdf

—_____, Minimum Security Standards for Safeguarding Biological Select Agents and Toxins, Office of the Chief of Naval Operations Instruction 5530.16A, May 11, 2011.

, "Physical Requirements for Non-Submarine Personnel Embarked on Submarines," Office of the Chief of Naval Operations Instruction 6420.1A, April 26, 2012a. As of February 9, 2014: http://doni.daps.dla.mil/Directives/06000%20Medical%20and%20Dental%20Services/ 06-400%20Special%20Medical%20Fields%20Support/6420.1A.pdf

, Regulations Governing the Issuance of Entry Authorizations for Naval Defensive Sea Areas, Naval Airspace Reservations, and Areas Under Navy Administration, Office of the Chief of Naval Operations Instruction 5500.11F, July 17, 2012b. As of February 9, 2014: http://doni.daps.dla.mil/Directives/ 05000%20General%20Management%20Security%20and%20Safety%20Services/

05-500%20Security%20Services/5500.11F.pdf

—_____, Child and Youth Program, Office of the Chief of Naval Operations Instruction 1700.9E CH-1, September 24, 2012c. As of February 9, 2014: http://doni.daps.dla.mil/Directives/01000%20Military%20Personnel%20Support/ 01-700%20Morale,%20Community%20and%20Religious%20Services/1700.9E%20CH-1.pdf

Office of Equal Opportunity and Diversity, Washington Headquarters Service, "Equal Employment Opportunity (EEO) and Diversity Programs," Administrative Instruction 31, August 19, 2013a. As of February 7, 2014:

http://www.dtic.mil/whs/directives/corres/pdf/a031p.pdf

——, *Processing Complaints of Discrimination*, Administrative Instruction 9, November 6, 2013b. As of February 7, 2014:

http://www.dtic.mil/whs/directives/corres/pdf/a009p.pdf

Office of the Deputy Under Secretary of Defense for Policy, *Personnel Security Program*, Department of Defense Regulation 5200.2-R, January 1987; administrative reissuance incorporating through change 3 February 23, 1996. As of February 7, 2014: http://www.dtic.mil/whs/directives/corres/pdf/520002r.pdf

Office of the Director for Administration and Management, Office of the Secretary of Defense, *Department of Defense Privacy Program*, Department of Defense Regulation 5400.11-R, May 14, 2007. As of February 7, 2014: http://www.dtic.mil/whs/directives/corres/pdf/540011r.pdf

Office of the Surgeon General, U.S. Army Medical Command, *Joint Mental Health Advisory Team 7* (*J-MHAT 7*) Operation Enduring Freedom 2010 Afghanistan, Fort Sam Houston, Texas, February 22, 2011. As of January 31, 2014:

http://www.dtic.mil/cgi-bin/GetTRDoc?AD=ADA543997

Office of the Under Secretary of Defense for Acquisition, Technology, and Logistics, *Nuclear Weapons Personnel Reliability Program (PRP) Regulation*, Department of Defense Regulation 5210.42-R, June 2006; incorporating change 1 November 10, 2009. As of February 7, 2014:

http://www.dtic.mil/whs/directives/corres/pdf/521042r.pdf

Office of the Under Secretary of Defense for Personnel and Readiness, Total Force Planning and Requirements Office, *Defense Manpower Requirements Report: Fiscal Year 2012*, April 2012. As of February 6, 2014:

http://prhome.defense.gov/Portals/52/Documents/RFM/TFPRQ/Docs/FY12%20Defense%20 Manpower%20Requirements%20Report%20(DMRR)%20--%20Apr%202012.pdf

Okumura, Y., and S. Sakamoto, "Depression Treatment Preferences Among Japanese Undergraduates: Using Conjoint Analysis," *International Journal of Social Psychiatry*, Vol. 58, No. 2, March 2012, pp. 195–203.

Oliver, Michael, *The Politics of Disablement: A Sociological Approach*, New York: St. Martin's Press, 1990.

O'Mahony, Joyce Maureen, Tam Truong Donnelly, Shelley Raffin Bouchal, and David Este, "Cultural Background and Socioeconomic Influence of Immigrant and Refugee Women Coping with Postpartum Depression," *Journal of Immigrant and Minority Health*, Vol. 8, No. 1, January 2006.

Operation Iraqi Freedom Mental Health Advisory Team, *Report*, U.S. Army Surgeon General and Headquarters, Department of the Army, G-1, December 16, 2003.

Operation Iraqi Freedom Mental Health Advisory Team II, *Report*, U.S. Army Surgeon General, January 30, 2005.

OPNAVINST 1100.4C CH-2—*See* Headquarters, Department of the Army; Headquarters, Department of the Navy; Headquarters, Department of the Air Force; Headquarters, Department of the Marine Corps; and Headquarters, Department of the Coast Guard, 2007 (2011).

OPNAVINST 1220.1D—See Office of the Chief of Naval Operations Instruction, 2009d.

OPNAVINST 1300.14D—See Office of the Chief of Naval Operations, 2007b.

OPNAVINST 1500.75B—See Office of the Chief of Naval Operations, 2010.

OPNAVINST 1700.9E—See Office of the Chief of Naval Operations, 2012c.

OPNAVINST 1720.4A—See Office of the Chief of Naval Operations, 2009f.

OPNAVINST 1754.7—See Office of the Chief of Naval Operations, 2009e.

OPNAVINST 3591.1F—See Office of the Chief of Naval Operations, 2009g.

OPNAVINST 3710.7U—See Office of the Chief of Naval Operations, 2009h.

OPNAVINST 3750.6R—See Office of the Chief of Naval Operations, 2009c.

OPNAVINST 5500.11F—See Office of the Chief of Naval Operations, 2012b.

OPNAVINST 5510.60M—See Office of the Chief of Naval Operations, 2009b.

OPNAVINST 5530.14E—See Office of the Chief of Naval Operations, 2009a.

OPNAVINST 5530.16A—See Office of the Chief of Naval Operations, 2011.

OPNAVINST 6100.2A—See Office of the Chief of Naval Operations, 2007a.

OPNAVINST 6420.1A—See Office of the Chief of Naval Operations, 2012a.

OPNAVINST 11200.5D—*See* Headquarters, Department of the Army; Headquarters, Department of the Navy; Headquarters, Department of the Air Force; Headquarters, Department of the Marine Corps; and Defense Logistics Agency, 2006.

O'Reilly, C. L., J. S. Bell, P. J. Kelly, and T. F. Chen, "Impact of Mental Health First Aid Training on Pharmacy Students' Knowledge, Attitudes and Self-Reported Behaviour: A Controlled Trial," *Australian and New Zealand Journal of Psychiatry*, Vol. 45, No. 7, July 2011, pp. 549–557.

Ouimette, Paige, Dawne Vogt, Michael Wade, Vanessa Tirone, Mark A. Greenbaum, Rachel Kimerling, Charlene Laffaye, Julie E. Fitt, and Craig S. Rosen, "Perceived Barriers to Care Among Veterans Health Administration Patients with Posttraumatic Stress Disorder," *Psychological Services*, Vol. 8, No. 3, August 2011, pp. 212–223.

PACAFDIR 90-254—See Commander, Pacific Air Forces, 2005 (2009).

Pachankis, John E., "The Psychological Implications of Concealing a Stigma: A Cognitive-Affective-Behavioral Model," *Psychological Bulletin*, Vol. 133, No. 2, March 2007, pp. 328–345.

Paluck, Elizabeth Levy, and Donald P. Green, "Prejudice Reduction: What Works? A Review and Assessment of Research and Practice," *Annual Review of Psychology*, Vol. 60, 2009, pp. 339–367.

Parcesepe, Angela M., and Leopoldo J. Cabassa, "Public Stigma of Mental Illness in the United States: A Systematic Literature Review," *Administration and Policy in Mental Health*, Vol. 40, No. 5, July 26, 2012, pp. 384–399.

Park, S., M. J. Cho, J. N. Bae, S. M. Chang, H. J. Jeon, B. J. Hahm, J. W. Son, S. G. Kim, A. Bae, and J. P. Hong, "Comparison of Treated and Untreated Major Depressive Disorder in a Nationwide Sample of Korean Adults," *Community Mental Health Journal*, Vol. 48, No. 3, June 2012, pp. 363–371.

Parle, Siobhan, "How Does Stigma Affect People with Mental Illness?" *Nursing Times*, Vol. 108, No. 28, July 10–16, 2012, pp. 12–14.

Parton, D., "Living with Stigma Is Still the Common Experience for Mental Health Service Users," *Mental Health Today*, July–August 2011, p. 5.

Paykel, E. S., D. Hart, and R. G. Priest, "Changes in Public Attitudes to Depression During the Defeat Depression Campaign," *British Journal of Psychiatry*, Vol. 173, 1998, pp. 519–522.

Penn, David L., and Shannon Couture, "Interpersonal Contact and the Stigma of Mental Illness: A Review of the Literature," *Journal of Mental Health*, Vol. 12, No. 3, June 2003, pp. 291–305.

Penn, David L., Kim Guynan, Tamara Daily, William D. Spaulding, Calvin P. Garbin, and Mary Sullivan, "Dispelling the Stigma of Schizophrenia: What Sort of Information Is Best?" *Schizophrenia Bulletin*, Vol. 20, No. 3, 1994, pp. 567–578.

Penn, David L., Samantha Kommana, Maureen Mansfield, and Bruce G. Link, "Dispelling the Stigma of Schizophrenia: II. The Impact of Information on Dangerousness," *Schizophrenia Bulletin*, Vol. 25, No. 3, 1999, pp. 437–446.

Penn, David L., and Amy Nowlin-Drummond, "Politically Correct Labels and Schizophrenia: A Rose by Any Other Name?" *Schizophrenia Bulletin*, Vol. 27, No. 2, 2001, pp. 197–203.

Perlick, Deborah A., Ann H. Nelson, Kate Mattias, James Selzer, Carla Kalvin, Charles H. Wilber, Brittney Huntington, Caroline S. Holman, and Patrick W. Corrigan, "In Our Own Voice—Family Companion: Reducing Self-Stigma of Family Members of Persons with Serious Mental Illness," *Psychiatric Services*, Vol. 62, No. 12, December 2011, pp. 1456–1462.

Perlick, Deborah A., R. A. Rosenheck, J. F. Clarkin, J. A. Sirey, J. Salahi, E. L. Struening, and B. G. Link, "Stigma as a Barrier to Recovery: Adverse Effects of Perceived Stigma on Social Adaptation of Persons Diagnosed with Bipolar Affective Disorder," *Psychiatric Services*, Vol. 52, No. 12, December 2001, pp. 1627–1632.

Perry, Brea L., "The Labeling Paradox: Stigma, the Sick Role, and Social Networks in Mental Illness," *Journal of Health and Social Behavior*, Vol. 52, No. 4, December 2011, pp. 460–477.

Pescosolido, Bernice A., "Of Pride and Prejudice: The Role of Sociology and Social Networks in Integrating the Health Sciences," *Journal of Health and Social Behavior*, Vol. 47, No. 3, September 2006, pp. 189–208.

, "A Fresh Look at Recovery," *Psychiatric Services*, Vol. 59, No. 1, January 1, 2008, p. 5.

, "The Public Stigma of Mental Illness: What Do We Think; What Do We Know; What Can We Prove?" *Journal of Health and Social Behavior*, Vol. 54, No. 1, March 2013, pp. 1–21.

Pescosolido, Bernice A., Jack K. Martin, J. Scott Long, Tait R. Medina, Jo C. Phelan, and Bruce G. Link, "A Disease Like Any Other? A Decade of Change in Public Reactions to Schizophrenia, Depression, and Alcohol Dependence," *American Journal of Psychiatry*, Vol. 167, No. 11, November 1, 2010, pp. 1321–1330.

Pescosolido, Bernice A., Tait R. Medina, Jack K. Martin, and J. Scott Long, "The 'Backbone' of Stigma: Identifying the Global Core of Public Prejudice Associated with Mental Illness," *American Journal of Public Health*, Vol. 103, No. 5, May 2013, pp. 853–860.

Pettigrew, Thomas F., and Linda R. Tropp, "A Meta-Analytic Test of Intergroup Contact Theory," *Journal of Personality and Social Psychology*, Vol. 90, No. 5, May 2006, pp. 751–783.

Phelan, Jo C., "Geneticization of Deviant Behavior and Consequences for Stigma: The Case of Mental Illness," *Journal of Health and Social Behavior*, Vol. 46, No. 4, December 2005, pp. 307–322.

Phelan, Jo C., Evelyn J. Bromet, and Bruce G. Link, "Psychiatric Illness and Family Stigma," *Schizophrenia Bulletin*, Vol. 24, No. 1, 1998, pp. 115–126.

Phelan, Jo C., and Bruce G. Link, "Fear of People with Mental Illnesses: The Role of Personal and Impersonal Contact and Exposure to Threat or Harm," *Journal of Health and Social Behavior*, Vol. 45, No. 1, March 2004, pp. 68–80.

, "The Stigma of Mental Illness and Psychiatric Disorders," in Clifton D. Bryant, ed., *The Routledge Handbook of Deviant Behavior*, New York: Routledge, 2011, pp. 572–579.

Phelan, Jo C., Bruce G. Link, Ann Stueve, and Bernice A. Pescosolido, "Public Conceptions of Mental Illness in 1950 and 1996: What Is Mental Illness and Is It to Be Feared?" *Journal of Health and Social Behavior*, Vol. 41, No. 2, June 2000, pp. 188–207.

Phelan, S. M., J. M. Griffin, W. L. Hellerstedt, N. A. Sayer, A. C. Jensen, D. J. Burgess, and M. van Ryn, "Perceived Stigma, Strain, and Mental Health Among Caregivers of Veterans with Traumatic Brain Injury," *Disability and Health Journal*, Vol. 4, No. 3, July 2011, pp. 177–184.

Phillips, Michael R., Veronica Pearson, Feifei Li, Minjie Xu, and Lawrence Yang, "Stigma and Expressed Emotion: A Study of People with Schizophrenia and Their Family Members in China," *British Journal of Psychiatry*, Vol. 181, No. 6, 2002, pp. 488–493.

Pietrzak, R. H., D. C. Johnson, M. B. Goldstein, J. C. Malley, and S. M. Southwick, "Perceived Stigma and Barriers to Mental Health Care Utilization Among OEF-OIF Veterans," *Psychiatric Services*, Vol. 60, No. 8, August 2009, pp. 1118–1122.

Pincus, Fred L., and Howard J. Ehrlich, *Race and Ethnic Conflict: Contending Views on Prejudice, Discrimination, and Ethnoviolence*, Boulder, Colo.: Westview Press, 1999.

Pinfold, V., P. Huxley, G. Thornicroft, P. Farmer, H. Toulmin, and T. Graham, "Reducing Psychiatric Stigma and Discrimination: Evaluating an Educational Intervention with the Police Force in England," *Social Psychiatry and Psychiatric Epidemiology*, Vol. 38, No. 6, June 2003, pp. 337–344.

Pinfold, V., G. Thornicroft, P. Huxley, and P. Farmer, "Active Ingredients in Anti-Stigma Programmes in Mental Health," *International Review of Psychiatry*, Vol. 17, No. 2, April 2005, pp. 123–131.

Pinto-Foltz, Melissa D., M. Cynthia Logsdon, and John A. Myers, "Feasibility, Acceptability, and Initial Efficacy of a Knowledge-Contact Program to Reduce Mental Illness Stigma and Improve Mental Health Literacy in Adolescents," *Social Science and Medicine*, Vol. 72, No. 1, June 2011, pp. 2011–2209.

Pittman, James O. E., Sunghwan Noh, and Daniel Coleman, "Evaluating the Effectiveness of a Consumer Delivered Anti-Stigma Program: Replication with Graduate-Level Helping Professionals," *Psychiatric Rehabilitation Journal*, Vol. 33, No. 3, January 2010, pp. 236–238.

Pope, William Stuart, "Another Face of Health Care Disparity: Stigma of Mental Illness," *Journal of Psychosocial Nursing and Mental Health Services*, Vol. 49, No. 9, September 2011, pp. 27–31.

Post, E. P., M. Metzger, P. Dumas, and L. Lehmann, "Integrating Mental Health into Primary Care Within the Veterans Health Administration," *Families, Systems, and Health*, Vol. 28, No. 2, June 2010, pp. 83–90.

"Post-Traumatic Stress Disorders; U.S. Army Special Forces Operators Reluctant to Seek Treatment for Posttraumatic Stress Disorder," *Mental Health Weekly Digest*, 2009, p. 186.

Price, Treloar A., *Stigma Threat and Psychological Help-Seeking Attitudes in Military Personnel*, Regent University, doctoral thesis, 2011.

Principal Deputy Under Secretary of Defense for Personnel and Readiness, *Department of Defense* (*DoD*) Youth Programs (YPs), Department of Defense Instruction 6060.4, August 23, 2004. As of February 7, 2014:

http://www.dtic.mil/whs/directives/corres/pdf/606004p.pdf

Prior, Lindsay, Fiona Wood, Glyn Lewis, and Roisin Pill, "Stigma Revisited, Disclosure of Emotional Problems in Primary Care Consultations in Wales," *Social Science and Medicine*, Vol. 56, No. 10, May 2003, pp. 2191–2200.

Prochaska, J. O., and C. C. DiClemente, "The Transtheoretical Approach," in J. C. Norcross and M. R. Goldfried, eds., *Handbook of Psychotherapy Integration*, 2nd ed., New York: Oxford University Press, 2005, pp. 147–171.

Public Law 94-329, Arms Control Export Act, June 30, 1976.

Public Law 101-336, Americans with Disabilities Act of 1990, July 26, 1990.

Public Law 103-3, Family and Medical Leave Act of 1993, February 5, 1993.

Public Law 104-191, Health Insurance Portability and Accountability Act of 1996, August 21, 1996.

Public Law 109-163, National Defense Authorization Act for Fiscal Year 2006, January 6, 2006. As of February 6, 2014:

http://www.dod.mil/dodgc/olc/docs/PL109-163.pdf

Public Law 110-417, Duncan Hunter National Defense Authorization Act for Fiscal Year 2009, October 14, 2008.

Quinn, N., A. Shulman, L. Knifton, and P. Byrne, "The Impact of a National Mental Health Arts and Film Festival on Stigma and Recovery," *Acta Psychiatrica Scandinavica*, Vol. 123, No. 1, January 2011, pp. 71–81.

Rae Olmsted, Kristine L., Janice M. Brown, J. Russ Vandermaas-Peeler, Stephen J. Tueller, Ruby E. Johnson, and Deborah A. Gibbs, "Mental Health and Substance Abuse Treatment Stigma Among Soldiers," *Military Psychology*, Vol. 23, No. 1, January 2011, pp. 52–64.

Raguram, R., M. G. Weiss, S. M. Channabasavanna, and G. M. Devins, "Stigma, Depression, and Somatization in South India," *American Journal of Psychiatry*, Vol. 153, No. 8, August 1996, pp. 1043–1049.

Rahman, A., M. H. Mubbashar, R. Gater, and D. Goldberg, "Randomised Trial of Impact of School Mental-Health Programme in Rural Rawalpindi, Pakistan," *Lancet*, Vol. 352, No. 9133, September 26, 1998, pp. 1022–1025.

Ramchand, Rajeev, Joie Acosta, Rachel M. Burns, Lisa H. Jaycox, and Christopher G. Pernin, *The War Within: Preventing Suicide in the U.S. Military*, Santa Monica, Calif.: RAND Corporation, MG-953-OSD, 2011. As of June 9, 2014:

http://www.rand.org/pubs/monographs/MG953.html

Randolph, Whitney, and K. Viswanath, "Lessons Learned from Public Health Mass Media Campaigns: Marketing Health in a Crowded Media World," *Annual Review of Public Health*, Vol. 25, 2004, pp. 419–437.

Real Warriors Campaign, undated home page; referenced September 13, 2013. As of February 3, 2014:

http://www.realwarriors.net

Reavley, Nicola J., and Anthony F. Jorm, "Recognition of Mental Disorders and Beliefs About Treatment and Outcome: Findings from an Australian National Survey of Mental Health Literacy and Stigma," *Australian and New Zealand Journal of Psychiatry*, Vol. 45, No. 11, November 2011a, pp. 947–956.

———, "Young People's Stigmatizing Attitudes Towards People with Mental Disorders: Findings from an Australian National Survey," *Australian and New Zealand Journal of Psychiatry*, Vol. 45, No. 12, December 2011b, pp. 1033–1039.

———, "Public Recognition of Mental Disorders and Beliefs About Treatment: Changes in Australia over 16 Years," *British Journal of Psychiatry*, Vol. 200, No. 5, May 2012a, pp. 419–425.

, "Stigmatising Attitudes Towards People with Mental Disorders: Changes in Australia over 8 Years," *Psychiatry Research*, Vol. 197, No. 3, May 30, 2012b, pp. 302–306.

Regier, D. A., W. E. Narrow, D. S. Rae, R. W. Manderscheid, B. Z. Locke, and F. K. Goodwin, "The De Facto US Mental and Addictive Disorders Service System: Epidemiologic Catchment Area Prospective 1-Year Prevalence Rates of Disorders and Services," *Archives of General Psychiatry*, Vol. 50, No. 2, February 1993, pp. 85–94.

Resnick, Sandra G., and Robert A. Rosenheck, "Who Attends Vet-to-Vet? Predictors of Attendance in Mental Health Mutual Support," *Psychiatric Rehabilitation Journal*, Vol. 33, No. 4, 2010, pp. 262–268.

Rickwood, Debra, Sarah Cavanagh, Leigh Curtis, and Rebecca Sakrouge, "Educating Young People About Mental Health and Mental Illness: Evaluating a School-Based Programme," *International Journal of Mental Health Promotion*, Vol. 6, No. 4, 2004, pp. 23–32.

Rickwood, Debra, Frank P. Deane, Coralie J. Wilson, and Joseph Ciarrochi, "Young People's Help-Seeking for Mental Health Problems," *Australian e-Journal for the Advancement of Mental Health*, Vol. 4, No. 3, 2005.

Ritsher, Jennifer Boyd, and Jo C. Phelan, "Internalized Stigma Predicts Erosion of Morale Among Psychiatric Outpatients," *Psychiatry Research*, Vol. 129, No. 3, December 30, 2004, pp. 257–265.

Rizzo, Albert, Thomas D. Parsons, Belinda Lange, Patrick Kenny, John G. Buckwalter, Barbara Rothbaum, JoAnn Difede, John Frazier, Brad Newman, Josh Williams, and Greg Reger, "Virtual Reality Goes to War: A Brief Review of the Future of Military Behavioral Healthcare," *Journal of Clinical Psychology in Medical Settings*, Vol. 18, No. 2, June 2011, pp. 176–187.

Roberts, Glenn, John Somers, Jocelyn Dawe, Rowena Passy, Carly Mays, Graham Carr, David Shiers, and Jo Smith, "On the Edge: A Drama-Based Mental Health Education Programme on Early Psychosis for Schools," *Early Intervention in Psychiatry*, Vol. 1, No. 2, May 2007, pp. 168–176.

Rochlen, Aaron B., Jonathan J. Mohr, and Byron K. Hargrove, "Development of the Attitudes Toward Career Counseling Scale," *Journal of Counseling Psychology*, Vol. 46, No. 2, March 1999, pp. 196–206.

Rosen, Craig S., Mark A. Greenbaum, Julie E. Fitt, Charlene Laffaye, Virginia A. Norris, and Rachel Kimerling, "Stigma, Help-Seeking Attitudes, and Use of Psychotherapy in Veterans with Diagnoses of Posttraumatic Stress Disorder," *Journal of Nervous and Mental Disease*, Vol. 199, No. 11, November 2011, pp. 879–885.

Rosenfield, Sarah, "Labeling Mental Illness: The Effects of Received Services and Perceived Stigma on Life Satisfaction," *American Sociological Review*, Vol. 62, No. 4, August 1997, pp. 660–672.

Rukavina, T. V., A. Nawka, O. Brborović, N. Jovanović, M. R. Kuzman, L. Nawková, B. Bednárová, S. Zuchová, M. Hrodková, and Z. Lattová, "Development of the PICMIN (Picture of Mental Illness in Newspapers): Instrument to Assess Mental Illness Stigma in Print Media," *Social Psychiatry and Psychiatric Epidemiology*, Vol. 47, No. 7, July 2012, pp. 1131–1144.

Rule of Courts-Martial 706, Inquiry into the Mental Capacity or Mental Responsibility of the Accused. As of February 4, 2014: http://www.loc.gov/rr/frd/Military_Law/pdf/MCM-2012.pdf

Rusch, Laura C., Jonathan W. Kanter, Alisa F. Angelone, and Robyn C. Ridley, "The Impact of *In Our Own Voice* on Stigma," *American Journal of Psychiatric Rehabilitation*, Vol. 11, No. 4, 2008, pp. 373–389.

Rusch, Laura C., Jonathan W. Kanter, Michael J. Brondino, Cristal E. Weeks, and William M. Bowe, "Biomedical Stigma Reduction Programs Produce Negative but Transient Effects on a Depressed Low-Income Community Sample," *Journal of Social and Clinical Psychology*, Vol. 29, No. 9, 2010, pp. 1020–1030.

Rüsch, Nicolas, Matthias C. Angermeyer, and Patrick W. Corrigan, "Mental Illness Stigma: Concepts, Consequences, and Initiatives to Reduce Stigma," *European Psychiatry*, Vol. 20, No. 8, December 2005, pp. 529–539.

Sacca, Rose, and Christopher Ryan, "Relationships Between Interpersonal Contact as a Volunteer Companion and Stigma," *Australasian Psychiatry*, Vol. 19, No. 5, October 2011, pp. 439–443.

Sadow, Dolly, and Marie Ryder, "Reducing Stigmatizing Attitudes Held by Future Health Professionals: The Person Is the Message," *Psychological Services*, Vol. 5, No. 4, November 2008, pp. 362–372.

Safe Schools Coalition, "Suicide," updated June 27, 2013; referenced September 13, 2013. As of February 3, 2014:

http://www.safeschoolscoalition.org/RG-suicide.html

SAMHSA—See Substance Abuse and Mental Health Services Administration.

Sansone, R. A., V. G. Matheson, G. A. Gaither, and N. Logan, "Concerns About Career Stigma by Military Parents of Children with Psychiatric Illness," *Military Medicine*, Vol. 173, No. 2, February 2008, pp. 134–137.

Sansone, R. A., and L. A. Sansone, "Responses of Mental Health Clinicians to Patients with Borderline Personality Disorder," *Innovations in Clinical Neuroscience*, Vol. 10, No. 5–6, May 2013, pp. 39–43.

Sartorius, Norman, "Short-Lived Campaigns Are Not Enough," *Nature*, Vol. 468, No. 7321, November 11, 2010, pp. 163–165.

Savoca, E., and R. Rosenheck, "The Civilian Labor Market Experiences of Vietnam-Era Veterans: The Influence of Psychiatric Disorders," *Journal of Mental Health Policy and Economics*, Vol. 3, No. 4, December 1, 2000, pp. 199–207.

Sayce, Liz, "Stigma, Discrimination and Social Exclusion: What's in a Word?" *Journal of Mental Health*, Vol. 7, No. 4, 1998, pp. 331–343.

Schell, Terry L., and Grant N. Marshall, "Survey of Individuals Previously Deployed for OEF/ OIF," in Terri Tanielian and Lisa H. Jaycox, eds., *Invisible Wounds of War: Psychological and Cognitive Injuries, Their Consequences, and Services to Assist Recovery*, Santa Monica, Calif.: RAND Corporation, MG-720-CCF, 2008, pp. 87–116. As of February 8, 2013: http://www.rand.org/pubs/monographs/MG720.html

Schmetzer, Alan D., and Joan E. Lafuze, "Overcoming Stigma: Involving Families in Medical Student and Psychiatric Residency Education," *Academic Psychiatry*, Vol. 32, No. 2, March 2008, pp. 127–131.

Schneider, Justine, Chris Beeley, and Julie Repper, "Campaign Appears to Influence Subjective Experience of Stigma," *Journal of Mental Health*, Vol. 20, No. 1, February 2011, pp. 89–97.

Schnittker, Jason, "An Uncertain Revolution: Why the Rise of a Genetic Model of Mental Illness Has Not Increased Tolerance," *Social Science and Medicine*, Vol. 67, No. 9, November 2008, pp. 1370–1381.

Schnurr, P. P., M. J. Friedman, C. C. Engel, E. B. Foa, M. T. Shea, B. K. Chow, P. A. Resick, V. Thurston, S. M. Orsillo, R. Haug, C. Turner, and N. Bernardy, "Cognitive Behavioral Therapy for Posttraumatic Stress Disorder in Women: A Randomized Controlled Trial," *Journal of the American Medical Association*, Vol. 297, No. 8, February 28, 2007, pp. 820–830.

Schomerus, Georg, and M. C. Angermeyer, "Stigma and Its Impact on Help-Seeking for Mental Disorders: What Do We Know?" *Epidemiologia e psichiatria sociale*, Vol. 17, No. 1, January–March 2008, pp. 31–37.

Schomerus, Georg, Charlotte Auer, Dieter Rhode, Melanie Luppa, Harald J. Freyberger, and Silke Schmidt, "Personal Stigma, Problem Appraisal and Perceived Need for Professional Help in Currently Untreated Depressed Persons," *Journal of Affective Disorders*, Vol. 139, No. 1, June 2012, pp. 94–97.

Schomerus, Georg, H. Matschinger, and M. C. Angermeyer, "The Stigma of Psychiatric Treatment and Help-Seeking Intentions for Depression," *European Archives of Psychiatry and Clinical Neuroscience*, Vol. 259, No. 5, August 2009, pp. 298–306.

Schomerus, Georg, C. Schwahn, A. Holtzinger, Patrick W. Corrigan, H. J. Grabe, M. G. Carta, and M. C. Angermeyer, "Evolution of Public Attitudes About Mental Illness: A Systematic Review and Meta-Analysis," *Acta Psychiatrica Scandinavica*, Vol. 125, No. 6, June 2012, pp. 440–452.

Schulze, Beate, "Mental-Health Stigma: Expanding the Focus, Joining Forces," *Lancet*, Vol. 373, No. 9661, January 31, 2009, pp. 362–363.

Schulze, Beate, M. Richter-Werling, H. Matschinger, and M. C. Angermeyer, "Crazy? So What! Effects of a School Project on Students' Attitudes Towards People with Schizophrenia," *Acta Psychiatrica Scandinavica*, Vol. 107, No. 2, February 2003, pp. 142–150.

Seal, K. H., T. J. Metzler, K. S. Gima, D. Bertenthal, S. Maguen, and C. R. Marmar, Trends and Risk Factors for Mental Health Diagnoses Among Iraq and Afghanistan Veterans Using Department of Veterans Affairs Health Care, 2002–2008," *American Journal of Public Health*, Vol. 99, No. 9, September 2009, pp. 1651–1658.

SECNAVINST 1730.9—See Secretary of the Navy, 2008.

SECNAVINST 1850.4E—See Secretary of the Navy, 2002.

SECNAVINST 1920.6C CH-4—See Secretary of the Navy, 2011b.

SECNAVINST 5312.12C—See Secretary of the Navy, 2006a.

SECNAVINST 5510.35B—See Secretary of the Navy, 2011a.

SECNAVINST 5815.3J—See Secretary of the Navy, 2003.

SECNAVINST 6320.24A—See Secretary of the Navy, 1999.

SECNAVINST 8120.1—See Secretary of the Navy, 2010.

SECNAV M-5510.30—See Secretary of the Navy, 2006b.

Secretary of the Air Force, *Discipline and Adverse Actions*, Air Force Instruction 36-704, July 22, 1994. As of February 20, 2014:

http://static.e-publishing.af.mil/production/1/af_a1/publication/afi36-704/afi36-704.pdf

———, "Biographical Evaluation and Screening of Troops (BEST) Program," Air Force Instruction 40-404, November 1, 1997.

Regular Air Force and Special Category Accessions, Air Force Instruction 36-2002, April 7, 1999; incorporating through change 3, October 20, 2011. As of February 20, 2014: http://static.e-publishing.af.mil/production/1/af_a1/publication/afi36-2002/afi36-2002.pdf

———, "Mental Health, Confidentiality, and Military Law," Air Force Instruction 44-109, March 1, 2000a; certified current September 20, 2010. As of February 20, 2014: http://static.e-publishing.af.mil/production/1/af_sg/publication/afi44-109/afi44-109.pdf

———, Air Force Motor Vehicle Traffic Supervision, Air Force Instruction 31-204, July 14, 2000b; incorporating change 1, July 20, 2007; U.S. Air Forces in Europe Supplement January 5, 2006; incorporating change 1, September 3, 2008; Incirlik Air Base Supplement January 28, 2010; incorporating change 1, February 16, 2011. As of February 20, 2014: http://static.e-publishing.af.mil/production/1/incirlikab/publication/ afi31-204_usafesup_incirlikabsup_i/afi31-204_usafesup_incirlikabsup_i.pdf

-, High-Risk Response, Air Force Manual 31-201, Vol. 4, March 20, 2002a.

—____, Line of Duty (Misconduct) Determination, Air Force Instruction 36-2910, October 4, 2002b; incorporating through change 2, April 5, 2010. As of June 10, 2014: http://static.e-publishing.af.mil/production/1/af_a1/publication/afi36-2910/afi36-2910.pdf

———, "Security Forces Investigations Program," Air Force Instruction 31-206, Air Combat Command Supplement 1, December 17, 2002c; certified current December 15, 2008. As of February 20, 2014:

http://static.e-publishing.af.mil/production/1/acc/publication/afi31-206_accsup/afi31-206_accsup.pdf

—_____, Aeromedical Evacuation Patent [sic] Considerations and Standards of Care, Air Force Instruction 41-307, August 20, 2003; incorporating through change 2, January 28, 2014; certified current on July 5, 2011. As of February 20, 2014: http://static.e-publishing.af.mil/production/1/af_sg/publication/afi41-307/afi41-307.pdf

——, *The Air Force Corrections System*, Air Force Instruction 31-205, April 7, 2004a; incorporating

______, *The Air Force Corrections System*, Air Force Instruction 31-205, April 7, 2004a; incorporating change 1, July 6, 2007; certified current May 17, 2010. As of February 20, 2014: http://static.e-publishing.af.mil/production/1/incirlikab/publication/afi31-205_incirlikabsup_i/ afi31-205_incirlikabsup_i.pdf

———, Administrative Discharge Procedures for Commissioned Officers, Air Force Instruction 36-3206, June 9, 2004b; incorporating through change 7, July 2, 2013. As of February 20, 2014:

http://static.e-publishing.af.mil/production/1/af_a1/publication/afi36-3206/afi36-3206.pdf

—_____, Administrative Separation of Airmen, Air Force Instruction 36-3208, July 9, 2004c; incorporating through change 7, July 2, 2013. As of February 20, 2014: http://static.e-publishing.af.mil/production/1/af_a1/publication/afi36-3208/afi36-3208.pdf

—_____, Personnel Security Program Management, Air Force Instruction 31-501, January 27, 2005a; incorporating through change 2, November 29, 2012. As of February 20, 2014: http://static.e-publishing.af.mil/production/1/af_a4_7/publication/afi31-501/afi31-501.pdf

———, Separation and Retirement Procedures for Air National Guard and Air Force Reserve Members, Air Force Instruction 36-3209, April 14, 2005b; incorporating through change 3, September 20, 2011. As of February 20, 2014:

http://static.e-publishing.af.mil/production/1/af_a1/publication/afi36-3209/afi36-3209.pdf

—_____, Assignments, Air Force Instruction 36-2110, April 20, 2005c; Air Force District of Washington Supplement December 17, 2007. As of February 20, 2014: http://static.e-publishing.af.mil/production/1/afdw/publication/afi36-2110_afdwsup_i/afi36-2110_afdwsup_i.pdf ——, Protection of Human Subjects in Biomedical and Behavioral Research, Air Force Instruction 40-402, May 5, 2005d; Headquarters U.S. Air Force Academy Supplement, February 14, 2011. As of February 25, 2014:

http://static.e-publishing.af.mil/production/1/usafa/publication/afi40-402_usafasup_i/afi40-402_usafasup_i.pdf

——, *Nuclear Weapons Personnel Reliability Program (PRP)*, Air Force Manual 10-3902, November 13, 2006b; incorporating through change 2, November 2, 2010; Air Force Global Strike Command Supplement December 5, 2012. As of February 20, 2014: http://static.e-publishing.af.mil/production/1/afgsc/publication/ dodr5210.42_afman10-3902_afgscsup/dodr5210.42_afman10-3902_afgscsup.pdf

—______, Explosive Ordnance Disposal (EOD) Program, Air Force Instruction 32-3001, October 10, 2007a; Air Force District of Washington Supplement December 9, 2009. As of February 20, 2014: http://static.e-publishing.af.mil/production/1/afdw/publication/aff32-3001_afdwsup_i/aff32-3001_afdwsup_i.pdf

———, Administering Aeromedical Staging Facilities, Air Force Instruction 44-165, November 6, 2007b. As of June 10, 2014:

http://static.e-publishing.af.mil/production/1/af_sg/publication/afi44-165/afi44-165.pdf

—____, The First Sergeant, Air Force Instruction 36-2113, December 19, 2007c; certified current January 4, 2010.

, "Administration of Military Justice," Air Force Instruction 51-201, August 15, 2008a.

——, Sexual Assault Prevention and Response (SAPR) Program, Air Force Instruction 36-6001, September 29, 2008b; incorporating change 1, September 30, 2009; certified current October 14, 2010. As of February 20, 2014:

http://www.afpc.af.mil/shared/media/document/AFD-130510-040.pdf

—_____, Arming and Use of Force by Air Force Personnel, Air Force Instruction 31-207, January 29, 2009a; Air Force District of Washington Supplement May 18, 2012. As of February 20, 2014: http://static.e-publishing.af.mil/production/1/afdw/publication/afi31-207_afdwsup_i/afi31-207_afdwsup_i.pdf

——____, Family Advocacy, Air Force Instruction 40-301, November 30, 2009b; Air Force Instruction 40-301; Headquarters U.S. Air Force Academy Supplement, July 27, 2011. As of June 10, 2014:

http://static.e-publishing.af.mil/production/1/usafa/publication/afi40-301_usafasup_i/afi40-301_usafasup_i.pdf

—______, Airfield Operations Procedures and Programs, Air Force Instruction 13-204, Vol. 3, September 1, 2010a; incorporating change 1, January 9, 2012; Air Force Special Operations Command Supplement October 26, 2012. As of February 20, 2014: http://static.e-publishing.af.mil/production/1/afsoc/publication/afi13-204v3_afsocsup_i/afi13-204v3_afsocsup_i.pdf

—______, Equal Opportunity Program Military and Civilian, Air Force Instruction 36-2706, October 5, 2010b; incorporating change 1, October 5, 2011. As of February 21, 2014: http://static.e-publishing.af.mil/production/1/af_a1/publication/afi36-2706/afi36-2706.pdf

———, Aviation and Parachutist Service, Aeronautical Ratings and Aviation Badges, Air Force Instruction 11-402, December 13, 2010c; Air Force Special Operations Command Supplement June 21, 2012. As of February 20, 2014:

http://static.e-publishing.af.mil/production/1/afsoc/publication/afi11-402_afsocsup_i/afi11-402_afsocsup_i.pdf

—_____, Mental Health, Air Force Instruction 44-172, March 14, 2011a. As of February 20, 2014: http://static.e-publishing.af.mil/production/1/af_sg/publication/afi44-172/afi44-172.pdf

———, Traumatic Stress Response, Air Force Instruction 44-153, August 29, 2011b. As of February 20, 2014:

http://static.e-publishing.af.mil/production/1/af_sg/publication/afi44-153/afi44-153.pdf

——, Base Support and Expeditionary (BAS&E) Site Planning, Air Force Instruction 10-404, October 11, 2011c; Air National Guard Supplement, August 9, 2012. As of June 10, 2014: http://static.e-publishing.af.mil/production/1/ang/publication/afi10-404_angsup_i/ afi10-404_angsup_i.pdf

TRICARE Operations and Patient Administration Functions, Air Force Instruction 41-210, June 6, 2012a. As of February 20, 2014:

http://static.e-publishing.af.mil/production/1/af_sg/publication/afi41-210/afi41-210.pdf

———, "Air Force Guidance Memorandum to AFI 36-2501, Officer Promotions and Selective Continuation," Air Force Instruction 36-2501 Air Force Guidance Memorandum 1, July 2, 2012b.

—____, *Suicide Prevention Program*, Air Force Instruction 90-505, August 10, 2012c. As of February 20, 2014:

http://www.afms.af.mil/shared/media/document/AFD-130423-030.pdf

—_____, Deployment Planning and Execution, Air Force Instruction 10-403, September 20, 2012d; incorporating change 1, April 29, 2013. As of February 20, 2014: http://static.e-publishing.af.mil/production/1/af_a4_7/publication/afi10-403/afi10-403.pdf

—______, Occupational and Environmental Health Program Management, Air Force Manual 48-146, October 9, 2012e; incorporating change 1, December 5, 2012. As of February 20, 2014: http://static.e-publishing.af.mil/production/1/af_ja/publication/afman48-146/afman48-146.pdf

, "Readiness," Air Force Policy Directive 10-2, November 6, 2012f. As of February 21, 2014: http://static.e-publishing.af.mil/production/1/af_a3_5/publication/afpd10-2/afpd10-2.pdf

—____, Duty Limiting Conditions, Air Force Instruction 10-203, January 15, 2013a. As of February 21, 2014:

http://static.e-publishing.af.mil/production/1/af_a3_5/publication/afi10-203/afi10-203.pdf

—_____, Medical Examinations and Standards, Air Force Instruction 48-123, November 8, 2013b. As of February 20, 2014:

http://static.e-publishing.af.mil/production/1/af_sg/publication/afi48-123/afi48-123.pdf

, "AFSC 5R0X1 Chaplain Assistant," Career Field Education and Training Plan 5R0X1 change 1, January 1, 2014. As of February 7, 2014:

 $http://static.e-publishing.af.mil/production/1/af_hc/publication/cfetp5r0x1c1/cfetp5r0x1c1.pdf$

Secretary of the Navy, *Mental Health Evaluations of Members of the Armed Forces*, Secretary of the Navy Instruction 6320.24A, February 16, 1999. As of February 9, 2014: http://doni.daps.dla.mil/Directives/06000%20Medical%20and%20Dental%20Services/ 06-300%20General%20Medicine%20Support/6320.24A.pdf , Department of the Navy (DON) Disability Evaluation Manual, Secretary of the Navy Instruction 1850.4E, April 30, 2002. As of February 9, 2014: http://doni.daps.dla.mil/Directives/01000%20Military%20Personnel%20Support/ 01-800%20Millitary%20Retirement%20Services%20and%20Support/1850.4E.pdf

——, Department of the Navy Clemency and Parole Systems, Secretary of the Navy Instruction 5815.3J, June 12, 2003. As of February 9, 2014: http://doni.daps.dla.mil/Directives/ 05000%20General%20Management%20Security%20and%20Safety%20Services/ 05-800%20Laws%20and%20Legal%20Services/5815.3J.pdf

, "Selection of Department of the Navy Military and Civilian Personnel and Contractor Employees for Assignment to Presidential Support Activities," Secretary of the Navy Instruction 5312.12C, April 17, 2006a. As of February 9, 2014: http://doni.daps.dla.mil/Directives/ 05000%20General%20Management%20Security%20and%20Safety%20Services/ 05-300%20Manpower%20Personnel%20Support/5312.12C.pdf

—_____, Department of the Navy Personnel Security Program, Secretary of the Navy Manual 5510.30, June 2006b. As of February 9, 2014: http://doni.daps.dla.mil/secnav%20manuals1/5510.30.pdf

———, "Confidential Communications to Chaplains," Secretary of the Navy Instruction 1730.9, February 7, 2008. As of February 9, 2014: https://www.navyreserve.navy.mil/Documents/SECNAVINST%201730.9%20 CONFIDENTIAL%20COMMUNICATIONS%20TO%20CHAPLAINS.pdf

———, Department of the Navy Nuclear Weapon Personnel Reliability Program Instruction, Secretary of the Navy Instruction 5510.35B, March 1, 2011a.

—______, Administrative Separation of Officers, Secretary of the Navy Instruction 1920.6C CH-4, September 20, 2011b. As of February 9, 2014: http://doni.daps.dla.mil/Directives/01000%20Military%20Personnel%20Support/ 01-900%20Military%20Separation%20Services/1920.6C%20w%20CH-4.pdf

Shea, Munyi, and Christine J. Yeh, "Asian American Students' Cultural Values, Stigma, and Relational Self-Construal: Correlates of Attitudes Toward Professional Help Seeking," *Journal of Mental Health Counseling*, Vol. 30, No. 2, April 2008, pp. 157–172.

Shechtman, Zipora, David Vogel, and Neta Maman, "Seeking Psychological Help: A Comparison of Individual and Group Treatment," *Psychotherapy Research*, Vol. 20, No. 1, 2010, pp. 30–36.

Shin, Sun-Kyung, and Ellen P. Lukens, "Effects of Psychoeducation for Korean Americans with Chronic Mental Illness," *Psychiatric Services*, Vol. 53, No. 9, September 1, 2002, pp. 1125–1131.

Shrivastava, A., Y. Bureau, N. Rewari, and M. Johnston, "Clinical Risk of Stigma and Discrimination of Mental Illnesses: Need for Objective Assessment and Quantification," *Indian Journal of Psychiatry*, Vol. 55, No. 2, April 2013, pp. 178–182.

Silton, N. R., K. J. Flannelly, G. Milstein, and M. L. Vaaler, "Stigma in America: Has Anything Changed? Impact of Perceptions of Mental Illness and Dangerousness on the Desire for Social Distance: 1996 and 2006," *Journal of Nervous and Mental Disease*, Vol. 199, No. 6, June 2011, pp. 361–366.

Sirey, Jo Anne, Martha L. Bruce, George S. Alexopoulos, Deborah A. Perlick, Steven J. Friedman, and Barnett S. Meyers, "Stigma as a Barrier to Recovery: Perceived Stigma and Patient-Rated Severity of Illness as Predictors of Antidepressant Drug Adherence," *Psychiatric Services*, Vol. 52, No. 12, December 2001, pp. 1615–1620.

Sirey, Jo Anne, Martha L. Bruce, George S. Alexopoulos, Deborah A. Perlick, Patrick Raue, Steven J. Friedman, and Barnett S. Meyers, "Perceived Stigma as a Predictor of Treatment Discontinuation in Young and Older Outpatients with Depression," *American Journal of Psychiatry*, Vol. 158, No. 3, March 1, 2001, pp. 479–481.

Skopp, N. A., N. E. Bush, D. L. Vogel, N. G. Wade, A. P. Sirotin, R. A. McCann, and M. J. Metzger-Abamukong, "Development and Initial Testing of a Measure of Public and Self-Stigma in the Military," *Journal of Clinical Psychology*, Vol. 68, No. 9, September 2012, pp. 1036–1047.

Smart, Laura, and Daniel M. Wegner, "Covering Up What Can't Be Seen: Concealable Stigma and Mental Control," *Journal of Personality and Social Psychology*, Vol. 77, No. 3, September 1999, pp. 474–486.

———, "The Hidden Costs of Hidden Stigma," in Todd F. Heatherton, ed., *The Social Psychology of Stigma*, New York: Guilford Press, 2000, pp. 220–242.

Søgaard, A. J., and V. Fønnebø, "The Norwegian Mental Health Campaign in 1992, Part II: Changes in Knowledge and Attitudes," *Health Education Research*, Vol. 10, No. 3, 1995, pp. 267–278.

Spagnolo, A. B., A. A. Murphy, and L. A. Librera, "Reducing Stigma by Meeting and Learning from People with Mental Illness," *Psychiatric Rehabilitation Journal*, Vol. 31, No. 3, Winter 2008, pp. 186–193.

SpeakUp ReachOut, "You Can't Fix Your Mental Health with Duct Tape," c. 2012; referenced September 13, 2013. As of February 3, 2014: http://www.suicidepreventioncolorado.org/uploads/8/7/8/4/8784346/ invite_to_town_hall_meeting.pdf

Spitz, Jonathan M., "Lessening the Impact of the Stigma of Mental Illness Within a Young Adult African American Male Culture: A Strategic Intervention," Chicago School of Professional Psychology, doctoral thesis, 2009.

Steadman, H. J., M. W. Deane, R. Borum, and J. P. Morrissey, "Comparing Outcomes of Major Models of Police Responses to Mental Health Emergencies," *Psychiatric Services*, Vol. 51, No. 5, May 2000, pp. 645–649.

Stecker, Tracy, John Fortney, Francis Hamilton, Cathy D. Sherbourne, and Icek Ajzen, "Engagement in Mental Health Treatment Among Veterans Returning from Iraq," *Patient Preference and Adherence*, Vol. 201, No. 4, March 2010, pp. 45–49.

Stengler-Wenzke, Katarina, Johanna Trosbach, Sandra Dietrich, and Mattias C. Angermeyer, "Experience of Stigmatization by Relatives of Patients with Obsessive Compulsive Disorder," *Archives* of *Psychiatric Nursing*, Vol. 18, No. 3, June 2004, pp. 88–96.

Stewart, Victoria, Perceived Social Support, Suicidal Ideation, Stigma, and Help Seeking Behaviors Among College Students, Scranton, Pa.: Marywood University, doctoral thesis, 3357205, 2009.

Stickney, S., D. Yanosky, D. R. Black, and N. L. Stickney, "Socio-Demographic Variables and Perceptual Moderators Related to Mental Health Stigma," *Journal of Mental Health*, Vol. 21, No. 3, June 2012, pp. 244–256.

Stone, Fred P., *Attitudinal and Perceptual Barriers to Accessing Mental Health Services Among Members of the U.S. Air Force*, Salt Lake City, Utah: University of Utah, doctoral thesis, 98-015D, 1998.

Straits-Tröster, Kristy A., Mira Brancu, Bob Goodale, Sheryl Pacelli, Charlotte Wilmer, Erin M. Simmons, and Harold Kudler, "Developing Community Capacity to Treat Post-Deployment Mental Health Problems: A Public Health Initiative," *Psychological Trauma: Theory, Research, Practice, and Policy*, Vol. 3, No. 3, September 2011, pp. 283–291.

Stromwall, Layne K., Lynn C. Holley, and Kathy E. Bashor, "Stigma in the Mental Health Workplace: Perceptions of Peer Employees and Clinicians," *Community Mental Health Journal*, Vol. 47, No. 4, August 2011, pp. 472–481.

Stromwall, Layne K., Lynn C. Holley, and David C. Kondrat, "Peer Employees' and Clinicians' Perceptions of Public Mental Illness Stigma and Discrimination," *Psychiatric Rehabilitation Journal*, Vol. 35, No. 5, September 2012, pp. 406–408.

Stuart, Heather, "Stigma and Work," HealthcarePapers, Vol. 5, No. 2, 2004, pp. 100-111.

———, "Media Portrayal of Mental Illness and Its Treatments: What Effect Does It Have on People with Mental Illness?" *CNS Drugs*, Vol. 20, No. 2, 2006, pp. 99–106.

Stull, Laura Grace, *Implicit Stigma of Mental Illness: Attitudes in an Evidence-Based Practice*, Purdue University, doctoral thesis, August 7, 2012. As of February 3, 2014: http://hdl.handle.net/1805/2891

Substance Abuse and Mental Health Services Administration, *Results from the 2010 National Survey on Drug Use and Health: Mental Health Findings*, Rockville, Md., National Survey on Drug Use and Health Series H-42, U.S. Department of Health and Human Services Publication (SMA) 11-4667, January 2012. As of February 3, 2014:

http://www.samhsa.gov/data/nsduh/2k10MH_Findings/2k10MHResults.htm

———, "People Recover" Comic Book: Transcription, 2013. As of February 3, 2014: http://store.samhsa.gov/shin/content/SMA13-4712/People_Recover_Comic_Book_Text.pdf

, "Mental Health Parity and Addiction Equity," last updated January 10, 2014; referenced 2013. As of February 3, 2014:

http://beta.samhsa.gov/health-reform/parity

Sudom, Kerry, Mark Zamorski, and Bryan Garber, "Stigma and Barriers to Mental Health Care in Deployed Canadian Forces Personnel," *Military Psychology*, Vol. 24, No. 4, January 2012, pp. 414–431.

Szmukler, G. I., P. Burgess, H. Herrman, S. Bloch, A. Benson, and S. Colusa, "Caring for Relatives with Serious Mental Illness: The Development of the Experience of Caregiving Inventory," *Social Psychiatry and Psychiatric Epidemiology*, Vol. 31, No. 3–4, 1996, pp. 137–148.

Tanielian, Terri, and Lisa H. Jaycox, *Invisible Wounds of War: Psychological and Cognitive Injuries, Their Consequences, and Services to Assist Recovery*, Santa Monica, Calif.: RAND Corporation, MG-720-CCF, 2008. As of February 3, 2014: http://www.rand.org/pubs/monographs/MG720.html

Taylor, S. Martin, and Michael J. Dear, "Scaling Community Attitudes Toward the Mentally Ill," *Schizophrenia Bulletin*, Vol. 7, No. 2, 1981, pp. 225–240.

Teplin, Linda A., "Criminalizing Mental Disorder: The Comparative Arrest Rate of the Mentally Ill," *American Psychologist*, Vol. 39, No. 7, July 1984, pp. 794–803.

Thoits, Peggy A., "Resisting the Stigma of Mental Illness," *Social Psychology Quarterly*, Vol. 74, No. 1, March 2011, pp. 6–28.

Thornicroft, Graham, *Shunned: Discrimination Against People with Mental Illness*, Oxford: Oxford University Press, 2006.

——, "No Time to Lose: Onset and Treatment Delay for Mental Disorders," *Epidemiology and Psychiatric Sciences*, Vol. 21, No. 1, March 2012, pp. 59–61.

Time to Change, undated home page; referenced September 13, 2013. As of January 31, 2014: http://www.time-to-change.org.uk/

TPR 630—See National Guard Bureau, 2010.

Tsang, Hector Wing-hong, Kelvin Mang-tak Fung, and Raymond Chi-keung Chung, "Self-Stigma and Stages of Change as Predictors of Treatment Adherence of Individuals with Schizophrenia," *Psychiatry Research*, Vol. 180, No. 1, November 30, 2010, pp. 10–15.

Tucker, Jeritt R., Joseph H. Hammer, David L. Vogel, Rachel L. Bitman, Nathaniel G. Wade, and Emily J. Maier, "Disentangling Self-Stigma: Are Mental Illness and Help-Seeking Self-Stigmas Different?" *Journal of Counseling Psychology*, Vol. 60, No. 4, October 2013, pp. 520–531.

Tuerk, Peter W., Maria Steenkamp, and Sheila A. M. Rauch, "Combat-Related PTSD: Scope of the Current Problem, Understanding Effective Treatment, and Barriers to Care," *Developments in Mental Health Law*, Vol. 29, No. 1, January 2010, pp. 49–58.

Turner, Erlanger A., *Parental Attitudes Toward Child Mental Health Services: The Influence of Ethnicity and Child Characteristics on Help-Seeking Intentions*, College Station, Tex.: Texas A&M University, doctoral thesis, 2009.

Under Secretary of Defense for Acquisition, Technology, and Logistics, *Occupational Medical Examinations and Surveillance Manual*, Department of Defense Manual 6055.05-M, May 2, 2007; incorporating change 1 September 16, 2008. As of February 7, 2014: http://www.dtic.mil/whs/directives/corres/pdf/605505mp.pdf

—_____, Operational Contract Support (OCS), Department of Defense Instruction 3020.41, December 20, 2011. As of February 7, 2014: http://www.dtic.mil/whs/directives/corres/pdf/302041p.pdf

———, "Nuclear Weapons Personnel Reliability Program (PRP)," Department of Defense Instruction 5210.42, July 16, 2012. As of February 4, 2014: http://www.dtic.mil/whs/directives/corres/pdf/521042p.pdf

Under Secretary of Defense for Intelligence, *Minimum Security Standards for Safeguarding Biological Select Agents and Toxins*, Department of Defense Instruction 5210.89, April 18, 2006. As of February 7, 2014:

http://www.dtic.mil/whs/directives/corres/pdf/521089p.pdf

—_____, Minimum Security Standards for Safeguarding Chemical Agents, Department of Defense Instruction 5210.65, March 12, 2007. As of February 7, 2014: http://www.dtic.mil/whs/directives/corres/pdf/521065p.pdf

———, Sensitive Compartmented Information (SCI) Administrative Security Manual: Administration of Information and Information Systems Security, Department of Defense Manual 5105.21, Vol. 1, October 19, 2012a. As of February 5, 2014: http://www.dtic.mil/whs/directives/corres/pdf/510521m_vol1.pdf

———, Sensitive Compartmented Information (SCI) Administrative Security Manual: Administration of Personnel Security, Industrial Security, and Special Activities, Department of Defense Manual 5105.21, Vol. 3, October 19, 2012b. As of February 7, 2014: http://www.dtic.mil/whs/directives/corres/pdf/510521m_vol3.pdf

Under Secretary of Defense for Personnel and Readiness, *Department of Defense Military Equal Opportunity (MEO) Program*, Department of Defense Directive 1350.2, August 18, 1995; incorporating change 1, May 7, 1997; certified current as of November 21, 2003. As of February 4, 2014:

http://www.dtic.mil/whs/directives/corres/pdf/135002p.pdf

, "Force Health Protection (FHP)," Department of Defense Directive 6200.04, October 9, 2004; certified current as of April 23, 2007. As of February 6, 2014: http://www.dtic.mil/whs/directives/corres/pdf/620004p.pdf

——, Provision of Early Intervention and Special Education Services to Eligible DoD Dependents, Department of Defense Instruction 1342.12, April 11, 2005a. As of February 7, 2014: http://www.dtic.mil/whs/directives/corres/pdf/134212p.pdf

——, *DoD Civilian Personnel Management System: Injury Compensation*, Department of Defense Instruction 1400.25-V810, April 12, 2005b; administratively reissued April 16, 2009. As of June 10, 2014:

http://www.dtic.mil/whs/directives/corres/pdf/1400.25-V810.pdf

—_____, Deployment Health, Department of Defense Instruction 6490.03, August 11, 2006; certified current as of September 30, 2011. As of February 7, 2014: http://www.dtic.mil/whs/directives/corres/pdf/649003p.pdf

——, *Domestic Abuse Involving DoD Military and Certain Affiliated Personnel*, Department of Defense Instruction 6400.06, August 21, 2007; incorporating change 1, September 20, 2011. As of February 4, 2014:

http://www.dtic.mil/whs/directives/corres/pdf/640006p.pdf

——, Department of Defense (DoD) Personnel Casualty Matters, Policies, and Procedures, Department of Defense Instruction 1300.18, January 8, 2008; incorporating change 1 August 14, 2009. As of February 7, 2014: http://www.dtic.mil/whs/directives/corres/pdf/130018p.pdf

———, "Counseling Services for DoD Military, Guard and Reserve, Certain Affiliated Personnel, and Their Family Members," Department of Defense Instruction 6490.06, April 21, 2009; incorporating change 1, July 21, 2011. As of February 4, 2014: http://www.dtic.mil/whs/directives/corres/pdf/649006p.pdf

———, "Deployment-Limiting Medical Conditions for Service Members and DoD Civilian Employees," Department of Defense Instruction 6490.07, February 5, 2010a. As of February 7, 2014: http://www.dtic.mil/whs/directives/corres/pdf/649007p.pdf

——, *Public Health Emergency Management Within the Department of Defense*, Department of Defense Instruction 6200.03, March 5, 2010b; incorporating change 2, effective October 2, 2013. As of February 4, 2014:

http://www.dtic.mil/whs/directives/corres/pdf/620003p.pdf

——, Medical Standards for Appointment, Enlistment, or Induction in the Military Services, Department of Defense Instruction 6130.03, April 28, 2010c; incorporating change 1, September 13, 2011. As of February 4, 2014:

http://www.dtic.mil/whs/directives/corres/pdf/613003p.pdf

—_____, Service Treatment Record (STR) and Non-Service Treatment Record (NSTR) Life Cycle Management, Department of Defense Instruction 6040.45, October 28, 2010d. As of February 7, 2014:

http://www.dtic.mil/whs/directives/corres/pdf/604045p.pdf

——, *Defense-Based Incident Reporting System (DIBRS): Data Segments and Elements*, Department of Defense Manual 7730.47-M, Vol. 1, December 7, 2010e. As of February 7, 2014: http://www.dtic.mil/whs/directives/corres/pdf/773047m_vol1.pdf

, "DoD Law Enforcement Officers (LEOs) Flying Armed," Department of Defense Instruction 5525.14, March 22, 2011a. As of February 7, 2014: http://www.dtic.mil/whs/directives/corres/pdf/552514p.pdf , "Mental Health Assessments for Members of the Military Services Deployed in Connection with a Contingency Operation," Directive-Type Memorandum 11-011, August 12, 2011b; incorporating change 3 December 14, 2012. As of February 7, 2014: https://www.fas.org/irp/doddir/dod/dtm-11-011.pdf

———, "Command Notification Requirements to Dispel Stigma in Providing Mental Health Care to Service Members," Department of Defense Instruction 6490.08, August 17, 2011c. As of February 3, 2014:

http://www.dtic.mil/whs/directives/corres/pdf/649008p.pdf

———, "DoD Patient Bill of Rights and Responsibilities in the Military Health System (MHS)," Department of Defense Instruction 6000.14, September 26, 2011d; incorporating change 1 October 3, 2013. As of February 7, 2014: http://www.dtic.mil/whs/directives/corres/pdf/600014p.pdf

———, "Maintenance of Psychological Health in Military Operations," Department of Defense Instruction 6490.05, November 22, 2011e; incorporating change 1, October 2, 2013. As of February 7, 2014:

http://www.dtic.mil/whs/directives/corres/pdf/649005p.pdf

—_____, Integrated Disability Evaluation System (IDES), Directive-Type Memorandum 11-015, December 19, 2011f; incorporating change 3, August 6, 2013. As of February 7, 2014: http://www.dtic.mil/whs/directives/corres/pdf/DTM-11-015.pdf

———, "DoD Directors of Psychological Health," Department of Defense Instruction 6490.09, February 27, 2012a; incorporating change 1, October 2, 2013. As of February 7, 2014: http://www.dtic.mil/whs/directives/corres/pdf/649009p.pdf

———, "Continuity of Behavioral Health Care for Transferring and Transitioning Service Members," Department of Defense Instruction 6490.10, March 26, 2012b. As of February 7, 2014: http://www.dtic.mil/whs/directives/corres/pdf/649010p.pdf

———, "Mental Health Evaluations of Members of the Military Services," Department of Defense Instruction 6490.04, March 4, 2013a. As of February 7, 2014: http://www.dtic.mil/whs/directives/corres/pdf/649004p.pdf

———, Administration of Military Correctional Facilities and Clemency and Parole Authority, Department of Defense Instruction 1325.07, March 11, 2013b. As of February 7, 2014: http://www.dtic.mil/whs/directives/corres/pdf/132507p.pdf

———, Sexual Assault Prevention and Response (SAPR) Program Procedures, Department of Defense Instruction 6495.02, March 28, 2013c. As of February 7, 2014: http://www.dtic.mil/whs/directives/corres/pdf/649502p.pdf

Under Secretary of Defense for Policy, *International Transfers of Technology, Articles, and Services*, Department of Defense Instruction 2040.2, July 10, 2008. As of February 9, 2014: http://www.dtic.mil/whs/directives/corres/pdf/204002p.pdf

——, Procedures for Sharing Information with and Providing Support to the U.S. Secret Service (USSS), Department of Homeland Security (DHS), Department of Defense Instruction 3025.19, November 29, 2011. As of February 7, 2014: http://www.dtic.mil/whs/directives/corres/pdf/302519p.pdf

University of Bristol Students' Union, "Look After Your Mate," undated; referenced September 13, 2013. As of February 3, 2014: http://www.ubu.org.uk/voice/laym/

U.S. Air Force, *The Air Force Suicide Prevention Program: A Description of Program Initiatives and Outcomes*, Air Force Pamphlet 44-160, April 2001.

———, Home and Community Care Program, "Chaplain Assistant Professional Development," Air Force Instruction 52-102, Vol. 2, August 1, 2012. As of February 4, 2014: http://static.e-publishing.af.mil/production/1/af hc/publication/afi52-102v2/afi52-102v2.pdf

U.S. Army, Suicide Risk Management and Surveillance Office, Army Behavioral Health Technology Office, *Army Suicide Event Report: Calendar Year 2006*, Tacoma, Wash.: Madigan Army Medical Center, 2007.

U.S. Army Center for Health Promotion and Preventive Medicine, *Investigation of Homicides at Fort Carson, Colorado, November 2008–May 2009*, Preventive Medicine Survey 40-5fl, July 2009. As of February 6, 2014:

http://armymedicine.mil/Documents/FinalRedactedEpiconReport14July2009.pdf

U.S. Code, Title 10, Armed Forces.

U.S. Department of Defense, TRICARE Assistance Program (TRIAP), July 27, 2011.

——, Joint Service Committee on Military Justice, *Manual for Courts-Martial, United States* (2012 Edition), Washington, D.C., 2012. As of June 10, 2014: http://purl.access.gpo.gov/GPO/LPS65043

U.S. Department of Health and Human Services, Public Health Service, Centers for Disease Control and Prevention, and National Institutes of Health, *Biosafety in Microbiological and Biomedical Laboratories*, 5th ed., (CDC) 21-1112, December 2009. As of February 9, 2014: http://www.cdc.gov/biosafety/publications/bmbl5/

U.S. Department of the Navy, *Chaplains Religious Enrichment Development Operation: CREDO*—*CREDO Retreats*, December 2011.

U.S. Department of Veterans Affairs and U.S. Department of Defense Joint Executive Council, *Joint Strategic Plan Fiscal Years 2011–2013*, c. 2011. As of June 14, 2014: http://www.va.gov/op3/docs/StrategicPlanning/VA_DOD_JEC_Joint_Strategic_Plan_ FY_2011_2013_final_with_Addendum.pdf

Vaughan, G., and C. Hansen, "Like Minds, Like Mine': A New Zealand Project to Counter the Stigma and Discrimination Associated with Mental Illness," *Australasian Psychiatry*, Vol. 12, No. 2, June 2004, pp. 113–117.

Vega, William A., Michael A. Rodriguez, and Alfonso Ang, "Addressing Stigma of Depression in Latino Primary Care Patients," *General Hospital Psychiatry*, Vol. 32, No. 2, March–April 2010, pp. 182–191.

Verhaeghe, M., and P. Bracke, "Stigma and Trust Among Mental Health Service Users," *Archives of Psychiatric Nursing*, Vol. 25, No. 4, August 2011, pp. 294–302.

Veteran Recovery, "Peer Support/Peer Education," 2009; referenced December 15, 2011. No longer available online.

Vets 4 Vets, "Peer Support for and by Iraq and Afghanistan-Era Veterans," June 23, 2011.

Vogel, David L., Zipora Shechtman, and Nathaniel G. Wade, "The Role of Public and Self-Stigma in Predicting Attitudes Toward Group Counseling," *Counseling Psychologist*, Vol. 38, No. 7, 2010, pp. 904–922.

Vogel, David L., Nathaniel G. Wade, and Shawn Haake, "Measuring the Self-Stigma Associated with Seeking Psychological Help," *Journal of Counseling Psychology*, Vol. 53, No. 3, July 2006, pp. 325–337.

Vogel, David L., Nathaniel G. Wade, and Ashley H. Hackler, "Perceived Public Stigma and the Willingness to Seek Counseling: The Mediating Roles of Self-Stigma and Attitudes Toward Counseling," *Journal of Counseling Psychology*, Vol. 54, No. 1, January 2007, pp. 40–50.

Vogel, D. L., Nathaniel G. Wade, S. R. Wester, L. Larson, and Ashley H. Hackler, "Seeking Help from a Mental Health Professional: The Influence of One's Social Network," *Journal of Clinical Psychology*, Vol. 63, No. 3, March 2007, pp. 233–245.

Vogel, David L., Stephen R. Wester, Meifen Wei, and Guy A. Boysen, "The Role of Outcome Expectations and Attitudes on Decisions to Seek Professional Help," *Journal of Counseling Psychology*, Vol. 52, No. 4, 2005, pp. 459–470.

Vogt, Dawne, "Mental Health–Related Beliefs as a Barrier to Service Use for Military Personnel and Veterans: A Review," *Psychiatric Services*, Vol. 62, No. 2, 2011, pp. 135–142.

Voorend, C. G., W. H. Van Brakel, H. Cross, V. Augustine, and B. Ebenso, "Report of the Stigma Research Workshop for the Development of Scientific Consensus Papers and Field Guidelines on Health-Related Stigma, Held in Amsterdam, the Netherlands from 11–14 October 2010," *Leprosy Review*, Vol. 82, No. 2, June 2011, pp. 188–201.

Wade, Nathaniel G., and Ashley H. Hackler, "Emotional Expression and the Decision to Seek Therapy: The Mediating Roles of the Anticipated Benefits and Risks," *Journal of Social and Clinical Psychology*, Vol. 27, No. 3, 2008, pp. 254–278.

Wade, Nathaniel G., Brian C. Post, Marilyn A. Cornish, David L. Vogel, and Jeritt R. Tucker, "Predictors of the Change in Self-Stigma Following a Single Session of Group Counseling," *Journal of Counseling Psychology*, Vol. 58, No. 2, April 2011, pp. 170–182.

Wahl, Otto F., "Mental Health Consumers' Experience of Stigma," *Schizophrenia Bulletin*, Vol. 25, No. 3, 1999, pp. 467–478.

———, "Stigma as a Barrier to Recovery from Mental Illness," *Trends in Cognitive Sciences*, Vol. 16, No. 1, January 2012, pp. 9–10.

Wang, J. L., "Perceived Work Stress, Imbalance Between Work and Family/Personal Lives, and Mental Disorders," *Social Psychiatry and Psychiatric Epidemiology*, Vol. 41, No. 7, July 2006, pp. 541–548.

——, "Mental Health Literacy and Stigma Associated with Depression in the Working Population," in Izabela Z. Schultz and E. Sally Rogers, eds., *Work Accommodation and Retention in Mental Health*, New York: Springer, 2011, pp. 341–351.

Wang, Min-Ling, *The Cultural Beliefs and Mental Health Help-Seeking Behaviors of People with Schizophrenia in Taiwan*, Washington, D.C.: Catholic University of America, doctoral thesis, 2011.

Wang, Philip S., Michael Lane, Mark Olfson, Harold A. Pincus, Kenneth B. Wells, and Ronald C. Kessler, "Twelve-Month Use of Mental Health Services in the United States: Results from the National Comorbidity Survey Replication," *Archives of General Psychiatry*, Vol. 62, No. 6, 2005, pp. 629–640.

Warner, Christopher H., George N. Appenzeller, K. Mullen, C. M. Warner, and Thomas Grieger, "Soldier Attitudes Toward Mental Health Screening and Seeking Care upon Return from Combat," *Military Medicine*, Vol. 173, No. 6, June 2008, pp. 563–569.

Warner, Christopher H., George N. Appenzeller, Jessica R. Parker, Carolynn Warner, Carroll J. Diebold, and Thomas Grieger, "Suicide Prevention in a Deployed Military Unit," *Psychiatry: Interpersonal and Biological Processes*, Vol. 74, No. 2, 2011, pp. 127–141.

Watson, Amy C., and Shaun M. Eack, "Oppression and Stigma and Their Effects," in Nina Rovinelli Heller and Alex Gitterman, eds., *Mental Health and Social Problems: A Social Work Perspective*, London: Routledge, 2011, p. 21–43.

Wegner, Daniel M., "Ironic Processes of Mental Control," *Psychological Review*, Vol. 101, No. 1, 1994, pp. 34–52.

Werner, P., I. Stein-Shvachman, and J. Heinik, "Perceptions of Self-Stigma and Its Correlates Among Older Adults with Depression: A Preliminary Study," *International Psychogeriatrics*, Vol. 21, No. 6, December 2009, pp. 1180–1189.

Westphal, Richard J., "Fleet Leaders' Attitudes About Subordinates' Use of Mental Health Services," *Military Medicine*, Vol. 172, No. 11, November 2007, pp. 1138–1143.

Whaley, A. L., "Ethnic and Racial Differences in Perceptions of Dangerousness of Persons with Mental Illness," *Psychiatric Services*, Vol. 48, No. 10, October 1997, pp. 1328–1330.

Wig, N. N., M. A. Suleiman, R. Routledge, R. Srinivasa Murthy, L. Ladrido-Ignacio, H. H. A. Ibrahim, and T. W. Harding, "Community Reactions to Mental Disorders: A Key Informant Study in Three Developing Countries," *Acta Psychiatrica Scandinavica*, Vol. 61, No. 2, February 1980, pp. 111–126.

Wong, E. C., G. N. Marshall, T. L. Schell, M. N. Elliott, K. Hambarsoomians, C. Chun, and S. M. Berthold, "Barriers to Mental Health Care Utilization for U.S. Cambodian Refugees," *Journal of Consulting and Clinical Psychology*, Vol. 74, No. 6, December 2006, pp. 1116–1120.

Wong, E. C., T. L. Schell, L. H. Jaycox, G. N. Marshall, T. Tanielian, and J. N. Miles, "Mental Health Treatment Experiences of U.S. Service Members Previously Deployed to Iraq and Afghanistan," *Psychiatric Services*, Vol. 64, No. 3, March 1, 2013, pp. 277–279.

Wood, Amy L., and Otto F. Wahl, "Evaluating the Effectiveness of a Consumer-Provided Mental Health Recovery Education Presentation," *Psychiatric Rehabilitation Journal*, Vol. 30, No. 1, 2006, pp. 46–53.

Wright, Annemarie, Anthony F. Jorm, and Andrew J. Mackinnon, "Labeling of Mental Disorders and Stigma in Young People," *Social Science and Medicine*, Vol. 73, No. 4, August 2011, pp. 498–506.

Wright, Annemarie, Patrick D. McGorry, Meredith G. Harris, Anthony F. Jorm, and Kerryn Pennell, "Development and Evaluation of a Youth Mental Health Community Awareness Campaign: The Compass Strategy," *BMC Public Health*, Vol. 6, 2006, p. 215.

Wright, Eric R., William P. Gronfein, and Timothy J. Owens, "Deinstitutionalization, Social Rejection, and the Self-Esteem of Former Mental Patients," *Journal of Health and Social Behavior*, Vol. 41, No. 1, March 2000, pp. 68–90.

Wrigley, S., H. Jackson, F. Judd, and A. Komiti, "Role of Stigma and Attitudes Toward Help-Seeking from a General Practitioner for Mental Health Problems in a Rural Town," *Australian and New Zealand Journal of Psychiatry*, Vol. 39, No. 6, June 2005, pp. 514–521.

Yamaguchi, Sosei, Yoshio Mino, and Shahir Uddin, "Strategies and Future Attempts to Reduce Stigmatization and Increase Awareness of Mental Health Problems Among Young People: A Narrative Review of Educational Interventions," *Psychiatry and Clinical Neurosciences*, Vol. 65, No. 5, August 2011, pp. 405–415.

Yap, Marie Bee Hui, and Anthony Francis Jorm, "The Influence of Stigma on First Aid Actions Taken by Young People for Mental Health Problems in a Close Friend or Family Member: Findings from an Australian National Survey of Youth," *Journal of Affective Disorders*, Vol. 134, No. 1–3, November 2011, pp. 473–477. Yap, Marie Bee Hui, Annemarie Wright, and Anthony Francis Jorm, "The Influence of Stigma on Young People's Help-Seeking Intentions and Beliefs About the Helpfulness of Various Sources of Help," *Social Psychiatry and Psychiatric Epidemiology*, Vol. 46, No. 12, December 2011, pp. 1257–1265.

Yau, Sania S. W., Katrina H. W. Pun, and Jessica P. S. Tang, "Outcome Study of School Programmes for Reducing Stigma and Promoting Mental Health," *Journal of Youth Studies*, Vol. 14, No. 1, January 2011, pp. 30–40.

Yen, Cheng-Fang, Cheng-Chun Chen, Yu Lee, Tze-Chun Tang, Ju-Yu Yen, and Chih-Hung Ko, "Self-Stigma and Its Correlates Among Outpatients with Depressive Disorders," *Psychiatric Services*, Vol. 56, No. 5, May 1, 2005, pp. 599–601.

Zamorski, M. A., *Towards a Broader Conceptualization of Need, Stigma, and Barriers to Mental Health Care in Military Organizations: Recent Research Findings from the Canadian Forces*, Ottawa, Ont.: Department of National Defence, 2011.

Zeber, J. E., P. H. Noel, M. J. Pugh, L. A. Copeland, and M. L. Parchman, "Family Perceptions of Post-Deployment Healthcare Needs of Iraq/Afghanistan Military Personnel," *Mental Health in Family Medicine*, Vol. 7, No. 3, September 2010, pp. 135–143.

Zinsow, Heidi M., Thomas W. Britt, Cynthia L. S. Pury, Mary Anne Raymond, Anna C. McFadden, and Crystal M. Burnette, "Barriers and Facilitators of Mental Health Treatment Seeking Among Active-Duty Army Personnel," *Military Psychology*, Vol. 25, No. 5, September 2013, pp. 514–535.

Zrihen, Ilan, Isaac Ashkenazi, Gad Lubin, and Racheli Magnezi, "The Cost of Preventing Stigma by Hospitalizing Soldiers in a General Hospital Instead of a Psychiatric Hospital," *Military Medicine*, Vol. 172, No. 7, July 2007, pp. 686–689.

Despite the efforts of both the U.S. Department of Defense (DoD) and the Veterans Health Administration to enhance mental health services, many service members are not regularly seeking needed care when they have mental health problems. Without appropriate treatment, these mental health problems can have wide-ranging and negative impacts on the quality of life and the social, emotional, and cognitive functioning of affected service members. The services have been actively engaged in developing policies, programs, and campaigns designed to reduce stigma and increase service members' help-seeking behavior. However, there has been no comprehensive assessment of these efforts' effectiveness and the extent to which they align with service members' needs or evidence-based practices. The goal of this research was to assess DoD's approach to stigma reduction—how well it is working and how it might be improved. To address these questions, RAND researchers used five complementary methods: (1) literature review, (2) a microsimulation modeling of costs, (3) interviews with program staff, (4) prospective policy analysis, and (5) an expert panel. The priorities outlined in this report identify ways in which program and policy development and research and evaluation can improve understanding of how best to efficiently and effectively provide needed treatment to service members with mental illness.



www.rand.org

\$29.50

